Alzheimer's Disease: Drug Discovery

Alzheimer's Disease: Drug Discovery

Edited by XUDONG HUANG, PhD



Alzheimer's Disease: Drug Discovery

ISBN: 978-0-6450017-0-9 DOI: https://dx.doi.org/10.36255/exonpublications.alzheimersdisease.2020

Edited by

Xudong Huang, PhD, Neurochemistry Laboratory, Department of Psychiatry, Massachusetts General Hospital and Harvard Medical School, Charlestown, MA, USA

Published by

Exon Publications, Brisbane, Australia

Copyright© 2020 Exon Publications

Copyright of individual chapters belongs to the respective authors. The authors grant unrestricted publishing and distribution rights to the publisher. The electronic versions of the chapters are published under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0). https://creativecommons.org/ licenses/by-nc/4.0/. Users are allowed to share and adapt the chapters for any non-commercial purposes as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source. The book in its entirety is subject to copyright by the publisher. The reproduction, modification, replication and display of the book in its entirety, in any form, by anyone, for commercial purposes are strictly prohibited without the written consent of the publisher.

Notice to the user

The views and opinions expressed in this book are believed to be accurate at the time of publication. The publisher, editors or authors cannot be held responsible or liable for any errors, omissions or consequences arising from the use of the information contained in this book. The publisher makes no warranty, implicit or explicit, with respect to the contents of this book, or its use.

First Published in December 2020 Printed in Australia

CONTENTS

Foreword Preface List of Contributors		vii ix xi
1	Alzheimer's Disease: Etiology, Neuropathology and Pathogenesis Olivia Sheppard, Michael Coleman	1
2	Immunotherapy Targeting Amyloid-β Peptides in Alzheimer's Disease Silvia Zampar, Oliver Wirths	23
3	Amyloid β-Targeted Inhibitory Peptides for Alzheimer's Disease: Current State and Future Perspectives Safura Jokar, Saeedeh Khazaei, Xaniar Esmailzadeh Gameshgoli, Mona Khafaji, Behnam Yarani, Mohammad Sharifzadeh, Davood Beiki, Omid Bavi	51
4	Tau Protein-Targeted Therapies in Alzheimer's Disease: Current State and Future Perspectives Ryszard Pluta, Marzena Ułamek-Kozioł	69
5	Targeting Metal Homeostasis as a Therapeutic Strategy for Alzheimer's Disease Antonietta Vilella, Eleonora Daini, Chiara A. De Benedictis, Andreas M. Grabrucker	83
6	Atypical Protein Kinase C Hyperactivity in Insulin-Resistant and Insulin-Sensitive Forms of Alzheimer's Disease: A Potential Therapeutic Target Robert V. Farese, Mark A. Kindy, Mini P. Sajan	99

vi	Contents

7	Modeling the Blood–Brain Barrier to Understand Drug Delivery in Alzheimer's Disease Joanna M. Wasielewska, Juliana C Da Silva Chaves, Anthony R. White, Lotta E. Oikari	117
8	Blood–Brain Barrier Degradation and the Implication of SPARC Protein as a Potential Therapeutic Target for Alzheimer's Disease Alexander Pilozzi, Caitlin Carro, Michael Whalen, Xudong Huang	135
9	The Therapeutic Potential of Epigenetic Modifications in Alzheimer's Disease Enric Bufill, Roser Ribosa-Nogué, Rafael Blesa	151
10	Chimeric Conjugates for Alzheimer's Disease Illham Dhala, Tabassum Khan, Arati Prabhu	165
11	Multifunctional Ligand Approach: Search for Effective Therapy Against Alzheimer's Disease Anna Pasieka, Dawid Panek, Barbara Malawska	181
12	Natural Products for Neurocognitive Disorders Adriana Servello, Vincenzo Leccese, Evaristo Ettorre	205
Index		221

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

FOREWORD

Alzheimer's disease (AD) is the most common cause of age-related dementia, accounting for up to 70% of all dementia cases. There are 50 million individuals living with AD worldwide and its global prevalence is expected to grow due to predicted expansion of the older population. AD presents with irreversible cognitive decline, which commences as insidious short-term memory dysfunction and gradually spreads to other cognitive domains, rendering patients mute and nonambulatory after 10-15 years of progressive course. AD is a genetically complex disease. The majority of AD cases are sporadic and their risk is predominantly controlled by the APOE genotype. The APOE ε 4 allele increases AD risk in alleledose dependent fashion while the ε^2 allele has a risk-mitigating effect. Early-onset familial AD (FAD) accounts for 2-5% of all AD cases and is caused by mutations of the amyloid precursor protein (APP) or presenilin 1 (PS1) or 2 (PS2) genes, which are inherited in autosomal dominant fashion. Neuropathological hallmarks of AD include accumulation of insoluble β -amyloid (A β) peptides in the form of parenchymal plaques and vascular deposits, intraneuronal neurofibrillary tangles (NFTs) composed of misfolded and hyperphosphorylated microtubule-associated τ protein, activated astrocytes and microglia and widespread loss of synapses and nerve cell bodies. Unfortunately, there are neither effective preventive measures nor efficacious treatments available for this devastating disease.

Identification of Aβ peptides as the main constituents of Aβ plaques and vascular deposits by the late Dr. George Glenner in 1984 and later by Drs. Konrad Beyreuther and Colin Masters in 1985 has led to the development of the Alzheimer's A cascade hypothesis, which proposes that the accumulation of A β is the primary culprit of AD pathogenesis and is thus a critical therapeutic target. Building on this premise, development of AB-directed immunotherapeutics and inhibitors of APP proteases, β -site cleaving enzyme 1 (BACE1) and γ -secretase complex (γ -SC), whose synergistic action generates $A\beta$ has been pursued over the past 20 years. However, a series of setbacks these approaches have encountered during clinical trials testing hampered progress toward their successful clinical development. At the time this book is set for print, Aducanumab an Aβ-directed monoclonal antibody remains under evaluation by regulatory agencies for an approval as a possible first AD modifying treatment, while two similar antibodies remain in advanced phase III testing. Unfortunately, clinical efficacy of anti-A β antibodies remains limited to moderating disease progression, while also causing side-effects in the form of amyloid related imaging abnormalities (ARIA). Such a situation undoubtedly calls for the diversification of AD druggable targets and a search for poly-targets or drug combination therapeutic strategies.

Contributed by an assembly of clinicians and translational and basic AD research scientists, this book intends to provide a brief overview of AD drug discovery field. It covers the underlying AD pathogenic mechanisms and provides a review of A β amyloid- and τ protein-targeted immunotherapies, and peptide inhibitors for anti-A β amyloidosis. In addition, it also examines therapeutic

potentials of various AD drug targets such as brain metal homeostasis, protein kinase C, the blood-brain barrier, epigenetic therapies, as well as discusses chimeric conjugates, multifunctional ligands, and natural products as interventional approaches. I am convinced that this book will be valuable to healthcare professionals caring for AD patients, AD researchers and interested readers as it will provoke thoughts about identifying novel and more efficacious therapeutic agents for AD and related dementias.

Martin Sadowski, MD, PhD Professor of Neurology, Psychiatry, Biochemistry and Molecular Pharmacology at NYU Grossman School of Medicine Lead, NYU AD Clinical Trial Program Attending Neurologist, NYU Langone Health New York, USA Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.fr

PREFACE

In 1906, German psychiatrist Alois Alzheimer first reported the histological features of unique plaques and neurofibrillary tangles in the post-mortem brain of a 50-year-old female patient, whom he had followed up for 5 years from admission until her death, for progressive sleep disturbance, memory dysfunction and behavioral changes such as paranoia, aggression, delusion, and confusion. The disease is now recognized as Alzheimer's disease (AD), the most common form of senile dementia. The plaques are now known as amyloid beta (A β) plaques, and the neurofibrillary tangles are known as microtubule-associated hyperphosphorylated protein τ in the paired helical filaments.

More than 200 clinical research programs and clinical trials targeting $A\beta$ and protein τ , either directly or indirectly, as a therapeutic strategy for AD have failed thus far. A growing database of the etiopathological, genetic, and biochemical features of AD indicate that it is a heterogeneous, polygenic, multifactorial, and complex disease. Hence, more rational therapeutic strategies for AD should be druggable target diversification, multi-targeting, or drug combinations. While current drug discovery programs for AD continue to focus on anti-A β and anti- τ strategies, a deeper understanding of the disease in recent years has opened drug discovery avenues involving neuroinflammation, metabolic derangements, stem cells, gene therapy and alternative therapies. Despite the high attrition rate in AD drug discovery and development, and serial failures of AD drug trials, we remain hopeful for effective AD therapeutics to come in the near future.

This book takes a snapshot of current AD drug discovery approaches to satisfy interested readers' curiosity for diversity and complexity of AD drug discovery. Chapter 1 provides an overview of the underlying pathogenic mechanisms of AD. Chapters 2-4 summarize A β immunotherapy, A β -targeted inhibitory peptides, and τ protein immunotherapy, respectively. Chapter 5 reviews AD therapeutic strategy targeting brain metal homeostasis, while Chapter 6 examines atypical protein kinase C as a potential AD drug target. Chapters 7-8 discuss blood-brain barrier (BBB) models in AD drug delivery and the BBB degradation-related protein- secreted protein acidic and rich in cysteine (SPARC) as a potential AD druggable target, respectively. Chapter 9 examines the therapeutic potential of epigenetic therapies for AD. Chapters 10 and 11 discuss the search for effective AD therapies using chimeric conjugate and multifunctional ligand approaches, and Chapter 12 examines natural products as potential interventions for neurocognitive disorders such as AD.

I am grateful for all the authors' intellectual contributions and diligence toward the fruition of this book. The 12 chapters cover diverse AD therapeutic approaches, but by no means do they completely reflect the dynamic and challenging field of AD drug discovery. I hope this book will encourage interested readers to dive into this field and appreciate both the challenges and excitement of developing effective therapeutics for AD.

Xudong Huang, PhD Associate Professor of Psychiatry at the Psychiatry Department Massachusetts General Hospital and Harvard Medical School Charlestown, MA, USA Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.pr

LIST OF CONTRIBUTORS

ADRIANA SERVELLO, MD, PHD

Department of Clinical Internal, Anesthesiological and Cardiovascular Sciences, University of Rome, Sapienza, Italy

ALEXANDER PILOZZI, MSC

Neurochemistry Laboratory, Department of Psychiatry, Massachusetts General Hospital and Harvard Medical School, Charlestown, MA 02129, USA

ANDREAS M. GRABRUCKER, PHD

Cellular Neurobiology and Neuro-Nanotechnology Lab, Department of Biological Sciences, Bernal Institute of University of Limerick, Limerick, Ireland

ANNA PASIEKA, MD

Department of Physicochemical Drug Analysis, Faculty of Pharmacy, Jagiellonian University Medical College, Kraków 30-688, Poland

ANTHONY R WHITE, PHD

Mental Health Program, Cell and Molecular Biology Department, QIMR Berghofer Medical Research Institute, Brisbane, QLD, Australia

ANTONIETTA VILELLA, PHD

Department of Biomedical, Metabolic and Neural Sciences, University of Modena and Reggio Emilia, Modena, Italy

ARATI PRABHU, PHD

SVKM's Dr. Bhanuben Nanavati College of Pharmacy, Mumbai, India

BARBARA MALAWSKA, PHD

Department of Physicochemical Drug Analysis, Faculty of Pharmacy, Jagiellonian University Medical College, Kraków 30-688, Poland

BEHNAM YARANI, MSC

Division of Nanobiotechnology, Department of Life Sciences Engineering, Faculty of New Sciences and Technologies, University of Tehran, Tehran, Iran

CAITLIN CARRO, BS (C)

Neurochemistry Laboratory, Department of Psychiatry, Massachusetts General Hospital and Harvard Medical School, Charlestown, MA 02129, USA

CHIARA A. DE BENEDICTIS, MSC

Cellular Neurobiology and Neuro-Nanotechnology Lab, Department of Biological Sciences, University of Limerick, Limerick, Ireland

DAVOOD BEIKI, PHD

Research Center for Nuclear Medicine, Tehran University of Medical Sciences, Tehran, Iran

DAWID PANEK, PHD

Department of Physicochemical Drug Analysis, Faculty of Pharmacy, Jagiellonian University Medical College, Kraków 30-688, Poland

ELEONORA DAINI, PHD

Department of Biomedical, Metabolic and Neural Sciences, University of Modena and Reggio Emilia, Modena, Italy

ENRIC BUFILL, MD, PHD

Neurology Department, Vic University Hospital, Vic, Spain

EVARISTO ETTORRE, MD, PHD

Department of Clinical Internal, Anesthesiological and Cardiovascular Sciences, University of Rome, Sapienza, Italy

ILLHAM DHALA, B. PHARM

SVKM's Dr. Bhanuben Nanavati College of Pharmacy, Mumbai, India

JOANNA M WASIELEWSKA, MSC

Mental Health Program, Cell and Molecular Biology Department, QIMR Berghofer Medical Research Institute, Brisbane, QLD, Australia

JULIANA C DA SILVA CHAVES, BSC

Mental Health Program, Cell and Molecular Biology Department, QIMR Berghofer Medical Research Institute, Brisbane, QLD, Australia

LOTTA E OIKARI, MSC, PHD

Mental Health Program, Cell and Molecular Biology Department, QIMR Berghofer Medical Research Institute, Brisbane, QLD, Australia

MARK A. KINDY, PHD, FAHA

James A. Haley Veterans Medical Center, 13000 Bruce B. Downs Blvd, Tampa, FL, USA

MARZENA UŁAMEK-KOZIOŁ, MD, PHD

Laboratory of Ischemic and Neurodegenerative Brain Research, Mossakowski Medical Research Centre, Polish Academy of Sciences, Warsaw, Poland

MICHAEL COLEMAN, PHD

John van Geest Centre for Brain Repair, Department of Clinical Neurosciences, University of Cambridge, UK

MICHAEL WHALEN, MD

Neuroscience Center, Department of Pediatrics, Massachusetts General Hospital and Harvard Medical School, Boston, MA 02114, USA

MINI P. SAJAN, PHD

Department of Internal Medicine, University of South Florida College of Medicine, Tampa, FL, USA

MOHAMMAD SHARIFZADEH, PHD

Department of Toxicology and Pharmacology, Faculty of Pharmacy, Toxicology and Poisoning Research Centre, Tehran University of Medical Sciences, Tehran, Iran

MONA KHAFAJI, PHD

Institute for Nanoscience and Nanotechnology, Sharif University of Technology, Tehran, 14588, Iran

OLIVER WIRTHS, PHD

Department of Psychiatry and Psychotherapy, University Medical Center (UMG), Georg-August-University, Von-Siebold-Str. 5, 37075 Goettingen, Germany

OLIVIA SHEPPARD, MSC

John van Geest Centre for Brain Repair, Department of Clinical Neurosciences, University of Cambridge, UK

OMID BAVI, PHD

Department of Mechanical and Aerospace Engineering, Shiraz University of Technology, Shiraz, Iran

RAFAEL BLESA, MD, PHD

Memory Unit, Neurology Department, Hospital de la Santa Creu i Sant Pau, Biomedical Research Institute Sant Pau, Universitat Autònoma de Barcelona, Barcelona, Spain

ROBERT V. FARESE, MD

James A. Haley Veterans Medical Center, 13000 Bruce B. Downs Blvd, Tampa, FL, USA

ROSER RIBOSA-NOGUÉ, MD, PHD

Memory Unit, Neurology Department, Hospital de la Santa Creu i Sant Pau, Biomedical Research Institute Sant Pau, Universitat Autònoma de Barcelona, Barcelona, Spain

RYSZARD PLUTA, MD, PHD

Laboratory of Ischemic and Neurodegenerative Brain Research, Mossakowski Medical Research Centre, Polish Academy of Sciences, Warsaw, Poland

SAEEDEH KHAZAEI, MSC

Department of Pharmaceutical Biomaterials, Faculty of Pharmacy, Tehran University of Medical Sciences, Tehran, Iran

SAFURA JOKAR, MSC

Department of Nuclear Pharmacy, Faculty of Pharmacy, Tehran University of Medical Sciences, Tehran, Iran

SILVIA ZAMPAR, MSC

Department of Psychiatry and Psychotherapy, University Medical Center (UMG), Georg-August-University, Von-Siebold-Str. 5, 37075 Goettingen, Germany

TABASSUM KHAN, PHD

SVKM's Dr. Bhanuben Nanavati College of Pharmacy, Mumbai, India

VINCENZO LECCESE, MD

Department of Clinical Internal, Anesthesiological and Cardiovascular Sciences, University of Rome, Sapienza, Italy

XANIAR ESMAILZADEH GAMESHGOLI, MSC

Colloids and Polymers Physics Group, Particle Physics Department, 15782 Campus Vida, Universidade de Santiago de Compostela, Spain

XUDONG HUANG, PHD

Neurochemistry Laboratory, Department of Psychiatry, Massachusetts General Hospital and Harvard Medical School, Charlestown, MA 02129, USA

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.cont

Alzheimer's Disease: Etiology, Neuropathology and Pathogenesis

Olivia Sheppard • Michael Coleman

John van Geest Centre for Brain Repair, Department of Clinical Neurosciences, University of Cambridge, Cambridge, UK

Author for correspondence: Olivia Sheppard, John van Geest Centre for Brain Repair, Department of Clinical Neurosciences, University of Cambridge, Cambridge, UK. Email: os356@cam.ac.uk

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch1

Abstract: Alzheimer's disease is the most common form of dementia and the most common neurodegenerative disease. It manifests as a decline in short-term memory and cognition that impairs daily behavior. Most cases of Alzheimer's disease are sporadic, but a small minority of inherited forms allow gene identification which, together with neuropathology, yields important clues about the wider causes. Environmental and metabolic risk factors, including inflammation and vascular impairment, play a role in disease onset and progression. While neuronal atrophy and a loss of synapses occur throughout the cerebral cortex, we lack a full understanding of how this arises. The known hallmarks of Alzheimer's disease include amyloid- β plaques and neurofibrillary tau tangles and while extensive research has been carried out throughout the past few decades, the exact role of these protein aggregates in the disease remains elusive. In this chapter, we discuss mechanisms that have been implicated, including inflammation, mitochondrial dysfunction, oxidative stress and changes in protein clearance.

Keywords: amyloid- β plaques; etiology of Alzheimer's disease; dementia; neurodegeneration in Alzheimer's disease; neurofibrillary tau tangles

Copyright: The Authors.

In: *Alzheimer's Disease: Drug Discovery.* Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

INTRODUCTION

Around 50 million people worldwide suffer from dementia (1). About two thirds have Alzheimer's disease (AD) (2), an irreversible neurodegenerative disorder involving a decline in memory and executive function, and personality change (3). It is named after Alois Alzheimer who first characterized AD in 1906 (4). AD results in synapse loss and neuronal atrophy predominately throughout the hippocampus and cerebral cortex. It is characterized by amyloid plaques and neurofibrillary tau tangles (NFTs), aggregates of misfolded proteins, throughout the brain. Both genetics and environmental factors are believed to play a role in AD. While there are a small number of cases due to dominant genetic mutations (5-7), a majority of AD cases are sporadic and have no single genetic cause. Environmental and metabolic risk factors such as diabetes, cerebrovascular disease, poor diet, head injury and stress are linked to increased dementia risk. The leading hypothesis as to how AD begins and progresses, the amyloid hypothesis, though quite widely accepted, leaves many questions. In particular it remains unclear "what is the best drug target?" and "what lies upstream of the rise in amyloid- β (A β) in sporadic cases?." We still lack a fundamental understanding of how AD comes to fruition, and therapies to help individuals fight the disease. AD is a chronic disease manifesting as loss of memory, language, cognition and problem-solving skills, changes in behavior and ultimately death. While the primary signs are memory loss and executive dysfunction, they are often preceded by changes in language and vision (8). Additionally, not all types of memory are equally affected. People with AD have severely impaired episodic, semantic and working memory, vet long-term memory, such as procedural memory, tends to remain intact (9, 10). Clinically, AD is classified into seven stages (Table 1) (11). Patients often die 3–10 years after onset of symptoms (12) with complications arising from immobility, such as pneumonia or blood clots (13, 14).

TABLE 1

The seven clinical stages of Alzheimer's disease (Global Deterioration Scale) (11)

	Symptoms and characteristics
Stage 1	Persons appear cognitively normal, but pathological changes are happening in the brain.
Stage 2	Prodromal stage: mild memory loss, but generally this is indistinguishable from normal forgetfulness.
Stage 3	Progression into mild cognitive impairment (MCI). Individuals may get lost or have difficulty in finding correct wording.
Stage 4	Moderate dementia; poor short-term memory. Individuals forget some of their personal history.
Stage 5	Cognition continues to decline and at this point individuals need help in their daily lives. They suffer from confusion and forget many personal details.
Stage 6	Severe dementia. Requiring constant supervision and care. Patients fail to recognize many of their family and friends and have personality changes.
Stage 7	Individuals are nearing death. They show motor symptoms, have difficulty communicating, are incontinent and require assistance in feeding.

ETIOLOGY

Both genetic and environmental risk factors play a role in the manifestation of AD. The greatest risk factor is age. At age 65, the likelihood of having AD is about 3%, rising to over 30% by age 85 (15). The incidence of AD under the age of 65 is less certain, but estimates suggest that this age group accounts for around 3% of AD cases (15). Although overall numbers are increasing with the ageing population, age-specific incidence appears to be falling in several countries (16–18).

AD can be classified by when the disease manifests, and whether it is inherited. Early-onset Alzheimer's disease (EOAD) occurs before age 65, whereas late-onset Alzheimer's disease (LOAD) accounts for over 95% of cases (19) and manifests beyond age 65. Familial AD shows Mendelian (usually dominant) inheritance, while sporadic AD shows no simple familial link (20). Nearly all EOAD are familial as these cases are due to mutations in *APP*, *PSEN1* or *PSEN2*, and a vast majority of LOAD are sporadic. Genome wide association studies (GWAS) and sequencing have now provided more than 20 risk loci in total that contribute to sporadic cases (21), but often there is no identifiable genetic cause.

Aβ precursor protein

A β precursor protein (APP) was the first gene shown to have autosomal dominant mutations causing AD. As the precursor of the aggregated peptide in amyloid plaques, its discovery in 1991 by John Hardy and colleagues (5) led to the "amyloid hypothesis," which states that the toxic build-up of A β starts a cascade of events, leading to neuronal death and disease (22, 23). There are now over 50 known *APP* mutations, accounting for approximately 10% of familial cases. Widely studied ones include the London (V717I) (24), Swedish (KM670/671NL) (25), Indiana (V717F) (26) and Artic (E693G) (27) mutations, and most cluster around cleavage sites for β and γ -secretase (28). Research suggests that many of these mutations increase A β production, or the A β 42:40 ratio, leading to increased amyloid accumulation. In very rare instances, *APP* duplication or promoter mutations can cause AD (29, 30). Interestingly, studies have also found that there is an *APP* mutation (Icelandic—A673T) which lowers A β and protects against AD (31).

Presenilins

Presenilin 1 (*PSEN1*) and Presenilin 2 (*PSEN2*) encode the catalytic components of γ -secretase, an enzyme complex involved in APP processing (32). Presenilin mutations cause autosomal dominant AD, with *PSEN1* variants being the most commonly known Mendelian genetic cause, estimated to account for around 30–50% of familial EOAD cases (33, 34). Research shows that *PSEN1* and *PSEN2* mutations alter A β production, similar to *APP* mutations (35) but paradoxically tend to confer loss of function, raising questions as to how this fits the amyloid hypothesis (36, 37).

Other genetic risk factors

Other genes known to have variants associated with AD risk include TREM2 (38), APOE (39), CLU (40–42), SORL1 (43), BIN1 (42) and PICALM (40, 42). APOE

(apolipoprotein E) is a protein involved in fat metabolism, and its E4 allele is the most common genetic risk factor for AD with an allele frequency of ~13.7% (44, 45). Heterozygosity for this allele increases the risk 3-fold (39). Although rarer, the variant $TREM2^{R47H}$ (triggering receptor expressed on myeloid cells 2) has a similar effect size (46). TREM2 is a receptor expressed on multiple cell types of the immune response, and its association supports a role for inflammation in AD pathogenesis.

Down syndrome

By age 65, up to 80% of Down syndrome (DS) individuals develop dementia (47). As with other instances of EOAD, amyloid and tau pathology begin much earlier than in LOAD, even at <40 years of age (48–50). DS results from the trisomy of chromosome 21, where the *APP* gene is located, and having three copies of this gene is sufficient to increase A β levels. However, the increased risk of developing the disease may also be due in part to triplication of other genes on chromosome 21 (47, 51, 52).

Inflammation

Sporadic AD often results from a combination of genetic and environmental risk factors, with cerebral hypoperfusion (53) and inflammation (54) being among the most common. Inflammation due to trauma, sepsis and infection has been linked to both short- and long-term cognitive impairment (55–57). Traumatic brain injury, and even bone fractures in the elderly, are implicated in dementia risk (58, 59). Higher levels of inflammatory markers such as interleukin 6 (IL-6) associate with greater risk of AD and vascular dementia (60). AD patients often have higher levels of certain inflammatory markers and activated microglia and astrocytes in the brain, which tend to surround plaques and tangles (61, 62). Finally, higher levels of these markers are associated with faster cognitive decline (63).

Cerebral, cardiovascular disease and diabetes

There is a strong link between vascular disease and dementia. Cardiovascular disease, including high blood pressure and heart attack, and cerebrovascular disease such as ischemia are associated with increased risk of AD (64). Metabolic and lifestyle risk factors for developing vascular diseases, including poor diet, obesity, high cholesterol and sedentary lifestyle, are also risk factors for dementia (65, 66). Poor diet and high cholesterol can produce metabolic changes both systemically and in the brain, and alter oxygen levels (67). Additionally, type 2 diabetes approximately doubles the risk for dementia (68–70).

Other environmental risk factors

The list of environmental and metabolic risk factors discussed here is not intended to be comprehensive, especially as the nature of epidemiology in populations with diverse genetics and lifestyle means that important mechanisms will not always generate conclusive evidence. Other risk factors implicated include pollution, stress and heavy metal exposure (71–76). Many of these risk factors share some common characteristics with one another which can thus make it difficult to determine how their presence affects the brain. Some may act through similar mechanisms, such as inflammation or oxidative stress, which will be discussed later in this chapter.

NEUROPATHOLOGY

AD is characterised by synapse loss, followed by the atrophy of neurons throughout the cerebral cortex, with the medial temporal lobe being the most severely affected (77–79). Pathology appears to start within the hippocampus and entorhinal regions and spreads subsequently throughout the fronto-temporal cortices. It reaches as far as the striatum and thalamus, usually with sparing of the cerebellum (80–83). On a macroscale level, MRI scans show shrinkage of these regions (84). In particular, pyramidal cells of the CA1 of the hippocampus are vulnerable to morphological changes and cell death, consistent with the main symptom of memory loss (85, 86). The appearance of A β plaques and NFTs precedes clinical symptoms suggesting that by symptom onset, there have been years of pathological changes making early intervention difficult.

Aβ plaques

Senile plaques are primarily made of a variety of 36–43 residue-long amyloid peptides that undergo fibrilization to form Aβ sheets that are resistant to degradation (87). They often co-localize with neuronal debris and activated microglia and astrocytes (88), and first appear in the frontal, temporal and occipital lobes of the neocortex. They spread throughout neocortical areas as well as the hippocampal formation and entorhinal region, and eventually spread further throughout the cerebral cortex to the striatum and thalamus (83) (Figure 1). Amyloid pathology appears to precede that of tau, with NFTs only being found in regions where amyloid was already present. Numerous studies have shown that cognitively unimpaired elderly individuals can also have significant A β deposition (89–91), while on the contrary, others have reported a correlation of deposition to cognitive decline (92) and dementia severity (93). A recent study has more specifically shown that differences in A β oligomer concentration may be a better correlate of disease (94, 95). It is likely that differences in methodology are responsible for the varying conclusions from these studies. It has also been suggested that cognitively normal persons with high plaque levels may have "prodromal" disease, with $A\beta$ pathology that precedes cognitive changes (96, 97).

Neuronal fibrillary tau tangles

NFTs are intraneuronal aggregates of hyperphosphorylated tau protein, encoded by the microtubule associated protein tau (*MAPT*) gene (98) (Figure 1). NFTs are



Figure 1. Amyloid and tau pathology. (A) Thioflavin S staining of Aβ plaques in the cortex of a CRND8 APP transgenic mouse. (B) AT8 staining of neurofibrillary tau tangles (NFTs) within an aged human CA1 region of the hippocampus. (C) The spread of amyloid and tau pathology throughout the brain during AD, adapted from Braak and Braak 1991 (83).

composed of paired helical fragments (PHFs) of tau fibrils approximately 20 nm in diameter (Figure 2). Like plaques, they spread throughout the brain as disease progresses, beginning near the entorhinal cortex. Braak staging is commonly used as a means of defining the progression of disease as determined by tau pathology. In stages I–II, tangles appear in the trans-entorhinal region; in stages III–IV, tangles have spread to the limbic system and start to show in the neocortex; in stages V–VI, pathology is present throughout the neocortex (83) (Figure 1). In addition to AD, several other neurodegenerative diseases are classified as tauopathies due to the presence of NFTs; these include Parkinson's disease, progressive supranuclear palsy, corticobasal degeneration and frontotemporal dementia (FTD) (99). While aggregates of amyloid and tau have both been associated with neuronal loss and toxicity, they have a poor correlation with cognitive decline as AD progresses. On the contrary, the loss of synapses is one of the strongest correlates to cognitive decline in AD (100). Familial cases and PET imaging have allowed us to identify changes in both AB and tau prior to changes in brain structure and symptom onset (101). A combination of psychological and cognitive testing, scans and CSF and blood tests (to rule out other neurological disorders) are required to obtain the diagnosis of AD. Ultimately though, definitive confirmation of the disease requires post-mortem histopathology.



Figure 2. Microtubule-associated protein Tau (MAPT) aggregation results in the accumulation of neurofibrillary tau tangles (NFTs). Tau is believed to play a role in the stabilization of microtubules. Hyperphosphorylated tau polymerization leads to the creation of insoluble paired helical fragments (PHFs), which further aggregate into NFTs.

PATHOGENESIS

The mechanism of AD pathology and neuronal loss remains elusive. The roles of both A β and tau have been extensively researched in the past few decades, yet we are still unsure of their role in disease. A variety of mechanisms have been proposed to explain what occurs in the pathogenesis of AD. It is possible that different combinations of risk factors in different patients activate the disease in different ways, and that these converge on a common pathway of degeneration.

Aβ and APP

The amyloid hypothesis remains the dominant hypothesis in AD research due to the causal mutations found in both APP and presenilin genes. APP is processed via either the amyloidogenic or non-amyloidogenic pathway. For A β , APP is sequentially cleaved by the β - and γ -secretases, releasing the peptide into the cytosol (Figure 3). Functions of APP and $A\beta$ are largely unknown, but they are thought to play a role in signal transduction for neuronal development, growth and survival (102, 103). While genetic mutations may explain A β accumulation in EOAD, it is still unclear how this occurs in LOAD. A β accumulation has been proposed to cause neuronal death via a number of mechanisms, including excitotoxicity, synaptic disruption, oxidative stress and mitochondrial dysfunction. Excitotoxicity can occur when NMDA receptors are continually activated, either by A β directly or by a downstream mechanism. In conjunction with synapse loss, both AD patients and animal models show reductions in the synaptic proteins synaptophysin and PSD-95 (104–108). A β oligomers accumulating in an AD brain (109) may be even more toxic than fibrils or plaques. Soluble oligomers appear to amass in a different manner compared to plaques and appear early in pathogenesis (110). Oligomers can disrupt cognitive function (111) and inhibit long-term potentiation (LTP) (112) in vivo, and can be neurotoxic (113) in vitro. Interestingly, oligomers tend to cluster near synapses (114) and can induce synapse loss and dysfunction (115). It has also been suggested that changes in another APP



Figure 3. Post-translational processing of A β precursor protein (APP) is thought to occur at the cell surface or within endosomes. It includes cleavage by either α - then γ -secretase (non-amyloidogenic), or β - then γ -secretase (amyloidogenic pathway).

processing product could be a contributor to AD (103). Though many APP mouse models present with aspects of AD pathology, most fail to fully recapitulate the neurodegeneration seen in the human AD brain. While this most likely reflects inter-species differences, it also raises questions about the relative importance of APP/A β in driving dementia (116).

NFTs and Tau

While no MAPT mutations are associated with AD, causal mutations in tau have been found for other neurodegenerative diseases such as FTD, suggesting that tau dysfunction and aggregation can be neurotoxic. Tau's major role is thought to be that of a cytoskeletal protein, interacting with tubulin to help assemble and stabilize microtubules (117). In humans there are six isoforms of tau generated by alternative splicing of exons 2, 3 and 10. The incorporation of exon 10 leads to four microtubule-binding repeats (4R tau) instead of three (3R tau), altering how tightly the protein binds to microtubules and its propensity to aggregate (118). Healthy adult humans express similar amounts of 3R and 4R tau. Research has shown that the ratio between the two may impact disease, with higher 4R isoforms leading to greater degeneration. In AD, there is a higher ratio of 4R to 3R, and reported downstream consequences include transcriptional alterations in the Wnt signaling pathway (119) and altered axonal transport (120). Prior to NFT formation, tau becomes hyperphosphorylated, and tau phosphorylation not only plays a large role in regulating tau function, but could be the key change resulting in the accumulation, and potential toxicity, of this protein. In fact, multiple tauopathy mutations cause tau to be more readily phosphorylated (117).

Mutant tau mouse models have shown that mutations in this gene can result in severe neurological phenotypes (121, 122). Tau has been hypothesized to induce neurotoxicity via loss of function, gain of function and/or mis-localization. Loss of function of tau occurs when tau is no longer able to stabilize microtubes having an impact on neuronal cytoskeleton, and similarly could lead to deficiencies in axonal transport (123, 124). Higher levels of tau have also been shown to inhibit vesicle and organelle trafficking, including those carrying APP, and increase levels of oxidative stress (125), as well as have an effect on axonal transport (126). The mis-localization of tau to dendritic spines has been shown to effect cognition and synapses *in vivo* (127, 128). As with APP, it remains unclear as to exactly how tau influences disease progression, but interestingly, $A\beta$ induced toxicity and impairment in LTP has been found to be a requirement for the presence of endogenous tau (129, 130). It has also been suggested that tau and $A\beta$ work together to result in transcriptional deficits (131) and synaptic changes (132) in AD.

Mitochondrial dysfunction and oxidative stress

One of the many processes that is compromised in AD is mitochondrial function. Alterations in mitochondrial morphology, number and transport, reduced cytochrome oxidase activity, deficiencies in metabolic proteins, changes in mitochondrial membrane potential and an increase in oxidative stress have been observed in AD (133, 134). Neurons are highly dependent on mitochondria, and mitochondria accumulate at synapses, helping to power their high metabolic demand. The high level of ROS production which occurs at synapses, in conjunction with insufficient antioxidants, can lead to oxidative stress (134). In addition, the brain is composed of high levels of cholesterols, which are also very vulnerable to oxidative damage (135). Thus, the high energy demands of the brain and its high lipid concentration naturally put it at risk for oxidative damage. Rather than aging driving amyloid pathology, as in the case of the amyloid hypothesis, the mitochondrial cascade hypothesis proposes that genetic and environmental factors determine the rate of mitochondrial decline, which in turn determines the rate of aging and subsequently AD (133). In terms of EOAD, APP or Aβ induces mitochondrial deficits, inducing an increase in the rate of aging, thus making some people susceptible to AD. This has been suggested as a potential link between EOAD and LOAD pathogenesis (136). Supporting this hypothesis, Thy-1-APP mice show reduced mitochondrial membrane potential and ATP synthesis and increased ROS production (137). Similarly, transgenic APP mice have shown an increase in AB within synaptic mitochondria, leading to dysfunction and oxidative stress prior to plague accumulation (138). Paradoxically, oxidative stress, a by-product of mitochondrial deficiency, has been known to affect β -secretase activity (139), which in turn could alter $A\beta$ production.

Insulin

Insulin resistance and a decrease in insulin receptors have been observed in the AD brain (140). Late stages of diabetes also result in insulin resistance in the brain. As cells are heavily dependent upon glucose metabolism for energy production, this can lead to energy deficiencies, potentially leading to oxidative stress. It has also been shown that insulin plays a role in neurotransmission (141) and can be neuroprotective during insults such as ischemia (142). Additionally, it has been reported that insulin and metabolic inhibitors result in increased levels of β -secretase in both wild-type and Tg2576 mice (an APP transgenic model). In Tg2576 mice, this also resulted in an increase in A β levels (143). Yet, as others report a protective role of insulin, it is likely that there is a certain level of this hormone which allows the brain to function optimally.

Hypoglycemia and vascular dysfunction

In addition to insulin resistance, the link between diabetes and AD could be due to changes in metabolic proteins, glucose receptors/transporters or even hypoglycemia due to over-medication. Glucose metabolism decreases in the normal aging brain (144) and even further in the AD brain (145). It has also been reported that there is a decline in the expression of glucose transporter at the blood brain barrier (BBB) in both AD patients and animal models of AD (146, 147), as well as in aged wild-type mice (147, 148). In addition, insulin-induced hypoglycemia has also been shown to cause neuronal death *in vitro* and *in vivo* (149). Glucose deprivation can elevate tau levels *in vitro* (150), and hypoglycemia has also been linked to increases in oxidative stress (151). Hypoglycemia could also be the link between cardiovascular and cerebral-vascular diseases and dementia, but whether it be hypoglycemia, hypoxia, a change in another blood component or a combination of these which increases one's risk of disease is still unknown. Finally, abnormal angiogenesis and alterations of vasculature, including changes in blood flow, have been shown in AD patients and animal models of the disease (152–154).

Inflammation

The role of inflammation is a more recent topic of interest in the AD field. As discussed previously, people with inflammation are more likely to develop dementia, and dementia patients with higher levels of inflammatory markers tend to deteriorate more rapidly. Studies in animal models have shown that inflammation can result in cognitive impairment (155), as well as neuronal damage and synaptic loss *in vivo* and *in vitro* (156–159). Although inflammation and the activation of microglia are thought to play a neuroprotective role in acute circumstances, in the long term, this may lead to neurotoxicity, and an increase in A β load (155, 160, 161). A β itself is thought to activate microglia, attracting them to plaques and enhancing phagocytosis (162–164). Potentially, microglial response to A β is protective, but after chronic activation, the microglia begin to play a detrimental role, resulting in a feed-forward loop of degradation (54). Similarly, it has been shown that increased ROS levels increase inflammatory markers, and that immune cells influence the production of ROS (165–168), demonstrating the complex interplay between A β , oxidative stress and inflammation.

Tau pathology also appears to be influenced by (169, 170), and have an effect upon (171, 172), inflammation. Research looking at the ability of microglia to phagocytose tau aggregates is conflicting, potentially due to microglia playing an initial role in clearance, but losing their ability to maintain this over extended periods (173). And finally, it has been reported that altering expression of TREM2, which plays a role in inflammation, may have an effect on A β levels and plaque-associated macrophages (174).

Ubiquitin-proteasome system

The ubiquitin-proteasome system (UPS) is involved in the degradation of misfolded and excess proteins. It is particularly important for synapse function, where there is high protein turnover (175). Proteins to be degraded go through an enzymatic process where they are labelled with a polyubiquitin chain which is recognized by the proteasome (176), and subsequently broken down. The proteasome targets monomeric proteins, so is not thought to break down plaques or tangles, but both have been shown to potentially inhibit proteasome activity (177). This could lead to a toxic build-up of excess and misfolded proteins in the brain, and more specifically synapses.

Autophagy lysosome pathway

Autophagy and lysosomal dysfunction are also proposed mechanisms of AD pathogenesis. Autophagy is involved in tau clearance (178), and plays a role in both the generation and clearance of A β . APP amyloidogenic processing involves trafficking through the endo-lysosomal pathway (179). Several genes implicated in AD including *BIN1*, *SORL1* and *PICALM* are involved in endosomal recycling, and studies have reported that each may directly play a role in APP endosomal processing (95, 180, 181).

Cholinergic hypothesis

The cholinergic hypothesis was one of the first proposed theories on the manifestation of AD (182, 183). This came to fruition due to abnormal levels of acetylcholine in the AD brain. Cholinergic neurons of the basal forebrain are one of the earliest affected by AD and there is a decrease in choline acetyltransferase (ChAT) transcription and activity in remaining neurons. Studies have also shown a relationship between acetylcholinesterase (AChE) and A β accumulation (182). However, as the AD field has moved forward there has been difficulty in linking acetylcholine with other AD pathologies. Indeed, pyramidal neurons are lost in greatest numbers in regions with plaques and tangles and these are, for the most part, glutamatergic neurons (184).

CONCLUSION

Although we have amassed a vast amount of knowledge in the search for a central, unifying mechanism behind dementia and AD, we are still lacking suitable therapies to help slow down the progression of disease. The amyloid hypothesis remains the dominant theory, yet drugs aimed at lowering A β levels have been largely unsuccessful. The possibility of NFT and plaque-load being correlative rather than causative with disease progression is entirely possible. There is much overlap between many of the risk factors, both genetic and environmental, and the known pathogenesis, highlighting the complexity of dementia. Similarly, we lack a firm understanding of how familial EOAD and sporadic LOAD ultimately produce the same neurodegenerative outcome. By enhancing our understanding of AD etiology, pathology and pathogenesis, we hope to one day find an effective therapy.

Acknowledgements: Olivia Sheppard is currently funded by an Alzheimer's Research UK project grant. The authors thank Dr. Robert Adalbert for contributing the $A\beta$ plaque image in Figure 1A. The authors also thank Dr. Antonina Kouli,

Dr. Caroline Williams-Gray and the Cambridge Brain Bank for the NFT image in Figure 1B. The Cambridge Brain Bank is supported by the NIHR Cambridge Biomedical Research Centre.

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this chapter.

Copyright and Permission: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- 1. World Health Organization. Risk reduction of cognitive decline and dementia: WHO guidelines. Geneva: World Health Organization; 2019.
- World Health Organization. Global action plan on the public health response to dementia 2017– 2025. Geneva: World Health Organization; 2017.
- 3. Wisniewski T, editor. Alzheimer's disease. Brisbane: Codon Publications; 2019. https://doi.org/10.15586/ alzheimersdisease.2019
- Hippius H, Neundörfer G. The discovery of Alzheimer's disease. Dialogues Clin Neurosci. 2003;5(1):101–8.
- Goate A, Chartier-Harlin MC, Mullan M, Brown J, Crawford F, Fidani L, et al. Segregation of a missense mutation in the amyloid precursor protein gene with familial Alzheimer's disease. Nature. 1991 Feb 21;349(6311):704–6. http://dx.doi.org/10.1038/349704a0
- Sherrington R, Rogaev EI, Liang Y, Rogaeva EA, Levesque G, Ikeda M, et al. Cloning of a gene bearing missense mutations in early-onset familial Alzheimer's disease. Nature. 1995;375(6534):754–60. http://dx.doi.org/10.1038/375754a0
- Levy-Lahad E, Wasco W, Poorkaj P, Romano DM, Oshima J, Pettingell WH, et al. Candidate gene for the chromosome 1 familial Alzheimer's disease locus. Science (80-). 1995;269(5226):973–7. http:// dx.doi.org/10.1126/science.7638622
- 8. Schachter AS, Davis KL. Alzheimer's disease. Dialogues Clin Neurosci. 2000 Jun;2(2):91-100.
- 9. Gold CA, Budson AE. Memory loss in Alzheimer's disease: Implications for development of therapeutics. Expert Rev Neurother. 2008 Dec 9;8(12):1879–91. http://dx.doi.org/10.1586/14737175.8.12.1879
- Budson AE, Price BH. Memory dysfunction. N Engl J Med. 2005 Feb 17;352(7):692–9. http://dx.doi. org/10.1056/NEJMra041071
- Reisberg B, Ferris S, De Leon M, Crook T. The Global Deterioration Scale for assessment of primary degenerative dementia. Am J Psychiatry: 1982 Sep;139(9):1136–9. http://dx.doi.org/10.1176/ajp.139.9.1136
- Zanetti O, Solerte SB, Cantoni F. Life expectancy in Alzheimer's disease (AD). Arch Gerontol Geriatr. 2009 Jan;49:237–43. http://dx.doi.org/10.1016/j.archger.2009.09.035
- Alzheimer's Association. 2016 Alzheimer's disease facts and figures. Alzheimer's Dement. 2016 Apr;12(4):459–509. http://dx.doi.org/10.1016/j.jalz.2016.03.001
- Burns A, Jacoby R, Luthert P, Levy R. Cause of death in Alzheimer's disease. Age Ageing. 1990 Sep;19(5):341–4.
- Alzheimer's Association. 2019 Alzheimer's disease facts and figures. Alzheimer's Dement. 2019 Mar;15(3):321–87. http://dx.doi.org/10.1016/j.jalz.2019.01.010
- Matthews FE, Arthur A, Barnes LE, Bond J, Jagger C, Robinson L, et al. A two-decade comparison of prevalence of dementia in individuals aged 65 years and older from three geographical areas of England: Results of the cognitive function and ageing study I and II. Lancet. 2013;382(9902):1405–12. http://dx.doi.org/10.1016/S0140-6736(13)61570-6

- Matthews FE, Stephan BCM, Robinson L, Jagger C, Barnes LE, Arthur A, et al. A two decade dementia incidence comparison from the cognitive function and ageing studies I and II. Nat Commun. 2016;7:11398. http://dx.doi.org/10.1038/ncomms11398
- Schrijvers EMC, Verhaaren BFJ, Koudstaal PJ, Hofman A, Ikram MA, Breteler MMB. Is dementia incidence declining? Trends in dementia incidence since 1990 in the Rotterdam study. Neurology. 2012;78(19):1456–63. http://dx.doi.org/10.1212/WNL.0b013e3182553be6
- Masters CL, Bateman R, Blennow K, Rowe CC, Sperling RA, Cummings JL. Alzheimer's disease. Nat Rev Dis Prim. 2015 Dec 15;1(1):15056. http://dx.doi.org/10.1038/nrdp.2015.56
- Talwar P, Sinha J, Grover S, Rawat C, Kushwaha S, Agarwal R, et al. Dissecting complex and multifactorial nature of Alzheimer's disease pathogenesis: A clinical, genomic, and systems biology perspective. Mol Neurobiol. 2016;53(7):4833–64. http://dx.doi.org/10.1007/s12035-015-9390-0
- Cuyvers E, Sleegers K. Genetic variations underlying Alzheimer's disease: Evidence from genomewide association studies and beyond. Lancet Neurol. 2016;15(8):857–68. http://dx.doi.org/10.1016/ S1474-4422(16)00127-7
- Hardy J, Higgins G. Alzheimer's disease: The amyloid cascade hypothesis. Science (80-). 1992 Apr 10;256(5054):184–5. http://dx.doi.org/10.1126/science.1566067
- 23. Hardy J, Allsop D. Amyloid deposition as the central event in the aetiology of Alzheimer's disease. Trends Pharmacol Sci. 1991 Jan;12:383–8. http://dx.doi.org/10.1016/0165-6147(91)90609-V
- Eckman CB, Mehta ND, Crook R, Perez-tur J, Prihar G, Pfeiffer E, et al. A new pathogenic mutation in the APP gene (I716V) increases the relative proportion of A beta 42(43). Hum Mol Genet. 1997 Nov;6(12):2087–9. http://dx.doi.org/10.1093/hmg/6.12.2087
- Mullan M, Crawford F, Axelman K, Houlden H, Lilius L, Winblad B, et al. A pathogenic mutation for probable Alzheimer's disease in the APP gene at the N–terminus of β–amyloid. Nat Genet. 1992 Aug;1(5):345–7. http://dx.doi.org/10.1038/ng0892-345
- Murrell J, Farlow M, Ghetti B, Benson M. A mutation in the amyloid precursor protein associated with hereditary Alzheimer's disease. Science (80-). 1991 Oct 4;254(5028):97–9. http://dx.doi.org/10.1126/ science.1925564
- Nilsberth C, Westlind-Danielsson A, Eckman CB, Condron MM, Axelman K, Forsell C, et al. The "Arctic" APP mutation (E693G) causes Alzheimer's disease by enhanced Aβ protofibril formation. Nat Neurosci. 2001 Sep 20;4(9):887–93. http://dx.doi.org/10.1038/nn0901-887
- De Jonghe C, Esselens C, Kumar-Singh S, Craessaerts K, Serneels S, Checler F, et al. Pathogenic APP mutations near the gamma-secretase cleavage site differentially affect Abeta secretion and APP C-terminal fragment stability. Hum Mol Genet. 2001 Aug 1;10(16):1665–71. http://dx.doi.org/10.1093/hmg/10.16.1665
- McNaughton D, Knight W, Guerreiro R, Ryan N, Lowe J, Poulter M, et al. Duplication of amyloid precursor protein (APP), but not prion protein (PRNP) gene is a significant cause of early onset dementia in a large UK series. Neurobiol Aging. 2012 Feb;33(2):426.e13–21. http://dx.doi.org/10.1016/j. neurobiolaging.2010.10.010
- Theuns J, Brouwers N, Engelborghs S, Sleegers K, Bogaerts V, Corsmit E, et al. Promoter mutations that increase amyloid precursor-protein expression are associated with Alzheimer disease. Am J Hum Genet. 2006 Jun;78(6):936–46. http://dx.doi.org/10.1086/504044
- Jonsson T, Atwal JK, Steinberg S, Snaedal J, Jonsson PV, Bjornsson S, et al. A mutation in APP protects against Alzheimer's disease and age-related cognitive decline. Nature. 2012 Aug 2;488(7409):96–9. http://dx.doi.org/10.1038/nature11283
- 32. Brunkan AL, Goate AM. Presenilin function and gamma-secretase activity. J Neurochem. 2005 May;93(4):769–92. http://dx.doi.org/10.1111/j.1471-4159.2005.03099.x
- 33. Singleton AB, Hall R, Ballard CG, Perry RH, Xuereb JH, Rubinsztein DC, et al. Pathology of earlyonset Alzheimer's disease cases bearing the Thr113-114ins presenilin-1 mutation. Brain. 2000 Dec 1;123 Pt 12(12):2467–74. http://dx.doi.org/10.1093/brain/123.12.2467
- 34. Cruts M, van Duijn CM, Backhovens H, Van den Broeck M, Wehnert A, Serneels S, et al. Estimation of the genetic contribution of presenilin-1 and -2 mutations in a population-based study of presenile Alzheimer disease. Hum Mol Genet. 1998 Jan;7(1):43–51. http://dx.doi.org/10.1093/hmg/7.1.43
- 35. Borchelt DR, Thinakaran G, Eckman CB, Lee MK, Davenport F, Ratovitsky T, et al. Familial Alzheimer's disease–linked presenilin 1 variants elevate Aβ1–42/1–40 ratio in vitro and in vivo. Neuron. 1996 Nov;17(5):1005–13. http://dx.doi.org/10.1016/S0896-6273(00)80230-5

- De Strooper B. Loss-of-function presenilin mutations in Alzheimer disease. EMBO Rep. 2007 Feb;8(2):141–6. http://dx.doi.org/10.1038/sj.embor.7400897
- Xia D, Watanabe H, Wu B, Lee SH, Li Y, Tsvetkov E, et al. Presenilin-1 knockin mice reveal lossof-function mechanism for familial Alzheimer's disease. Neuron. 2015 Mar 4;85(5):967–81. http:// dx.doi.org/10.1016/j.neuron.2015.02.010
- Guerreiro R, Wojtas A, Bras J, Carrasquillo M, Rogaeva E, Majounie E, et al. TREM2 variants in Alzheimer's disease. N Engl J Med. 2013 Jan 10;368(2):117–27. http://dx.doi.org/10.1056/ NEJMoa1211851
- Corder EH, Saunders AM, Strittmatter WJ, Schmechel DE, Gaskell PC, Small GW, et al. Gene dose of apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families. Science (80-). 1993;261(5123):921–3. http://dx.doi.org/10.1126/science.8346443
- Harold D, Abraham R, Hollingworth P, Sims R, Gerrish A, Hamshere ML, et al. Genome-wide association study identifies variants at CLU and PICALM associated with Alzheimer's disease. Nat Genet. 2009 Oct 6;41(10):1088–93.
- Lambert J-C, Heath S, Even G, Campion D, Sleegers K, Hiltunen M, et al. Genome-wide association study identifies variants at CLU and CR1 associated with Alzheimer's disease. Nat Genet. 2009 Oct 6;41(10):1094–9. http://dx.doi.org/10.1038/ng.439
- 42. Lee JH, Cheng R, Barral S, Reitz C, Medrano M, Lantigua R, et al. Identification of novel loci for Alzheimer disease and replication of CLU, PICALM, and BIN1 in Caribbean Hispanic individuals. Arch Neurol. 2011 Mar 1;68(3):320–8. http://dx.doi.org/10.1001/archneurol.2010.292
- Rogaeva E, Meng Y, Lee JH, Gu Y, Kawarai T, Zou F, et al. The neuronal sortilin-related receptor SORL1 is genetically associated with Alzheimer disease. Nat Genet. 2007 Feb 14;39(2):168–77.
- 44. Farrer LA, Cupples LA, Haines JL, Hyman B, Kukull WA, Mayeux R, et al. Effects of age, sex, and ethnicity on the association between apolipoprotein E genotype and Alzheimer disease. A meta-analysis. APOE and Alzheimer Disease Meta Analysis Consortium. JAMA. 1997 Oct 22;278(16):1349–56. http://dx.doi.org/10.1001/jama.278.16.1349
- 45. Liu C-C, Kanekiyo T, Xu H, Bu G. Apolipoprotein E and Alzheimer disease: Risk, mechanisms and therapy. Nat Rev Neurol. 2013 Feb 8;9(2):106–18. http://dx.doi.org/10.1038/nrneurol.2012.263
- Jonsson T, Stefansson H, Steinberg S, Jonsdottir I, Jonsson PV, Snaedal J, et al. Variant of TREM2 associated with the risk of Alzheimer's disease. N Engl J Med. 2013 Jan 10;368(2):107–16. http://dx.doi. org/10.1056/NEJMoa1211103
- Wiseman FK, Al-Janabi T, Hardy J, Karmiloff-Smith A, Nizetic D, Tybulewicz VLJ, et al. A genetic cause of Alzheimer disease: Mechanistic insights from Down syndrome. Nat Rev Neurosci. 2015 Sep;16(9):564–74. http://dx.doi.org/10.1038/nrn3983
- Mann DMA. Alzheimer's disease and Down's syndrome. Histopathology. 1988 Aug;13(2):125–37. http://dx.doi.org/10.1111/j.1365-2559.1988.tb02018.x
- Allsop D, Haga SI, Haga C, Ikeda SI, Mann DMA, Ishii T. Early senile plaques in down's syndrome brains show a close relationship with cell bodies of neurons. Neuropathol Appl Neurobiol. 1989;15(6):531–42. http://dx.doi.org/10.1111/j.1365-2990.1989.tb01252.x
- Abrahamson EE, Head E, Lott IT, Handen BL, Mufson EJ, Christian BT, et al. Neuropathological correlates of amyloid PET imaging in down syndrome. Dev Neurobiol. 2019;79(7):750–66. http://dx.doi. org/10.1002/dneu.22713
- 51. Kimura R, Kamino K, Yamamoto M, Nuripa A, Kida T, Kazui H, et al. The DYRK1A gene, encoded in chromosome 21 down syndrome critical region, bridges between β-amyloid production and tau phosphorylation in Alzheimer disease. Hum Mol Genet. 2007;16(1):15–23. http://dx.doi.org/10.1093/ hmg/ddl437
- Sheppard O, Plattner F, Rubin A, Slender A, Linehan JM, Brandner S, et al. Altered regulation of tau phosphorylation in a mouse model of down syndrome aging. Neurobiol Aging. 2012 Apr;33(4):828. e31–828.e44. http://dx.doi.org/10.1016/j.neurobiolaging.2011.06.025
- 53. Duncombe J, Kitamura A, Hase Y, Ihara M, Kalaria RN, Horsburgh K. Chronic cerebral hypoperfusion: A key mechanism leading to vascular cognitive impairment and dementia. Closing the translational gap between rodent models and human vascular cognitive impairment and dementia. Clin Sci. 2017;131(19):2451–68. http://dx.doi.org/10.1042/CS20160727

- Kinney JW, Bemiller SM, Murtishaw AS, Leisgang AM, Salazar AM, Lamb BT. Inflammation as a central mechanism in Alzheimer's disease. Alzheimer's Dement Transl Res Clin Interv. 2018 Jan 5;4(1):575–90. http://dx.doi.org/10.1016/j.trci.2018.06.014
- Iwashyna TJ, Ely EW, Smith DM, Langa KM. Long-term cognitive impairment and functional disability among survivors of severe sepsis. JAMA. 2010 Oct 27;304(16):1787–94. http://dx.doi.org/10.1001/ jama.2010.1553
- Annane D, Sharshar T. Cognitive decline after sepsis. Lancet Respir Med. 2015;3(1):61–9. http:// dx.doi.org/10.1016/S2213-2600(14)70246-2
- Benros ME, Sørensen HJ, Nielsen PR, Nordentoft M, Mortensen PB, Petersen L. The association between infections and general cognitive ability in young men—A nationwide study. PLoS One. 2015;10(5):1–13. http://dx.doi.org/10.1371/journal.pone.0124005
- Fann JR, Ribe AR, Pedersen HS, Fenger-Grøn M, Christensen J, Benros ME, et al. Long-term risk of dementia among people with traumatic brain injury in Denmark: A population-based observational cohort study. Lancet Psychiatry. 2018;5(5):424–31. http://dx.doi.org/10.1016/ S2215-0366(18)30065-8
- Tsai C-H, Chuang C-S, Hung C-H, Lin C-L, Sung F-C, Tang C-H, et al. Fracture as an independent risk factor of dementia: A nationwide population-based cohort study. Medicine (Baltimore). 2014 Nov;93(26):e188. http://dx.doi.org/10.1097/MD.00000000000188
- Engelhart MJ, Geerlings MI, Meijer J, Kiliaan A, Ruitenberg A, van Swieten JC, et al. Inflammatory proteins in plasma and the risk of dementia. Arch Neurol. 2004 May 1;61(5):668. http://dx.doi. org/10.1001/archneur.61.5.668
- Beach TG, Walker R, McGeer EG. Patterns of gliosis in alzheimer's disease and aging cerebrum. Glia. 1989;2(6):420–36. http://dx.doi.org/10.1002/glia.440020605
- Navarro V, Sanchez-Mejias E, Jimenez S, Muñoz-Castro C, Sanchez-Varo R, Davila JC, et al. Microglia in Alzheimer's disease: Activated, dysfunctional or degenerative. Front Aging Neurosci. 2018 May 11;10(5):1005–13. http://dx.doi.org/10.3389/fnagi.2018.00140
- Passamonti L, Tsvetanov KA, Jones PS, Bevan-Jones WR, Arnold R, Borchert RJ, et al. Neuroinflammation and functional connectivity in Alzheimer's disease: Interactive influences on cognitive performance. J Neurosci. 2019;39(36):7218–26. http://dx.doi.org/10.1523/JNEUROSCI.2574-18.2019
- Stampfer MJ. Cardiovascular disease and Alzheimer's disease: Common links. J Intern Med. 2006 Sep;260(3):211–23. http://dx.doi.org/10.1111/j.1365-2796.2006.01687.x
- Akbaraly TN, Singh-Manoux A, Dugravot A, Brunner EJ, Kivimäki M, Sabia S. Association of midlife diet with subsequent risk for dementia. JAMA—J Am Med Assoc. 2019;321(10):957–68. http:// dx.doi.org/10.1001/jama.2019.1432
- de Bruijn RFAG, Ikram MA. Cardiovascular risk factors and future risk of Alzheimer's disease. BMC Med. 2014 Nov 11;12(1):130. http://dx.doi.org/10.1186/s12916-014-0130-5
- Uranga RM, Keller JN. Diet and age interactions with regards to cholesterol regulation and brain pathogenesis. Curr Gerontol Geriatr Res. 2010;2010(Vldl):219683. http://dx.doi.org/10.1155/2010/219683
- Ott A, Stolk RP, van Harskamp F, Pols HAP, Hofman A, Breteler MMB. Diabetes mellitus and the risk of dementia: The Rotterdam study. Neurology. 1999 Dec 1;53(9):1937. http://dx.doi.org/10.1212/ WNL.53.9.1937
- Biessels GJ, Staekenborg S, Brunner E, Brayne C, Scheltens P. Risk of dementia in diabetes mellitus: A systematic review. Lancet Neurol. 2006 Jan;5(1):64–74. http://dx.doi.org/10.1016/ S1474-4422(05)70284-2
- Butterfield DA, Di Domenico F, Barone E. Elevated risk of type 2 diabetes for development of Alzheimer disease: A key role for oxidative stress in brain. Biochim Biophys Acta. 2014 Sep;1842(9):1693–706. http://dx.doi.org/10.1016/j.bbadis.2014.06.010
- Armstrong RA. Risk factors for Alzheimer's disease. Folia Neuropathol. 2019;57(2):87–105. http:// dx.doi.org/10.5114/fn.2019.85929
- Chen H, Kwong JC, Copes R, Tu K, Villeneuve PJ, van Donkelaar A, et al. Living near major roads and the incidence of dementia, Parkinson's disease, and multiple sclerosis: A population-based cohort study. Lancet (London, England). 2017;389(10070):718–26. http://dx.doi.org/10.1016/ S0140-6736(16)32399-6

16

- 73. Justice NJ. The relationship between stress and Alzheimer's disease. Neurobiol Stress. 2018 Feb;8(April):127–33. http://dx.doi.org/10.1016/j.ynstr.2018.04.002
- 74. Neri LC, Hewitt D. Aluminium, Alzheimer's disease, and drinking water. Lancet (London, England). 1991 Aug 10;338(8763):390. http://dx.doi.org/10.1016/0140-6736(91)90531-S
- Forbes WF, McAiney CA. Aluminum and dementia. Lancet (London, England). 1992 Sep 12; 340(8820):668–9. http://dx.doi.org/10.1016/0140-6736(92)92198-0
- 76. Cacciottolo M, Wang X, Driscoll I, Woodward N, Saffari A, Reyes J, et al. Particulate air pollutants, APOE alleles and their contributions to cognitive impairment in older women and to amyloidogenesis in experimental models. Transl Psychiatry. 2017;7(1):e1022. http://dx.doi.org/10.1038/tp.2016.280
- Sheng M, Sabatini BL, Südhof TC. Synapses and Alzheimer's disease. Cold Spring Harb Perspect Biol. 2012 May 1;4(5):1–18. http://dx.doi.org/10.1101/cshperspect.a005777
- Kashyap G, Bapat D, Das D, Gowaikar R, Amritkar RE, Rangarajan G, et al. Synapse loss and progress of Alzheimer's disease—A network model. Sci Rep. 2019;9(1):1–9. http://dx.doi.org/10.1038/ s41598-019-43076-y
- Chen M-K, Mecca AP, Naganawa M, Finnema SJ, Toyonaga T, Lin S-F, et al. Assessing synaptic density in Alzheimer disease with synaptic vesicle glycoprotein 2A positron emission tomographic imaging. JAMA Neurol. 2018;75(10):1215–24. http://dx.doi.org/10.1001/jamaneurol.2018.1836
- Duara R, Loewenstein DA, Potter E, Appel J, Greig MT, Urs R, et al. Medial temporal lobe atrophy on MRI scans and the diagnosis of Alzheimer disease. Neurology. 2008 Dec 9;71(24):1986–92. http:// dx.doi.org/10.1212/01.wnl.0000336925.79704.9f
- McDonald CR, McEvoy LK, Gharapetian L, Fennema-Notestine C, Hagler DJ, Holland D, et al. Regional rates of neocortical atrophy from normal aging to early Alzheimer disease. Neurology. 2009 Aug 11;73(6):457–65. http://dx.doi.org/10.1212/WNL.0b013e3181b16431
- Mann DMA. The topographic distribution of brain atrophy in Alzheimer's disease. Acta Neuropathol. 1991 Dec;83(1):81–6. http://dx.doi.org/10.1007/BF00294434
- Braak H, Braak E. Neuropathological staging of Alzheimer-related changes. Acta Neuropathol. 1991 Sep;82(4):239–59. http://dx.doi.org/10.1007/BF00308809
- Perl DP. Neuropathology of Alzheimer's disease. Mt Sinai J Med A J Transl Pers Med. 2010 Jan;77(1): 32–42. http://dx.doi.org/10.1002/msj.20157
- Braak E, Braak H. Alzheimer's disease: Transiently developing dendritic changes in pyramidal cells of sector CA1 of the Ammon's horn. Acta Neuropathol. 1997;93(4):323–5. http://dx.doi.org/10.1007/ s004010050622
- Masurkar AV. Towards a circuit-level understanding of hippocampal CA1 dysfunction in Alzheimer's disease across anatomical axes. J Alzheimer's Dis Park. 2018;8(1):100–6. http://dx.doi. org/10.4172/2161-0460.1000412
- Thal DR, Walter J, Saido TC, Fändrich M. Neuropathology and biochemistry of Aβ and its aggregates in Alzheimer's disease. Acta Neuropathol. 2015 Feb 23;129(2):167–82. http://dx.doi.org/10.1007/ s00401-014-1375-y
- Ries M, Sastre M. Mechanisms of Aβ clearance and degradation by glial cells. Front Aging Neurosci. 2016;8(Jun):1–9. http://dx.doi.org/10.3389/fnagi.2016.00160
- Aizenstein HJ, Nebes RD, Saxton JA, Price JC, Mathis CA, Tsopelas ND, et al. Frequent amyloid deposition without significant cognitive impairment among the elderly. Arch Neurol. 2008 Nov 10;65(11):1509. http://dx.doi.org/10.1001/archneur.65.11.1509
- Dickson DW, Crystal HA, Mattiace LA, Masur DM, Blau AD, Davies P, et al. Identification of normal and pathological aging in prospectively studied nondemented elderly humans. Neurobiol Aging. 1992;13(1):179–89. http://dx.doi.org/10.1016/0197-4580(92)90027-U
- 91. Katzman R, Terry R, DeTeresa R, Brown T, Davies P, Fuld P, et al. Clinical, pathological, and neurochemical changes in dementia: A subgroup with preserved mental status and numerous neocortical plaques. Ann Neurol. 1988;23(2):138–44. http://dx.doi.org/10.1002/ana.410230206
- Cummings B, Cotman C. Image analysis of β-amyloid load in Alzheimer's disease and relation to dementia severity. Lancet. 1995 Dec;346(8989):1524–8. http://dx.doi.org/10.1016/ S0140-6736(95)92053-6
- Nielson KA, Cummings BJ, Cotman CW. Constructional apraxia in Alzheimer's disease correlates with neuritic neuropathology in occipital cortex. Brain Res. 1996;741(1–2):284–93. http://dx.doi. org/10.1016/S0006-8993(96)00983-3

- 94. Esparza TJ, Zhao H, Cirrito JR, Cairns NJ, Bateman RJ, Holtzman DM, et al. Amyloid-beta oligomerization in Alzheimer dementia versus high-pathology controls. Ann Neurol. 2013 Jan;73(1):104–19. http://dx.doi.org/10.1002/ana.23748
- Selkoe DJ, Hardy J. The amyloid hypothesis of Alzheimer's disease at 25 years. EMBO Mol Med. 2016 Nov;8(6):595–608. http://dx.doi.org/10.15252/emmm.201606210
- 96. Price JL, Morris JC. Tangles and plaques in nondemented aging and "preclinical" alzheimer's disease. Ann Neurol. 1999;45(3):358–68. http://dx.doi.org/10.1002/1531-8249(199903)45:3%3C358:: AID-ANA12%3E3.0.CO;2-X
- Hulette CM, Welsh-Bohmer KA, Murray MG, Saunders AM, Mash DC, Mcintyre LM. Neuropathological and neuropsychological changes in "normal" aging. J Neuropathol Exp Neurol. 1998 Dec;57(12):1168–74. http://dx.doi.org/10.1097/00005072-199812000-00009
- Binder LI, Guillozet-Bongaarts AL, Garcia-Sierra F, Berry RW. Tau, tangles, and Alzheimer's disease. Biochim Biophys Acta—Mol Basis Dis. 2005 Jan;1739(2–3):216–23. http://dx.doi.org/10.1016/j. bbadis.2004.08.014
- Irwin DJ. Tauopathies as clinicopathological entities. Parkinsonism Relat Disord. 2016 Jan;22 Suppl 1(3):S29–33. http://dx.doi.org/10.1016/j.parkreldis.2015.09.020
- 100. Terry RD, Masliah E, Salmon DP, Butters N, DeTeresa R, Hill R, et al. Physical basis of cognitive alterations in Alzheimer's disease: Synapse loss is the major correlate of cognitive impairment. Ann Neurol. 1991 Oct;30(4):572–80. http://dx.doi.org/10.1002/ana.410300410
- 101. Thal DR, Attems J, Ewers M. Spreading of amyloid, tau, and microvascular pathology in alzheimer's disease: Findings from neuropathological and neuroimaging studies. J Alzheimer's Dis. 2014;42: S421–9. http://dx.doi.org/10.3233/JAD-141461
- 102. Dawkins E, Small DH. Insights into the physiological function of the β-amyloid precursor protein: Beyond Alzheimer's disease. J Neurochem. 2014 Jun;129(5):756–69. http://dx.doi.org/10.1111/ jnc.12675
- 103. Zhang Y, Thompson R, Zhang H, Xu H. APP processing in Alzheimer's disease. Mol Brain. 2011 Nov;4(1):3. http://dx.doi.org/10.1186/1756-6606-4-3
- 104. Sza C-I, Troncoso JC, Kawas C, Moution P, Price DL, Martin LJ. Loss of the presynaptic vesicle protein synaptophysin in hippocampus correlates with cognitive decline in Alzheimer disease. J Neuropathol Exp Neurol. 1997 Aug;56(8):933–44. http://dx.doi.org/10.1097/00005072-199708000-00011
- 105. Gylys KH, Fein JA, Yang F, Wiley DJ, Miller CA, Cole GM. Synaptic changes in Alzheimer's disease. Am J Pathol. 2004 Nov;165(5):1809–17. http://dx.doi.org/10.1016/S0002-9440(10)63436-0
- 106. Mucke L, Masliah E, Yu GQ, Mallory M, Rockenstein EM, Tatsuno G, et al. High-level neuronal expression of Aβ(1-42) in wild-type human amyloid protein precursor transgenic mice: Synaptotoxicity without plaque formation. J Neurosci. 2000;20(11):4050–8. http://dx.doi.org/10.1523/JNEUROSCI. 20-11-04050.2000
- 107. Shao CY, Mirra SS, Sait HBR, Sacktor TC, Sigurdsson EM. Postsynaptic degeneration as revealed by PSD-95 reduction occurs after advanced Aβ and tau pathology in transgenic mouse models of Alzheimer's disease. Acta Neuropathol. 2011;122(3):285–92. http://dx.doi.org/10.1007/ s00401-011-0843-x
- 108. Harwell CS, Coleman MP. Synaptophysin depletion and intraneuronal Aβ in organotypic hippocampal slice cultures from huAPP transgenic mice. Mol Neurodegener. 2016;11(1):44. http://dx.doi. org/10.1186/s13024-016-0110-7
- 109. Gong Y, Chang L, Viola KL, Lacor PN, Lambert MP, Finch CE, et al. Alzheimer's disease-affected brain: Presence of oligomeric A ligands (ADDLs) suggests a molecular basis for reversible memory loss. Proc Natl Acad Sci. 2003 Sep 2;100(18):10417–22. http://dx.doi.org/10.1073/pnas.1834302100
- Viola KL, Klein WL. Amyloid β oligomers in Alzheimer's disease pathogenesis, treatment, and diagnosis. Acta Neuropathol. 2015 Feb;129(2):183–206. http://dx.doi.org/10.1007/s00401-015-1386-3
- 111. Cleary JP, Walsh DM, Hofmeister JJ, Shankar GM, Kuskowski MA, Selkoe DJ, et al. Natural oligomers of the amyloid-beta protein specifically disrupt cognitive function. Nat Neurosci. 2005 Jan;8(1): 79–84. http://dx.doi.org/10.1038/nn1372
- 112. Walsh DM, Klyubin I, Fadeeva JV, Cullen WK, Anwyl R, Wolfe MS, et al. Naturally secreted oligomers of amyloid beta protein potently inhibit hippocampal long-term potentiation in vivo. Nature. 2002 Apr 4;416(6880):535–9. http://dx.doi.org/10.1038/416535a

- 113. Lambert MP, Barlow AK, Chromy BA, Edwards C, Freed R, Liosatos M, et al. Diffusible, nonfibrillar ligands derived from Abeta1-42 are potent central nervous system neurotoxins. Proc Natl Acad Sci U S A. 1998 May 26;95(11):6448–53. http://dx.doi.org/10.1073/pnas.95.11.6448
- 114. Lacor PN. Synaptic targeting by Alzheimer's-related amyloid oligomers. J Neurosci. 2004 Nov 10;24(45):10191–200. http://dx.doi.org/10.1523/JNEUROSCI.3432-04.2004
- 115. Shankar GM, Bloodgood BL, Townsend M, Walsh DM, Selkoe DJ, Sabatini BL. Natural oligomers of the Alzheimer amyloid-beta protein induce reversible synapse loss by modulating an NMDA-type glutamate receptor-dependent signaling pathway. J Neurosci. 2007 Mar 14;27(11):2866–75. http:// dx.doi.org/10.1523/JNEUROSCI.4970-06.2007
- 116. Kent SA, Spires-Jones TL, Durrant CS. The physiological roles of tau and Aβ: Implications for Alzheimer's disease pathology and therapeutics. Acta Neuropathol. 2020 Jul 29;140:417–47. http:// dx.doi.org/10.1007/s00401-020-02196-w
- 117. Iqbal K, Liu F, Gong C-X, Grundke-Iqbal I. Tau in Alzheimer disease and related tauopathies. Curr Alzheimer Res. 2010 Dec;7(8):656–64. http://dx.doi.org/10.2174/156720510793611592
- 118. Gao Y, Tan L, Yu J-T, Tan L. Tau in Alzheimer's disease: Mechanisms and therapeutic strategies. Curr Alzheimer Res. 2018 Jan 23;15(3):1005–13. http://dx.doi.org/10.2174/1567205014666170417111 859
- 119. Chen S, Townsend K, Goldberg TE, Davies P, Conejero-Goldberg C. MAPT isoforms: Differential transcriptional profiles related to 3R and 4R splice variants. J Alzheimer's Dis. 2011 Jan 7;22(4):1313–29. http://dx.doi.org/10.3233/JAD-2010-101155
- 120. Combs B, Mueller RL, Morfini G, Brady ST, Kanaan NM. Tau and axonal transport misregulation in tauopathies. Adv Exp Med Biol. 2019;1184(3):81–95. http://dx.doi.org/10.1007/978-981-32-9358-8_7
- 121. Yoshiyama Y, Higuchi M, Zhang B, Huang S-M, Iwata N, Saido TC, et al. Synapse loss and microglial activation precede tangles in a P301S tauopathy mouse model. Neuron. 2007 Feb;53(3):337–51. http://dx.doi.org/10.1016/j.neuron.2007.01.010
- 122. Allen B, Ingram E, Takao M, Smith MJ, Jakes R, Virdee K, et al. Abundant tau filaments and nonapoptotic neurodegeneration in transgenic mice expressing human P301S tau protein. J Neurosci. 2002 Nov 1;22(21):9340–51. http://dx.doi.org/10.1523/JNEUROSCI.22-21-09340.2002
- Noble W, Pooler AM, Hanger DP. Advances in tau-based drug discovery. Expert Opin Drug Discov. 2011 Aug 24;6(8):797–810. http://dx.doi.org/10.1517/17460441.2011.586690
- 124. Brunden KR, Trojanowski JQ, Lee VM-Y. Advances in tau-focused drug discovery for Alzheimer's disease and related tauopathies. Nat Rev Drug Discov. 2009 Oct;8(10):783–93. http://dx.doi. org/10.1038/nrd2959
- 125. Stamer K, Vogel R, Thies E, Mandelkow E, Mandelkow E-M. Tau blocks traffic of organelles, neurofilaments, and APP vesicles in neurons and enhances oxidative stress. J Cell Biol. 2002 Mar 18;156(6):1051–63. http://dx.doi.org/10.1083/jcb.200108057
- 126. Dixit R, Ross JL, Goldman YE, Holzbaur ELF. Differential regulation of dynein and kinesin motor proteins by tau. Science. 2008 Feb 22;319(5866):1086–9. http://dx.doi.org/10.1126/science.1152993
- 127. Hoover BR, Reed MN, Su J, Penrod RD, Kotilinek LA, Grant MK, et al. Tau mislocalization to dendritic spines mediates synaptic dysfunction independently of neurodegeneration. Neuron. 2010 Dec;68(6):1067–81. http://dx.doi.org/10.1016/j.neuron.2010.11.030
- 128. Miller EC, Teravskis PJ, Dummer BW, Zhao X, Huganir RL, Liao D. Tau phosphorylation and tau mislocalization mediate soluble Aβ oligomer-induced AMPA glutamate receptor signaling deficits. Eur J Neurosci. 2014 Apr;39(7):1214–24. http://dx.doi.org/10.1111/ejn.12507
- 129. Wang Y, Mandelkow E. Tau in physiology and pathology. Nat Rev Neurosci. 2016;17(1):5–21. http:// dx.doi.org/10.1038/nrn.2015.1
- Shipton OA, Leitz JR, Dworzak J, Acton CEJ, Tunbridge EM, Denk F, et al. Tau protein is required for amyloid β-induced impairment of hippocampal long-term potentiation. J Neurosci. 2011;31(5):1688– 92. http://dx.doi.org/10.1523/JNEUROSCI.2610-10.2011
- 131. Pickett EK, Herrmann AG, McQueen J, Abt K, Dando O, Tulloch J, et al. Amyloid beta and tau cooperate to cause reversible behavioral and transcriptional deficits in a model of Alzheimer's disease. Cell Rep. 2019 Dec;29(11):3592–604.e5. http://dx.doi.org/10.1016/j.celrep.2019.11.044

- 132. Spires-Jones TL, Hyman BT. The intersection of amyloid beta and tau at synapses in Alzheimer's disease. Neuron. 2014 May;82(4):756–71. http://dx.doi.org/10.1016/j.neuron.2014.05.004
- 133. Swerdlow RH, Burns JM, Khan SM. The Alzheimer's disease mitochondrial cascade hypothesis. Zhu X, Beal MF, Wang X, Perry G, Smith MA, editors. J Alzheimer's Dis. 2010 Jun 3;20(s2):S265–79. http://dx.doi.org/10.3233/JAD-2010-100339
- 134. Cenini G, Voos W. Mitochondria as potential targets in Alzheimer disease therapy: An update. Front Pharmacol. 2019;10(July):1–20. http://dx.doi.org/10.3389/fphar.2019.00902
- Vetrivel KS, Thinakaran G. Membrane rafts in Alzheimer's disease beta-amyloid production. Biochim Biophys Acta. 2010 Aug;1801(8):860–7. http://dx.doi.org/10.1016/j.bbalip.2010.03.007
- 136. Swerdlow RH. Pathogenesis of Alzheimer's disease. Clin Interv Aging. 2007;2(3):347-59.
- 137. Hauptmann S, Scherping I, Dröse S, Brandt U, Schulz KL, Jendrach M, et al. Mitochondrial dysfunction: An early event in Alzheimer pathology accumulates with age in AD transgenic mice. Neurobiol Aging. 2009 Oct;30(10):1574–86. http://dx.doi.org/10.1016/j.neurobiolaging.2007.12.005
- Du H, Guo L, Yan S, Sosunov AA, McKhann GM, Yan SSD. Early deficits in synaptic mitochondria in an Alzheimer's disease mouse model. Proc Natl Acad Sci U S A. 2010;107(43):18670–5. http://dx.doi. org/10.1073/pnas.1006586107
- 139. Tamagno E, Guglielmotto M, Aragno M, Borghi R, Autelli R, Giliberto L, et al. Oxidative stress activates a positive feedback between the gamma- and beta-secretase cleavages of the beta-amyloid precursor protein. J Neurochem. 2008 Feb;104(3):683–95. http://dx.doi.org/10.3389/fnins.2018.00830
- 140. Ferreira LSS, Fernandes CS, Vieira MNN, De Felice FG. Insulin resistance in Alzheimer's disease. Front Neurosci. 2018;12(Nov):1–11. http://dx.doi.org/10.3389/fnins.2018.00830
- 141. Smythe GA, Bradshaw JE, Nicholson MV, Grunstein HS, Storlien LH. Rapid bidirectional effects of insulin on hypothalamic noradrenergic and serotoninergic neuronal activity in the rat: Role in glucose homeostasis. Endocrinology. 1985;117(4):1590–7. http://dx.doi.org/10.1210/endo-117-4-1590
- 142. Shuaib A, Ijaz MS, Waqar T, Voll C, Kanthan R, Miyashita H, et al. Insulin elevates hippocampal GABA levels during ischemia. This is independent of its hypoglycemic effect. Neuroscience. 1995;67(4): 809–14. http://dx.doi.org/10.1016/0306-4522(95)00093-X
- 143. Velliquette RA, O'Connor T, Vassar R. Energy inhibition elevates beta-secretase levels and activity and is potentially amyloidogenic in APP transgenic mice: Possible early events in Alzheimer's disease pathogenesis. J Neurosci. 2005 Nov 23;25(47):10874–83. http://dx.doi.org/10.1523/ JNEUROSCI.2350-05.2005
- 144. Goyal MS, Vlassenko AG, Blazey TM, Su Y, Couture LE, Durbin TJ, et al. Loss of brain aerobic glycolysis in normal human aging. Cell Metab. 2017 Aug;26(2):353–60.e3. http://dx.doi.org/10.1016/j. cmet.2017.07.010
- 145. Mosconi L. Glucose metabolism in normal aging and Alzheimer's disease: Methodological and physiological considerations for PET studies. Clin Transl Imaging. 2013 Aug;1(4):1–7. http://dx.doi.org/10.1007/s40336-013-0026-y
- 146. Shah K, DeSilva S, Abbruscato T. The role of glucose transporters in brain disease: Diabetes and Alzheimer's disease. Int J Mol Sci. 2012;13(10):12629–55. http://dx.doi.org/10.3390/ijms131012629
- 147. Ding F, Yao J, Rettberg JR, Chen S, Brinton RD. Early decline in glucose transport and metabolism precedes shift to ketogenic system in female aging and Alzheimer's mouse brain: Implication for bioenergetic intervention. PLoS One. 2013;8(11):1–14. http://dx.doi.org/10.1371/journal.pone.0079977
- 148. Lee KY, Yoo DY, Jung HY, Baek L, Lee H, Kwon HJ, et al. Decrease in glucose transporter 1 levels and translocation of glucose transporter 3 in the dentate gyrus of C57BL/6 mice and gerbils with aging. Lab Anim Res. 2018;34(2):58. http://dx.doi.org/10.5625/lar.2018.34.2.58
- 149. Suh SW, Aoyama K, Chen Y, Garnier P, Matsumori Y, Gum E, et al. Hypoglycemic neuronal death and cognitive impairment are prevented by poly(ADP-ribose) polymerase inhibitors administered after hypoglycemia. J Neurosci. 2003;23(33):10681–90. http://dx.doi.org/10.1523/ JNEUROSCI.23-33-10681.2003
- 150. Cheng B, Mattson MP. Glucose deprivation elicits neurofibrillary tangle-like antigenic changes in hippocampal neurons: Prevention by NGF and bFGF. Exp Neurol. 1992 Aug;117(2):114–23. http:// dx.doi.org/10.1016/0014-4886(92)90120-F

- 151. Wang J, Alexanian A, Ying R, Kizhakekuttu TJ, Dharmashankar K, Vasquez-Vivar J, et al. Acute exposure to low glucose rapidly induces endothelial dysfunction and mitochondrial oxidative stress. Arterioscler Thromb Vasc Biol. 2012 Mar;32(3):712–20. http://dx.doi.org/10.1161/ ATVBAHA.111.227389
- 152. Durrant CS, Ruscher K, Sheppard O, Coleman MP, Özen I. Beta secretase 1-dependent amyloid precursor protein processing promotes excessive vascular sprouting through NOTCH3 signalling. Cell Death Dis. 2020;11(2):98. http://dx.doi.org/10.1038/s41419-020-2288-4
- 153. Roher AE, Debbins JP, Malek-Ahmadi M, Chen K, Pipe JG, Maze S, et al. Cerebral blood flow in Alzheimer's disease. Vasc Health Risk Manag. 2012;8:599–611. http://dx.doi.org/10.2147/VHRM. S34874
- 154. Navarro-Dorado J, Villalba N, Prieto D, Brera B, Martín-Moreno AM, Tejerina T, et al. Vascular dysfunction in a transgenic model of Alzheimer's disease: Effects of CB1R and CB2R cannabinoid agonists. Front Neurosci. 2016;10(Sep):422. http://dx.doi.org/10.3389/fnins.2016.00422
- 155. Lee JW, Lee YK, Yuk DY, Choi DY, Ban SB, Oh KW, et al. Neuro-inflammation induced by lipopolysaccharide causes cognitive impairment through enhancement of beta-amyloid generation. J Neuroinflammation. 2008;5:1–14. http://dx.doi.org/10.1186/1742-2094-5-37
- 156. Sheppard O, Coleman MP, Durrant CS. Lipopolysaccharide-induced neuroinflammation induces presynaptic disruption through a direct action on brain tissue involving microglia-derived interleukin 1 beta. J Neuroinflammation. 2019 May 18;16(1):106. http://dx.doi.org/10.1186/s12974-019-1490-8
- 157. Chen YY, Zhang L, Shi DL, Song XH, Shen YL, Zheng MZ, et al. Resveratrol attenuates subacute systemic inflammation-induced spatial memory impairment via inhibition of astrocyte activation and enhancement of synaptophysin expression in the hippocampus. Ann Clin Lab Sci. 2017;47(1):17–24.
- Rao JS, Kellom M, Kim H-W, Rapoport SI, Reese EA. Neuroinflammation and synaptic loss. Neurochem Res. 2012 May;37(5):903–10. http://dx.doi.org/10.1007/s11064-012-0708-2
- 159. Castaño A, Herrera AJ, Cano J, Machado A. Lipopolysaccharide intranigral injection induces inflammatory reaction and damage in nigrostriatal dopaminergic system. J Neurochem. 1998 Apr 14;70(4):1584–92. http://dx.doi.org/10.1046/j.1471-4159.1998.70041584.x
- 160. Sheng JG, Bora SH, Xu G, Borchelt DR, Price DL, Koliatsos VE. Lipopolysaccharide-inducedneuroinflammation increases intracellular accumulation of amyloid precursor protein and amyloid β peptide in APPswe transgenic mice. Neurobiol Dis. 2003;14(1):133–45. http://dx.doi.org/10.1016/ S0969-9961(03)00069-X
- 161. Wang L-M, Wu Q, Kirk RA, Horn KP, Ebada Salem AH, Hoffman JM, et al. Lipopolysaccharide endotoxemia induces amyloid-β and p-tau formation in the rat brain. Am J Nucl Med Mol Imaging. 2018 Nov;8(2):86–99.
- 162. Neniskyte U, Neher JJ, Brown GC. Neuronal death induced by nanomolar amyloid β is mediated by primary phagocytosis of neurons by microglia. J Biol Chem. 2011;286(46):39904–13. http://dx.doi. org/10.1074/jbc.M111.267583
- 163. Stalder M, Phinney A, Probst A, Sommer B, Staufenbiel M, Jucker M. Association of microglia with amyloid plaques in brains of APP23 transgenic mice. Am J Pathol. 1999;154(6):1673–84. http:// dx.doi.org/10.1016/S0002-9440(10)65423-5
- 164. Bolmont T, Haiss F, Eicke D, Radde R, Mathis CA, Klunk WE, et al. Dynamics of the microglial/ amyloid interaction indicate a role in plaque maintenance. J Neurosci. 2008;28(16):4283–92. http:// dx.doi.org/10.1523/JNEUROSCI.4814-07.2008
- 165. Yang W, Tao Y, Wu Y, Zhao X, Ye W, Zhao D, et al. Neutrophils promote the development of reparative macrophages mediated by ROS to orchestrate liver repair. Nat Commun. 2019 Dec 6;10(1):1076. http://dx.doi.org/10.1038/s41467-019-09046-8
- 166. Flohé L, Brigelius-Flohé R, Saliou C, Traber MG, Packer L. Redox regulation of NF-kappa B activation. Free Radic Biol Med. 1997;22(6):1115–26. http://dx.doi.org/10.1016/S0891-5849(96)00501-1
- 167. Anderson MT, Staal FJT, Gitler C, Herzenberg LA, Herzenberg LA. Separation of oxidant-initiated and redox-regulated steps in the NF-κB signal transduction pathway. Proc Natl Acad Sci U S A. 1994;91(24):11527–31. http://dx.doi.org/10.1073/pnas.91.24.11527
- 168. Tan BL, Norhaizan ME, Liew WPP, Rahman HS. Antioxidant and oxidative stress: A mutual interplay in age-related diseases. Front Pharmacol. 2018;9(Oct):1–28. http://dx.doi.org/10.3389/fphar. 2018.01162

- 169. Bemiller SM, McCray TJ, Allan K, Formica SV, Xu G, Wilson G, et al. TREM2 deficiency exacerbates tau pathology through dysregulated kinase signaling in a mouse model of tauopathy. Mol Neurodegener. 2017;12(1):1–12. http://dx.doi.org/10.1186/s13024-017-0216-6
- 170. Ising C, Venegas C, Zhang S, Scheiblich H, Schmidt SV, Vieira-Saecker A, et al. NLRP3 inflammasome activation drives tau pathology. Nature. 2019;575(7784):669–73. http://dx.doi.org/10.1038/ s41586-019-1769-z
- 171. Morales I, Jiménez JM, Mancilla M, Maccioni RB. Tau oligomers and fibrils induce activation of microglial cells. J Alzheimer's Dis. 2013;37(4):849–56. http://dx.doi.org/10.3233/JAD-131843
- 172. Lastres-Becker I, Innamorato NG, Jaworski T, Rábano A, Kügler S, Van Leuven F, et al. Fractalkine activates NRF2/NFE2L2 and heme oxygenase 1 to restrain tauopathy-induced microgliosis. Brain. 2014;137(1):78–91. http://dx.doi.org/10.1093/brain/awt323
- 173. Vogels T, Murgoci A-N, Hromádka T. Intersection of pathological tau and microglia at the synapse. Acta Neuropathol Commun. 2019 Nov 5;7(1):109. http://dx.doi.org/10.1186/s40478-019-0754-y
- 174. Jay TR, Miller CM, Cheng PJ, Graham LC, Bemiller S, Broihier ML, et al. TREM2 deficiency eliminates TREM2+ inflammatory macrophages and ameliorates pathology in Alzheimer's disease mouse models. J Exp Med. 2015;212(3):287–95. http://dx.doi.org/10.1084/jem.20142322
- 175. Thibaudeau TA, Anderson RT, Smith DM. A common mechanism of proteasome impairment by neurodegenerative disease-associated oligomers. Nat Commun. 2018;9(1):1097. http://dx.doi. org/10.1038/s41467-018-03509-0
- 176. Upadhya SC, Hegde AN. Role of the ubiquitin proteasome system in Alzheimer's disease. BMC Biochem. 2007 Nov 22;8(Suppl. 1):S12. http://dx.doi.org/10.1186/1471-2091-8-S1-S12
- 177. Oddo S. The ubiquitin-proteasome system in Alzheimer's disease. J Cell Mol Med. 2008;12(2): 363–73. http://dx.doi.org/10.1111/j.1582-4934.2008.00276.x
- 178. Liu J, Li L. Targeting autophagy for the treatment of Alzheimer's disease: Challenges and opportunities. Front Mol Neurosci. 2019;12(August):1–9. http://dx.doi.org/10.3389/fnmol.2019.00203
- 179. Haass C, Kaether C, Thinakaran G, Sisodia S. Trafficking and proteolytic processing of APP. Cold Spring Harb Perspect Med. 2012 May 1;2(5):a006270. http://dx.doi.org/10.1101/cshperspect. a006270
- 180. Yin RH, Yu JT, Tan L. The role of SORL1 in Alzheimer's disease. Mol Neurobiol. 2015;51(3):909–18. http://dx.doi.org/10.1007/s12035-014-8742-5
- 181. Glennon EBC, Whitehouse IJ, Miners JS, Kehoe PG, Love S, Kellett KAB, et al. BIN1 is decreased in sporadic but not familial Alzheimer's disease or in aging. PLoS One. 2013;8(10):1–11. http://dx.doi. org/10.1371/journal.pone.0078806
- 182. Francis PT, Palmer AM, Snape M, Wilcock GK. The cholinergic hypothesis of Alzheimer's disease: A review of progress. J Neurol Neurosurg Psychiatry. 1999;66(2):137–47. http://dx.doi.org/10.1136/ jnnp.66.2.137
- 183. Craig LA, Hong NS, McDonald RJ. Revisiting the cholinergic hypothesis in the development of Alzheimer's disease. Neurosci Biobehav Rev. 2011 May;35(6):1397–409. http://dx.doi.org/10.1016/j. neubiorev.2011.03.001
- 184. Hardy J. A hundred years of Alzheimer's disease research. Neuron. 2006 Oct 5;52(1):3–13. http:// dx.doi.org/10.1016/j.neuron.2006.09.016
Immunotherapy Targeting Amyloid-β Peptides in Alzheimer's Disease

Silvia Zampar • Oliver Wirths

Department of Psychiatry and Psychotherapy, University Medical Center (UMG), Georg-August-University, Goettingen, Germany

Author for correspondence: Oliver Wirths, Department of Psychiatry and Psychotherapy, University Medical Center Goettingen (UMG), Von-Siebold-Str. 5, 37075 Goettingen, Germany, Email: owirths@gwdg.de

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch2

Abstract: Neurodegenerative diseases, in particular Alzheimer's disease, represent significant unmet medical needs due to a lack of effective therapeutic treatment options and cause a substantial burden for health care systems. Accumulation of β -amyloid peptides within the brain is believed to be an initial trigger of the disease process. In the last 20 years, immunotherapy has emerged as a promising target-directed strategy to develop efficient treatment options with disease-modifying potential. Unfortunately, either active vaccination against β -amyloid or its fragments, as well as passive immunization using monoclonal antibodies, have largely failed to show a clinical benefit in a variety of clinical trials. This chapter addresses progress and developments with regard to active and passive immunization against A β and summarizes the current state of clinical trials.

Keywords: Alzheimer's disease; amyloid; immunization; immunotherapy; monoclonal antibodies

In: *Alzheimer's Disease: Drug Discovery.* Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

Copyright: The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

INTRODUCTION

With an estimated 50 million people affected worldwide, Alzheimer's disease (AD) is the most frequent cause of dementia, accounting for about 60–80% of all cases (1). The incidence is expected to increase in the next decades, due to the rapid increase of age in the population of the developing nations, possibly reaching 152 million cases by 2050 (1). Despite numerous and continuous efforts to find an effective cure, no drug has been approved for AD in the last 17 years (2). Additionally, the currently available therapies, comprising cholinesterase inhibitors and N-methyl-D-aspartate receptor agonist, do not modify the underlying pathophysiology of the disease and offer only modest, symptomatic and transient effects (3, 4). The amyloid cascade hypothesis is still widely considered the main theory for the pathology of AD (5), supported by the discovery of genetic autosomal dominant mutations in the amyloid precursor protein (APP), presenilin 1 (PSEN1) or presenilin 2 (PSEN2) genes in patients with early onset AD (EOAD), resulting in an enhanced formation and accumulation of amyloid- β (A β) peptides in plaques (6). AB accumulation in the brain, which starts 15-20 years before the manifestation of clinical symptoms, is believed to be the starting point for the progression of AD, driving tau phosphorylation and leading to synaptic and neuronal loss, which ultimately translates to cognitive impairment. The cascade hypothesis has been revised and modified due to, among other reasons, lack of correlation between fibrillary A β aggregates and AD severity (7). The focus shifted to intraneuronal AB accumulations as a site of AB toxicity (8) or oligometric forms of A β , which are considered the toxic and pathogenic driving force in AD (9). The cascade hypothesis is the rationale for the development of passive and active anti-A β immunotherapy strategies, targeting both fibrillary aggregates and soluble forms of A β . Reducing A β burden by employing monoclonal antibodies (mAb) appears a straightforward and appealing strategy to slow or prevent the progression of the disease. Numerous antibodies have been tested so far and are currently under investigation in clinical trials; however, the outcomes of the past two decades have been disappointing. Though some antibodies, such as bapineuzumab and aducanumab, appeared to clear parenchymal amyloid (10, 11), failure to meet the primary endpoints or the occurrence of adverse side effects such as vasogenic edema and/or microbleeding (12) caused the termination of the ongoing trials for most of the tested mAb. The reasons for the disappointing outcomes could also be imputable to factors independent from the actual mode of action of the tested mAbs. An inaccurate selection of trial patients, leading to huge variations in cognitive and clinical decline during the trial period, as well as a late intervention and insensitive efficacy measures are potentially confounding factors. Proper target engagement (e.g. soluble, monomeric, dimeric, oligomeric, fibrillary A β) is also a critical aspect that needs to be addressed.

Aβ GENERATION AND AMYLOID CASCADE HYPOTHESIS

The vast majority of AD cases are of sporadic origin, occurring beyond 65 years of age with an unknown cause. While mutations in *APP* or the *PSEN* genes have

been linked to early-onset autosomal dominant forms of familial AD (FAD) with an early disease onset (13, 14), so far, only genetic risk loci have been identified as potentially involved in APP processing or β-amyloid peptide generation in sporadic cases (15). APP is a single-pass transmembrane protein, and A β peptides are generated via a series of consecutive proteolytical cleavage steps from this larger precursor protein (16). The generation of $A\beta$ peptides from its precursor APP is linked to the so-called amyloidogenic processing pathway, which is initiated by β -secretase cleavage. This cleavage is predominantly carried out by an aspartic protease named β -site APP cleaving enzyme (BACE1) (17), resulting in the release of a soluble APP fragment (sAPP- β) and a slightly longer APP C-terminal fragment of 99 amino acids (CTF- β). Further cleavage by γ -secretase, a protein complex consisting of PSEN1/2 among others (18), releases AB peptides. This complex is able to cut APP at slightly different positions, mainly resulting in the production of ~90% of A β_{1-40} and less than 10% of A β_{1-42} under basal conditions (19), but also shorter as well as slightly elongated A β peptides (A β_{37} – A β_{43}) (20). Processing by BACE1 and γ -secretase generates full-length A β peptides starting with an aspartic acid residue at position 1 (mainly $A\beta_{1-40}$ and $A\beta_{1-42}$). While most research efforts have concentrated on the full-length peptide species $A\beta_{1-40}$ and $A\beta_{1-42}$, there is accumulating evidence that a variety of other N- and C-terminally modified A β peptides may play an important role in the disease process (21–23).

The accumulation of $A\beta$ peptides is regarded as one of the central processes underlying the neuropathological changes in AD. Almost 30 years ago, the amyloid cascade hypothesis was formulated, theorizing that $A\beta$ accumulation is the initial event triggering further pathological alterations such as tau phosphorylation and neurofibrillary tangle formation, neuron and synapse loss, as well as cognitive impairment (5). While research efforts initially focused on fibrillar $A\beta$ deposits in the form of extracellular plaques, the significance of soluble $A\beta$ species (24, 25), mainly in the form of oligomers, became more and more recognized. They may directly injure synapses and neurites of brain neurons (26, 27), in addition to activating microglia and astrocytes (9). These metastable oligomeric forms likely exist in an equilibrium with amyloid plaques and consist of cross- β -sheet $A\beta$ peptide units of variable size, including protofibrillar intermediates (28, 29) (Figure 1).

MECHANISM AND PRINCIPLES OF (AMYLOID-β) IMMUNOTHERAPY

Immunotherapy focuses on the generation (in case of active) or use (in case of passive) of antibodies targeting a specific antigen, $A\beta$ in this specific context, counteracting the disease by activation of the immune system. In active immunizations, a vaccine containing the $A\beta$ -antigen is administered usually intramuscularly. Depending on the type of antigen used, a humoral response with B-cell and helper T-cell (T_H) involvement, cytokine secretion and production of polyclonal antibodies, and/or a cell-mediated immunity response with the activation of phagocytes (antigen-specific cytotoxic T-lymphocytes) is induced. T-cell populations can be further divided into cytotoxic T-cells, which kill target cells by



Figure 1. Proposed targets of anti-amyloid- β (A β) drugs used in active and passive immunization approaches (modified from (2)).

inducing apoptosis, macrophage-activating proinflammatory Th1 cells, and Th2 cells that stimulate B-cells into antibody-producing cells (30).

In passive immunization, monoclonal antibodies (mAb) against specific AB forms are administered by intravenous infusions or subcutaneous injection. In both cases, the antibodies are at first peripherally located and are required to pass the blood-brain barrier (BBB), greatly restricting the transport of antibodies, in order to reach the brain parenchyma. The access route for immunoglobulins has not been clearly identified yet, but could comprise passive diffusion, the lymphatic system, and perivascular spaces. The absence of active transport systems for antibodies, the presence of receptors (such as the neonatal Fc receptor) acting as a pump to remove antibodies in the central nervous system (CNS), as well as other not yet understood clearance mechanisms, are reasons why only a small fraction of antibodies (approximately 0.1%) introduced into the peripheral circulation can be detected in the brain or cerebrospinal fluid (CSF) (31). The presence of a large number of antibodies in the periphery could also act as a driving force for the efflux of A β out of the CNS, likely by changing the dynamic equilibrium between Aβ in the blood and the brain. Antibodies might therefore act as a peripheral A β "sink," creating a concentration gradient that attracts monomeric A β out of the CNS via passive diffusion mechanisms (32). In the brain parenchyma there are several mechanisms, that are not mutually exclusive, by which the humoral response could exert its effects (33), and the A β epitope against which the antibody is directed (monomeric, oligomeric, fibrillary A β) may lead to a preferred mechanism over another. The antibodies could directly be responsible for the disassembly of A β deposits in the brain (34) or prevention of reassembly and inhibition of toxicity, as shown by *in-vitro* experiments (35, 36). Direct binding to A β oligomers, thus neutralizing their toxicity, is also a putative mechanism (37). The clearance of A β could also be enhanced by the antibodies through microglial activation, leading to Fc-mediated (32) or Fc-independent phagocytosis (38) (Figure 2). Peripherally, large immunoglobulin IgM, which is able to cross the BBB



Figure 2. Proposed mechanisms of anti-amyloid-\beta (A\beta) antibodies. Antibodies might either directly target A β assemblies, leading to a neutralization of A β toxicity (A), or activate microglia, resulting in Fc-receptor (FcR) mediated phagocytosis (B). Alternatively, antibodies might not enter the brain but create a concentration gradient between the brain and the blood, leading to A β removal via a peripheral sink mechanism (C) (modified from (121)).

to a lesser extent compared to IgG but is likely involved in the already-mentioned peripheral-sink effect, is believed to directly hydrolyze A β (39). The specific advantages and disadvantages of active and passive immunization are described in the next paragraphs.

ACTIVE AMYLOID-β-DIRECTED IMMUNOTHERAPIES

In active immunization, immunity is achieved following exposure to an A^β antigen that causes the generation of antibodies in the recipient. It engages the cellular and humoral immune system, including T and B cells. Typically, an active vaccine is comprised of an antigen (alone or conjugated to a non-self T helper cell epitope) combined with an immune boosting adjuvant to ensure high antibody production. An advantage of active immunization is that with few vaccinations the patient should be able to produce a prolonged antibody response. The variability of the induced response across patients is, on the other hand, a problematic aspect, especially when dealing with elderly individuals. Adverse side effects may occur after active immunization: when a T-cell response is induced, the risk of an abnormal immune response increases. With age, the competency of the immune system reduces and the probability of developing autoimmune responses is enhanced. Additionally, vaccines lead to the formation of polyclonal antibodies, which can recognize multiple and possibly overlapping epitopes on the target protein. Polyclonal antibodies may be problematic in case the goal is the recognition of a specific form of the antigen.

The first effort to explore active immunization as a possible therapy for AD was made in 2001 with a vaccine called AN1792, consisting of synthetic full-length $A\beta_{1-42}$ peptide with QS-21 adjuvant. Despite initial positive findings in an APPoverexpressing mouse model (40), the phase II clinical trial in individuals with mild-to-moderate AD was interrupted, as 6% of the treated patients developed a T-cell-mediated meningoencephalitis (41). Additionally, only 20% of patients produced antibodies above the preset therapeutic cut-off titration level and clinical outcomes were no better than those of the placebo-treated control subjects (42). Despite the cessation of the trial, several follow-up studies were carried out as post-mortem brain samples from trial participants became available. Neuropathological analyses from AN1792 recipients in general showed a lower mean A β load compared to an age-matched unimmunized control group. The degree of plaque removal varied among immunized patients along with mean antibody response, and no evidence of improved survival or delay in the development of severe dementia was observed (43). It was further reported that immunized patients showed several-fold increases in $A\beta_{42}$ -containing blood vessels in the cerebral cortex and leptomeninges, as well as a higher density of micro-hemorrhages. However, no major cerebral amyloid angiopathy (CAA)-related intracerebral hemorrhages were noted and, interestingly, two of the longest survivors showed a virtually complete absence of both plaques and CAA (44). Further studies revealed that active immunotherapy with AN1792 was associated with wall splitting in leptomeningeal vessels (45) and an accelerated loss of damaged degenerating neurons, an observation consistent with imaging data indicating an increased rate of cerebral atrophy among immunized AD individuals (46). A recent study reporting on post-mortem data from two AD patients who died 14 years after immunization revealed that these patients remained virtually plaque-free, however, an extensive overall distribution of neurofibrillary tangles (Braak stage V/VI) was observed (47).

In order to control the immune response by eliciting a strong antibody production but avoiding inflammatory T-cell activation, second-generation vaccines were designed to target more specific epitopes (48). One of these secondgeneration A β vaccines, ACC-001 (vanutide cridificar), studied by Janssen Immunotherapy and Pfizer, was discontinued in phase II clinical trials, as the primary efficacy-biomarker endpoints were found not statistically significant in the considered dosage groups (49, 50). The vaccine was composed of $A\beta_{1-7}$ with QS-21 adjuvant, designed to avoid the autoimmune meningoencephalitis caused by Th1 lymphocyte activation seen with AN1792, attributed to A β residues 15–42. CAD106 (Amilomotide) was another second-generation A β vaccine that reached phase II clinical trials involving patients with mild AD. CAD106 is composed of multiple copies of $A\beta_{1-6}$ peptide, coupled to a Q β virus-like particle. Phase II trials in the United States and Europe ended in 2010 and 2011, supporting the favorable safety profile found in phase I trials and reporting prolonged antibody titers in responders (51). In a phase IIb trial, 120 patients suffering from mild AD received up to 7 intramuscular injections of CAD106 or placebo over 60 weeks. The vaccine was generally well tolerated and elicited an Aβ-specific immune response with an acceptable safety profile and preliminary evidence of target engagement by amyloid positron emission tomography (PET) (52). Despite a phase II/III trial began in November 2015, set to run until 2023, in September 2019 Novartis noted in its quarterly financial report that it had "retired" the CAD106 program. Several other candidates have been investigated and reached different stages of clinical development (Table 1).

ABvac40

ABvac40 is an investigational vaccine targeting the C-terminus of $A\beta_{40}$. The agent comprises multiple repeats of a short C-terminal fragment of the $A\beta$ peptide ($A\beta_{33-40}$), conjugated to the keyhole limpet cyanine (KHL) carrier protein and formulated with the adjuvant alum hydroxide. The phase I clinical trials demonstrated a favorable safety and tolerability profile with no incidence of vasogenic edema nor microhemorrhage (53). A phase II clinical trial by Araclon Biotech S.L. is ongoing in several European countries to confirm the results and explore the clinical efficacy of ABvac40 in patient with amnestic MCI and very mild AD (Clinical Trial: NCT03461276) and is due to be completed in February 2022.

ACI-24

ACI-24 is a liposome vaccine that is designed to elicit an antibody response against aggregated A β peptides. ACI-24 is based on the truncated A β_{1-15} sequence, thus avoiding the T-cell epitopes. At each end of the peptide, a palmitoylated lysine residue was attached, enabling anchoring the peptide in the lipid bilayer of a liposome adjuvant thus adopting an aggregated β -sheet structure and forming a conformational epitope. After promising preclinical results (54), a phase I/II trial to assess safety, tolerability, immunogenicity as well as efficacy of the vaccine in patients with mild-to-moderate AD began in 2009 in Denmark, Finland and

TADICA		ine office leave	ممنا لمتعمدتك مالندا	and the second second			
		ncıpaı acuve amy	ioia-p-airectea im	imunotnerapy va	ccines		
Drug	Main ref.	Peptide	Adjuvant	Subjects	Phase	Clinical trial #	Outcome
ABvac40	53	Aβ ₃₃₋₄₀ (multiple copies) conjugated to KHL	Alum hydroxide	a-MCI or very mild AD	Phase II	NCT03461276	Ongoing trial
ACC-001	49	$A\beta_{1-7}$	QS-21	Mild-to-moderate AD	Phase II	NCT00479557 NCT00498602	Failed
ACI-24	54	$A\beta_{1-15}$ with palmitoylated lysine residues	Liposome adjuvant	AD in Down Syndrome	Phase II	NCT04373616	Trial scheduled late 2020
Affitope AD02	55, 56	$A\beta_{1-6}$ conjugated to KHL	Aluminum	Early AD	Phase II	NCT01117818	Failed
AN1792	42	$A\beta_{1-42}$	QS-21	Mild-to-moderate AD	Phase II	NCT00021723	Failed
CAD106	51, 52	$A\beta_{1-6}$ (multiple copies) conjugated to $Q\beta$	I	Mild AD	Phase II	NCT02565511	Failed
UB 311	57	two Aβ ₁₋₁₄ targeting peptides	Alum-containing Th2- biased delivery system	Mild AD	Phase II	NCT02551809	Completed
The list is morented	يمتيم المامات من			н С			

for

The list is presented in alphabetical order. The table shows the status of studies on 31 July 2020, as reported in ClinicalTrials.gov. $A\beta$, amyloid- β ; AD, Alzheimer disease; a-WCI, amnesic mild cognitive impairment.

Sweden. In 2016, ACI-24 became the first anti-A β vaccine to be evaluated for the treatment of Alzheimer's disease in Down's syndrome, and in late 2020 a doubleblind, randomized, placebo-controlled phase II trial to assess the safety, tolerability and target engagement in adults with Down syndrome is scheduled to start (Clinical Trial: NCT04373616).

Affitope AD02

Affitope AD02 consists of a synthetic peptide of six amino acids mimicking the N-terminus of A β , lacking the most common T-cell epitopes, but including the B cell epitope. This peptide induced an anti-A β antibody response when conjugated to Keyhole Limpet Hemocyanin and adjuvanted with aluminum (55). In 2009, a phase I study was conducted in Austria by AFFiRiS AG and showed a favorable safety and tolerability profile 1 year after treatment. A phase II trial of AD02 was conducted in Europe between 2010 and 2013 in patients with early AD, but no significant treatment effects were seen with AD02. Surprisingly, the placebo group receiving a dose of the immunomodulator aluminum oxihydroxide which was part of the formulation, then called AD04, showed a significantly reduced cognitive decline correlating with a reduced hippocampal shrinkage (56). The company declared to be interested in further investigating the potential therapeutic effects of AD04; however no further data have been disclosed yet and no further activities with regard to AD are listed on the company website.

UB 311

UB 311 is a synthetic peptide vaccine developed by United Neuroscience, coupling a helper T-cell epitope to the $A\beta_{1-14}$ sequence. The approach aims to stimulate a T helper type 2 regulatory immune response over a T helper type 1 proinflammatory response (57). In a transgenic AD mouse model (hAPP751), UB-311 reduced levels of A $\beta_{1,42}$ oligomers and protofibrils, as well as extracellular amyloid plaque load (57). In a first-in-human clinical trial in patients with mild-to-moderate AD, each participant received three immunizations (300 µg/dose) by intramuscular injection. The vaccine was well tolerated and showed encouraging improvement in ADAS-Cog scores in the subgroup of mild AD patients (57). As a result, a phase-II clinical trial started in Taiwan in October 2015 enrolling people with a clinical diagnosis of mild AD, which was followed by a safety extension in 2018. A press release from United Neuroscience at the beginning of 2019 reported a favorable safety profile and promising, yet not statistically significant, changes in the secondary endpoints (amyloid PET burden, CDR-SB, ADCS-ADL, ADAS-Cog and MMSE [Mini-Mental State Examination]) (58). United Biomedical, as of July 2020, lists UB 311 as investigational vaccine but no current clinical trials are registered.

Passive Immunotherapy with Monoclonal Antibodies

In passive immunization, externally produced antibodies are administered through intravenous infusions or subcutaneous injections. They can be humanized versions of murine mAb evaluated in previous preclinical trials (such as Bapineuzumab) or fully human mAbs (like Gantenerumab). In the first group, murine mAbs are modified so that a large part of their protein sequences is similar to naturally produced human antibody variants, in order to reduce the immunogenicity that (foreign) murine antibodies would cause. Fully human mAbs are produced for example with transgenic mice that have been genetically engineered with the human immunoglobulin locus, while in contrast, humanized mAbs are initially generated in wild type mice with the endogenous murine immunoglobulin locus (59). Avoiding some of the side effects that the humanized murine mAb still possess, fully human mAb are considered safer and more effective (60).

The passive immunization strategy allows for a precise titration of the administered antibodies and a possible rapid clearance in case adverse effects develop, but has the disadvantage that repeated infusions/injections over time are required to maintain a constant amount of therapeutic antibodies. Passive immunization might allow for targeting specific conformations of the AB peptide, presumably leading to the specific removal of distinct A β assemblies such as monomers, oligomers, or fibrils (61). The employment of mAbs against A β has been associated with the risk of developing amyloid-related imaging abnormalities (ARIAs) as severe adverse effects. These abnormalities seen in neuroimaging of AD patients comprise "vasogenic edema" and/or sulcal effusion (ARIA-E) and hemosiderin deposits (ARIA-H) including microhemorrhage and cortical superficial siderosis (62), and are believed to be the consequence of the removal of vascular amyloid leading to increased vascular permeability. The development of ARIAs after mAb treatment appears to be compound-dependent and dose-related. ARIAs represent a core safety issue in immunotherapy trials and challenged the progress of mAbs as a treatment for AD. Magnetic resonance imaging (MRI) imaging is used to detect active stages of ARIAs in clinical trials, but is not appropriate for predicting the risk of developing ARIAs during treatment (63). Currently, efforts to discover and use specific biomarkers for ARIAs in clinical trials are being made to better manage these severe side effects and reduce the delay caused by this side effect often seen in clinical trials (64).

Amyloid clearance in immunotherapy is largely correlated with IgG Fcy Receptor (FcyR)-mediated activation of microglia and antibody-mediated phagocytosis, however, these same effects are probably responsible for an increased inflammatory response and vascular side effects (ARIAs) observed in a variety of studies (65). FcyRs are activated by human IgG1 and mouse IgG2a with higher affinity compared to other IgG subclasses. Using a different class of immunoglobulin G (e.g. IgG4) could help prevent an excessive microglial activation, reducing the risk of vascular damage (66). Modification of the effector function, such as de-glycosylation, by antibody engineering, was also used as a strategy to reduce the incidence of adverse ARIAs (67). Even though the role of the antibody effector function in the development of vascular side effects is clear, the engaged epitope is also crucial. A comparative study of murine versions of therapeutic A β antibody candidates with a constant IgG2a region showed strong differences in their plaque-removing potential, demonstrating that the ability of an antibody to remove plaques and activate inflammation is critically dependent on its epitope and affinity (68). A variety of antibodies have been evaluated in passive immunotherapy approaches and reached different stages of clinical trials (Table 2).

TABLE 2	Prii	ncipal passive amy	/loid-β-directec	l immunotherap	y drugs		
Drug	Main ref.	Target	Antibody - subtype	Subjects	Phase	Clinical trial #	Outcome
Aducanumab	122	Αβ ₃₋₇	Humanized mAb - IgG1	Early AD	Phase III (ENGANGE and EMERGE)	NCT02477800 NCT02484547	Terminated
BAN2401	115	Soluble Aß protofibrils	Humanized mAb - IgG1	Early AD Preclinical AD	Phase III Phase III	NCT03887455 NCT04468659	Ongoing Trial Ongoing Trial
Bapineuzumab	12	$A\beta_{1-5}$ (fibrillary and soluble $A\beta)$	Fully human mAb - IgG1	Mild-to-moderate AD	Phase III	NCT00676143 NCT00667810 NCT00575055 NCT00574132	Failed
Crenezumab	81 82	Aβ ₁₃₋₂₄ (pentameric Aβ oligomers and fibrils)	Humanized mAb - IgG4	Prodromal-to-mild AD Preclinical AD with PSEN1 E280A mutation	Phase III (CREAD 1 and 2) Phase II	NCT02670083 NCT03114657 NCT01998841 NCT03977584	Failed Ongoing Trials
Gantenerumab	85, 86	Aβ fibrils	Humanized mAb - IgG1	Early AD	Phase III (GRADUATE 1 and 2)	NCT03443973 NCT03444870	Ongoing Trials
MEDI1814	129	Aβ ₄₂ C-terminus (Aβ monomers and low-n oligomers)	Fully human mAb - IgG1À	Mild-to-moderate AD	Phase I	NCT02036645	Completed
Ponezumab	95	Aβ ₄₀ C-terminus (Aβ monomers)	Humanized mAb - IgG2∆a	Mild-to-moderate AD	Phase II	NCT00722046 NCT00945672	Failed
SAR228810	130	Protofibrillary and fibrillary Aβ	Humanized mAb - IgG4	AD	Phase I	NCT01485302	Completed
Solanezumab	105, 106	Aβ ₁₆₋₂₆ (Aβ monomers)	Humanized mAb - IgG1	Mild-to-moderate AD Individuals with EOAD-associated mutations	Phase III (EXPEDITION 1,2 and 3) Phase II/III	NCT00905372 NCT00904683 NCT01900665 NCT01760005	Failed Ongoing Trial
The list is presented	in alphabetica	il order. The table shows the statu	s of studies on 31 July 2020), as reported in ClinicalTrials.	. gov.		

Bapineuzumab

Bapineuzumab is a humanized form of the murine monoclonal antibody 3D6, directed specifically towards the N-terminus of the A β sequence starting at Asp1 (69, 70). This antibody of the IgG1 subclass binds fibrillary and soluble $A\beta$ and activates microglial phagocytosis as well as cytokine production, aiming to reduce plaque formation and promote A β clearance (71). Preclinical studies and phase I-II clinical trials gave initial promising results. When 3D6 mAb was administered to 4-month-old PDAPP mice with i.p. injections of 10 mg/kg/week for 12 months, total A β deposition was reported to be almost completely reduced (72). Although the translatability of these preclinical studies was later questioned (73), bapineuzumab was tested in a phase I clinical trial where a single ascending dose was administered to patients with mild-to-moderate Alzheimer's disease in order to determine the safety, tolerability, and pharmacokinetics of the mAb (74). MRI abnormalities, consistent with vasogenic edema, were observed in 3 out of 10 patients receiving the higher dose of $\overline{5}$ mg/kg, but this resolved with time. MMSE scores improved at the lower doses (0.5 and 1.5 mg/kg) of bapineuzumab compared to the placebo, a finding not observed with the highest dose.

In a phase II clinical trial, patients with mild-to-moderate AD were randomly assigned to one of four dose cohorts (0.15, 0.5, 1.0, or 2.0 mg/kg) and received six infusions 13 weeks apart. The final assessments were performed at week 78 but no significant differences were found in co-primary efficacy endpoints, the ADAS-cog and Disability Assessment for Dementia (DAD). Exploratory analyses showed potential treatment differences on cognitive and functional endpoints. Differences based on APOE ϵ 4 carrier status were also observed. ARIA-E was found in 12/124 treated patients, with a dose and APOE ϵ 4 carrier-dependent incidence increase (71). Additional phase II studies reported a reduction in exploratory CSF biomarkers T-Tau and p-Tau, the latter being significantly different between treated and placebo groups (75). A reduced cortical ¹¹C-Pittsburgh compound B (PiB) average uptake, visualized by PET, was also found after 78 weeks of treatment with bapineuzumab (76). The feasible and tolerable administration of bapineuzumab, together with evidence that the mAb could be disease modifying, led to the actualization of phase III clinical trials.

A four-trial phase III program was launched in North America and Europe. The first two double-blind, randomized, placebo-controlled 18-month phase III trials tested bapineuzumab in patients with mild-to-moderate Alzheimer's disease, divided into APOE ϵ 4 carriers and non-carriers (71). Bapineuzumab was administered by intravenous infusion every 13 weeks for 78 weeks at a dose of 0.5 mg/kg in APOE ϵ 4 carriers and at 0.5 mg/kg, 1 mg/kg, and 2 mg/kg doses in non-carriers, even though the highest dose was soon discontinued due to ARIA-E and ARIA-H development. No significant differences were found in the primary outcome measures (ADAS-cog11 and DAD) between groups. The APOE ϵ 4 carriers group showed a modest reduction in PiB PET binding as well as a significant reduction of CSF p-Tau when compared to the placebo group. Consistent with the phase II data, a dose-related and APOE ϵ 4 carriers-dependent increase in ARIA-E was observed. The failure to meet the primary endpoints led to the discontinuation of two additional phase III clinical trials and the further evaluation of bapineuzumab as treatment for AD.

Crenezumab

Crenezumab is a humanized mAb designed on an IgG4 backbone targeting multiple species of A β . Its epitope is located in the central part (~ A β_{13-24}) of the peptide and it shows particular affinity for pentameric oligomeric and fibrillary l6mer assemblies of aggregated A β (77, 78). A recent study confirmed that it detects a variety of full-length and N-terminal truncated A β variants in postmortem human AD brain samples (70). Limited preclinical data are currently published on the efficacy of chronic treatments with crenezumab. The murine version of the antibody (mC2) was tested in 18-month Tg2576 transgenic mice with a single intracerebral injection of 2 µg of antibody, which did not cause significant inflammatory changes (68). *In vivo* imaging of 10-month-old transgenic hAPP^(V7171)/PS1 mice showed decreased plaque volumes over a period of 3 weeks after an intraperitoneal injection of 60 mg/kg antibody (77). The same study reported the results of a phase I clinical trial, performed in patients with mild-to-moderate AD. No ARIAs were observed either with a single or multiple ascending dosage.

Crenezumab was further tested in phase II clinical trials in patients with mild-to-moderate AD. A total of 431 patients were enrolled in the ABBY study, receiving either a low subcutaneous dose (300 mg) or placebo every 2 weeks, or an intravenous high dose (15 mg/kg) or placebo every 4 weeks, for a total period of 68 weeks (79). The primary endpoints (changes in ADAS-Cog12 and CDR-SB scores), measured at week 73, were not met. Exploratory analyses pointed towards a reduction in decline on the ADAS-Cog12 in the high-dose group, and the patients with mild AD showed the greatest deviation from the placebo group. This difference became significant in the group with MMSE scores ranging from 22 to 26. These trends were also observed in a smaller phase II brain imaging study (BLAZE), enrolling 91 patients. Even though no significant trends toward ADAS-Cog12 and CDR-SB score improvements were observed in the mild AD group receiving the higher dose of antibody (80). Throughout these studies, no ARIAs adverse effects were reported.

Two large phase III clinical trials, CREAD1 and CREAD2, started in 2016 and 2017 respectively, and enrolled patients with prodromal-to-mild AD. These double-blind, placebo-controlled global studies recruited overall more than 1500 patients, testing a 60 mg/kg dose by intravenous infusion every 4 weeks for a period of 100 weeks with the primary endpoint being changes in the CDR-SB score at 2 years (81). In January 2019 the company Roche announced the decision to discontinue both trials, based on preliminary analyses suggesting that the primary endpoint would unlikely be met. Crenezumab is, to date, being tested as a preventive treatment as part of the Alzheimer Prevention Initiative (API) in a randomized, double-blind, placebocontrolled phase II study by Genentech, estimated to end in 2022 (Clinical Trial: NCT01998841). The 5-year trial started in 2013 and recruited patients who carry the PSEN1 E280A autosomal-dominant mutation and are still in a preclinical phase of AD (82). In a subgroup of participants (carriers and noncarriers) the longitudinal tau burden will be evaluated with a tau positron emission tomography (PET) scan after IV injection of the probe [18F]GTP1 (Clinical Trial: NCT03977584).

Gantenerumab

Gantenerumab is a recombinant human IgG1 antibody, designed to recognize a conformational epitope present on A β fibrils, in order to disassemble and degrade aggregated A β peptides via recruiting microglia and activating phagocytosis (83). Using peptide mapping, N-terminal as well as central portions of A β were recognized and no evidence of altered plasma A β was detected. In a preclinical study, gantenerumab bound cerebral A β and significantly reduced small amyloid- β plaques in APP/PS2 transgenic mice with chronic treatment (83). An initial randomized study of AD patients receiving either 60 mg or 200 mg intravenous gantenerumab or placebo, showed a ~16% or ~36% reduction in Pittsburgh Compound B retention in the 60 mg and 200 mg gantenerumab group respectively. However, two patients in the 200 mg group showed vasogenic edema and focal areas of inflammation on MRI scans at sites with the highest level of amyloid removal (84).

The Scarlet RoAD trial assessed the efficacy and safety of gantenerumab in prodromal AD patients. Participants enrolled in this 2-year randomized doubleblind phase III study received 105 mg, 225 mg or placebo every 4 weeks subcutaneously. A dose- and APOE ϵ 4 genotype-dependent increase of generally asymptomatic ARIAs was noticed and the study was terminated for futility when no differences in primary or secondary endpoints were observed (85). Of note, significant reductions in total and phosphorylated tau in the CSF, as well as a dose-dependent reduction in brain amyloid on PET scans were observed in an exploratory biomarker analysis (85). A 2-year PET sub-study evaluating the effect of up to 1200 mg of gantenerumab every 4 weeks in patients with prodromal-tomoderate AD, revealed a 3.5-times greater reduction in amyloid-PET signal than seen after 2 years at a dose of 225 mg, with 51% of patients having amyloid- β plaque levels below the positivity threshold (86).

A phase I randomized, open-label study including healthy volunteers aged 40–80 years, evaluated different subcutaneous injection regimens of gantenerumab, with regard to pharmacokinetic properties and tolerability. The results of this study suggest that subcutaneous injections at speeds of 5 and 15 s were well-tolerated and might enable at-home administrations by AD patients or their caregivers (87). Gantenerumab is currently under investigation in two large phase III trials (GRADUATE 1 and 2), which started enrolling patients with early AD in 2018 with the goal of more than 1500 patients in up to 350 study centers with a data read-out expected in 2022 (Clinical Trial: NCT03443973 and NCT03444870).

Ponezumab

Ponezumab is a humanized monoclonal IgG2 Δa anti-A β antibody reported to bind to the C-terminus of the most abundant A β_{1-40} peptide. It contains two mutations that eliminate effector function and therefore potential cell toxicity depending on the antibody. Structural analyses revealed extensive contacts of ponezumab with the carboxyl moiety of A β_{40} (88). Preclinical analyses using the murine antibody 2H6, similarly binding to the C-terminal of A β_{1-40} , demonstrated a robust reduction of amyloid deposits in aged Tg2576 (89). Intraperitoneal injections of ponezumab increased plasma $A\beta_{1-x}$ and $A\beta_{x-40}$ levels in PS1xAPP mice in a concentration-dependent manner, while $A\beta_{x-42}$ plasma concentrations remained unchanged. This led to the suggestion that ponezumab removes brain $A\beta$ via a peripheral sink mechanism (88). Another preclinical study in cynomolgus monkeys, sharing the same $A\beta$ peptide sequence with humans, confirmed increased plasma $A\beta_{1-40}$ and $A\beta_{1-x}$ levels in treated animals versus controls (90).

An initial randomized, double-blind, single-dose-escalation study evaluated safety, pharmacokinetics and pharmacodynamics using doses of 0.1 mg/kg up to 10 mg/kg. The 2-h infusion was well-tolerated, and in individuals receiving the highest dose increases in CSF A β were observed, which is suggestive of altered central AB levels (91). A related study in a cohort of Japanese subjects yielded comparable results (92). A different administration protocol of a single 10-min intravenous infusion was evaluated and produced comparable effects on plasma Aß species (93). Individuals aged 50 and older with a diagnosis of mild-tomoderate AD and a MMSE score of 16 to 26 were enrolled in a placebo-controlled, multiple dose study (0.1 mg/kg up to 8.5 mg/kg) of ponezumab. The treatment was administered as 10 2-h infusion every 2 months, and was generally well tolerated with an acceptable safety profile and robust plasma A β increases but no evidence of a dose response with regard to CSF biomarkers (94). Effects on peripheral and central AB were characterized in small Swedish cohorts suffering from mildto-moderate AD. One cohort received ponezumab (10 mg/kg) or placebo quarterly over 1 year, whereas a second cohort started with an initial dose of 10 mg/kg or placebo, followed by monthly infusions of 7.5 mg/kg or placebo respectively. This phase II study again showed that ponezumab was generally safe and well tolerated, with dose-dependent increases in plasma A β . However, no apparent differences in brain amyloid burden assessed by PiB-PET were detected and changes in both cognitive and functional decline were observed during the course of the study without, however, differences between treatment arms (95). The potential effect of intravenous ponezumab was also investigated in patients with probable cerebral amyloid angiopathy (CAA) (96), a disease condition with amyloid deposition in the walls of leptomeningeal and intracortical blood vessels of the CNS (97, 98). In this study, again, ponezumab was safe and well tolerated; however, this antibody has been discontinued as prespecified efficacy criteria were not met in the majority of the trials.

Solanezumab

Solanezumab is a humanized monoclonal IgG1 antibody (mouse version m266), targeting the mid-region of A β . Co-crystallization studies revealed that solanezumab accommodates a large A β epitope (residues 16–26), forming extensive contacts and hydrogen bonds with the antibody (99). As administration of solanezumab as well as its murine precursor m266 cause substantial dose-dependent increases in plasma antibody-bound A β levels (100–102), it has been suggested that this antibody primarily targets soluble monomeric forms of A β . On the contrary, neuropathological studies employing human brain samples indicated that a recombinant biosimilar antibody of solanezumab showed a strong binding affinity to amyloid plaques (103), calling its assumed selectivity for monomeric A β into question. In transgenic PDAPP mice, administration of m266 resulted in a rapid reversal of memory deficits in the absence of amyloid plaque reductions (102); however, a more recent study in the J20 mouse model of AD reported no improvement of behavioral deficits and even a strongly increased mortality rate following m266 immunization (101).

Solanezumab has been investigated in several clinical trials in order to evaluate its disease-modifying potential. Following a phase II trial with 52 patients suffering from mild-to-moderate AD evaluating diverse dose regimens (104), two large phase III studies (EXPEDITION-1, EXPEDITION-2) were launched. These studies recruited 2,052 mild-to-moderate AD patients, who received monthly 400 mg infusions. However, both showed a lack of efficacy with regard to cognitive performance, the primary outcome measure of both studies (105). Pooled analyses of both studies suggested less functional and cognitive decline in the mild AD population; however, no significant differences in baseline-to-endpoint changes were found for a variety of secondary outcome measures such as activities of daily living (106). Following the review of the data obtained from the pooled mild AD population, a third phase III trial (EXPEDITION-3) was initiated. This trial enrolled 2129 patients with mild dementia and evidence of amyloid deposition, shown by either florbetapir PET or $A\beta_{1-42}$ measurements in CSF, and patients received 400 mg solanezumab or placebo every 4 weeks for 76 weeks. As a result, the secondary analyses of the previous EXPEDITION trials were not reproduced and solanezumab showed no benefit with regard to cognitive decline in patients with mild AD (107).

Solanezumab is being tested within the Dominantly Inherited Alzheimer Network (DIAN) trial in a phase II/III study as a potential disease-modifying treatment, together with gantenerumab, in individuals at risk for or with a mutation associated with EOAD. The trials are estimated to be completed by March 2021 (Clinical trial: NCT017660005).

BAN2401

BAN2401 is the humanized version (IgG1) of the mouse monoclonal antibody mAb158, which has been shown to primarily bind to large soluble Aβ protofibrils (108). Selectivity for this type of aggregate has been described to be at least 1000-fold higher than for monomers and 10–15 times better than for A β fibrils (109, 110). Administration of mAb158 to plaque-bearing AD transgenic mice carrying both the Arctic and Swedish APP mutations (tg-ArcSwe) resulted in lowered Aβ protofibrils, albeit unchanged insoluble Aβ levels. When treatment was started prior to extracellular plaque onset, a prevention of amyloid deposition and a reduction in protofibril levels was observed (111). Interestingly, individual performance of young tg-ArcSwe mice in a spatial memory test (Morris water maze) was inversely correlated with protofibril but not total A β levels (112). This antibody, as well as its humanized version BAN2401, efficiently precipitated soluble A β aggregates from the human brain, and more than 50% reduction of protofibrils/oligomers was observed after long-term mAb158 treatment in the CSF of tg-ArcSwe mice (113). A radiolabeled version of mAb158 conjugated to a transferrin receptor antibody has been shown to effectively visualize A β in the brain of two AD mouse models, enabled via receptor-mediated transcytosis across the BBB (114).

Safety and tolerability of BAN2401 were investigated in an ascending dose study (0.1 mg/kg up to 10 mg/kg biweekly) for 4 months in mild-to-moderate AD cases. The treatment was well-tolerated across all doses and a slight elevation of plasma $A\beta_{1-40}$ was noted, albeit in the absence of measurable effects on CSF biomarkers (115). A subsequent placebo-controlled, double-blind, randomized phase IIb study enrolling 856 patients with mild cognitive impairment (MCI) caused by AD or mild AD-dementia evaluated several doses in a Bayesian adaptive design. A statistically significant reduction in amyloid PET standard uptake value ratio (SUVR) was observed after 18 months at the highest dose, together with a significant clinical benefit measured by ADCOMS at 6 and 12 months. The drug was well-tolerated with an incidence of ARIA-E of not more than 10% in any treatment arm and less than 15% in APOE $\varepsilon4$ carriers at the highest dose (116). Assessment of amyloid PET status in patients in an ongoing open-label extension (OLE) of BAN2401-G000-201 revealed that all amyloid-negative, BAN2401treated individuals entering the OLE were also amyloid negative at OLE baseline. despite subjects being off treatment for 9–52 months (117).

A phase III trial for individuals with preclinical AD and elevated amyloid (AHEAD 3–45 study) is currently underway and participants are being recruited and is expected to be completed in October 2027 (Clinical Trial: NCT04468659).

Aducanumab

Aducanumab (BIIB037) is a recombinant human IgG1 antibody that has been isolated from blood lymphocytes of a healthy donor population of elderly subjects with unusually slow cognitive decline and lack of symptoms of cognitive impairment. Preclinical studies in the Tg2576 mouse model employing chronic dosing of a murine IgG2a/k chimeric aducanumab analogue showed significant reductions of AB in both soluble and insoluble protein extracts, as well as significantly reduced A β deposits in both the cortex and hippocampus; however, no data on behavioral performance was provided (11). Structural and biochemical analyses revealed that aducanumab binds a linear A β epitope comprised of amino acids 3–7 in an extended conformation, discriminating between monomers and higher molecular weight peptide assemblies, based on a strong avidity for epitope-rich aggregates and very weak monomer affinity (118). The linear sequence recognized by aducanumab substantially overlaps with other A β antibodies (such as bapineuzumab or gantenerumab), while specific interactions such as critical contacts formed with Phe-4 and His-6, are different and the interaction of Aβ and aducanumab is quite shallow (118).

An initial phase I study investigated the safety, tolerability, and pharmacokinetics of a single ascending aducanumab dose (0.3–60 mg/kg) or placebo in mild-tomoderate AD patients. While doses up to 30 mg/kg were generally well-tolerated, all three patients receiving 60 mg/kg developed serious adverse events (SAEs) of symptomatic ARIA, which completely resolved after several weeks (119).

A subsequent phase Ib, 12-month, double-blind placebo-controlled, multiple ascending-dose (1 - 10 mg/kg) study (PRIME) enrolled 165 patients with a clinical diagnosis of prodromal or mild AD (11). Amyloid PET imaging using florbetapir was used as an adjunct tool to identify and select patients for enrollment (120). Of the 165 dosed patients, 40 discontinued treatment, mainly due to adverse events

or withdrawal of consent. Aducanumab reduced brain A β plaques as quantified by florbetapir PET in a dose- and time-dependent manner, with significantly reduced SUVR composite scores in the 3, 6 and 10 mg/kg dose groups after 54 weeks of treatment. A slowing of clinical progression as measured by both the MMSE as well as the CDR-SB was observed in patients receiving the highest dose after 1 year of treatment (11). Although this represents the first study reporting an effect of lowering the brain A β load coupled to beneficial effects on cognitive outcomes, the small sample size, a staggered parallel-group design and potential unblinding due to ARIA-E in the treatment groups receiving higher antibody doses, impede interpretation of the results. In addition, the clinical stage of dropouts might bear a potential interpretation bias. Similar discontinuation rates were reported among prodromal and mild AD patients in the placebo group; however, more mild than prodromal AD patients at baseline dropped out in the 10 mg/kg group, with a potential impact on the observed slower cognitive decline (121).

Two large 18-month, randomized, double-blind, placebo-controlled phase III trials (ENGAGE & EMERGE) evaluated aducanumab in patients with early AD and MCI due to AD with PET-confirmed amyloid pathology (122). The participants were randomized to receive a low dose (3 mg/kg for ApoE ε 4 carriers, 6 mg/kg for non-carriers) or a high-dose of 6 or 10 mg/kg for 78 weeks. The protocol has been amended during the course of the study, allowing ApoE ε 4 carriers to receive up to 10 mg/kg and increasing the sample size of each trial to 1650 to compensate for larger than expected standard deviation. A planned futility analysis indicated little chance of treatment efficacy and the trials were terminated in March 2019 (123). Later in 2019, analyses of a more complete data set from both studies were presented, with 29% of patients in EMERGE and 22% in ENGAGE receiving the full possible 14 doses of 10 mg/kg and final participant numbers of 982 and 1084 respectively (124). In EMERGE, the high dose aducanumab group showed a significant 23% reduction in decline on the CDR-SB and 27% reduction on the AD Assessment Scale-Cognitive Subscale 13 items (ADAS-Cog 13) compared to placebo; however, only a 2% reduction on CDR-SB and a 12% reduction in ADAS-Cog 13 were observed in the high-dose group in the ENGAGE sister trial (124). This was explained by the greater exposure to high-dose aducanumab in the EMERGE trial; however, other possibilities such as greater worsening in the placebo group are conceivable as well (125). On July 8, 2020 Biogen announced that it had completed the submission of a Biologics License Application (BLA) to the U.S. Food and Drug Administration (FDA) for the approval of aducanumab (126). On August 7, 2020, Biogen announced that the agency accepted this BLA granting priority review, which means that the time to review is cut down to 6 months. In case of successful approval, aducanumab will be the first approved biological capable of removing amyloid plaques.

MEDI1814

MEDI1814 is a fully human monoclonal IgG1 λ antibody targeting the C-terminus of A β_{42} , with a triple mutation in the Fc tail to reduce its effector function. It aims to bind and remove monomers and low n-oligomers from circulation, thus preventing further aggregation of the peptide (127). MEDI1814 showed a dose-dependent suppression of up to 90% of free A β_{42} in the CSF of V717I transgenic mice, naive rats and cynomolgus monkeys (128).

AstraZeneca started a clinical trial in the United States in 2014 testing single and multiple ascending dose in subjects with mild-to-moderate AD. Safety, tolerability, pharmacokinetics, and pharmacodynamics were analyzed and none of the participants on the drug showed signs of ARIAs. In addition, pharmacokinetics and pharmacodynamics data provided evidence of dose-dependent and selective A β_{42} target engagement in the CNS (129).

SAR228810

SAR228810 is a humanized version of murine IgG1 SAR255952 antibody with an engineered human IgG4 backbone with two amino-acid substitutions to reduce the Fc effector function-dependent risk of ARIAs. It binds specifically to soluble protofibrillar and fibrillar forms of A β and it is relatively inactive against A β monomers and small oligomeric aggregates (130). Co-application of SAR228810 and oligomeric A β_{42} preparations significantly inhibited A β -induced neurotoxicity in primary neurons (131). Preclinical pharmacological studies of SAR255952 in APPSL mice showed that a chronic 4-month treatment dose-dependently prevented brain amyloid plaque formation. Even with high doses (up to 50 mg/kg/ week intravenously), SAR255952 did not increase brain micro-hemorrhages in old mice. In immunotolerized APPSL mice, in which CD4⁺ T lymphocytes have been transiently depleted, SAR228810 demonstrated the same efficacy as its murine precursor (130, 132).

A multi-center, double-blind, placebo-controlled phase I clinical trial testing escalating single and multiple doses by Sanofi has been completed and no further clinical trials are ongoing at the moment. SAR228810 has been administered by intravenous infusion or subcutaneous injection in patients with mild-to-moderate AD.

CONCLUSION

A multitude of preclinical biochemical, histopathological and animal studies, as well as a large number of genetic, biomarker and clinical reports support the central role of A β in AD pathogenesis. While the amyloid cascade hypothesis, with all its modifications, is still considered relevant, the continuous failures of late stage clinical trials with immunotherapy approaches raise questions about considering the right target. There is increasing evidence that A β peptides might also play important physiological roles, as neurotrophic effects (133) or improved synaptic function after application of picomolar AB concentrations in mice depleted of endogenous A β have been described (134, 135). The observation that $\dot{A}\beta$ is elevated in the CSF after sleep deprivation in healthy adults (136), together with its increased brain levels in a variety of other neurologic disease conditions such as traumatic brain injury (137) or cerebrovascular lesions (138) may indicate, that Aβ production in the case of neuronal stress or damage might represent response rather than origin. While immunotherapy trials targeting $A\beta$ have been regarded as the final proof of the validity of the amyloid cascade hypothesis, the aforementioned studies still paint a nebulous picture and alternative therapeutic strategies and approaches should be vigorously investigated.

Acknowledgements: This work has been supported by the Alzheimer Forschung Initiative (grant #16103 to OW), Gerhard Hunsmann Stiftung and Alzheimer Stiftung Göttingen (to OW). The figures were generated using the Smart Servier Medical Art website (https://smart.servier.com), licensed under a Creative Commons Attribution 3.0 Unported License.

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship, and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- 1. World Alzheimer Report 2019. Attitudes to dementia. London: Alzheimer's Disease International; 2019.
- Panza F, Lozupone M, Logroscino G, Imbimbo BP. A critical appraisal of amyloid-β-targeting therapies for Alzheimer disease. Nat Rev Neurol. 2019;15(2):73–88. http://dx.doi.org/10.1038/ s41582-018-0116-6
- Hansen RA, Gartlehner G, Webb AP, Morgan LC, Moore CG, Jonas DE. Efficacy and safety of donepezil, galantamine, and rivastigmine for the treatment of Alzheimer's disease: A systematic review and meta-analysis. Clin Interv Aging. 2008;3(2):211–25.
- Kaduszkiewicz H, Zimmermann T, Beck-Bornholdt H-P, van den Bussche H. Cholinesterase inhibitors for patients with Alzheimer's disease: Systematic review of randomised clinical trials. BMJ. 2005;331(7512):321–7. http://dx.doi.org/10.1136/bmj.331.7512.321
- 5. Hardy J, Allsop D. Amyloid deposition as the central event in the aetiology of Alzheimer's disease. Trends Pharmacol Sci. 1991;12(10):383–8. http://dx.doi.org/10.1016/0165-6147(91)90609-V
- 6. Hardy J. The amyloid hypothesis for Alzheimer's disease: A critical reappraisal. J Neurochem. 2009;110(4):1129–34. http://dx.doi.org/10.1111/j.1471-4159.2009.06181.x
- Giannakopoulos P, Herrmann FR, Bussiere T, Bouras C, Kovari E, Perl DP, et al. Tangle and neuron numbers, but not amyloid load, predict cognitive status in Alzheimer's disease. Neurology. 2003;60(9):1495–500. http://dx.doi.org/10.1212/01.WNL.0000063311.58879.01
- Wirths O, Multhaup G, Bayer TA. A modified beta-amyloid hypothesis: Intraneuronal accumulation of the beta-amyloid peptide – The first step of a fatal cascade. J Neurochem. 2004;91(3):513–20. http://dx.doi.org/10.1111/j.1471-4159.2004.02737.x
- 9. Selkoe DJ, Hardy J. The amyloid hypothesis of Alzheimer's disease at 25 years. EMBO Mol Med. 2016;8(6):595–608. http://dx.doi.org/10.15252/emmm.201606210
- Liu E, Schmidt ME, Margolin R, Sperling R, Koeppe R, Mason NS, et al. Amyloid-β 11C-PiB-PET imaging results from 2 randomized bapineuzumab phase 3 AD trials. Neurology. 2015;85(8): 692–700. http://dx.doi.org/10.1212/WNL.000000000001877
- Sevigny J, Chiao P, Bussière T, Weinreb PH, Williams L, Maier M, et al. The antibody aducanumab reduces Aβ plaques in Alzheimer's disease. Nature. 2016;537(7618):50–6. http://dx.doi.org/10.1038/ nature19323
- Penninkilampi R, Brothers HM, Eslick GD. Safety and efficacy of anti-amyloid-β immunotherapy in Alzheimer's disease: A systematic review and meta-analysis. J Neuroimmune Pharmacol. 2017;12(1):194–203. http://dx.doi.org/10.1007/s11481-016-9722-5
- 13. Schellenberg G, Montine T. The genetics and neuropathology of Alzheimer's disease. Acta Neuropathol. 2012;124(3):305–23. http://dx.doi.org/10.1007/s00401-012-0996-2

- 14. Tanzi RE. The genetics of Alzheimer disease. Cold Spring Harb Perspect Med. 2012;2(10):a006296. http://dx.doi.org/10.1101/cshperspect.a006296
- Dourlen P, Kilinc D, Malmanche N, Chapuis J, Lambert J-C. The new genetic landscape of Alzheimer's disease: From amyloid cascade to genetically driven synaptic failure hypothesis? Acta Neuropathol. 2019;138(2):221–36. http://dx.doi.org/10.1007/s00401-019-02004-0
- Nhan HS, Chiang K, Koo EH. The multifaceted nature of amyloid precursor protein and its proteolytic fragments: Friends and foes. Acta Neuropathol. 2015;129(1):1–19. http://dx.doi.org/10.1007/ s00401-014-1347-2
- Vassar R, Bennett BD, Babu-Khan S, Kahn S, Mendiaz EA, Denis P, et al. Beta-secretase cleavage of Alzheimer's amyloid precursor protein by the transmembrane aspartic protease BACE. Science. 1999;286(5440):735–41. http://dx.doi.org/10.1126/science.286.5440.735
- Golde TE, Koo EH, Felsenstein KM, Osborne BA, Miele L. Gamma-secretase inhibitors and modulators. Biochim Biophys Acta. 2013;1828(12):2898–907. http://dx.doi.org/10.1016/j.bbamem.2013.06.005
- Dovey HF, Suomensaari-Chrysler S, Lieberburg I, Sinha S, Keim PS. Cells with a familial Alzheimer's disease mutation produce authentic beta-peptide. Neuroreport. 1993;4(8):1039–42. http://dx.doi. org/10.1097/00001756-199308000-00011
- Steiner H, Fukumori A, Tagami S, Okochi M. Making the final cut: Pathogenic amyloid-β peptide generation by γ-secretase. Cell Stress. 2018;2(11):292–310. http://dx.doi.org/10.15698/cst2018.11.162
- Bayer TA, Wirths O. Focusing the amyloid cascade hypothesis on N-truncated Aβ peptides as drug targets against Alzheimer's disease. Acta Neuropathol. 2014;127(6):787–801. http://dx.doi.org/10.1007/ s00401-014-1287-x
- Dunys J, Valverde A, Checler F. Are N- and C-terminally truncated Aβ species key pathological triggers in Alzheimer's disease? J Biol Chem. 2018;293(40):15419–28. http://dx.doi.org/10.1074/jbc. R118.003999
- Wirths O, Zampar S. Emerging roles of N- and C-terminally truncated Aβ species in Alzheimer's disease. Expert Opin Ther Targets. 2019;23(12):991–1004. http://dx.doi.org/10.1080/14728222.2019. 1702972
- Chabrier MA, Blurton-Jones M, Agazaryan AA, Nerhus JL, Martinez-Coria H, Laferla FM. Soluble abeta promotes wild-type tau pathology in vivo. J Neurosci. 2012;32(48):17345–50. http://dx.doi. org/10.1523/JNEUROSCI.0172-12.2012
- McLean CA, Cherny RA, Fraser FW, Fuller SJ, Smith MJ, Beyreuther K, et al. Soluble pool of Abeta amyloid as a determinant of severity of neurodegeneration in Alzheimer's disease. Ann Neurol. 1999 Dec;46(6): 860–6. http://dx.doi.org/10.1002/1531-8249(199912)46:6%3C860::AID-ANA8%3E3.0.CO;2-M
- Shankar GM, Bloodgood BL, Townsend M, Walsh DM, Selkoe DJ, Sabatini BL. Natural oligomers of the Alzheimer amyloid-beta protein induce reversible synapse loss by modulating an NMDA-type glutamate receptor-dependent signaling pathway. J Neurosci. 2007;27(11):2866–75. http://dx.doi. org/10.1523/JNEUROSCI.4970-06.2007
- 27. Selkoe DJ. Soluble oligomers of the amyloid beta-protein impair synaptic plasticity and behavior. Behav Brain Res. 2008;192(1):106–13. http://dx.doi.org/10.1016/j.bbr.2008.02.016
- Benilova I, Karran E, De Strooper B. The toxic Abeta oligomer and Alzheimer's disease: An emperor in need of clothes. Nat Neurosci. 2012;15(3):349–57. http://dx.doi.org/10.1038/nn.3028
- Kayed R, Pensalfini A, Margol L, Sokolov Y, Sarsoza F, Head E, et al. Annular protofibrils are a structurally and functionally distinct type of amyloid oligomer. J Biol Chem. 2009;284(7):4230–7. http:// dx.doi.org/10.1074/jbc.M808591200
- 30. Janeway CAJ, Travers P, Walport M, Shlomchik MJ. Immunobiology: The immune system in health and disease. 5th ed. New York: Garland Science; 2001.
- Lannfelt L, Relkin NR, Siemers ER. Amyloid-β-directed immunotherapy for Alzheimer's disease. J Int Med. 2014;275(3):284–95. http://dx.doi.org/10.1111/joim.12168
- DeMattos RB, Bales KR, Cummins DJ, Dodart JC, Paul SM, Holtzman DM. Peripheral anti-Abeta antibody alters CNS and plasma Abeta clearance and decreases brain Abeta burden in a mouse model of Alzheimer's disease. Proc Natl Acad Sci U S A. 2001;98(15):8850–5. http://dx.doi.org/10.1073/ pnas.151261398
- Wisniewski T, Konietzko U. Amyloid-beta immunisation for Alzheimer's disease. Lancet Neurol. 2008;7(9):805–11. http://dx.doi.org/10.1016/S1474-4422(08)70170-4

- 34. Bacskai BJ, Kajdasz ST, Christie RH, Carter C, Games D, Seubert P, et al. Imaging of amyloid-β deposits in brains of living mice permits direct observation of clearance of plaques with immunotherapy. Nat Med. 2001;7(3):369–72. http://dx.doi.org/10.1038/85525
- Solomon B, Koppel R, Hanan E, Katzav T. Monoclonal antibodies inhibit in vitro fibrillar aggregation of the Alzheimer beta-amyloid peptide. Proc Nat Acad Sci. 1996;93(1):452–5. http://dx.doi. org/10.1073/pnas.93.1.452
- Frenkel D, Solomon B, Benhar I. Modulation of Alzheimer's β-amyloid neurotoxicity by sitedirected single-chain antibody. J Neuroimmunol. 2000;106(1):23–31. http://dx.doi.org/10.1016/ S0165-5728(99)00232-5
- 37. Brody DL, Holtzman DM. Active and passive immunotherapy for neurodegenerative disorders. Annu Rev Neurosci. 2008;31:175–93. http://dx.doi.org/10.1146/annurev.neuro.31.060407.125529
- Bacskai BJ, Kajdasz ST, McLellan ME, Games D, Seubert P, Schenk D, et al. Non-Fc-mediated mechanisms are involved in clearance of amyloid-beta in vivo by immunotherapy. J Neurosci. 2002;22(18):7873–8. http://dx.doi.org/10.1523/JNEUROSCI.22-18-07873.2002
- Taguchi H, Planque S, Nishiyama Y, Symersky J, Boivin S, Szabo P, et al. Autoantibody-catalyzed hydrolysis of amyloid β peptide. J Biol Chem. 2008;283(8):4714–22. http://dx.doi.org/10.1074/jbc.M707983200
- Schenk D, Barbour R, Dunn W, Gordon G, Grajeda H, Guido T, et al. Immunization with amyloid-beta attenuates Alzheimer-disease-like pathology in the PDAPP mouse. Nature. 1999;400(6740):173–7. http://dx.doi.org/10.1038/22124
- Orgogozo JM, Gilman S, Dartigues JF, Laurent B, Puel M, Kirby LC, et al. Subacute meningoencephalitis in a subset of patients with AD after Abeta42 immunization. Neurology. 2003;61(1):46–54. http:// dx.doi.org/10.1212/01.WNL.0000073623.84147.A8
- Gilman S, Koller M, Black RS, Jenkins L, Griffith SG, Fox NC, et al. Clinical effects of Aβ immunization (AN1792) in patients with AD in an interrupted trial. Neurology. 2005;64(9):1553–62. http:// dx.doi.org/10.1212/01.WNL.0000159740.16984.3C
- Holmes C, Boche D, Wilkinson D, Yadegarfar G, Hopkins V, Bayer A, et al. Long-term effects of Abeta42 immunisation in Alzheimer's disease: Follow-up of a randomised, placebo-controlled phase I trial. Lancet. 2008;372(9634):216–23. http://dx.doi.org/10.1016/S0140-6736(08)61075-2
- 44. Boche D, Zotova E, Weller RO, Love S, Neal JW, Pickering RM, et al. Consequence of Abeta immunization on the vasculature of human Alzheimer's disease brain. Brain. 2008;131(Pt 12):3299–310. http://dx.doi.org/10.1093/brain/awn261
- Sakai K, Boche D, Carare R, Johnston D, Holmes C, Love S, et al. Aβ immunotherapy for Alzheimer's disease: Effects on apoE and cerebral vasculopathy. Acta Neuropathol. 2014;128(6):777–89. http:// dx.doi.org/10.1007/s00401-014-1340-9
- 46. Paquet C, Amin J, Mouton-Liger F, Nasser M, Love S, Gray F, et al. Effect of active Aβ immunotherapy on neurons in human Alzheimer's disease. J Pathol. 2015;235(5):721–30. http://dx.doi.org/10.1002/ path.4491
- Nicoll JAR, Buckland GR, Harrison CH, Page A, Harris S, Love S, et al. Persistent neuropathological effects 14 years following amyloid-β immunization in Alzheimer's disease. Brain. 2019;142(7): 2113–26. http://dx.doi.org/10.1093/brain/awz142
- Sterner RM, Takahashi PY, Yu Ballard AC. Active vaccines for Alzheimer disease treatment. J Am Med Dir Assoc. 2016;17(9):862.e11–15. http://dx.doi.org/10.1016/j.jamda.2016.06.009
- 49. Ketter N, Liu E, Di J, Honig LS, Lu M, Novak G, et al. A randomized, double-blind, phase 2 study of the effects of the vaccine vanutide cridificar with QS-21 adjuvant on immunogenicity, safety and amyloid imaging in patients with mild to moderate Alzheimer's disease. J Prev Alzheimers Dis. 2016;3(4):192–201.
- Pasquier F, Sadowsky C, Holstein A, Leterme GLP, Peng Y, Jackson N, et al. Two phase 2 multiple ascending–dose studies of vanutide cridificar (ACC-001) and QS-21 adjuvant in mild-to-moderate Alzheimer's disease. J Alzheimers Dis. 2016;51:1131–43. http://dx.doi.org/10.3233/JAD-150376
- Farlow MR, Andreasen N, Riviere ME, Vostiar I, Vitaliti A, Sovago J, et al. Long-term treatment with active Aβ immunotherapy with CAD106 in mild Alzheimer's disease. Alzheimers Res Ther. 2015;7(1):23. http://dx.doi.org/10.1186/s13195-015-0108-3
- Vandenberghe R, Riviere M-E, Caputo A, Sovago J, Maguire RP, Farlow M, et al. Active Aβ immunotherapy CAD106 in Alzheimer's disease: A phase 2b study. Alzheimers Dementia. 2016;3(1):10–22. http://dx.doi.org/10.1016/j.trci.2016.12.003

- 53. Lacosta A-M, Pascual-Lucas M, Pesini P, Casabona D, Pérez-Grijalba V, Marcos-Campos I, et al. Safety, tolerability and immunogenicity of an active anti-Aβ40 vaccine (ABvac40) in patients with Alzheimer's disease: A randomised, double-blind, placebo-controlled, phase I trial. Alzheimers Res Ther. 2018;10(1):12. http://dx.doi.org/10.1186/s13195-018-0340-8
- Muhs A, Hickman DT, Pihlgren M, Chuard N, Giriens V, Meerschman C, et al. Liposomal vaccines with conformation-specific amyloid peptide antigens define immune response and efficacy in APP transgenic mice. Proc Nat Acad Sci. 2007;104(23):9810–15. http://dx.doi.org/10.1073/pnas.0703137104
- Mandler M, Santic R, Gruber P, Cinar Y, Pichler D, Funke SA, et al. Tailoring the antibody response to aggregated Aß using novel Alzheimer-vaccines. PLoS One. 2015;10(1):e0115237. http://dx.doi. org/10.1371/journal.pone.0115237
- 56. Schneeberger A, Hendrix S, Mandler M, Ellison N, Burger V, Brunner M, et al. Results from a Phase II study to assess the clinical and immunological activity of AFFITOPE(R) AD02 in patients with early Alzheimer's disease. J Prev Alzheimers Dis. 2015;2(2):103–14.
- 57. Wang CY, Wang P-N, Chiu M-J, Finstad CL, Lin F, Lynn S, et al. UB-311, a novel UBITh(®) amyloid β peptide vaccine for mild Alzheimer's disease. Alzheimers Dementia. 2017;3(2):262–72. http://dx.doi. org/10.1016/j.trci.2017.03.005
- 58. United Neuroscience announces positive top-line results from Phase 2a clinical study of UB-311 vaccine in patients with Alzheimer's disease. United Neuroscience, Press Release. 2019:Jan 16, 2019.
- Mallbris L, Davies J, Glasebrook A, Tang Y, Glaesner W, Nickoloff BJ. Molecular insights into fully human and humanized monoclonal antibodies: What are the differences and should dermatologists care? J Clin Aesthet Dermatol. 2016;9(7):13–15.
- 60. Nelson AL, Dhimolea E, Reichert JM. Development trends for human monoclonal antibody therapeutics. Nature Rev Drug Discov. 2010;9(10):767–74. http://dx.doi.org/10.1038/nrd3229
- van Dyck CH. Anti-amyloid-β monoclonal antibodies for Alzheimer's disease: Pitfalls and promise. Biol Psychiatry. 2018;83(4):311–19. http://dx.doi.org/10.1016/j.biopsych.2017.08.010
- Sperling RA, Jack CR, Black SE, Frosch MP, Greenberg SM, Hyman BT, et al. Amyloid-related imaging abnormalities in amyloid-modifying therapeutic trials: Recommendations from the Alzheimer's Association Research Roundtable Workgroup. Alzheimers Dementia. 2011;7(4):367–85. http:// dx.doi.org/10.1016/j.jalz.2011.05.2351
- 63. DiFrancesco J, Longoni M, Piazza F. Anti-Aβ autoantibodies in amyloid related imaging abnormalities (ARIA): Candidate biomarker for immunotherapy in Alzheimer's disease and cerebral amyloid angiopathy. Front Neurol. 2015;6:207. http://dx.doi.org/10.3389/fneur.2015.00207
- Piazza F, Winblad B. Amyloid-Related Imaging Abnormalities (ARIA) in immunotherapy trials for Alzheimer's disease: Need for prognostic biomarkers? J Alzheimers Dis. 2016;52(2):417–20. http:// dx.doi.org/10.3233/JAD-160122
- Wilcock DM, Colton CA. Immunotherapy, vascular pathology, and microhemorrhages in transgenic mice. CNS Neurol Disord Drug Targets. 2009;8(1):50–64. http://dx.doi.org/10.2174/187152709787601858
- 66. Bruhns P. Properties of mouse and human IgG receptors and their contribution to disease models. Blood. 2012;119(24):5640–9. http://dx.doi.org/10.1182/blood-2012-01-380121
- Freeman GB, Brown TP, Wallace K, Bales KR. Chronic administration of an aglycosylated murine antibody of ponezumab does not worsen microhemorrhages in aged Tg2576 mice. Curr Alzheimer Res. 2012;9(9):1059–68. http://dx.doi.org/10.2174/156720512803569064
- Fuller JP, Stavenhagen JB, Christensen S, Kartberg F, Glennie MJ, Teeling JL. Comparing the efficacy and neuroinflammatory potential of three anti-abeta antibodies. Acta Neuropathol. 2015;130(5): 699–711. http://dx.doi.org/10.1007/s00401-015-1484-2
- Miles LA, Crespi GA, Doughty L, Parker MW. Bapineuzumab captures the N-terminus of the Alzheimer's disease amyloid-beta peptide in a helical conformation. Sci Rep. 2013;3:1302. http:// dx.doi.org/10.1038/srep01302
- Zampar S, Klafki HW, Sritharen K, Bayer TA, Wiltfang J, Rostagno A, et al. N-terminal heterogeneity of parenchymal and vascular amyloid-β deposits in Alzheimer's disease. Neuropathol Appl Neurobiol. 2020;46:673–85. http://dx.doi.org/10.1111/nan.12637
- Salloway S, Sperling R, Fox NC, Blennow K, Klunk W, Raskind M, et al. Two phase 3 trials of bapineuzumab in mild-to-moderate Alzheimer's disease. New Engl J Med. 2014;370(4):322–33. http://dx.doi.org/10.1056/NEJMoa1304839

- 72. Seubert P, Barbour R, Khan K, Motter R, Tang P, Kholodenko D, et al. Antibody capture of soluble Aβ does not reduce cortical Aβ amyloidosis in the PDAPP mouse. Neurodegen Dis. 2008;5(2):65–71. http://dx.doi.org/10.1159/000112834
- Karran E, Hardy J. A critique of the drug discovery and phase 3 clinical programs targeting the amyloid hypothesis for Alzheimer disease. Ann Neurol. 2014;76(2):185–205. http://dx.doi.org/10.1002/ ana.24188
- 74. Black RS, Sperling RA, Safirstein B, Motter RN, Pallay A, Nichols A, et al. A single ascending dose study of bapineuzumab in patients with Alzheimer disease. Alzheimer Dis Assoc Disord. 2010;24(2): 198–203. http://dx.doi.org/10.1097/WAD.0b013e3181c53b00
- 75. Blennow K, Zetterberg H, Rinne JO, Salloway S, Wei J, Black R, et al. Effect of immunotherapy with bapineuzumab on cerebrospinal fluid biomarker levels in patients with mild to moderate Alzheimer disease. Arch Neurol. 2012;69(8):1002–10. http://dx.doi.org/10.1001/archneurol.2012.90
- 76. Rinne JO, Brooks DJ, Rossor MN, Fox NC, Bullock R, Klunk WE, et al. 11C-PiB PET assessment of change in fibrillar amyloid-beta load in patients with Alzheimer's disease treated with bapineuzumab: A phase 2, double-blind, placebo-controlled, ascending-dose study. Lancet Neurol. 2010;9(4): 363–72. http://dx.doi.org/10.1016/S1474-4422(10)70043-0
- 77. Adolfsson O, Pihlgren M, Toni N, Varisco Y, Buccarello AL, Antoniello K, et al. An effectorreduced anti-β-amyloid (Aβ) antibody with unique Aβ binding properties promotes neuroprotection and glial engulfment of Aβ. J Neurosci. 2012;32(28):9677–89. http://dx.doi.org/10.1523/ JNEUROSCI.4742-11.2012
- Zhao J, Nussinov R, Ma B. Mechanisms of recognition of Aβ monomer, oligomer, and fibril by homologous antibodies. J Biol Chem. 2017;292(44):18325–43. http://dx.doi.org/10.1074/jbc.M117.801514
- Cummings JL, Cohen S, van Dyck CH, Brody M, Curtis C, Cho W, et al. ABBY A phase 2 randomized trial of crenezumab in mild to moderate Alzheimer disease. Neurology. 2018;90(21):e1889–e97. http://dx.doi.org/10.1212/WNL.00000000005550
- Salloway S, Honigberg LA, Cho W, Ward M, Friesenhahn M, Brunstein F, et al. Amyloid positron emission tomography and cerebrospinal fluid results from a crenezumab anti-amyloid-beta antibody double-blind, placebo-controlled, randomized phase II study in mild-to-moderate Alzheimer's disease (BLAZE). Alzheimers Res Ther. 2018;10(1):96. http://dx.doi.org/10.1186/s13195-018-0424-5
- Blaettler T, Smith J, Smith J, Paul R, Asnaghi V, Fuji R, et al. P2-003: Clinical trial design of cread: A randomized, double-blind, placebo-controlled, parallel-group phase 3 study to evaluate crenezumab treatment in patients with prodromal-to-mild Alzheimer's disease. Alzheimers Dementia. 2016;12:P609. http://dx.doi.org/10.1016/j.jalz.2016.06.1207
- 82. Tariot PN, Lopera F, Langbaum JB, Thomas RG, Hendrix S, Schneider LS, et al. The Alzheimer's prevention initiative autosomal-dominant Alzheimer's disease trial: A study of crenezumab versus placebo in preclinical PSEN1 E280A mutation carriers to evaluate efficacy and safety in the treatment of autosomal-dominant Alzheimer's disease, including a placebo-treated noncarrier cohort. Alzheimer's Dementia. 2018;4:150–60. http://dx.doi.org/10.1016/j.trci.2018.02.002
- Bohrmann B, Baumann K, Benz J, Gerber F, Huber W, Knoflach F, et al. Gantenerumab: A novel human anti-abeta antibody demonstrates sustained cerebral amyloid-beta binding and elicits cell-mediated removal of human amyloid-beta. J Alzheimers Dis. 2011;28(1):49–69. http://dx.doi.org/10.3233/ JAD-2011-110977
- Ostrowitzki S, Deptula D, Thurfjell L, Barkhof F, Bohrmann B, Brooks DJ, et al. Mechanism of amyloid removal in patients with Alzheimer disease treated with gantenerumab. Arch Neurol. 2012;69(2): 198–207. http://dx.doi.org/10.1001/archneurol.2011.1538
- Ostrowitzki S, Lasser RA, Dorflinger E, Scheltens P, Barkhof F, Nikolcheva T, et al. A phase III randomized trial of gantenerumab in prodromal Alzheimer's disease. Alzheimers Res Ther. 2017;9(1):95. http://dx.doi.org/10.1186/s13195-017-0318-y
- 86. Klein G, Delmar P, Voyle N, Rehal S, Hofmann C, Abi-Saab D, et al. Gantenerumab reduces amyloid-β plaques in patients with prodromal to moderate Alzheimer's disease: A PET substudy interim analysis. Alzheimers Res Ther. 2019;11(1):101. http://dx.doi.org/10.1186/s13195-019-0559-z
- 87. Portron A, Jordan P, Draper K, Muenzer C, Dickerson D, van Iersel T, et al. A phase I study to assess the effect of speed of injection on pain, tolerability, and pharmacokinetics after high-volume

subcutaneous administration of gantenerumab in healthy volunteers. Clin Ther. 2020;42(1):108–20. http://dx.doi.org/10.1016/j.clinthera.2019.11.015

- La Porte SL, Bollini SS, Lanz TA, Abdiche YN, Rusnak AS, Ho WH, et al. Structural basis of C-terminal beta-amyloid peptide binding by the antibody ponezumab for the treatment of Alzheimer's disease. J Mol Biol. 2012;421(4–5):525–36. http://dx.doi.org/10.1016/j.jmb.2011.11.047
- Karlnoski RA, Rosenthal A, Kobayashi D, Pons J, Alamed J, Mercer M, et al. Suppression of amyloid deposition leads to long-term reductions in Alzheimer's pathologies in Tg2576 mice. J Neurosci. 2009;29(15):4964–71. http://dx.doi.org/10.1523/JNEUROSCI.4560-08.2009
- Freeman GB, Lin JC, Pons J, Raha NM. 39-week toxicity and toxicokinetic study of ponezumab (PF-04360365) in cynomolgus monkeys with 12-week recovery period. J Alzheimers Dis. 2012;28(3):531–41. http://dx.doi.org/10.3233/JAD-2011-110869
- Landen JW, Zhao Q, Cohen S, Borrie M, Woodward M, Billing CBJ, et al. Safety and pharmacology of a single intravenous dose of ponezumab in subjects with mild-to-moderate Alzheimer disease: A phase I, randomized, placebo-controlled, double-blind, dose-escalation study. Clin Neuropharmacol. 2013;36(1):14–23. http://dx.doi.org/10.1097/WNF0b013e31827db49b
- 92. Miyoshi I, Fujimoto Y, Yamada M, Abe S, Zhao Q, Cronenberger C, et al. Safety and pharmacokinetics of PF-04360365 following a single-dose intravenous infusion in Japanese subjects with mild-to-moderate Alzheimer's disease: A multicenter, randomized, double-blind, placebocontrolled, dose-escalation study. Int J Clin Pharmacol Ther. 2013;51(12):911–23. http://dx.doi. org/10.5414/CP201816
- Burstein AH, Zhao Q, Ross J, Styren S, Landen JW, Ma WW, et al. Safety and pharmacology of ponezumab (PF-04360365) after a single 10-minute intravenous infusion in subjects with mild to moderate Alzheimer disease. Clin Neuropharmacol. 2013;36(1):8–13. http://dx.doi.org/10.1097/ WNF0b013e318279bcfa
- Landen JW, Cohen S, Billing CB, Cronenberger C, Styren S, Burstein AH, et al. Multiple-dose ponezumab for mild-to-moderate Alzheimer's disease: Safety and efficacy. Alzheimers Dementia. 2017;3(3):339–47. http://dx.doi.org/10.1016/j.trci.2017.04.003
- Landen JW, Andreasen N, Cronenberger CL, Schwartz PF, Börjesson-Hanson A, Östlund H, et al. Ponezumab in mild-to-moderate Alzheimer's disease: Randomized phase II PET-PIB study. Alzheimers Dementia. 2017;3(3):393–401. http://dx.doi.org/10.1016/j.trci.2017.05.003
- Leurent C, Goodman JA, Zhang Y, He P, Polimeni JR, Gurol ME, et al. Immunotherapy with ponezumab for probable cerebral amyloid angiopathy. Ann Clin Transl Neurol. 2019;6(4):795–806. http:// dx.doi.org/10.1002/acn3.761
- Revesz T, Ghiso J, Lashley T, Plant G, Rostagno A, Frangione B, et al. Cerebral amyloid angiopathies: A pathologic, biochemical, and genetic view. J Neuropathol Exp Neurol. 2003;62(9):885–98. http:// dx.doi.org/10.1093/jnen/62.9.885
- 98. AttemsJ.JellingerK, ThalDR, VanNostrandW. Review: Sporadiccerebralamyloidangiopathy. Neuropathol Appl Neurobiol. 2011;37(1):75–93. http://dx.doi.org/10.1111/j.1365-2990.2010.01137.x
- Crespi GAN, Hermans SJ, Parker MW, Miles LA. Molecular basis for mid-region amyloid-β capture by leading Alzheimer's disease immunotherapies. Sci Rep. 2015;5(1):9649. http://dx.doi.org/10.1038/ srep09649
- 100. Siemers ER, Friedrich S, Dean RA, Gonzales CR, Farlow MR, Paul SM, et al. Safety and changes in plasma and cerebrospinal fluid amyloid β after a single administration of an amyloid β monoclonal antibody in subjects with Alzheimer disease. Clin Neuropharmacol. 2010;33(2):67–73. http://dx.doi. org/10.1097/WNE0b013e3181cb577a
- 101. Mably AJ, Liu W, Mc Donald JM, Dodart J-C, Bard F, Lemere CA, et al. Anti-Aβ antibodies incapable of reducing cerebral Aβ oligomers fail to attenuate spatial reference memory deficits in J20 mice. Neurobiol Dis. 2015;82:372–84. http://dx.doi.org/10.1016/j.nbd.2015.07.008
- 102. Dodart J-C, Bales KR, Gannon KS, Greene SJ, DeMattos RB, Mathis C, et al. Immunization reverses memory deficits without reducing brain Aβ burden in Alzheimer's disease model. Nat Neurosci. 2002;5:452–7. http://dx.doi.org/10.1038/nn842
- 103. Bouter Y, Noguerola JS, Tucholla P, Crespi GA, Parker MW, Wiltfang J, et al. Abeta targets of the biosimilar antibodies of Bapineuzumab, Crenezumab, Solanezumab in comparison to an antibody

against N-truncated Abeta in sporadic Alzheimer disease cases and mouse models. Acta Neuropathol. 2015;130:713–29. http://dx.doi.org/10.1007/s00401-015-1489-x

- 104. Farlow M, Arnold SE, van Dyck CH, Aisen PS, Snider BJ, Porsteinsson AP, et al. Safety and biomarker effects of solanezumab in patients with Alzheimer's disease. Alzheimers Dementia. 2012;8(4):261–71. http://dx.doi.org/10.1016/j.jalz.2011.09.224
- 105. Doody RS, Thomas RG, Farlow M, Iwatsubo T, Vellas B, Joffe S, et al. Phase 3 trials of solanezumab for mild-to-moderate Alzheimer's disease. New Engl J Med. 2014;370(4):311–21. http://dx.doi. org/10.1056/NEJMoa1312889
- 106. Siemers ER, Sundell KL, Carlson C, Case M, Sethuraman G, Liu-Seifert H, et al. Phase 3 solanezumab trials: Secondary outcomes in mild Alzheimer's disease patients. Alzheimers Dementia. 2016;12(2):110–20. http://dx.doi.org/10.1016/j.jalz.2015.06.1893
- 107. Honig LS, Vellas B, Woodward M, Boada M, Bullock R, Borrie M, et al. Trial of solanezumab for mild dementia due to Alzheimer's disease. New Engl J Med. 2018;378(4):321–30. http://dx.doi. org/10.1056/NEJMoa1705971
- 108. Englund H, Sehlin D, Johansson AS, Nilsson LN, Gellerfors P, Paulie S, et al. Sensitive ELISA detection of amyloid-beta protofibrils in biological samples. J Neurochem. 2007;103(1):334–45.
- 109. Sehlin D, Englund H, Simu B, Karlsson M, Ingelsson M, Nikolajeff F, et al. Large aggregates are the major soluble Aβ species in AD brain fractionated with density gradient ultracentrifugation. PLoS One. 2012;7(2):e32014. http://dx.doi.org/10.1371/journal.pone.0032014
- 110. Sehlin D, Hedlund M, Lord A, Englund H, Gellerfors P, Paulie S, et al. Heavy-chain complementarity-determining regions determine conformation selectivity of anti-Aβ antibodies. Neurodegen Dis. 2011;8(3):117–23. http://dx.doi.org/10.1159/000316530
- 111. Lord A, Gumucio A, Englund H, Sehlin D, Sundquist VS, Soderberg L, et al. An amyloid-beta protofibril-selective antibody prevents amyloid formation in a mouse model of Alzheimer's disease. Neurobiol Dis. 2009;36(3):425–34. http://dx.doi.org/10.1016/j.nbd.2009.08.007
- 112. Lord A, Englund H, Soderberg L, Tucker S, Clausen F, Hillered L, et al. Amyloid-beta protofibril levels correlate with spatial learning in Arctic Alzheimer's disease transgenic mice. FEBS J. 2009;276(4): 995–1006. http://dx.doi.org/10.1111/j.1742-4658.2008.06836.x
- 113. Tucker S, Möller C, Tegerstedt K, Lord A, Laudon H, Sjödahl J, et al. The murine version of BAN2401 (mAb158) selectively reduces amyloid-β protofibrils in brain and cerebrospinal fluid of tg-ArcSwe mice. J Alzheimers Dis. 2015;43:575–88. http://dx.doi.org/10.3233/JAD-140741
- 114. Sehlin D, Fang XT, Cato L, Antoni G, Lannfelt L, Syvänen S. Antibody-based PET imaging of amyloid beta in mouse models of Alzheimer's disease. Nat Commun. 2016;7(1):10759. http://dx.doi. org/10.1038/ncomms10759
- 115. Logovinsky V, Satlin A, Lai R, Swanson C, Kaplow J, Osswald G, et al. Safety and tolerability of BAN2401 – a clinical study in Alzheimer's disease with a protofibril selective Aβ antibody. Alzheimers Res Ther. 2016;8(1):14. http://dx.doi.org/10.1186/s13195-016-0181-2
- 116. Osswald G. BioArctic announces positive topline results of BAN2401 phase 2b at 18 months in early Alzheimer's disease. BioArctic Press Release. 2018;Jul 6, 2018.
- 117. Swanson CJ, Zhang Y, Dhadda S, Wang J, Kaplow J, Bradley H, et al. Persistence of BAN2401mediated amyloid reductions post-treatment: A preliminary comparison of amyloid status between the core phase of BAN2401-G000-201 and baseline of the open-label extension phase in subjects with early Alzheimer's disease (1330). Neurology. 2020;94(15 Suppl.):1330.
- 118. Arndt JW, Qian F, Smith BA, Quan C, Kilambi KP, Bush MW, et al. Structural and kinetic basis for the selectivity of aducanumab for aggregated forms of amyloid-β. Sci Rep. 2018;8(1):6412. http://dx.doi. org/10.1038/s41598-018-24501-0
- 119. Ferrero J, Williams L, Stella H, Leitermann K, Mikulskis A, O'Gorman J, et al. First-in-human, double-blind, placebo-controlled, single-dose escalation study of aducanumab (BIIB037) in mild-to-moderate Alzheimer's disease. Alzheimers Dementia. 2016;2(3):169–76. http://dx.doi.org/10.1016/j. trci.2016.06.002
- 120. Sevigny J, Suhy J, Chiao P, Chen T, Klein G, Purcell D, et al. Amyloid PET screening for enrichment of early-stage Alzheimer disease clinical trials: Experience in a phase 1b clinical trial. Alzheimer Dis Assoc Disord. 2016;30(1):1–7. http://dx.doi.org/10.1097/WAD.00000000000144

- 121. Panza F, Lozupone M, Seripa D, Imbimbo BP. Amyloid-β immunotherapy for alzheimer disease: Is it now a long shot? Ann Neurol. 2019;85(3):303–15. http://dx.doi.org/10.1002/ana.25410
- 122. Budd Haeberlein S, O'Gorman J, Chiao P, Bussière T, von Rosenstiel P, Tian Y, et al. Clinical development of aducanumab, an anti-Aβ human monoclonal antibody being investigated for the treatment of early Alzheimer's disease. J Prev Alzheimers Dis. 2017;4(4):255–63.
- 123. Howard R, Liu KY. Questions EMERGE as Biogen claims aducanumab turnaround. Nat Rev Neurol. 2020;16(2):63–4. http://dx.doi.org/10.1038/s41582-019-0295-9
- 124. Budd Haeberlein S, Von Hehn C, Tian Y, Chalkias S, Muralidharan KK, Chen T, et al. EMERGE and ENGAGE topline results: Two phase 3 studies to evaluate aducanumab in patients with early Alzheimer's disease [Internet]. San Diego, CA: CTAD; Dec 5, 2019. Available from: http://investors.biogen.com/ static-files/ddd45672-9c7e-4c99-8a06-3b557697c06f
- 125. Schneider L. A resurrection of aducanumab for Alzheimer's disease. Lancet Neurol. 2020;19(2): 111–12. http://dx.doi.org/10.1016/S1474-4422(19)30480-6
- 126. Caouette D. Biogen completes submission of biologics license application to FDA for aducanumab as a treatment for Alzheimer's disease. Biogen Press Release. 2020:July 8, 2020.
- 127. Bogstedt A, Groves M, Tan K, Narwal R, McFarlane M, Höglund K. Development of immunoassays for the quantitative assessment of amyloid-β in the presence of therapeutic antibody: Application to pre-clinical studies. J Alzheimers Dis. 2015;46:1091–101. http://dx.doi.org/10.3233/JAD-142988
- 128. Billinton A, Newton P, Lloyd C, Groves M, Welsh F, Bogstedt A, et al. Preclinical discovery and development of MEDI1814, a monoclonal antibody selectively targeting β-amyloid 42 (Aβ42). Alzheimers Dementia. 2017;13:P266. http://dx.doi.org/10.1016/j.jalz.2017.06.141
- 129. Ostenfeld T, Pomfret M, Billinton A, Chessell I, Chessell T, Lindqvist E, et al. Evaluation of safety, tolerability, pharmacokinetics and pharmacodynamics of MEDI1814, a β-amyloid 42 (Aβ42)-specific antibody, in patients with mild-moderate Alzheimer's disease. Alzheimers Dementia. 2017;13:P574. http://dx.doi.org/10.1016/j.jalz.2017.07.193
- 130. Pradier L, Blanchard-Brégeon V, Bohme A, Debeir T, Menager J, Benoit P, et al. SAR228810: An antibody for protofibrillar amyloid β peptide designed to reduce the risk of amyloid-related imaging abnormalities (ARIA). Alzheimers Res Ther. 2018;10(1):117. http://dx.doi.org/10.1186/s13195-018-0447-y
- 131. Taupin V, Vaucher N, Loux C, Le Parc J, Michoux F, Colonna C, et al. Anti-amyloid antibodies as a potential therapeutic for Alzheimer's disease: Neuroprotective activity of SAR228810 against amyloidinduced neurotoxicity in vitro. Alzheimers Dementia. 2017;13:P944. http://dx.doi.org/10.1016/j. jalz.2017.06.1851
- 132. Pradier L, Cohen C, Veronique B, Debeir T, Barneoud P, Canton T, et al. SAR228810: An antiprotofibrillar beta-amyloid antibody designed to reduce risk of amyloid-related imaging abnormalities (ARIA). Alzheimers Dementia. 2013;9:P808. http://dx.doi.org/10.1016/j.jalz.2013.05.1678
- Whitson JS, Selkoe DJ, Cotman CW. Amyloid beta protein enhances the survival of hippocampal neurons in vitro. Science. 1989;243(4897):1488–90. http://dx.doi.org/10.1126/science.2928783
- 134. Puzzo D, Privitera L, Leznik E, Fa M, Staniszewski A, Palmeri A, et al. Picomolar amyloid-beta positively modulates synaptic plasticity and memory in hippocampus. J Neurosci. 2008;28(53): 14537–45. http://dx.doi.org/10.1523/JNEUROSCI.2692-08.2008
- 135. Puzzo D, Privitera L, Fa M, Staniszewski A, Hashimoto G, Aziz F, et al. Endogenous amyloid-beta is necessary for hippocampal synaptic plasticity and memory. Ann Neurol. 2011;69(5):819–30. http:// dx.doi.org/10.1002/ana.22313
- 136. Lucey BP, Hicks TJ, McLeland JS, Toedebusch CD, Boyd J, Elbert DL, et al. Effect of sleep on overnight cerebrospinal fluid amyloid β kinetics. Ann Neurol. 2018;83(1):197–204. http://dx.doi.org/10.1002/ana.25117
- 137. Szczygielski J, Mautes A, Steudel WI, Falkai P, Bayer TA, Wirths O. Traumatic brain injury: Cause or risk of Alzheimer's disease? A review of experimental studies. J Neural Transm. 2005 Nov;112(11): 1547–64. http://dx.doi.org/10.1007/s00702-005-0326-0
- 138. Garcia-Alloza M, Gregory J, Kuchibhotla KV, Fine S, Wei Y, Ayata C, et al. Cerebrovascular lesions induce transient β-amyloid deposition. Brain. 2011;134(12):3697–707. http://dx.doi.org/10.1093/ brain/awr300

Amyloid β-Targeted Inhibitory Peptides for Alzheimer's Disease: Current State and Future Perspectives

Safura Jokar¹ • Saeedeh Khazaei² • Xaniar Esmailzadeh Gameshgoli³ • Mona Khafaji⁴ • Behnam Yarani⁵ • Mohammad Sharifzadeh⁶ • Davood Beiki⁷ • Omid Bavi⁸

¹Department of Nuclear Pharmacy, Faculty of Pharmacy, Tehran University of Medical Sciences, Tehran, Iran; ²Department of Pharmaceutical Biomaterials, Faculty of Pharmacy, Tehran University of Medical Sciences, Tehran, Iran; ³Colloids and Polymers Physics Group, Particle Physics Department, Universidade de Santiago de Compostela, Santiago de Compostela, Spain; ⁴Institute for Nanoscience and Nanotechnology, Sharif University of Technology, Tehran, Iran; ⁵Division of Nanobiotechnology, Department of Life Sciences Engineering, Faculty of New Sciences and Technologies, University of Tehran, Tehran, Iran; ⁶Department of Toxicology and Pharmacology, Faculty of Pharmacy; Toxicology and Poisoning Research Centre, Tehran University of Medical Sciences, Tehran, Iran; ⁸Department of Nuclear Medicine, Tehran University of Medical Sciences, Tehran, Iran; ⁸Department of Mechanical and Aerospace Engineering, Shiraz University of Technology, Shiraz, Iran

Authors for correspondence: Omid Bavi, Department of Mechanical and Aerospace Engineering, Shiraz University of Technology, Shiraz, Iran. Email: o.bavi@sutech.ac.ir; Davood Beiki, Research Center for Nuclear Medicine, Tehran University of Medical Sciences, Shariati Hospital, North Kargar Ave. 1411713135, Tehran, Iran. Email: beikidav@tums.ac.ir Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch3

Abstract: Alzheimer's disease is the most common irreversible neurodegenerative disorder. To date, there is no cure for Alzheimer's disease. While multiple pathological mechanisms have been proposed for the onset and progression of Alzheimer's disease, the hypothesis that attracted much attention is the amyloid hypothesis. The senile plaques that accumulate in the brain of Alzheimer's disease

In: *Alzheimer's Disease: Drug Discovery.* Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

Copyright: The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

patients are predominantly composed of beta amyloid (A β). A β deposition in the brain is thought to occur years before the emergence of clinical symptoms. The overproduction, aggregation, and fibrillation of A β , combined with reduced clearance, eventually lead to amyloid plaque formation and subsequent neurotoxicity. Hence, inhibition of A β aggregation and the promotion of A β clearance have been actively explored as therapeutic strategies for Alzheimer's disease. This chapter provides an overview of the current knowledge on one such strategy, A β -targeted inhibitory peptides.

Keywords: Aβ aggregation in Alzheimer's disease; biopanning; inhibitory peptides for Alzheimer's disease; peptide–nanostructure conjugates; peptidomimetics

INTRODUCTION

Alzheimer's disease is an age-dependent disorder that is the fifth leading cause of death in people aged 65 years and older. It is estimated that over 50 million people worldwide suffer from Alzheimer's disease, and this figure is set to increase to 152 million by 2050 with a financial burden of 1.1 trillion US dollars by 2050 (1–3). Several hypotheses, including the amyloid, cholinergic (4), and Tau protein hypotheses have been proposed to explain the pathophysiology and etiology of Alzheimer's disease (5). Because of the presence of $A\beta$ in the brain tissue, cerebrospinal fluid, and plasma, the amyloid cascade hypothesis is the most widely accepted. The amyloid cascade hypothesis states that neurodegeneration in Alzheimer's disease is the result of amyloid plaque and neurofibrillary tangle formations (6, 7). The overproduction, clearance failure, aggregation, and fibrillation of AB eventually leads to amyloid plaque formation. These factors also contribute to neuroinflammation and cell death. A β deposition in the brain is likely to be the first pathological incident that occurs years before the emergence of clinical symptoms. A β is produced through the proteolytic cleavage of the amyloid precursor protein (APP), a transmembrane glycoprotein, which is made up of a cytoplasmic domain with 55 amino acids and a long extracellular domain with 590–680 amino acids (8). APP cleavage by the proteases β - and γ -secretases produce A β fragments of varying size depending on the cleavage site (9), of which $A\beta_{40}$ (about 90%) and $A\beta_{42}$ (about 5–10%) are the most prevalent (Figure 1). $A\beta_{42}$ is more toxic than $A\beta_{40}$. After production, the A β peptides aggregate to form amyloid deposits. There are different aggregation forms such as low molecular weight oligomers, protofibrils, as well as mature fibrils that eventually come together to form amyloid deposits in the brain parenchyma and cerebrovascular spaces (10, 11).

Therefore, inhibition of $A\beta$ aggregation and the promotion of $A\beta$ clearance have been investigated as therapeutic strategies for Alzheimer's disease. Some of these strategies include the use of metal chelators (12), peptides (13), organic molecules (14), and biomolecules (15, 16). Peptides are considered a better option than small molecule-based compounds because of their high affinity for $A\beta$ and low toxicity (17). Although natural amino acid-based peptides are effective inhibitors of $A\beta$ aggregation, they are prone to faster enzymatic degradation and show a tendency for self-assembly into fibrils during administration (15). To overcome these problems, modified peptides have been generated (18) with

53



Figure 1. Amyloid- β **fibrillation and neuronal damage.** First, amyloid precursor protein is cleaved by β and γ secretases, respectively. A peptide fragment of 39-42 amino acid is formed depending on the site of cleavage. After cleavage, A β monomers start to self-assemble to form soluble toxic aggregates, and finally into insoluble fibrils, which subsequently cause synaptic dysfunction and neuronal death.

D-amino acids, retro-inverso cyclization, fluorination, as well as N-methylation of the ester bond (19). With this knowledge, peptides could be potential candidates for inhibiting A β conformational transitions, self-assembly, and toxicity against neurons, and promotion of the pathways of the nontoxic fibrillation and early diagnosis of Alzheimer's disease (20). This chapter provides an overview of the therapeutic potential peptides as A β aggregation inhibitors.

PEPTIDIC INHIBITORS

Luhrs and co-workers first experimentally described the structure of the $A\beta_{42}$ fibril (Figure 2, right) (21). At least four specific structural sites for interaction have been identified on the A β fibril (22): (i), hydrophobic regions of Ala30–Val36, and Leu17–Ala21 residues from the C and N-terminal β -sheets respectively; (ii), hydrophilic part using electrostatic interactions between Asp23 and Lys28 residues; (iii), central cleft in the interior of the U-shaped turn; and (iv), Glu22 ladder between the side chains of the Glu22 residues of the adjacent β -strands (Figure 2, bottom right). The formation of the salt-bridge between Asp23 and Lys28 is an essential β -sheet conformation stabilizer. Moreover, it might stimulate the oligomerization of A β via stabilizing the Val24–Asn27 turn (23). The hydrophobic residue of Met35 in the C-terminus domain could support

54



Figure 2. The structure of A β . Right, schematic of the A β_{42} amyloid fibril. Left, binding sites of inhibitory peptides of LPFFD and KVLFF on A β_{40} fibril.

fibril stability by hydrophobic interactions. The Met35 binding site can potentially inhibit protein-protein interactions and prevent amyloid fibril formation (24). These sites are probably critical regions in the initiation of A β nucleation, conformational transition promotion, and fibril formation. The residues ¹⁶KLVFFA²¹ (Figure 2) of the central hydrophobic core (CHC) region is a critical nucleation site, or self-recognition sequence. The Ile41 and Ala₄₂ residues can modulate A β_{42} oligomer formation (25) by interacting with the N and C-terminus of A β_{42} (26). Figure 2, left, shows the binding sites of the most common inhibitory peptides, such as LPFFD and KVLFF, on the A β_{40} fibril structure.

Peptide inhibitors are generally divided into $A\beta$ -based peptide inhibitors and non- $A\beta$ -based peptide inhibitors. A list of select $A\beta$ inhibitory peptides are presented in Table 1.

Aβ-based peptide inhibitors

These are based on the structure of the C-terminal fragments (CTFs) and the CHC sequences of the A β peptide. They bind to the A β peptide at specific sites and prevent its assembly into amyloid fibrils. Peptides consisting of D-enantiomeric amino acids exhibit greater stability against proteases and show a higher binding

∠	A select list of the select list	of inhibitory peptic	des against Aβ peptid	e aggregation	
v	Name/sequence of inhibitor peptic	le Configuration	Type	Therapeutic findings	Ref.
-	RR (RYYAAFFARR)	L-enantiomer	Rational design based on hydrophobic core and salt bridge region	- Inhibition of AB $_{40}$ fibrillation and decrease of AB $_{40}$ induced toxicity (in vitro)	(27)
2	D-(PGKLVYA) and D(KKLVFFARRRA)	D-enantiomer	CHC-derived sequence	 Inhibition of Aβ aggregation Increase life time in an animal model of expressing Aβ₊₂ 	(28)
$\tilde{\mathbf{c}}$	CP-2 (ILwHsK)	Cyclic d,l-α-peptide	Screening of the randomly library members (500)	 Nontoxic Stabilize small Aβ oligomers Disassemble formed Aβ fibrils 	(29)
4	D3D3 (D-RPRTRLHTHRNRRPRvTRLHTHF	D-enantiomer NR)	Not derived from Aβ sequence	 More inhibitory effect against Aβ₄₂ oligomerization than D3 peptide Formation of nontoxic amorphous instead of toxic oligomers (<i>In vivo</i>) 	(30)
ري ا	RD2 (DPTLHTHNRRR-NH ₂)	D-enantiomer	 Not derived from Aβ sequence 	 Good binding affinity to Aβ Reduce Aβ fibrillation formation High stability in mouse plasma High bioavailability (<i>In vivo</i>) 	(31, 32)
0	AOEP2 (FDYKAEFMPWDT)	D-enantiomer	 A mimotope of the Aβ oligomer Selected by phage display 	• Targeting of all forms of A β • The considrable reduction of TNF- α	(33)
~	LPYFD-amide	L-enantiomer	CHC-derived sequence	 Neuroprotective Decrease tau aggregation (In vivo) 	(34)
8	Ac-LPFFN-NH ₂	L-enantiomer	CHC-derived sequence	 Inhibit of Aβ₄₀ aggregation Stabilize the α-helical conformation of Aβ₄₀ Prolong the fibril formation 	(35)
0	D-4F peptide	D-enantiomer	Not derived from Aβ sequence	 Inhibit Aβ deposition Improve cognitive function 	(36)
				Table continued on fol	lowing page

TABLE 1

A select list of inhibitory peptides against Aß peptide aggregation (Continued)

°Z	Name/sequence of inhibitor peptide	Configuration	Type	Therapeutic findings	Ref.
10	Diazirine-equipped cyclo-KLVF(b-Ph)F	cyclic	CHC-derived sequence	 Inhibit Aβ aggregation Prevent Aβ₄₂ induced toxicity 	(37)
11	Attached flavin to an A β -binding peptide	cyclic	CHC-derived sequence	 Inhibition of Aβ aggregation Decrease of the aggregation potency and Aβ induced neurotoxicity 	(38)
12	ZAb3 affibody	L-enantiomer	Selected by phage display technique	 Inhibit Aβ aggregation Disassemble preformed oligomers 	(39)
13	H102 (HKQLPFFEED)	L-enantiomer	CHC-derived sequence	• Inhibit $A\beta_{42}$ fibrillization	(40)
14	Th-NT, Th-CT, and Th-SC	Trehalose conjugated peptides, L-enantiomer	CHC-derived sequence	Inhibit fibril formation	(41)
15	Fc-KLVFF	Ferrocene- conjugated peptide, L-enantiomer	CHC-derived sequence	• Inhibit A β aggregation	(42)
16	N-methylated A $\beta(32-37)$	L-enantiomer	C-terminal fragment	 Inhibits Aβ aggregation Improve Drosophila longevity Increase locomotion 	(43)
17	D-Trp-Aib	D-enantiomer	Rational design	 Inhibits Aß oligomerization Decrease cerebral amyloid deposits in transgenic model Improves cognitive activity (<i>in vivo</i>) 	(44)
18	GABA-FPLIAIMA	D-enantiomer	C-terminal fragment	• Inhibit and reduce the A β aggregation	(22)
19	Tyr(Allyl-RCM)-Xaa-Gly(Allyl-RCM) and 'Gly(Allyl-RCM)-Xaa-Tyr(Allyl-RCM)	cyclic	CHC-derived sequence	- Inhibit significantly the $A\beta_{42}$ aggregation	(45)
20	D(KLVFW)- aminobutyric acid (Aib)	D-enantiomer	CHC-derived sequence	 High specific interaction with Aβ₄₂ monomers and oligomers Remarkable inhibition of A Aβ₄₂ fibril formation No cytotoxic effects 	(46)

affinity for $A\beta$ compared with their L-enantiomeric counterparts. Moreover, D-peptides inhibit A β aggregation in animal models (28). Retro-inverso peptides are a special class of modified peptides that contain D-amino acids and reversed NH and CO groups in the peptide bonds. These peptides could keep the same spatial position in the side chain of the residues and preserve the desirable 3D structure compared to unchanged L-peptides (47). They also displayed advantages in terms of Aβ aggregation inhibition, higher proteolytic stability, lower selfassembly, and better blood-brain-barrier (BBB) permeability when compared with L-peptides in an animal model (48, 49). Fluorinated hydrophobic valine or phenylalanine in the LVFFA-based peptides can considerably delay the formation of A β aggregation. Fluorinated amino acids can also inhibit A β aggregation (50). Modification of amide functional groups with a methyl group is another strategy in the development of new inhibitors. N-methylated amide groups could enhance the peptide's solubility in aqueous solutions and decrease AB-induced toxicity. Cyclic peptides have a higher inhibitory activity than acyclic derivatives (51). Because of their high enzymatic resistance, they are degraded slowly. Residues of lysine and glutamic acid have been known to be effective stabilizing and enhancing agents of A β fibrillation due to their ability to improve surface tension. In contrast, arginine residues have been reported as aggregation inhibitors or destabilizers (chaotropes) (52).

The ¹⁶KLVFF²⁰-based peptides play a crucial role in disrupting AB aggregation by binding to full-length A β peptides and preventing fibril formation (53, 54). Ac-LVFFARK-NH2 (LK7), designed by adding arginine and lysine to KLVFF, induced a dose-dependent inhibition on $A\beta_{42}$ fibrillation; however, it was cytotoxic due to high self-assembling properties (55). When conjugated with poly (lactic-co-glycolic acid) nanoparticles (NPs), the LK7-PLGA-NPs complex resulted in the elimination of the LK7 self-assembly feature while inhibiting A β_{42} fibrillation (55). Binding β -cyclodextrin to LK7 (56) improved LK7 peptide solubility, inhibited its tendency to self-aggregate, improved its binding to A β , and inhibited A β aggregation. Head-to-tail cyclization of LK7 peptide also resulted in a decrease in self-assembly of the LK7, an increase in binding affinity to the A β_{40} peptide, and proteolytic stability in serum. This derivative also can stabilize the A β_{40} secondary structure and inhibit $A\beta_{40}$ –mediated cytotoxicity. Another derivative of LK7 peptide is Ac-LVFFARKHH-NH2 (LK7-HH), in which LK7 has been conjugated to the HH ligand as a chelator for reducing reactive oxygen species (ROS) production and capturing free and complexed ions of Cu^{2+} (57). This chelator also improved the anti-aggregate effects of LK7 on A β peptide and reduced its self-aggregation properties.

Proline and aspartic acid were exchanged for valine and alanine, respectively, in KVLFFA (58, 59). The derived peptide, referred to as 5-mer iA β 5 with sequence LPFFD, inhibited A β aggregation, neurotoxicity, and reduced plaque load (58). Due to the lack of a proton on the secondary substituted nitrogen in the peptide bond of proline residue, it could inhibit the formation of the intramolecular hydrogen bonds into fibrils. Since these small peptides are prone to faster enzymatic degradation and have reduced BBB permeability *in vivo*, iA β 5p was modified by N-methylation between Pro and Phe residues to improve its stability (60). The results from *in vitro* and *in vivo* studies showed that it has the same inhibitory activity as the parental iA β 5 peptide against amyloid fibril formation and neurotoxicity but with improved protease resistance. Also, molecular dynamics

58

simulations show that this peptide has more durable binding and enhanced activity against $A\beta_{40}$ aggregation in comparison to the iA β 5 peptide. In a similar study, the RIVFF sequence was produced by residue mutations of lysine16 (K) to arginine (R) and leucine17 (L) to isoleucine (I) on the KLVFF segment (61). The results indicated that this peptide could self-aggregate into β -sheet structures by reducing the surface tension of water and at higher concentrations (>250 μ M) enhanced the A β -induced cytotoxicity.

The peptide D-GRKKRRQRRR-GGGG-DVEFRH ($A\beta_{1-6}$ A2V-TAT) was investigated *in vivo* (62, 63). It was generated by modifying the N-terminal fragment of ¹DAEFRH⁶ through mutation of alanine in position 2 to valine and conjugating with the HIV protein transduction domain GRKKRRQRRR (TAT). The resulting peptide showed strong anti-amyloidogenic effects *in vitro* and A β aggregation inhibition in mouse models of Alzheimer's disease (64). The KLVFWAK motif was designed based on the ¹⁶KLVFFAE²² sequence with mutations introduced at phenylalanine and glutamic acid residues to tryptophan and lysine respectively to enhance solubility and disrupt self-assembly via electrostatic repulsion. Results showed that the designed motif could only target the C-terminal region of A β oligomers. The designed motif exhibited a lower self-aggregation tendency in comparison to other KLVFF-related sequences. Moreover, it demonstrated a higher binding affinity to A β aggregates and fibrils than monomers (65).

RGKLVFFGR (OR1) and RGKLVFFGR-NH (OR2) are retro-inverso peptides (66), designed by the addition of arginine (R) and glycine (G) to the KLVFF sequence. They exhibit high solubility and stability against enzymes. However, only the OR2 peptide showed inhibitory effects on A β oligomer formation and cytotoxicity. OR2 was modified to HN-rGklvffGr-Ac (RI-OR2) by acetylation of the C-terminal residue (49). The result illustrated that the peptide has a high resistance to proteolysis, while maintaining the same inhibitory activity *in vivo*. In a follow-up study, the RI-OR2 peptide was attached to the TAT peptide to improve its permeability into cells and the BBB (48). The results showed the peptide was able to decrease A β aggregation, plaque levels, and oxidative damages as well as increase the number of young neurons in the brain.

³¹IIGLMVGGVVIA⁴² and ³⁹VVIA⁴² sequences were designed based on the C-terminal domain of A β_{42} (67). The ³⁹VVIA⁴² sequences could interact with A β_{42} monomers and smaller oligomers at several sites, specifically at the N-terminal domain. At micromolar concentrations, the VVIA-NH₂ peptide inhibited A β_{42} aggregation, exhibited less toxicity, and protected synaptic activity. However, these effects were not observed for the acetylated Ac-VVIA sequence (68). The non-acetylated VVIA-NH₂ sequence particularly interacts with the C-terminal domain while the Ac-VVIA peptide has a dispersed binding distribution (68). The Ac-³²IGLMVG³⁷-NH₂ sequence, a hexapeptide from the C-terminal fragment, has been shown to have a moderate efficacy with less toxicity (69).

O-acyl isopeptide and NMe-b-Ala26 (70) were derived from the full-length A β sequence with modification of an ester bond at the Gly25-Ser26 moiety and an N-methyl amide- β -Ala26, respectively. O-acyl isopeptide inhibited A β_{42} fibrillation at equimolar concentrations through an inhibitory mechanism distinct from any other peptidic inhibitors reported previously. Also, this derivative was more soluble than A β_{42} peptides and rapidly decomposed to A β_{42} monomers under physiological conditions through an O-to-N acyl rearrangement reaction whereas NMe-b-Ala26 showed higher chemical stability at physiological conditions.
Carnosine, a natural imidazole-containing dipeptide is a metal ion chelator (71). It inhibits the fibrillation and toxicity of amyloidogenic species such as glycated α -Crystallin, A β peptide, and prions. This peptide also inhibits the intramolecular salt bridge formation, which is vital to the stability and elongation of fibrils (71). Peptide D1, QSHYRHISPAQV (72), is another non-A β peptide that reduces A β aggregation and A β -associated cytotoxicity at high concentrations. N-methylated proprietary peptides such as D-NH₂ (SEN304) and SEN1576 can inhibit A β -associated toxicity *in vivo* (73). Furthermore, SEN304 is a more potent inhibitor than customized versions of the KLVFF peptide. These peptides could interfere with the nucleation of A β , convert them into non-toxic forms, and eliminate toxic oligomers.

PEPTIDE LIBRARY SCREENING

There are many screening approaches to identify target-specific ligands (74, 75). Phage display is one such efficient high-throughput screening method that allows the screening of a wide variety of peptide libraries to identify specific peptide sequences against the desired target (76, 77). Wang et al. synthesized a linear peptide with sequence PYRWQLWWHNWS selected based on the screening of a randomized 12-mer peptide library against the target $A\beta_{1-10}$ sequence (78). After screening, specific phages were selected and their binding affinity to $A\beta_{1-10}$ was evaluated by real-time biomolecular interaction analysis. This peptide could specifically bind to $A\beta_{1-10}$, inhibit the aggregation of $A\beta$ into plaques, and reduce $A\beta_{1-42}$ induced apoptosis. Furthermore, it illustrated a protective effect against $A\beta_{1-42}$ -induced memory and learning impairments in animal models (59).

Larbanoix et al. utilized the phage display method to discover a linear hexapeptide against $A\beta_{1-42}$ aggregation (79). Two of the selected clones, Pep1: LIAIMA and Pep2: IFALMG, corresponding to fragment ³¹IIGLMV³⁶ from $A\beta_{1-42}$ peptide, demonstrated the highest binding affinities to $A\beta_{1-42}$ with Kd values in the micromolar range. Their specific interactions with $A\beta_{1-42}$ plaques were identified by immunohistochemistry on harvested brain tissue from an animal model of Alzheimer's disease. The peptides did not induce any toxicity in neurons *in vitro*. Moreover, the thioflavin T aggregation assay indicated that the designed peptides could suppress the amyloid fibril formation.

In 2010, a random heptapeptide library (XX–P–XXXX) on T7 phage was reported by Kawasaki et al. (80). The library was designed based on the LPFFD sequence XX-P-XXXX, where P is proline, and X is any amino acid. After the fifthround of biopanning against $A\beta_{1-42}$ soluble oligomers, eight new peptides containing arginine residues were obtained. The peptide with the strongest affinity to $A\beta$ (RGPRGRV) suppressed the formation of 37–48 kDa oligomers and maintained the monomeric form of $A\beta_{1-42}$ for up to 24 h. In follow-up studies, to assess the effect of the peptide length on the inhibition of soluble oligomers formation, random libraries containing 3-residue and 4-residue peptides were prepared by phage display and evaluated. The results demonstrated that the 3-residue peptides could not significantly inhibit oligomers formation because of their **60**

smaller size. In contrast, the 4-residue peptide with the RFRK sequence inhibited the soluble oligomer formation like the heptapeptide (RGPRGRV). It also showed a slight decrease on A β fibrillation (81), similar to the inhibitory activity of the N-Methylated Peptide (SEN304), against A β_{42} aggregation (25, 82, 83). Tsuji-Ueno et al. utilized the all-steps-all-combinations (ASAC) method to explore $A\beta_{42}$ -binding peptide aptamers. The identified peptides from the primary and secondary libraries showed a weak binding affinity to $A\beta_{42}$ (Kd values in the μ m range) (84). To further improve the peptide aptamers, Gautam et al. applied the mRNA display technique and paired-peptide library method. The library was assembled by a random shuffling method on selected peptide blocks taken from the formed primary and secondary peptide libraries by Tsuji-Ueno et al. (84). They reported two peptides with high binding affinity to $A\beta_{42}$ (Kd in the nM range) which significantly inhibited the A β_{42} aggregation (85). The improved P84 (CGILDPIPWGGSGGSCGILDPIPW) aptamers. and P131 peptide (GCPCIGIIGGSGGSDCSSDLTPS), where GGSGGS is the linker sequence, demonstrated a higher binding affinity for the $A\beta_{42}$ peptide (Kd values in the nanomolar range) compared to the primary and secondary $A\beta_{42}$ -binding peptides (86). The results showed that both peptides could inhibit the $A\beta_{42}$ aggregation and result in the reduction of the cytotoxic effects of A β_{42} fibrils and A β_{42} oligomers in PC12 cells; P84 showed better efficacy than P131 on the cell line.

Groen and co-workers employed mirror-image phage display to identify selective and high-affinity D-peptide ligands for A β_{1-42} . The D-enantiomer A β_{1-42} was used as a target for selection from a randomized 12- amino acid peptide library with more than 1 billion different peptides. After six rounds of biopanning, they identified a specific D-enantiomeric peptide, RPRTRLHTHRNR, called D3 (73). The D3 ligand inhibited A β aggregation, and dissolved pre-formed A β fibrils. Additionally, D3 ligand could disaggregate pre-existing amyloid plaques in the brain and result in an increase in the amount of $A\beta$ monomeric form, which has high clearance from the brain (87). FITC fluorescence data demonstrated that A β –D3 clearance might have been associated with pericytes, which have a major role in the clearance of different $A\beta_{40/42}$ species (88, 89). Glial fibrillary acidic protein (GFAP) staining of astrocytes and CD11b staining of microglia in brain sections revealed that the D3 significantly decreased the amount of plaque-related inflammation markers (active astrocytes and microglia) around the A β plaques in comparison to the untreated animals. In addition to the anti-inflammatory properties, this peptide ligand could drastically reduce the A β plaque load in brain tissue of transgenic APP-PSD mice after a 30-day treatment with administration of 9 mg D3 per day per mouse. Computational simulation studies demonstrated strong electrostatic interactions between the arginine-rich D3 and negatively charged groups of A β nonamer; D3 binding to A β nonamer could change the topology of the A β oligomers by inducing a twist in them and consequently promote the formation of A β nonfibrillar aggregations (73, 90, 91).

Luo et al. applied peptoid chemistry, N-substituted glycine oligomers as a class of peptidomimetics, to develop and improve selective high-affinity ligands for $A\beta_{42}$ (92). They constructed an on-bead peptoid library of 38,416 unique peptoids. After screening for $A\beta_{42}$ -selective peptoid ligands, the IAM1 ligand and its dimeric form were selected and further evaluated. IAM1 peptide showed about 10-fold more affinity for $A\beta_{42}$ -binding than for $A\beta_{40}$, and inhibited $A\beta_{42}$ aggregation *in vitro*. The dimeric derivative (IAM1)₂ demonstrated a 7.4-fold higher

affinity for A β_{42} (60 nM) than the monomeric form. Moreover, (IAM1)₂ demonstrated neuroprotective effects on primary hippocampal neurons against A β_{42} -induced toxicity.

Due to the considerable similarities between the self-assembly of cyclic d,l- α -peptides and amyloid structures, it is possible such peptides can bind to A β non-toxic forms and stabilize them (29). Richman et al. described the cyclic peptide CP-2, cyclo-[l–J–w– H–s–K]s (J denotes l-norleucine), by screening a 6-residue library of head-to-tail cyclic d,l- α -peptides consisting of residues Lys, Glu, Ser, Leu, Trp, and His using a one-bead-one-peptide combinatorial approach (29, 93). The selected peptide strongly interacted with A $\beta_{40}/A\beta_{42}$ and prevented their assembly, entirely disassembled A β_{40} fibrils, and protected PC12 cells against A $\beta_{40}/A\beta_{42}$ -induced toxicity, without having any toxic effects of its own. NMR spectroscopy revealed that the CP-2 peptide, in a self-assembled form, interacted with monomeric and low-oligomeric structures of A β_{40} and induced weak α -helix structures during the initial stage of A β_{40} aggregation and subsequently promoted the conformational transition shift from a more toxic antiparallel β -sheet conformation to the less toxic parallel β -sheet.

In another study, Acerra et al. utilized an intracellular protein-fragment complementation assay (PCA) methodology for the screening of selective high-affinity peptides to A β (94). The A β_{25-35} sequence, known to self-assemble into toxic fibrils (95), was inserted into one half of the murine dihydrofolate reductase enzyme as a target, and the A β 29–35 sequence-based peptide was inserted on the other half of the enzyme (96). After the screening of primary and secondary libraries, two new targeting peptides L2P1, FSKATSN, and L2P2, PVKATTA were selected. These peptides shared no homology with the starting template A β_{29-35} . The results showed that all selected peptides could bind A β_{42} , inhibit fibril formation, and disaggregate pre-formed fibrils. To further improve the metabolic stability of selected peptides from primary and secondary libraries, their retro-inverso (RI) analogs were evaluated (86). All RI peptide ligands, such as KAR-R1, L2P1a-RI, L2P1b-RI, L2P2a-RI, and L2P2b-RI, inhibited A β fibrillation and disaggregated pre-formed fibrils, and reduced A β_{42} -induced toxicity in PC-12 cells.

THE CURRENT STATE AND FUTURE DIRECTIONS OF Aβ INHIBITORY PEPTIDES IN ALZHEIMER'S DISEASE

A wide range of peptide-based inhibitors has been evaluated in cellular and animal models as new therapeutic compounds for inhibition of A β aggregation. While experimental studies generated promising results, only a few of these inhibitory peptides have been successful enough to enter clinical trials. NAP or Davunetide peptide with NAPVSIPQ sequence, derived from the activitydependent neuroprotective protein (ADNP), was reported in 2003 by Gozes et al. (97). NAP was able to inhibit A β aggregation, disassemble pre-formed fibrils, and protect the neuronal cells from A β induced toxicity. Though NAP demonstrated benefits in phase II clinical trials for mild cognitive impairment, it failed in a phase III trial (98–100). PPI-1019 peptide (APAN), with a sequence of D-(H-[(Me-L)-VFFL]NH₂), is an N-methylated peptide which is derived from the D-enantiomeric Cholyl-LVFFA-NH₂ that could inhibit A β aggregation and the induce toxicity in experimental studies (101). The phases I and II clinical trials of APAN was completed in patients with mild-moderate Alzheimer's disease in 2005 (<u>NCT00100282</u>, <u>NCT00100334</u>), but the outcome of this study is still unknown (https://clinical-trials.gov/, last assessed 28 October 2020). Other reported inhibitory peptides including, D3 (102), D-Trp-Aib–OH (44), D-4F (36), TAT-R1-OR2 (48), NL-RI-OR2-TAT90 (103), and R1-OR2 (49) have shown considerable efficacy in preclinical trials, but they have not yet entered clinical trials.

The reality is that, to date, there is no cure for Alzheimer's disease. Only optimism remains. Therefore, it is necessary to discover potential peptides for testing in clinical trials. The current inhibitory peptides have certain limitations such as poor BBB permeability and high cytotoxicity. To overcome these problems and further improve the inhibitory activity, a number of studies have focused on peptide–nanostructure conjugates (PNCs) approach that provides an opportunity to increase the capabilities of both these classes of materials (55, 104, 105). Nanostructures could be considered as a potential vehicle to overcome poor BBB permeability and bring hope for neurodegenerative diseases therapy due to their size and various surface modifications (106). As an example, multivalent inhibitors can be developed against A β aggregation by decorating gold nanoparticles with VVIA and LPFFD (107). The PNCs approach gives a fascinating insight into the fields of diagnosis and treatment, and provides new opportunities for the design of high-performance peptides (108, 109).

CONCLUSION

Despite a better understanding of the pathogenic mechanisms of Alzheimer's disease, finding efficient therapeutic compounds to prevent or halt the progression of Alzheimer's disease continues to be a challenge. A β aggregation inhibition-based approaches are being developed with the aim to stop disease progression. While the reported inhibitory peptides have considerable advantages over other compounds, and experimental evidence has been encouraging, bench-to-bedside has not yet become a reality. Therefore, adequate knowledge of binding interactions of these peptides with their biological targets, the ligand-target complex, is required to design more accurate therapeutic biomolecules. Peptide inhibitors have unique properties, particularly, high selectivity, low accumulation in tissues, low side-effects and toxicity, and different chemical and biological synthesis routes when compared with other compounds. As researchers continue to focus on rational design, characterization, optimization, and interaction between the inhibitor and the A β peptide complex, more peptide inhibitors are expected to succeed in clinical trials.

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this chapter.

Copyright and Permission Statement: We confirm that the materials included in this chapter do not violate copyright laws. Where relevant, appropriate permissions have been obtained from the original copyright holder(s). All original sources have been appropriately acknowledged and/or referenced.

REFERENCES

- 1. 2009 Alzheimer's disease facts and figures. Alzheimer's & Dementia. 2009;5(3):234–70. https://doi. org/10.1016/j.jalz.2009.03.001
- 2018 Alzheimer's disease facts and figures. Alzheimer's Dementia. 2018;14(3):367–429. https://doi. org/10.1016/j.jalz.2018.02.001
- 3. Prince M, Comas-Herrera A, Knapp M, Guerchet M, Karagiannidou M. World Alzheimer report 2016: improving healthcare for people living with dementia: coverage, quality and costs now and in the future. Alzheimer's Disease International (ADI); London. 2016.
- Bartus RT, Emerich DF. Cholinergic markers in Alzheimer disease. JAMA. 1999;282(23):2208–9. https://doi.org/10.1001/jama.282.23.2208
- Mandelkow E-M, Mandelkow E. Tau in Alzheimer's disease. Trends Cell Biol. 1998;8(11):425–7. https://doi.org/10.1016/S0962-8924(98)01368-3
- Finder VH, Glockshuber R. Amyloid-β aggregation. Neurodegen Dis. 2007;4(1):13–27. https://doi. org/10.1159/000100355
- Hardy J, Selkoe DJ. The amyloid hypothesis of Alzheimer's disease: Progress and problems on the road to therapeutics. Science. 2002;297(5580):353–6. https://doi.org/10.1126/science.1072994
- Singh M, Singh S, Prasad S, Gambhir I. Nanotechnology in medicine and antibacterial effect of silver nanoparticles. Dig J Nanomater Biostruct. 2008;3(3):115–22.
- Haass C, Selkoe DJ. Cellular processing of β-amyloid precursor protein and the genesis of amyloid β-peptide. Cell. 1993;75(6):1039–42. https://doi.org/10.1016/0092-8674(93)90312-E
- Hall D, Edskes H. A model of amyloid's role in disease based on fibril fracture. Biophys Chem. 2009;145(1):17–28. https://doi.org/10.1016/j.bpc.2009.08.004
- Hall D, Edskes H. Computational modeling of the relationship between amyloid and disease. Biophys Rev. 2012;4(3):205–22. https://doi.org/10.1007/s12551-012-0091-x
- Liu H, Dong X, Liu F, Zheng J, Sun Y. Iminodiacetic acid-conjugated nanoparticles as a bifunctional modulator against Zn2+-mediated amyloid β-protein aggregation and cytotoxicity. J Colloid Interf Sci. 2017;505:973–82. https://doi.org/10.1016/j.jcis.2017.06.093
- Wang W, Han Y, Fan Y, Wang Y. Effects of gold nanospheres and nanocubes on amyloid-β peptide fibrillation. Langmuir. 2019;35(6):2334–42. https://doi.org/10.1021/acs.langmuir.8b04006
- Brahmachari S, Paul A, Segal D, Gazit E. Inhibition of amyloid oligomerization into different supramolecular architectures by small molecules: Mechanistic insights and design rules. Fut Med Chem. 2017;9(8):797–810. https://doi.org/10.4155/fmc-2017-0026
- Jokar S, Khazaei S, Behnammanesh H, Shamloo A, Erfani M, Beiki D, et al. Recent advances in the design and applications of amyloid-β peptide aggregation inhibitors for Alzheimer's disease therapy. Biophys Rev. 2019;11:1–25. https://doi.org/10.1007/s12551-019-00606-2
- Rahimi F, Li H, Sinha S, Bitan G. Modulators of amyloid β-protein (Aβ) self-assembly. Developing Therapeutics for Alzheimer's Disease. Elsevier; Boston. 2016. p. 97–191. https://doi.org/10.1016/ B978-0-12-802173-6.00006-X
- Danho W, Swistok J, Khan W, Chu X-J, Cheung A, Fry D, et al. Opportunities and challenges of developing peptide drugs in the pharmaceutical industry. Peptides for Youth. Springer; New York. 2009. p. 467–9. https://doi.org/10.1007/978-0-387-73657-0_201
- Bruno BJ, Miller GD, Lim CS. Basics and recent advances in peptide and protein drug delivery. Therap Deliv. 2013;4(11):1443–67. https://doi.org/10.4155/tde.13.104
- Goyal D, Shuaib S, Mann S, Goyal B. Rationally designed peptides and peptidomimetics as inhibitors of amyloid-β (Aβ) aggregation: Potential therapeutics of Alzheimer's disease. ACS Combinator Sci. 2017;19(2):55–80. https://doi.org/10.1021/acscombsci.6b00116
- Jokar S, Behnammanesh H, Erfani M, Sharifzadeh M, Gholami M, Sabzevari O, et al. Synthesis, biological evaluation and preclinical study of a novel 99mTc-peptide: A targeting probe of amyloid-β plaques as a possible diagnostic agent for Alzheimer's disease. Bioorgan Chem. 2020;99:103857. https://doi.org/10.1016/j.bioorg.2020.103857
- Lührs T, Ritter C, Adrian M, Riek-Loher D, Bohrmann B, Döbeli H, et al. 3D structure of Alzheimer's amyloid-β (1–42) fibrils. Proc Natl Acad Sci. 2005;102(48):17342–7. https://doi.org/10.1073/ pnas.0506723102

- Jokar S, Erfani M, Bavi O, Khazaei S, Sharifzadeh M, Hajiramezanali M, et al. Design of peptide-based inhibitor agent against amyloid-β aggregation: Molecular docking, synthesis and in vitro evaluation. Bioorgan Chem. 2020;102:104050. https://doi.org/10.1016/j.bioorg.2020.104050
- Reddy G, Straub JE, Thirumalai D. Influence of preformed asp23– lys28 salt bridge on the conformational fluctuations of monomers and dimers of Aβ peptides with implications for rates of fibril formation. J Phys Chem B. 2009;113(4):1162–72. https://doi.org/10.1021/jp808914c
- Friedemann M, Helk E, Tiiman A, Zovo K, Palumaa P, Tõugu V. Effect of methionine-35 oxidation on the aggregation of amyloid-β peptide. Biochem Biophys Rep. 2015;3:94–9. https://doi.org/10.1016/j. bbrep.2015.07.017
- Li H, Du Z, Lopes DH, Fradinger EA, Wang C, Bitan G. C-terminal tetrapeptides inhibit Aβ42induced neurotoxicity primarily through specific interaction at the N-terminus of Aβ42. J Med Chem. 2011;54(24):8451–60. https://doi.org/10.1021/jm200982p
- Urbanc B, Cruz L, Yun S, Buldyrev SV, Bitan G, Teplow DB, et al. In silico study of amyloid β-protein folding and oligomerization. Proc Natl Acad Sci. 2004;101(50):17345–50. https://doi.org/10.1073/ pnas.0408153101
- 27. Liu J, Wang W, Zhang Q, Zhang S, Yuan Z. Study on the efficiency and interaction mechanism of a decapeptide inhibitor of β -amyloid aggregation. Biomacromolecules. 2014;15(3):931–9. https://doi.org/10.1021/bm401795e
- Jagota S, Rajadas J. Synthesis of d-amino acid peptides and their effect on beta-amyloid aggregation and toxicity in transgenic Caenorhabditis elegans. Med Chem Res. 2013;22(8):3991–4000. https:// doi.org/10.1007/s00044-012-0386-2
- Richman M, Wilk S, Chemerovski M, Wärmländer SK, Wahlström A, Gräslund A, et al. In vitro and mechanistic studies of an antiamyloidogenic self-assembled cyclic d, l-α-peptide architecture. J Am Chem Soc. 2013;135(9):3474–84. https://doi.org/10.1021/ja310064v
- Brener O, Dunkelmann T, Gremer L, Van Groen T, Mirecka EA, Kadish I, et al. QIAD assay for quantitating a compound's efficacy in elimination of toxic Aβ oligomers. Sci Rep. 2015;5:13222. https://doi.org/10.1038/srep13222
- Leithold LH, Jiang N, Post J, Niemietz N, Schartmann E, Ziehm T, et al. Pharmacokinetic properties of tandem d-peptides designed for treatment of Alzheimer's disease. Eur J Pharmaceut Sci. 2016;89: 31–8. https://doi.org/10.1016/j.ejps.2016.04.016
- Klein AN, Ziehm T, Tusche M, Buitenhuis J, Bartnik D, Boeddrich A, et al. Optimization of the all-D peptide D3 for Aβ oligomer elimination. PLoS One. 2016;11(4):e0153035. https://doi.org/10.1371/ journal.pone.0153035
- Zhang YX, Wang SW, Lu S, Zhang LX, Liu DQ, Ji M, et al. A mimotope of Aβ oligomers may also behave as a β-sheet inhibitor. FEBS Lett. 2017;591(21):3615–24. https://doi.org/10.1002/1873-3468.12871
- Granic I, Masman MF, Nijholt IM, Naude PJ, de Haan A, Borbély E, et al. LPYFDa neutralizes amyloidβ-induced memory impairment and toxicity. J Alzheimers Dis. 2010;19(3):991–1005. https://doi. org/10.3233/JAD-2010-1297
- 35. Minicozzi V, Chiaraluce R, Consalvi V, Giordano C, Narcisi C, Punzi P, et al. Computational and experimental studies on β -sheet breakers targeting A β 1–40 fibrils. J Biol Chem. 2014;289(16):11242–52. https://doi.org/10.1074/jbc.M113.537472
- 36. Handattu SP, Garber DW, Monroe CE, van Groen T, Kadish I, Nayyar G, et al. Oral apolipoprotein AI mimetic peptide improves cognitive function and reduces amyloid burden in a mouse model of Alzheimer's disease. Neurobiol Dis. 2009;34(3):525–34. https://doi.org/10.1016/j.nbd.2009.03.007
- Kino R, Araya T, Arai T, Sohma Y, Kanai M. Covalent modifier-type aggregation inhibitor of amyloid-β based on a cyclo-KLVFF motif. Bioorgan Med Chem Lett. 2015;25(15):2972–5. https://doi. org/10.1016/j.bmcl.2015.05.027
- Sohma Y. Medicinal chemistry focusing on aggregation of amyloid-β. Chem Pharmaceut Bull. 2016; 64(1):1–7. https://doi.org/10.1248/cpb.c15-00742
- Luheshi LM, Hoyer W, de Barros TP, van Dijk Härd I, Brorsson A-C, Macao B, et al. Sequestration of the Aβ peptide prevents toxicity and promotes degradation in vivo. PLoS Biol. 2010;8(3):e1000334.
- 40. Lin L-X, Bo X-Y, Tan Y-Z, Sun F-X, Song M, Zhao J, et al. Feasibility of β -sheet breaker peptide-H102 treatment for Alzheimer's disease based on β -amyloid hypothesis. PLoS One. 2014;9(11):e112052. https://doi.org/10.1371/journal.pone.0112052

- Sinopoli A, Giuffrida A, Tomasello MF, Giuffrida ML, Leone M, Attanasio F, et al. Ac-LPFFD-Th: A trehalose-conjugated peptidomimetic as a strong suppressor of amyloid-β oligomer formation and cytotoxicity. ChemBioChem. 2016;17(16):1541–9. https://doi.org/10.1002/cbic.201600243
- 42. Wei C-W, Peng Y, Zhang L, Huang Q, Cheng M, Liu Y-N, et al. Synthesis and evaluation of ferrocenoyl pentapeptide (Fc-KLVFF) as an inhibitor of Alzheimer's Aβ1–42 fibril formation in vitro. Bioorgan Med Chem Lett. 2011;21(19):5818–21. https://doi.org/10.1016/j.bmcl.2011.07.111
- Pratim Bose P, Chatterjee U, Nerelius C, Govender T, Norstrom T, Gogoll A, et al. Poly-N-methylated amyloid β-peptide (Aβ) C-terminal fragments reduce Aβ toxicity in vitro and in Drosophila melanogaster. J Med Chem. 2009;52(24):8002–9. https://doi.org/10.1021/jm901092h
- 44. Frydman-Marom A, Rechter M, Shefler I, Bram Y, Shalev DE, Gazit E. Cognitive-performance recovery of Alzheimer's disease model mice by modulation of early soluble amyloidal assemblies. Angewandte Chem. 2009;121(11):2015–20. https://doi.org/10.1002/ange.200802123
- 45. Tanaka F, Shibata K, Monobe Y, Akagi K-I, Masuda Y. Design and synthesis of β-strand-fixed peptides inhibiting aggregation of amyloid β-protein. Bioorgan Med Chem. 2020;28(18):115676. https://doi. org/10.1016/j.bmc.2020.115676
- 46. Horsley JR, Jovcevski B, Wegener KL, Yu J, Pukala TL, Abell AD. Rationally designed peptide-based inhibitor of Aβ42 fibril formation and toxicity: A potential therapeutic strategy for Alzheimer's disease. Biochem J. 2020;477(11):2039–54. https://doi.org/10.1042/BCJ20200290
- Chorev M, Goodman M. Recent developments in retro peptides and proteins An ongoing topochemical exploration. Trends Biotechnol. 1995;13(10):438–45. https://doi.org/10.1016/ S0167-7799(00)88999-4
- Parthsarathy V, McClean PL, Hölscher C, Taylor M, Tinker C, Jones G, et al. A novel retro-inverso peptide inhibitor reduces amyloid deposition, oxidation and inflammation and stimulates neurogenesis in the APPswe/PS1∆E9 mouse model of Alzheimer's disease. PLoS One. 2013;8(1):e54769. https:// doi.org/10.1371/journal.pone.0054769
- 49. Taylor M, Moore S, Mayes J, Parkin E, Beeg M, Canovi M, et al. Development of a proteolytically stable retro-inverso peptide inhibitor of β-amyloid oligomerization as a potential novel treatment for Alzheimer's disease. Biochemistry. 2010;49(15):3261–72. https://doi.org/10.1021/bi100144m
- 50. Loureiro JA, Crespo R, Börner H, Martins PM, Rocha FA, Coelho M, et al. Fluorinated beta-sheet breaker peptides. J Mater Chem B. 2014;2(16):2259–64. https://doi.org/10.1039/C3TB21483D
- Luo J, Abrahams JP. Cyclic peptides as inhibitors of amyloid fibrillation. Chem A Eur J. 2014;20(9):2410–19. https://doi.org/10.1002/chem.201304253
- 52. Han X, He G. Toward a rational design to regulate β-amyloid fibrillation for alzheimer's disease treatment. ACS Chem Neurosci. 2018;9(2):198–210. https://doi.org/10.1021/acschemneuro.7b00477
- Tjernberg LO, Näslund J, Lindqvist F, Johansson J, Karlström AR, Thyberg J, et al. Arrest of-amyloid fibril formation by a pentapeptide ligand. J Biol Chem. 1996;271(15):8545–8. https://doi. org/10.1074/jbc.271.15.8545
- Matsunaga Y, Fujii A, Awasthi A, Yokotani J, Takakura T, Yamada T. Eight-residue Aβ peptides inhibit the aggregation and enzymatic activity of Aβ42. Regul Peptides. 2004;120(1–3):227–36. https://doi. org/10.1016/j.regpep.2004.03.013
- 55. Xiong N, Dong X-Y, Zheng J, Liu F-F, Sun Y. Design of LVFFARK and LVFFARK-functionalized nanoparticles for inhibiting amyloid β-protein fibrillation and cytotoxicity. ACS Appl Mater Interf. 2015;7(10):5650–62. https://doi.org/10.1021/acsami.5b00915
- Zhang H, Dong X, Liu F, Zheng J, Sun Y. Ac-LVFFARK-NH2 conjugation to β-cyclodextrin exhibits significantly enhanced performance on inhibiting amyloid β-protein fibrillogenesis and cytotoxicity. Biophys Chem. 2018;235:40–7. https://doi.org/10.1016/j.bpc.2018.02.002
- Zhang H, Zhang C, Dong XY, Zheng J, Sun Y. Design of nonapeptide LVFFARKHH: A bifunctional agent against Cu2+-mediated amyloid β-protein aggregation and cytotoxicity. J Mol Recogn. 2018;31(6):e2697. https://doi.org/10.1002/jmr.2697
- Soto C, Sigurdsson EM, Morelli L, Kumar RA, Castaño EM, Frangione B. β-sheet breaker peptides inhibit fibrillogenesis in a rat brain model of amyloidosis: Implications for Alzheimer's therapy. Nat Med. 1998;4(7):822–6. https://doi.org/10.1038/nm0798-822
- 59. Wood SJ, Wetzel R, Martin JD, Hurle MR. Prolines and aamyloidogenicity in fragments of the Alzheimer's peptide. beta./A4. Biochemistry. 1995;34(3):724–30. https://doi.org/10.1021/bi00003a003

- Cruz M, Tusell J, Grillo-Bosch D, Albericio F, Serratosa J, Rabanal F, et al. Inhibition of β-amyloid toxicity by short peptides containing N-methyl amino acids. J Peptide Res. 2004;63(3):324–8. https:// doi.org/10.1111/j.1399-3011.2004.00156.x
- Ramaswamy K, Kumaraswamy P, Sethuraman S, Krishnan UM. Self-assembly characteristics of a structural analogue of Tjernberg peptide. RSC Adv. 2014;4(32):16517–23. https://doi.org/10.1039/ C3RA47754A
- 62. Cimini S, Sclip A, Mancini S, Colombo L, Messa M, Cagnotto A, et al. The cell-permeable Aβ1-6A2VTAT (D) peptide reverts synaptopathy induced by Aβ1-42wt. Neurobiol Dis. 2016;89:101–11. https://doi.org/10.1016/j.nbd.2015.12.013
- 63. Di Fede G, Catania M, Morbin M, Rossi G, Suardi S, Mazzoleni G, et al. A recessive mutation in the APP gene with dominant-negative effect on amyloidogenesis. Science. 2009;323(5920):1473–7. https://doi.org/10.1126/science.1168979
- Di Fede G, Catania M, Maderna E, Morbin M, Moda F, Colombo L, et al. Tackling amyloidogenesis in Alzheimer's disease with A2V variants of Amyloid-β. Sci Rep. 2016;6:20949. https://doi.org/10.1038/ srep20949
- 65. Aoraha E, Candreva J, Kim JR. Engineering of a peptide probe for β-amyloid aggregates. Mol BioSyst. 2015;11(8):2281–9. https://doi.org/10.1039/C5MB00280J
- Austen BM, Paleologou KE, Ali SA, Qureshi MM, Allsop D, El-Agnaf OM. Designing peptide inhibitors for oligomerization and toxicity of Alzheimer's β-amyloid peptide. Biochemistry. 2008;47(7):1984– 92. https://doi.org/10.1021/bi701415b
- Fradinger EA, Monien BH, Urbanc B, Lomakin A, Tan M, Li H, et al. C-terminal peptides coassemble into Aβ42 oligomers and protect neurons against Aβ42-induced neurotoxicity. Proc Natl Acad Sci. 2008;105(37):14175–80. https://doi.org/10.1073/pnas.0807163105
- 68. Zheng X, Wu C, Liu D, Li H, Bitan G, Shea J-E, et al. Mechanism of C-terminal fragments of amyloid β-protein as Aβ inhibitors: Do C-terminal interactions play a key role in their inhibitory activity? J Phys Chem B. 2016;120(8):1615–23. https://doi.org/10.1021/acs.jpcb.5b08177
- Bansal S, Maurya IK, Yadav N, Thota CK, Kumar V, Tikoo K, et al. C-terminal fragment, Aβ32–37, analogues protect against aβ aggregation-induced toxicity. ACS Chem Neurosci. 2016;7(5):615–23. https://doi.org/10.1021/acschemneuro.6b00006
- Kawashima H, Sohma Y, Nakanishi T, Kitamura H, Mukai H, Yamashita M, et al. A new class of aggregation inhibitor of amyloid-β peptide based on an O-acyl isopeptide. Bioorgan Med Chem. 2013;21(21):6323–7. https://doi.org/10.1016/j.bmc.2013.08.062
- Aloisi A, Barca A, Romano A, Guerrieri S, Storelli C, Rinaldi R, et al. Anti-aggregating effect of the naturally occurring dipeptide carnosine on aβ1-42 fibril formation. PLoS One. 2013;8(7):e68159. https://doi.org/10.1371/journal.pone.0068159
- 72. Wiesehan K, Buder K, Linke RP, Patt S, Stoldt M, Unger E, et al. Selection of D-amino-acid peptides that bind to Alzheimer's disease amyloid peptide Aβ1–42 by mirror image phage display. Chembiochem. 2003;4(8):748–53. https://doi.org/10.1002/cbic.200300631
- 73. Van Groen T, Wiesehan K, Funke SA, Kadish I, Nagel-Steger L, Willbold D. Reduction of Alzheimer's disease amyloid plaque load in transgenic mice by D3, A D-enantiomeric peptide identified by mirror image phage display. ChemMedChem. 2008;3(12):1848. https://doi.org/10.1002/cmdc.200800273
- 74. Biyani M, Nishigaki K, Biyani M. Biomolecular display technology: A new tool for drug discovery. Animal Biotechnology: Elsevier, 2014. p. 369–84. https://doi.org/10.1016/B978-0-12-416002-6.00020-1
- Zwick MB, Shen J, Scott JK. Phage-displayed peptide libraries. Curr Opin Biotechnol. 1998;9(4): 427–36. https://doi.org/10.1016/S0958-1669(98)80017-7
- 76. Kang CK, Jayasinha V, Martin PT. Identification of peptides that specifically bind Aβ1–40 amyloid in vitro and amyloid plaques in Alzheimer's disease brain using phage display. Neurobiol Dis. 2003;14(1):146–56. https://doi.org/10.1016/S0969-9961(03)00105-0
- 77. Smith GP. Filamentous fusion phage: Novel expression vectors that display cloned antigens on the virion surface. Science. 1985;228(4705):1315–17. https://doi.org/10.1126/science.4001944
- 78. Wang F, Zhou X-L, Yang Q-G, Xu W-H, Wang F, Chen Y-P, et al. A peptide that binds specifically to the β-amyloid of Alzheimer's disease: Selection and assessment of anti-β-amyloid neurotoxic effects. PLoS One. 2011;6(11):e27649. https://doi.org/10.1371/journal.pone.0027649

- 79. Larbanoix L, Burtea C, Ansciaux E, Laurent S, Mahieu I, Vander Elst L, et al. Design and evaluation of a 6-mer amyloid-beta protein derived phage display library for molecular targeting of amyloid plaques in Alzheimer's disease: Comparison with two cyclic heptapeptides derived from a randomized phage display library. Peptides. 2011;32(6):1232–43. https://doi.org/10.1016/j.peptides.2011.04.026
- Kawasaki T, Onodera K, Kamijo S. Selection of peptide inhibitors of soluble Aβ1-42 oligomer formation by phage display. Biosci Biotechnol Biochem. 2010;74;2214–19. https://doi.org/10.1271/ bbb.100388
- Kawasaki T, Onodera K, Kamijo S. Identification of novel short peptide inhibitors of soluble 37/48 kDa oligomers of amyloid β42. Biosci Biotechnol Biochem. 2011;75(8):1496–501. https://doi. org/10.1271/bbb.110198
- Kokkoni N, Stott K, Amijee H, Mason JM, Doig AJ. N-methylated peptide inhibitors of β-amyloid aggregation and toxicity. Optimization of the inhibitor structure. Biochemistry. 2006;45(32):9906–18. https://doi.org/10.1021/bi060837s
- Zheng X, Liu D, Klärner F-G, Schrader T, Bitan G, Bowers MT. Amyloid β-protein assembly: The effect of molecular tweezers CLR01 and CLR03. J Phys Chem B. 2015;119(14):4831–41. https://doi. org/10.1021/acs.jpcb.5b00692
- 84. Tsuji-Ueno S, Komatsu M, Iguchi K, Takahashi M, Yoshino S, Suzuki M, et al. Novel high-affinity Aβ-binding peptides identified by an advanced in vitro evolution, progressive library method. Protein Peptide Lett. 2011;18(6):642–50. https://doi.org/10.2174/092986611795222678
- Ghimire Gautam S, Komatsu M, Nishigaki K. Strong inhibition of beta-amyloid peptide aggregation realized by two-steps evolved peptides. Chem Biol Drug Design. 2015;85(3):356–68. https://doi. org/10.1111/cbdd.12400
- Acerra N, Kad NM, Griffith DA, Ott S, Crowther DC, Mason JM. Retro-inversal of intracellular selected β-amyloid-interacting peptides: Implications for a novel Alzheimer's disease treatment. Biochemistry. 2014;53(13):2101–11. https://doi.org/10.1021/bi5001257
- Tanzi R, Moir R, Wagner S. Clearance of Alzheimer's Aβ peptide: The many roads to perdition. Neuron. 2004;43(5):605–8. https://doi.org/10.1016/S0896-6273(04)00533-1
- Sagare AP, Bell RD, Zhao Z, Ma Q, Winkler EA, Ramanathan A, et al. Pericyte loss influences Alzheimer-like neurodegeneration in mice. Nat Commun. 2013;4(1):1–14. https://doi.org/10.1038/ ncomms3932
- Ma Q, Zhao Z, Sagare AP, Wu Y, Wang M, Owens NC, et al. Blood-brain barrier-associated pericytes internalize and clear aggregated amyloid-β42 by LRP1-dependent apolipoprotein E isoform-specific mechanism. Mol Neurodegen. 2018;13(1):1–13. https://doi.org/10.1186/s13024-018-0286-0
- Funke SA, van Groen T, Kadish I, Bartnik D, Nagel-Steger L, Brener O, et al. Oral treatment with the D-enantiomeric peptide D3 improves the pathology and behavior of Alzheimer's disease transgenic mice. ACS Chem Neurosci. 2010;1(9):639–48. https://doi.org/10.1021/cn100057j
- Funke SA, Willbold D. Mirror image phage display A method to generate D-peptide ligands for use in diagnostic or therapeutical applications. Mol BioSyst. 2009;5(8):783–6. https://doi.org/10.1039/ b904138a
- Luo Y, Vali S, Sun S, Chen X, Liang X, Drozhzhina T, et al. Aβ42-binding peptoids as amyloid aggregation inhibitors and detection ligands. ACS Chem Neurosci. 2013;4(6):952–62. https://doi. org/10.1021/cn400011f
- Lam KS, Salmon SE, Hersh EM, Hruby VJ, Kazmierski WM, Knapp RJ. A new type of synthetic peptide library for identifying ligand-binding activity. Nature. 1991;354(6348):82–4. https://doi. org/10.1038/354082a0
- Remy I, Campbell-Valois F, Michnick SW. Detection of protein–protein interactions using a simple survival protein-fragment complementation assay based on the enzyme dihydrofolate reductase. Nature Protocols. 2007;2(9):2120. https://doi.org/10.1038/nprot.2007.266
- Pike CJ, Walencewicz-Wasserman AJ, Kosmoski J, Cribbs DH, Glabe CG, Cotman CW. Structureactivity analyses of β-amyloid peptides: Contributions of the β25–35 region to aggregation and neurotoxicity. J Neurochem. 1995;64(1):253–65. https://doi.org/10.1046/j.1471-4159.1995.64010253.x
- Acerra N, Kad NM, Mason JM. Combining intracellular selection with protein-fragment complementation to derive Aβ interacting peptides. Protein Eng Design Select. 2013;26(7):463–70. https://doi. org/10.1093/protein/gzt021

- 97. Ashur-Fabian O, Segal-Ruder Y, Skutelsky E, Brenneman DE, Steingart RA, Giladi E, et al. The neuroprotective peptide NAP inhibits the aggregation of the beta-amyloid peptide. Peptides. 2003;24(9):1413–23. https://doi.org/10.1016/j.peptides.2003.08.005
- Boxer AL, Lang AE, Grossman M, Knopman DS, Miller BL, Schneider LS, et al. Davunetide in patients with progressive supranuclear palsy: A randomised, double-blind, placebo-controlled phase 2/3 trial. Lancet Neurol. 2014;13(7):676–85. https://doi.org/10.1016/S1474-4422(14)70088-2
- Matsuoka Y, Jouroukhin Y, Gray AJ, Ma L, Hirata-Fukae C, Li H-F, et al. A neuronal microtubuleinteracting agent, NAPVSIPQ, reduces tau pathology and enhances cognitive function in a mouse model of Alzheimer's disease. J Pharmacol Exp Therapeut. 2008;325(1):146–53. https://doi. org/10.1124/jpet.107.130526
- 100. Gozes I, Morimoto BH, Tiong J, Fox A, Sutherland K, Dangoor D, et al. NAP: Research and development of a peptide derived from activity-dependent neuroprotective protein (ADNP). CNS Drug Rev. 2005;11(4):353–68. https://doi.org/10.1111/j.1527-3458.2005.tb00053.x
- 101. Findeis MA. Peptide inhibitors of beta amyloid aggregation. Curr Topics Med Chem. 2002;2(4): 417–23. https://doi.org/10.2174/1568026024607508
- 102. Jiang N, Leithold LH, Post J, Ziehm T, Mauler J, Gremer L, et al. Preclinical pharmacokinetic studies of the tritium labelled D-enantiomeric peptide D3 developed for the treatment of Alzheimer s disease. PLoS One. 2015;10(6):e0128553. https://doi.org/10.1371/journal.pone.0128553
- 103. Gregori M, Taylor M, Salvati E, Re F, Mancini S, Balducci C, et al. Retro-inverso peptide inhibitor nanoparticles as potent inhibitors of aggregation of the Alzheimer's Aβ peptide. Nanomed Nanotechnol Biol Med. 2017;13(2):723–32. https://doi.org/10.1016/j.nano.2016.10.006
- 104. Yang J, Liu W, Sun Y, Dong X. LVFFARK-PEG-stabilized black phosphorus nanosheets potently inhibit amyloid-β fibrillogenesis. Langmuir. 2020;36(7):1804–12. https://doi.org/10.1021/acs. langmuir.9b03612
- 105. Hao S, Li X, Han A, Yang Y, Fang G, Liu J, et al. CLVFFA-functionalized gold nanoclusters inhibit aβ40 fibrillation, fibrils' prolongation, and mature fibrils' disaggregation. ACS Chem Neurosci. 2019;10(11):4633–42. https://doi.org/10.1021/acschemneuro.9b00469
- 106. Niu X, Chen J, Gao J. Nanocarriers as a powerful vehicle to overcome blood-brain barrier in treating neurodegenerative diseases: Focus on recent advances. Asian J Pharmaceut Sci. 2019;14(5):480–96. https://doi.org/10.1016/j.ajps.2018.09.005
- 107. Xiong N, Zhao Y, Dong X, Zheng J, Sun Y. Design of a molecular hybrid of dual peptide inhibitors coupled on AuNPs for enhanced inhibition of amyloid β-protein aggregation and cytotoxicity. Small. 2017;13(13):1601666. https://doi.org/10.1002/smll.201601666
- Jeong W-J, Bu J, Kubiatowicz LJ, Chen SS, Kim Y, Hong S. Peptide–nanoparticle conjugates: A next generation of diagnostic and therapeutic platforms? Nano Convergence. 2018;5(1):38. https://doi. org/10.1186/s40580-018-0170-1
- 109. Khafaji M, Zamani M, Golizadeh M, Bavi O. Inorganic nanomaterials for chemo/photothermal therapy: A promising horizon on effective cancer treatment. Biophys Rev. 2019;11(3):335–52. https://doi. org/10.1007/s12551-019-00532-3

Tau Protein-Targeted Therapies in Alzheimer's Disease: Current State

Ryszard Pluta • Marzena Ułamek-Kozioł

and Future Perspectives

Laboratory of Ischemic and Neurodegenerative Brain Research, Mossakowski Medical Research Centre, Polish Academy of Sciences, Warsaw, Poland

Author for correspondence: Ryszard Pluta, Laboratory of Ischemic and Neurodegenerative Brain Research, Mossakowski Medical Research Centre, Polish Academy of Sciences, Warsaw, Poland. Email: pluta@imdik.pan.pl

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch4

Abstract: Drugs available on the market for the treatment of Alzheimer's disease show only low symptomatic efficacy and phase 3 clinical trials against amyloid have been negative over the past 20 years. As dysfunctional tau protein is more closely correlated with dementia than amyloid, targeting tau protein may be more effective in improving cognitive function in cases of Alzheimer's disease. It should be emphasized that the development of tau protein therapy is in many ways more complicated than the development of anti-amyloid therapy. Several antibodies to the tau protein and two vaccines are currently undergoing clinical trials. Relatively speaking, tau protein therapy for Alzheimer's disease is still in its infancy. The purpose of this chapter is to draw the readers' attention to the various uncertainties and barriers to the success of tau protein therapy in treating Alzheimer's disease, and to show how future research and clinical trials can avoid previous limitations or mistakes.

Keywords: Alzheimer's disease; immunotherapy; neurofibrillary tangles; tau protein; vaccine for Alzheimer's disease

Copyright: The Authors.

In: *Alzheimer's Disease: Drug Discovery.* Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

INTRODUCTION

Alzheimer's disease is an age-related disorder characterized clinically by gradual memory loss and cognitive impairment. The disease affects more than 47 million people worldwide, and this number is expected to reach over 131 million by 2050. Alzheimer's disease is considered the most common cause of dementia (1, 2), covering about 60–80% of dementia cases worldwide (3). Available data indicate that the incidence of sporadic Alzheimer's disease is common and reaches 10-50% persons over the age of 65 (4). Factors that may be involved in the development of the disease include lifestyle habits such as diet, exercise, education, cognition and aging, immunosenescence, chronic infections and inflammation, latent infections, vascular factors, sleep problems and more (5–9). It has been suggested that intestinal microorganisms and ischemic episodes may also be involved in the development of this disease (10-13). Heritability of this form of dementia is high and estimated at 70-79% based on twin-studies. However, most evidence points to a heterogeneous etiology, with the disease resulting from a combination of many genetic, environmental, vascular and other currently unknown factors (5, 6, 10, 14–16). One particular genetic factor, the epsilon allele in the apolipoprotein E gene, has been identified as being associated with an increased risk of sporadic Alzheimer's disease, but a large percentage of the genetic risk remains unidentified. Alzheimer's disease is the leading cause of acquired disability in the world, affecting 1 in 2 in women and 1 in 3 in men (17). Alzheimer's disease is described as one of the unsolved problems of modern medicine (18), one which has a significant impact on the global economy, society and the families of the sick (19). In addition to significant personal costs, the total estimated worldwide financial burden due to dementia in 2010 was USD 604 billion (19). This is a serious public health problem that can grow to epidemic proportions over the next few decades if the disease cannot be prevented or slowed down (18).

Neuropathologically, Alzheimer's disease is characterized by the accumulation of amyloid in the form of plaques in the extracellular space and tau protein dysfunction in the form of neurofibrillary tangles present in the intracellular space, which are important in the final post-mortem diagnosis. The most neurotoxic forms of amyloid and tau protein are believed to be oligomeric forms that spread extracellularly as soluble oligomers through a prion-like mechanism (20, 21). The causes or mechanisms of amyloid plaques and neurofibrillary tangles formation are not yet well understood, but they are generally considered to be the result of a process of misfolding of proteins that leads to the development of pathological phenomena. Alzheimer's disease develops due to a combination of different neuropathological processes in the brain that cause massive neuronal death and loss of synapses. The resulting atrophy causes patients' brains to weigh about a third less than age-matched non-demented people (3). Today we know that the onset of Alzheimer's disease begins between 15 (in familial cases) and 20-30 years (in sporadic cases) before any clinical symptoms appear (2, 8, 22). By the time the clinical phenotype is recognized, significant neuronal and synaptic degeneration and massive neuroinflammatory changes have already occurred. Though the cause of Alzheimer's disease is not completely certain, it has been proven by diagnostic methods based on the analysis of the patient's brain images that the accumulation of amyloid in the brain precedes the appearance of clinical symptoms and indicates a number of pathological factors that are ultimately not defined.

The amyloid hypothesis of Alzheimer's disease suggests that increased amyloid aggregation causes Alzheimer's disease by triggering toxic events leading to progressive neurodegeneration. However, no drug candidate targeting amyloid has yet led to effective treatment (3, 23). It is currently speculated that treatment requires early targeting of amyloid when the changes remain reversible, and clinical trials should focus on assessing amyloid compounds in pro-dromal Alzheimer's disease. There is no prophylactic or causal therapy for the disease, and the lack of knowledge about etiology and when or why the disease really begins, significantly complicates the work of physicians (24). Currently available treatments for Alzheimer's disease are only aimed at mitigating clinical symptoms and delaying cognitive decline. The development of a therapy for Alzheimer's disease has resulted in only a few approved drugs that provide temporary symptomatic relief in some patients. None of these clinically used drugs stop or slow the progression of the disease. Currently, the only drugs that have an impact, albeit modest and transient, on the main symptoms in patients with mild to moderate dementia of Alzheimer's disease are acetylcholinesterase inhibitors and N-methyl-D-aspartate (NMDA) receptor antagonists. For therapeutic measures to have a significant effect on the delay or actual prevention of Alzheimer's disease, it is likely that patients will need to be diagnosed at the stage of preclinical Alzheimer's disease (i.e., presence of Alzheimer's disease neuropathology, but without clear clinical symptoms) or during early signs of Alzheimer's disease that could be treated with disease modifying agents. Treatments that target the etiological mechanisms of Alzheimer's disease are urgently needed.

That treatments targeting amyloid plaques have proven ineffective draws attention towards another pathologic hallmark of Alzheimer's disease, neurofibrillary tangles arising from hyperphosphorylation of the tau protein. Experimental evidence indicates that the symptoms of Alzheimer's disease appear in the presence of both accumulating amyloid and dysfunctional tau protein (3), and research has shown a strong correlation between the accumulation of neurofibrillary tangles and the deterioration of cognitive functions (3, 25, 26). Thus, it is believed that tau protein dysfunction cannot be ignored as an etiological factor of Alzheimer's disease. Compounds that prevent tau protein hyperphosphorylation may therefore affect disease progression; however, the failure of previous trials to treat tauopathy in progressive supranuclear palsy (26) gives a strong warning of a possible failure. Nonetheless, the importance of tau protein as a potential independent cause of Alzheimer's disease, and therefore a potential target for treatment, serves as the basis for ongoing clinical trials against the protein. Despite the lack of an in-depth understanding of the role of tau protein in the pathology of Alzheimer's disease, efforts are underway to develop new therapies targeting tau protein in Alzheimer's disease. This chapter presents some of the proposed therapeutic compounds in preclinical and clinical studies that may affect the development of the next generation of anti-Alzheimer's disease drugs.

SUBSTANCES PREVENTING TAU PROTEIN POST-TRANSLATIONAL CHANGES

The strong correlation between tau protein phosphorylation and its influence on the development of pathological processes gave rise to the search for tau protein kinases inhibitors as potentially effective therapeutic agents in Alzheimer's disease. The most advanced strategy for inhibiting protein kinase in the clinic is currently directed at glycogen synthase kinase-3 (GSK-3) (27). To this end, pilot studies were performed to determine the clinical effects of lithium in patients with Alzheimer's disease; however, the results of this study were inconsistent, perhaps due to the small number of patients, low susceptibility and narrow therapeutic range of lithium (28).

Tideglusib (NP031112, NP-12) is a GSK-3 inhibitor that, in preclinical studies, reduced neuronal loss, gliosis and tau protein phosphorylation, and improved spatial memory deficit in transgenic mice (29). Further investigation with human trials included a pilot study conducted on 30 patients affected with Alzheimer's disease who were treated for 5 months with Tideglusib in a placebo-controlled phase IIa increasing dose clinical trial (NCT00948259). The results showed overall positive, but not significant, trends in their cognitive health, though the study affirmed the safety of the drug (30). Another 6-month phase IIb study on 308 patients affected with Alzheimer's disease was conducted at 55 centers in Europe (NCT01350362). Tideglusib proved to be safe in the study, but those treated with the drug did not show significant clinical benefit (31).

Another protein kinase that is increasingly being considered as a potential therapeutic target is Fyn tyrosine kinase, which phosphorylates tau protein at the N-terminal domain, and also plays a role in the amyloid signaling pathway (32). Saracatinib (AZD0530) is a Fyn inhibitor that improves memory deficiencies in transgenic mice and is considered safe and well tolerated based on a phase I clinical trial (NCT01864655) (33). A multicenter phase IIa study in 159 Alzheimer's disease patients treated with Saracatinib is still ongoing (NCT02167256).

The tau protein is also modified post-translation by lysine acetylation, which leads to impaired protein activity and triggers pathological aggregation. This suggests that acetyltransferase inhibitors may be a potential therapeutic strategy for Alzheimer's disease (34). A phase 1 clinical trial of salsalate has recently started to assess its safety and tolerability in patients with Alzheimer's disease (NCT03277573). Patients will be randomly assigned to receive salsalate or placebo twice a day for 1 year. At present, the results of this study have not been published.

Phase I studies on substance AZP2006 are coming to an end for Alzheimer's disease cases in France, but no detailed results are available as yet. The oral substance has been proposed to block phosphorylation of the tau protein thereby preventing the tau protein from folding incorrectly. In addition, it appears to stimulate macrophages, inducing the removal and elimination of an incorrectly folded tau protein.

Nilotinib is a c-Abl tyrosine kinase inhibitor, and the rationale for the use of nilotinib in cases of Alzheimer's disease is based on the clearance of tau protein and amyloid accumulated in the brain in neurodegenerative processes. Although the exact molecular mechanism is uncertain, nilotinib appears to cross the blood–brain barrier and trigger autophagy in neurons to remove both tau protein and amyloid. The properties of nilotinib mentioned formed the basis of a randomized, double-blind, placebo-controlled, phase II trial to assess the effect of nilotinib on safety and clinical outcomes in patients diagnosed with Alzheimer's disease (NCT02947893).

SUBSTANCES THAT PREVENT MICROTUBULE DESTABILIZATION BY AMYLOID

The observation that amyloid oligomers destabilize microtubules and interfere with rapid axonal transport by activating calcineurin in tau protein-deficient mice has led to the conclusion that microtubule destabilization may be a key process during neurodegeneration (35). Epothilone D (BMS-241027), a small molecule stabilizer of microtubules that can pass through the blood–brain barrier, was able to increase the density of microtubules in axons, and improved cognitive function in a mouse transgenic tauopathy model; only insignificant changes in tau protein pathology were noted in the study (36). The substance was also tested in a double-blind, randomized, placebo-controlled, multicenter phase I clinical trial (NCT01492374) intended to assess its safety and tolerability in patients with Alzheimer's disease, but no results have been published and use of the drug in Alzheimer's disease has been suspended.

Recently, a small molecule called abeotaxane (TPI-287) was tested in a phase I study to assess safety and tolerability in patients with Alzheimer's disease (NCT02133846). Abeotaxane was administered intravenously for 9 weeks once per 3 weeks, with the option of extending the open label to 3 months. Ultimately, treatment was not well tolerated by people with Alzheimer's disease, and exploratory cognitive endpoints showed no significant improvement.

SUBSTANCES THAT PREVENT TAU PROTEIN AGGREGATION

Methylthioninium chloride easily crosses the blood–brain barrier and prevents tau protein aggregation *in vitro*, as well as in cells and animals models (37). A double-blind clinical trial in which single-site, 6-month methylthioninium monotherapy was conducted on patients with Alzheimer's disease (NCT00515333) showed signs of benefit in moderate cases of the disease (38). Leucomethylthioninium bis, a stable, reduced form of the methylthioninium moiety, acts as a selective inhibitor of tau protein aggregation both *in vitro* and in transgenic mouse models. The primary analysis using leuco-methylthioninium bis derivative of methylthioninium chloride in the treatment of Alzheimer's disease was negative and the results did not suggest therapeutic benefits for Alzheimer's disease treated orally, twice daily, (NCT01689233) showed no effect on primary endpoints (40). The authors' explanations of the effectiveness after the secondary analysis of the post-hoc subgroup raised many doubts, mainly regarding the methodology used and the interpretation of the results. Although these studies did not produce positive results, they are nevertheless an important step in the development of anti-tau protein drugs.

Since positron emission tomography (PET) tau protein imaging is currently used in conjunction with amyloid PET imaging in an increasing number of Alzheimer's disease clinical trials, our knowledge of the ideal stage of the disease for testing anti-amyloid, anti-tau protein or a combination of both will undoubtedly be improved. There is growing optimism that we have the right tools to evaluate compounds that can stop and even prevent Alzheimer's disease.

IMMUNOTHERAPY AGAINST TAU PROTEIN

The first mention of data on possible immunotherapy in Alzheimer's disease appeared during research into the possibility that human β -amyloid peptide 1–42 may be able to cross the blood-brain barrier (41, 42). Despite the disappointing data from several advanced clinical studies on anti-amyloid immunotherapy in the treatment of Alzheimer's disease (23), immunotherapy in neurodegenerative diseases is very actively sought as a promising approach for the removal of pathological proteins, particularly in Alzheimer's disease. Recently tested anti-tau protein immunotherapy strategies in animal models have shown that immunotherapy may be clinically viable to remove toxic protein species in tauopathies such as Alzheimer's disease (43). Anti-tau protein immunotherapy strategies involve the removal of pathological species of tau protein with antibodies, which may ultimately improve neuronal function (26, 43). Thus, choosing the right epitope is crucial for obtaining effective immunotherapy (26). Because hyperphosphorylation is thought to be the cause of tau protein aggregation and the development of neurofibrillary tangles, many phospho-epitopes have been tested in animal models with final positive effects (13, 26, 27, 29, 32, 33, 35–37, 44). It should be noted that the tau protein undergoes modifications other than phosphorylation during the transformation from soluble protein to insoluble aggregates and deposits (13, 27, 44). As mentioned above, the cascade of events leading to the development of neurofibrillary tangles may include post-translational modifications such as phosphorylation, glycosylation, truncation and ubiquitination (13, 27, 29, 34, 44). This gave rise to a series of active and passive immunotherapy programs in the treatment of patients associated with Alzheimer's disease (26). Some of the clinical studies of tau proteintargeted immunotherapy described below target different protein domains rather than specific phospho-epitopes (26).

At present, studies of anti-tau protein immunotherapies in clinical trials are in their early stages. AADvac-1 is an active immunotherapy (vaccine) based on a synthetic tau protein peptide containing residues 294–305 derived from a fragment of a misfolded tau protein. Phase I clinical trials of the therapy have been recently completed (NCT01850238). In this first-ever, randomized, double-blind, placebo-controlled study, 30 Alzheimer's disease patients aged 50–85 received subcutaneous injections of AADvac-1 for 3 months; generally mild, if any side effects were reported, and high tau protein titers indicated effective immunogenicity (45). The study was then extended to patients who completed the AADvac-1 phase 1 study to administer additional immunization doses that they received for the next 18 months (NCT02031198) (46). The recruitment of 185 Alzheimer's disease patients in a 24-month, randomized, placebo-controlled, double-blind, parallel

TABLE 1	Clinical trials of tau protein-targeted immunotherapies (21, 26, 45–55)				
Compound	Isotype	Therapy type	Patients	Trial phase	
AADvacl	IgG1	Active	Alzheimer's disease	1/2	
ACI-35	n.a.	Active	Alzheimer's disease	1	
BIB076	IgG1	Passive	Alzheimer's disease	1	
BIB092	IgG4	Passive	Alzheimer's disease	2	
LY3303560	n.a.	Passive	Alzheimer's disease	2	
RO7105705	IgG4	Passive	Alzheimer's disease	2	
JNJ-63733657	n.a.	Passive	Alzheimer's disease	2	
LuAF87908	n.a.	Passive	Alzheimer's disease	1	
ABBV-8E12	IgG4	Passive	Alzheimer's disease	2	
UCB0107	IgG4	Passive	Alzheimer's disease	2	
IVIg	pIgG	Passive	Alzheimer's disease	2/3	

n.a, not available.

group study, multicenter clinical safety and phase II efficacy study for AADvac-1 began in 2016 with an option to end by 2019 (NCT02579252) (Table 1).

So far only limited data from clinical trials and only for the AADvac1 vaccine have been presented (45, 46). Administration of the AADvac1 vaccine induced an IgG response against tau protein in 29 of 30 patients and a response to the shortened form of tau protein (151-391/4R) was noted in 25 of 28 patients (45). Although the phase 1 trial was not designed as an efficacy study, inter-individual differences in AADvac1-induced antibody titers enabled an assessment of the relationship between antibody response potency and disease progression. Patients with higher antibody titers were characterized by slower cognitive performance and lower hippocampal atrophy (46). Adverse reactions were generally mild, with the most common being local injection-site reactions (45, 46).

ACI-35 (vaccine) is a synthetic peptide comprising the tau protein sequence of human protein 393–408, phosphorylated at S396 and S404. In animal models, its administration has been shown to reduce both the quantity of aggregates of phosphorylated tau protein and the total pathological protein, as well as improve some cognitive functions. The vaccine triggers a specific antibody against the tau protein and an immune response that is independent of T cells. There are ongoing multicenter, double-blind, randomized, placebo-controlled phase I studies to assess the safety, tolerability and immunogenicity of ACI-35 in patients with mild to moderate Alzheimer's disease (ISRCTN13033912) (Table 1).

Data for AADvac1 and the fact that no dangerous adverse effects were reported for ACI-35 indicate that active immunization may be a safe way to counteract tau protein pathology. The safety aspect will become more and more important as active tau protein-targeted immunotherapy switches from the treatment of neurodegeneration towards its prevention (46).

Intravenous immunoglobulin (IVIg) derived from human plasma, consisting of polyclonal serum IgG obtained from blood donors, is effective in antiinflammatory and immunomodulating therapy in cases of neurological diseases (50). A randomized, double-blind, placebo-controlled phase III study (NCT00818662) involving 390 patients with Alzheimer's disease found no improvement in cognitive ability and function after IVIg infusions every 2 weeks for 18 months (Table 1) (47). Two further trials, one phase II (NCT01300728) and one phase III (NCT01561053) for the treatment of Alzheimer's disease are underway. Interestingly, it was found that tau protein-specific antibodies are present in the IVIg Flebogamma® product that recognizes a recombinant tau protein fragment containing residues 155–421, as well as tau protein aggregates from the brains of Alzheimer's disease patients (51).

In addition to active immunotherapy, strategies for passive immunotherapy using various antibodies to the tau protein are also under investigation (Table 1). It has been shown that this approach can improve behavioral and cognitive impairment in mouse models (52). In the past few years, three passive immunotherapy programs based on tau protein antibodies have been opened in clinical trials, mainly for the treatment of Alzheimer's disease as outlined in Table 1. BIIB092 is a humanized IgG4 monoclonal antibody against the extracellular N-terminal of a tau protein isolated from pluripotent stem cells of a patient with familial Alzheimer's disease (Table 1) (53).

ABBV-8E12 is a humanized monoclonal antibody against tau protein for the treatment of Alzheimer's disease in clinical settings. Currently, recruitment for a phase II clinical trial is done to assess the effects of ABBV-8E12 in patients with Alzheimer's disease (NCT02880956) (Table 1). A multicenter, randomized, double-blind, placebo-controlled study to assess the efficacy and safety of ABBV-8E12 in 400 patients with Alzheimer's disease should be completed by 2020 (54). A final approach to passive immunotherapy in clinical trials is R07105705, an antibody against tau protein, whose features regarding its nature are very limited and its preclinical efficacy unknown (55). In 2016, patients with Alzheimer's disease were recruited for the first stage of the study (NCT02820896). Although no results have been published, recruitment was initiated at the end of 2017 to an 18-month, randomized, double-blind, placebo-controlled phase II study to evaluate the efficacy and safety of RO7105705 in 360 patients with Alzheimer's disease (NCT03289143) (Table 1). Patients who complete the double-blind therapy will be invited to an optional 24-month extension period.

In summary, many active and passive tau protein-targeted immunotherapies are already in the clinical trial stage for the treatment of Alzheimer's disease (Table 1). Due to the fact that new strategies for tau protein-directed immunotherapy are still being developed, a number of key questions need to be answered, in particular regarding the choice of the immunogen, the species of tau protein to be targeted, as well as mechanism of action and safety (56–58).

FUTURE PERSPECTIVES

Per the studies and data noted in this chapter, the status of tau protein targeting therapy in the treatment of Alzheimer's disease is unclear due to lack of evidence or mutually exclusive observations. The small number of studies, and variable non-uniform measures of results suggest the field to be in its infancy and limit the possibility of making generalized conclusions. Although Alzheimer's disease is a real challenge for the pharmaceutical industry, there has been no clear progress in treatment options in the past few years. Current drugs on the market show only low effectiveness, and clinical studies from the last 20 years have ultimately proven to be negative in phase 3. This was perhaps due to the hegemony of the amyloid hypothesis and focus of therapeutic strategies on amyloid, while the focus on tau protein, the main component of neurofibrillary tangles which correlates better with the degree of dementia than amyloid (25) has only emerged in recent years. A limited number of studies provide evidence of low or very low quality. Some studies have shown side effects, although the study times were short and the longterm risk of side effects was not determined.

To date, several drug trials targeting the tau protein have failed in clinical trials. Although there are various causes for these failures, the following points can help improve the results of future attempts. Firstly, the tau protein should be ideally targeted intracellularly, since most pathologies of the tau protein affect neurons inside. Secondly, as previous anti-amyloid immunotherapy attempts have taught us, it is important to continue to develop second- and third-generation methods in the field of tau protein immunotherapy. Smaller antibodies that are fragments of whole antibodies should have better access to the inside of both the brain in general and the neurons themselves, while also enabling them to bind to different epitopes of the tau protein than whole antibodies, providing greater and more effective therapeutic benefits. Thus, due to their smaller size, they will also be better suited for gene therapy than whole antibodies. We will be looking forward to future preclinical studies examining antibody fragments as a novel therapeutic approach. Thirdly, in recent years, a major focus has been on the implementation of drug-screening models that have focused on preventing seeding or spreading aggregation. Much less attention has been paid to the identification of compounds that inhibit the neurotoxicity of these aggregates, which is not necessarily associated with their seeding or spreading tendency. Ideally, all these markers should be readings in a unified test or model. Fourthly, the variety of conformer or strain of aggregates complicates the development of drugs for small molecule aggregation inhibitors but will probably not pose a problem in antibody-based therapy. Fifthly, other more general goals related to neurodegeneration should still be pursued. but in many ways, they are more difficult to solve than the removal of amyloid and tau protein, which are the hallmarks of Alzheimer's disease. Sixthly, shifting the time of therapeutic intervention to the very early stages of Alzheimer's disease should be a feature of long-term clinical studies. Lastly, targeting the tau protein is likely to provide better therapeutic benefits at a later stage in the development of the disease because tau protein dysfunction is more closely correlated with performance on cognitive tests than amyloid (25). Based on the data discussed in this chapter regarding the presence and abundance of intra- and extracellular pools of tau protein and its epitopes, and the finding that antibodies can have diseasespecific efficacy and separate efficacy against tau protein toxicity, are the right direction for future preclinical and clinical investigations.

Future randomized clinical trials are needed to demonstrate the effectiveness of test substances and provide necessary data on several unresolved practical problems, that is, how and how long these substances can be administered to patients with Alzheimer's disease. What part of the tau protein is affected by immunotherapy and other test substances? How? What effect do treatment attempts have on the physiological and neuropathophysiological function of the tau protein? What human data was obtained to confirm this? These questions can only be answered by undertaking large, multi-center clinical trials. At the moment, information on the use of anti-tau protein substances as drugs in the treatment of Alzheimer's disease seems interesting because of the possible effect on the accumulation of tau protein. The few clinical trials available to date have no genuine control and group randomization. Evidence has shown the causative factors and indicates the need for further research to show that anti-tau protein substances have a positive effect on patients with Alzheimer's disease. Though there is uncertainty regarding the potential of tau protein-targeting therapies in the treatment of Alzheimer's disease, and indeed the reasoning behind this skepticism is important to consider, it is still worthwhile to investigate. Though published studies on the subject are often flawed, and the findings should be taken with caution, the promising results should not be completely disregarded.

Another issue is the study design for substances directed against tau protein. Current ethical standards for clinical trials require that substances under investigation must be compared to the current treatment standard; this makes it difficult to conduct a true randomized, placebo-controlled study that is independent of potentially confounding treatments. This necessitates that the question of what standards of treatment serve as the best basis of comparison be answered prior to further study. Based on comprehensive pre-clinical results, as well as the initial clinical data, it is clear that the next step must be to test these substances in welldesigned and controlled clinical trials. However, further double-blind studies are also needed to determine the efficacy of these substances in treatment. In conclusion, future clinical trials should focus on the proper selection of patients. Accurate and definitive explanation of the therapeutic properties of substances against tau protein can give hope for a long-lasting therapeutic effect. Based on the results of verified clinical trials, it may seem that the clinical effectiveness of substances that target tau protein is promisingly high, but this is not to be considered certain. We must be patient and wait at least a few years for thorough confirmation. No substances against the tau protein have been approved for use in the clinic. We hope that the evidence from ongoing clinical trials will help us better understand the therapeutic efficacy of substances against the tau protein and put them at the forefront of new therapies that patients and their doctors are eagerly awaiting. While some open questions and challenges remain, the data presented here encourages and demonstrates the potential of tau protein-based therapeutic strategies in the future treatment of Alzheimer's disease. The complexity of abnormal tau protein folding, aggregation and propagation of neuropathology, as well as the immune response during aging and neurodegeneration should be considered so that we can design and develop safe and effective treatment in a better way. Lessons must be learned from the disappointing experience accumulated over the past 20 years to develop disease-modifying therapies so that we can continue to progress, with caution, translating results from preclinical models into the development of drugs at the clinic. The growing interest in the tau protein will certainly lead to a deeper understanding of its function and will give new insight into the precise mechanisms and nature of tau protein species responsible for neuronal dysfunction and its causal role in the development of Alzheimer's disease. Hopefully, this will lead to an extension of the range of potentially useful therapeutic tools for treating such a devastating condition as Alzheimer's disease.

CONCLUSION

Although both amyloid and tau protein are very important, their relationship in causing Alzheimer's disease remains unknown. Treatment directed to amyloid and tau protein may be individually effective, but the convergent progression of amyloid and tau protein pathology suggests that combination therapy may eventually be required, especially in late stages when both are abundant. While ongoing works focused on single-goal therapies, the approach to double-targeting amyloid and tau protein is more likely to lead to a breakthrough (3). Referring to the above observations, it should be stated that Alzheimer's disease is an age-related neuro-degenerative disease whose various neuropathological and therapeutic aspects are still being investigated and are not fully explained; pending success in the development of an effective treatment for Alzheimer's disease, it may be best to focus on preventive measures (59).

Acknowledgments: The authors acknowledge the financial support from the Mossakowski Medical Research Centre, Polish Academy of Sciences, Warsaw, Poland (T3-RP).

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship, and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- 1. Yoshiyama Y, Lee VMY, Trojanowski. JQ. Therapeutic strategies for tau mediated neurodegeneration. J Neurol Neurosur Psychiat. 2012;84:784–95. http://dx.doi.org/10.1136/jnnp-2012-303144
- Michael MS. Targeting tau protein in Alzheimer's disease. Lancet. 2016;388:2842–4. http://dx.doi. org/10.1016/S0140-6736(16)32107-9
- Lansdall CJ. An effective treatment for Alzheimer's disease must consider both amyloid and tau. Biosci Horizons. 2014;7:1–11. http://dx.doi.org/10.1093/biohorizons/hzu002
- Liu H, Xie A-M, Wang L, Xiang-Qun WS. Advances in recent patent and clinical trial drug development for Alzheimer's disease. Pharm Pat Anal. 2014;3:429–47. http://dx.doi.org/10.4155/ppa.14.22
- Pluta R, Lossinsky AS, Walski M, Wisniewski HM, Mossakowski MJ. Platelet occlusion phenomenon after short- and long-term survival following complete cerebral ischemia in rats produced by cardiac arrest. J Hirnforsch. 1994;35:463–71.
- Wiśniewski HM, Pluta R, Lossinsky AS, Mossakowski MJ. Ultrastructural studies of cerebral vascular spasm after cardiac arrest-related global cerebral ischemia in rats. Acta Neuropathol. 1995;90: 432–40. http://dx.doi.org/10.1007/BF00294802
- Ooms S, Overeem S, Besse K, Rikkert MO, Verbeek M, Claassen JAHR. Effect of 1 night of total sleep deprivation on cerebrospinal fluid β-amyloid 42 in healthy middle-aged men: A randomized clinical trial. JAMA Neurol. 2014;71:971–7. http://dx.doi.org/10.1001/jamaneurol.2014.1173

80

- Sochocka M, Donskow-Łysoniewska K, Satler Diniz B, Kurpas D, Brzozowska E, Leszek J. The gut microbiome alterations and inflammation-driven pathogenesis of Alzheimer's disease – a critical review. Mol Neurobiol. 2019;56:1841–51. http://dx.doi.org/10.1007/s12035-018-1188-4
- 9. Pluta R, Ułamek-Kozioł M, Januszewski S, Czuczwar SJ. Gut microbiota and pro/prebiotics in Alzheimer's disease. Aging (Albany NY). 2020;12:5539–50. http://dx.doi.org/10.18632/aging.102930
- Pluta R. Brain ischemia: Alzheimer's disease mechanisms. New York: Nova Science Publishers, Inc. 2019; 311 p.
- Pluta R, Ułamek-Kozioł M, Januszewski S, Czuczwar SJ. Common proteomic and genomic contribution to ischemic brain damage and Alzheimer's disease. In: Wisniewski T, editor. Alzheimer's disease. Brisbane, Australia: Codon Publications; 2019. p. 53–68.
- Pluta R, Ułamek-Kozioł M, Januszewski S, Czuczwar SJ. Shared genomic and proteomic contribution of amyloid and tau protein characteristic of Alzheimer's disease to brain ischemia. Int J Mol Sci. 2020;21:3186. http://dx.doi.org/10.3390/ijms21093186
- Ułamek-Kozioł M, Czuczwar SJ, Januszewski S, Pluta R. Proteomic and genomic changes in tau protein, which are associated with Alzheimer's disease after ischemia-reperfusion brain injury. Int J Mol Sci. 2020;21:892. http://dx.doi.org/10.3390/ijms21030892
- Gatz M, Reynolds CA, Fratiglioni L, Johansson B, Mortimer JA, Berg S, et al. Role of genes and environments for explaining Alzheimer disease. Arch Gen Psychiatry. 2006;63:168–74. http://dx.doi. org/10.1001/archpsyc.63.2.168
- Barber RC. The genetics of Alzheimer's disease. Scientifica (Cairo). 2012;2012:246210. http://dx.doi. org/10.6064/2012/246210
- Bellenguez C, Grenier-Boley B, Lambert JC. Genetics of Alzheimer's disease: Where we are, and where we are going. Curr Opin Neurobiol. 2020:61:40–8. http://dx.doi.org/10.1016/j. conb.2019.11.024
- 17. Seshadri S, Wolf PA. Lifetime risk of stroke and dementia: Current concepts, and estimates from the Framingham study. Lancet Neurol. 2007;6:1106–14. http://dx.doi.org/10.1016/S1474-4422 (07)70291-0
- Wortmann M. Dementia: A global health priority—Highlights from an ADI and World Health Organization report. Alzheimer Res Ther. 2012;4:40. http://dx.doi.org/10.1186/alzrt143
- Wimo A, Jonsson L, Bond J, Prince M, Bengt Winblad B, Alzheimer Disease International. The worldwide economic impact of dementia 2010. Alzheimers Dement. 2013;9:1–11e3. http://dx.doi. org/10.1016/j.jalz.2012.11.006
- Ashe KH, Aguzzi A. Prions, prionoids and pathogenic proteins in Alzheimer disease. Prion. 2013;7:55–9. http://dx.doi.org/10.4161/pri.23061
- Colin M, Dujardin S, Schraen-Maschke S, Meno-Tetang G, Duyckaerts C, et al. From the prionlike propagation hypothesis to therapeutic strategies of anti-tau immunotherapy. Acta Neuropathol. 2020;139:3–25. http://dx.doi.org/10.1007/s00401-019-02087-9
- Jansen WJ, Ossenkoppele R, Knol DL, Tijms BM, Scheltens P, Verhey FRJ, et al. Prevalence of cerebral amyloid pathology in persons without dementia. A meta-analysis. JAMA. 2015;313:1924–38. http:// dx.doi.org/10.1001/jama.2015.4668
- 23. Barrera-Ocampo A, Lopera F. Amyloid-immunotherapy: The hope for Alzheimer disease? Colomb Med. 2016:47:203–12. http://dx.doi.org/10.25100/cm.v47i4.2640
- 24. Lloret A, Esteve D, Monllor P, Cervera-Ferri A, Lloret A. The effectiveness of vitamin E treatment in Alzheimer's disease. Int J Mol Sci. 2019;20:879. http://dx.doi.org/10.3390/ijms20040879
- Nelson PT, Alafuzoff I, Bigio EH, Bouras C, Braak H, Cairns NJ, et al. Correlation of Alzheimer disease neuropathologic changes with cognitive status: A review of the literature. J Neuropathol Exp Neurol. 2012;71:362–81. http://dx.doi.org/10.1097/NEN.0b013e31825018f7
- Sandusky-Beltran LA, Sigurdsson EM. Tau immunotherapies: Lessons learned, current status and future considerations. Neuropharmacology. 2020;175:108104. http://dx.doi.org/10.1016/j. neuropharm.2020.108104
- Medina M, Avila J. New insights into the role of glycogen synthase kinase 3 in Alzheimer's disease. Expert Opin Ther Targets. 2014;18:69–77. http://dx.doi.org/10.1517/14728222.2013.843670
- Forlenza OV, Diniz BS, Radanovic M, Santos FS, Talib LL, Gattaz WF. Disease-modifying properties of long-term lithium treatment for amnestic mild cognitive impairment: Randomised controlled trial. Br J Psychiatry. 2011;198:351–6. http://dx.doi.org/10.1192/bjp.bp.110.080044

- Serenó L, Coma M, Rodríguez M, Sánchez-Ferrer P, Sánchez MB, Gich I, et al. A novel GSK-3 inhibitor reduces Alzheimer's pathology and rescues neuronal loss in vivo. Neurobiol Dis. 2009;35:359–67. http://dx.doi.org/10.1016/j.nbd.2009.05.025
- Del Ser T, Steinwachs KC, Gertz HJ, Andrés MV, Gómez-Carrillo B, Medina M, et al. Treatment of Alzheimer's disease with the GSK-3 inhibitor tideglusib: A pilot study. J Alzheimers Dis. 2013;33: 205–15. http://dx.doi.org/10.3233/JAD-2012-120805
- Lovestone S, Boada M, Dubois B, Hüll M, Rinne JO, Huppertz HJ, et al. A phase II trial of tideglusib in Alzheimer's disease. J Alzheimers Dis. 2015;45:75–88. http://dx.doi.org/10.3233/JAD-141959
- Nygaard HB, van Dyck CH, Strittmatter SM. Fyn kinase inhibition as a novel therapy for Alzheimer's disease. Alzheimers Res Ther. 2014;6:8. http://dx.doi.org/10.1186/alzrt238
- 33. Nygaard HB, Wagner AF, Bowen GS, Good SP, MacAvoy MG, Strittmatter KA, et al. A phase lb multiple ascending dose study of the safety, tolerability, and central nervous system availability of AZD0530 (saracatinib) in Alzheimer's disease. Alzheimers Res Ther. 2015;7:35. http://dx.doi.org/10.1186/s13195-015-0119-0
- 34. Medina M, Hernández F, Avila J. New features about tau function and dysfunction. Biomolecules. 2016;6:E21. http://dx.doi.org/10.3390/biom6020021
- Ramser EM, Gan KJ, Decker H, Fan EY, Suzuki MM, Ferreira ST, Silverman MA. Amyloid-oligomers induce tau-independent disruption of BDNF axonal transport via calcineurin activation in cultured hippocampal neurons. Mol Biol Cell. 2013;24:2494–505. http://dx.doi.org/10.1091/mbc.e12-12-0858
- Brunden KR, Zhang B, Carroll J, Yao Y, Potuzak JS, Hogan AM, et al. Epothilone D improves microtubule density, axonal integrity, and cognition in a transgenic mouse model of tauopathy. J Neurosci. 2010:30:13861–6. http://dx.doi.org/10.1523/JNEUROSCI.3059-10.2010
- Wischik CM, Edwards PC, Lai RY, Roth M, Harrington CR. Selective inhibition of Alzheimer diseaselike tau aggregation by phenothiazines. Proc Nat Aca Sci U S A. 1996;93:11213–18. http://dx.doi. org/10.1073/pnas.93.20.11213
- Wischik CM, Staff RT, Wischik DJ, Bentham P, Murray AD, Storey JM, et al. Tau aggregation inhibitor therapy: An exploratory phase 2 study in mild or moderate Alzheimer's disease. J Alzheimers Dis. 2015;44:705–20. http://dx.doi.org/10.3233/JAD-142874
- 39. Gauthier S, Feldman HH, Schneider LS, Wilcock GK, Frisoni GB, Hardlund JH, et al. Efficacy and safety of tau-aggregation inhibitor therapy in patients with mild or moderate Alzheimer's disease: A randomised, controlled, double-blind, parallel-arm, phase 3 trial. Lancet. 2016;388:2873–84. http://dx.doi.org/10.1016/S0140-6736(16)31275-2
- Wilcock GK, Gauthier S, Frisoni GB, Jia J, Hardlund JH, Moebius HJ, et al. Potential of low dose leuco-methylthioninium bis (Hydromethanesulphonate) (LMTM) monotherapy for treatment of mild Alzheimer's disease: Cohort analysis as modified primary outcome in a phase III clinical trial. J Alzheimers Dis. 2018;61:435–57. http://dx.doi.org/10.3233/JAD-170560
- Pluta R, Barcikowska M, Misicka A, Lipkowski AW, Spisacka S, Januszewski S. Ischemic rats as a model in the study of the neurobiological role of human β-amyloid peptide. Time-dependent disappearing diffuse amyloid plaques in brain. NeuroReport. 1999;10:3615–19. http://dx.doi. org/10.1097/00001756-199911260-00028
- 42. Pluta R, Misicka A, Barcikowska M, Spisacka S, Lipkowski AW, Januszewski S. Possible reverse transport of beta-amyloid peptide across the blood-brain barrier. Acta Neurochir Suppl. 2000;76:73–7. http://dx.doi.org/10.1007/978-3-7091-6346-7_15
- 43. Pedersen JT, Sigurdsson EM. Tau immunotherapy for Alzheimer's disease. Trends Mol Med. 2015;21:394–402. http://dx.doi.org/10.1016/j.molmed.2015.03.003
- Pluta R, Ułamek-Kozioł M, Januszewski S, Czuczwar SJ. Tau protein dysfunction after brain ischemia. J Alzheimers Dis. 2018;66:429–37. http://dx.doi.org/10.3233/JAD-180772
- Novak P, Schmidt R, Kontsekova E, Zilka N, Kovacech B, Skrabana R, et al. Safety and immunogenicity of the tau vaccine AADvac1 in patients with Alzheimer's disease: A randomised, doubleblind, placebo-controlled, phase 1 trial. Lancet Neurol. 2017;16:123–34. http://dx.doi.org/10.1016/ S1474-4422(16)30331-3
- 46. Novak P, Schmidt R, Kontsekova E, Kovacech B, Smolek T, Katina S, et al. FUNDAMANT: An interventional 72-week phase 1 follow-up study of AADvac1, an active immunotherapy against tau protein pathology in Alzheimer's disease. Alzheimers Res Ther. 2018;10:108. http://dx.doi.org/10.1186/s13195-018-0436-1

- Relkin NR, Thomas RG, Rissman RA, Brewer JB, Rafii MS, van Dyck CH, et al. A phase 3 trial of IV immunoglobulin for Alzheimer disease. Neurology. 2017;88:1768–75. http://dx.doi.org/10.1212/ WNL.000000000003904
- Medina M. An overview on the clinical development of tau-based therapeutics. Int J Mol Sci. 2018;19:1160. http://dx.doi.org/10.3390/ijms19041160
- Novak P, Kontsekova E, Zilka N, Novak M. Ten years of tau-targeted immunotherapy: The path walked and the roads ahead. Front Neurosci. 2018;12:798. http://dx.doi.org/10.3389/fnins.2018.00798
- Lünemann JD, Nimmerjahn F, Dalakas MC. Intravenous immunoglobulin in neurology—Mode of action and clinical efficacy. Nat Rev Neurol. 2015;11:80–9. http://dx.doi.org/10.1038/nrneurol. 2014.253
- Krestova M, Hromadkova L, Bilkova Z, Bartos A, Ricny J. Characterization of isolated tau-reactive antibodies from the IVIG product, plasma of patients with Alzheimer's disease and cognitively normal individuals. J Neuroimmunol. 2017;313:16–24. http://dx.doi.org/10.1016/j.jneuroim.2017.09.011
- 52. Sigurdsson EM. Tau immunotherapy. Neurodegener Dis. 2016;16:34–8. http://dx.doi.org/10.1159/ 000440842
- Mullard A. Stem-cell discovery platforms yield first clinical candidates. Nat Rev Drug Discov. 2015;14:589–91. http://dx.doi.org/10.1038/nrd4708
- 54. Budur KK, West T, Braunstein JB, Fogelman I, Bordelon YM, Litvan I, et al. Results of a phase 1, single ascending dose, placebo-controlled study of ABBV-8E12 in patients with progressive supranuclear palsy and phase 2 study design in early Alzheimer's disease. Alzheimers Dement. 2017;13:P599–600. http://dx.doi.org/10.1016/j.jalz.2017.07.241
- 55. Doody R. Developing disease-modifying treatments in Alzheimer's disease—A perspective from Roche and Genentech. J Prev Alzheimers Dis. 2017;4:264–72.
- Mably AJ, Kanmert D, Mc Donald JM, Liu W, Caldarone BJ, Lemere CA, et al. Tau immunization: A cautionary tale? Neurobiol Aging. 2015:36:1316–32. http://dx.doi.org/10.1016/j.neurobiolaging. 2014.11.022
- 57. Guo T, Noble W, Hanger DP. Roles of tau protein in health and disease. Acta Neuropathol. 2017;133:665–704. http://dx.doi.org/10.1007/s00401-017-1707-9
- Li C, Götz J. Tau-based therapies in neurodegeneration: Opportunities and challenges. Nat Rev Drug Discov. 2017;16:863–83. http://dx.doi.org/10.1038/nrd.2017.155
- Viña J, Sanz-Ros J. Alzheimer's disease: Only prevention makes sense. Eur J Clin Invest. 2018; 48:e13005. http://dx.doi.org/10.1111/eci.13005

Targeting Metal Homeostasis as a Therapeutic Strategy for Alzheimer's Disease

Antonietta Vilella^{1*} • Eleonora Daini^{1*} • Chiara A. De Benedictis^{2*} • Andreas M. Grabrucker^{2,3,4}

¹Department of Biomedical, Metabolic and Neural Sciences, University of Modena and Reggio Emilia, Modena, Italy; ²Cellular Neurobiology and Neuro-Nanotechnology Lab, Department of Biological Sciences, University of Limerick, Limerick, Ireland; ³Bernal Institute, University of Limerick, Limerick, Ireland; ⁴Health Research Institute (HRI), University of Limerick, Limerick, Ireland

Author for correspondence: Andreas M. Grabrucker, Cellular Neurobiology and Neuro-Nanotechnology Lab, Department of Biological Sciences, Bernal Institute of University of Limerick, Limerick, Ireland. Email: andreas.grabrucker@ul.ie

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch5

Abstract: Trace metals play an important role in the pathophysiology of amyloid precursor protein, amyloid beta, and tau, the key molecules involved in Alzheimer's disease. Altering trace metal concentrations in the brain has been explored as a therapeutic strategy for Alzheimer's disease. It is not only the accumulation of metals that drives amyloid beta aggregation, but also the lack of sufficient trace metals for other biological processes created through sequestration by amyloid beta that affects brain health and function. Thus, balancing metal levels to achieve therapeutic effects is an intricate process. This chapter summarizes the role of trace metals in Alzheimer's disease and highlights the preclinical and clinical studies targeting metal homeostasis in animal models and humans. It further discusses recent developments in pharmacological approaches targeting metals in Alzheimer's disease and on current translational research, for example, nanomedicine.

*These authors contributed equally.

Copyright: The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

In: *Alzheimer's Disease: Drug Discovery*. Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

Keywords: ionophores; metal protein attenuating compounds; polymeric nanoparticles; trace metals in Alzheimer's disease; zinc metallochaperones

INTRODUCTION

Over the last decade, the literature covering the roles of metals in biological systems has increased considerably. Metallobiology has become a useful tool in investigating the connection between metals, metalloproteins, and neurodegenerative diseases and other disorders of the central nervous system (CNS). The primary risk factor for Alzheimer's Disease (AD) is aging (1, 2). However, evidence shows that dysregulation of trace metals such as copper (Cu), zinc (Zn), iron (Fe), aluminum (Al), manganese (Mn), and other elements such as lead (Pb) and cadmium (Cd) is also associated with AD (3). Fe, Zn, Cu, and Mn are essential for various biochemical and physiological processes in humans, playing roles in structural and regulatory functions in proteins, cellular signaling pathways, and oxygen transport. In the human body, the distribution of metals is different for each organ. For example, in the human brain, the most abundant trace metal is iron, mostly found as heme iron (iron bound to hemoglobin in red blood cells). Iron is essential not only for the transport of oxygen but also as a cofactor for a series of enzymes that play a role in the synthesis of neurotransmitters such as tryptophan hydroxylase and tyrosine hydroxylase (4, 5). Additionally, iron has several functions in the CNS including the regulation of synaptic plasticity (6), the myelination of neuronal axons (7), and the regulation of the neuronal energy status (8). Zinc is the second most copious trace element in the human brain (9). Like iron, zinc is a modulator of synaptic plasticity. In addition to this, it can regulate neurogenesis, neuronal migration, and differentiation, and plays a role in neurotransmission (10–12). Zinc is a fundamental structural and catalytic component of proteins, acting as a cofactor for over 300 enzymes. It also gives stability to various transcription factors. Finally, zinc has neuroprotective functions, and can protect against oxidative stress (13, 14). Copper is the third most prevalent trace metal in the brain. Like zinc, copper acts as a cofactor for a series of enzymes and plays a vital role in the biosynthesis of neurotransmitters (15). Manganese is a cofactor for several enzymes, including glutamine synthetase, pyruvate carboxylase, arginase, MnSOD, and protein serine/threonine phosphatase-1. Mn is also involved in a series of Mn-sensitive pathways, such as the ATM-p53 pathway (16).

Thus, trace metals are involved in enzyme activation, catalysis, regulation of gene expression, and protection against reactive oxygen species (ROS). Therefore, they are fundamental to appropriate brain growth, development, and function. The homeostasis of these charged elements is tightly regulated at the blood–brain barrier (BBB); metal ions are actively transported across the BBB by several transport proteins. Any disruption of metal balance, which can be caused by impairments in the processes of absorption, distribution, and excretion, or mutations in metal transport or metal-binding proteins, or by the competition between metals for the binding sites (e.g., Zn and Cu), can lead to an imbalance of essential trace metals or increased levels of non-essential trace elements (17, 18).

METALS AND THE AMYLOID PRECURSOR PROTEIN

In AD, metal imbalances occur in the brain (3, 19) as a result of the interactions between the metals and the critical proteins amyloid precursor protein (APP) and its products, amyloid beta (A β), and tau (Figure 1). APP, which is mainly expressed



Figure 1. Trace metals interactions with APP, Aβ, and tau. (A) APP can be considered to act as a metalloprotein. It has two putative metal-binding sides with which it binds Cu and Zn ions. Fe is involved in the regulation of APP translation, and APP may mediate Fe export of neurons. Aβ is derived from APP cleavage through β - and γ -secretases. Cu and Zn indirectly influence Aβ generation by interacting with all three secretases (α , β , and γ) involved in the cleavage of APP. (B) A β directly binds Zn and Cu ions after enzymatic cleavage. Aggregation of A β into insoluble fibrils is mediated by the interaction of A β with Zn and Cu. Chelation of Zn and Cu results in the enrichment of these ions in amyloid plaques. A β may also bind Fe, and changes in Fe metabolism occur after the development plaques. Al has been detected in A β plaques suggesting an interaction of Al with A β , and Al and Hg promote the accumulation of A β aggregates *in vitro*. (C) NFTs include trace metals such as Zn, Cu, Fe, and Mn, but also Pb, Cd, Hg, and Al have been detected. Trace metals may promote tau hyperphosphorylation and induce tau aggregation. A direct interaction between Zn and tau has been reported. Similarly, a selective binding between tau and Cu was found. Fe, as Zn and Cu, interacts with some isoforms of tau. Al, Cd, Pb, and Hg may not directly bind tau, but influence NFT formation through secondary processes.

86

in the human brain, is involved in neuronal cell migration and neurite outgrowth (20). Copper binds to APP between residues 142 and 166 (21), playing essential roles in the structural stability of the protein, including folding stability and homodimerization, and expression levels (22). Zinc can also interact with APP using the site located between amino acid positions 170 and 188, regulating the homodimerization of APP (23, 24). In addition, there is bidirectional regulation between iron and APP: iron can directly control APP translation, and APP levels control the export of iron in neurons (25). The A β protein is derived from APP in the process of sequential proteolysis realized by $\hat{\beta}$ and γ - secretases. Insoluble amyloid fibrils of AB are associated with metals like Zn, Cu, and Fe and represent a pathological feature of AD (26). Zinc plays a role in the stabilization of amyloid fibrils (27), binding to AB at histidine (His)13 and His6 sites (28). High zinc concentrations have been found at the level of senile plaques in postmortem tissues of patients affected by AD and in plaques of genetic AD mouse models, underlining the critical role of zinc in this process (26). Copper is considered to play a significant role in the neurotoxicity of Aβ. The monomeric Aβ exhibits three highaffinity His Cu-binding sites: His6, His13, and His14. They are known to form a tetragonal complex with copper ions along with the Nterminal amino group and aspartate (29). In addition to this, copper is involved in the formation of β -sheet structures, the precursors of the toxic aggregates of the fibrillary form of A β . Several studies have shown that, in transgenic AD mouse models, Cu chelators could inhibit the accumulation of A β (30, 31). Iron also interacts with A β at the binding sites of Asp1, Glu3, and the three His residues: His6, His13, and His14. The consequence of this interaction is the release of free radicals via Fenton chemistry (32). Other metals such as Al and Hg have been identified in A β plaques. While the role of Al in the AD pathology is still not clear, Hg promotes the accumulation of AB deposits in vitro. Cd and Mn are also involved in increasing the AB accumulation (33).

Another pathological hallmark of AD is hyperphosphorylation of the protein tau that leads to the formation of abnormal intracellular structures known as neurofibrillary tangles (NFTs). Zn, Cu, Fe, Mg, Mn, Pb, Cd, Hg, and Al have been found in these structures. Zinc interacts with tau using the serine (Ser) and proline (Pro) binding sites, using two cysteine (Cys) residues (C291 and C322), or threonine (Thr) and Pro sites (34). Zinc also interacts indirectly with tau, through the activation of kinase and phosphatase pathways (35). The importance of copper in the hyperphosphorylation of protein tau via the activation of the cyclindependent kinase (CDK)5/p25 complex is still controversial, although high levels of Cu have been reported in amyloid plaques and NFTs (36, 37). Iron (Fe³⁺) is also reported to interact with NFTs through His residues (38). Like Cu, Fe can also indirectly contribute to hyperphosphorylation of tau, by activating the CDK5/ p25 complex and GSK-3 β and MAP kinases (39).

The abundance of $A\beta$ and the formation of NFTs lead to an imbalance in trace metals that interact with tau and $A\beta$. It is both the interaction with the hallmark proteins of AD and the loss of trace metals in their physiological processes due to excessive binding by $A\beta$ and tau that contributes to the pathogenesis of AD. For example, in postmortem AD brains, increased Mn concentration, abnormal tau aggregation, and consequent hyperphosphorylation of tau caused by GSK-3 β kinase, have been observed (40). The presence of toxic trace metals has also been reported in AD. For example, Cd is involved in the activation of GSK-3 β kinase, inducing hyperphosphorylation of tau (41). Pb plays a role in the modulation of tau by regulating the activity of (CDK)5/p25 complex and GSK-3 β kinase (42). To modify A β aggregation, plaques, and the formation of NFTs, and to return essential trace metals to their effector proteins, balancing levels of trace metals is necessary and has been pursued as therapeutic strategy with some success in the past (43).

NUTRITIONAL INTERVENTIONS

The essential metals (Na, K, Ca, Mg) and essential trace metals (Fe, Zn, Cu, Mn, Co, Mo) are obtained almost exclusively from the diet. Trace metal deficiencies may affect the elderly (44, 45) and alter immune function (46). While some metaanalyses of studies analyzing the nutrient status of patients with AD found no significant differences in Fe, Zn, and Cu status (47, 48), more recent meta-analyses revealed significantly lower brain Cu (49) and serum Mn levels (50). However, subclinical trace metal deficiencies are hard to diagnose (51) and may not be detected in many individuals. Considering this, nutritional supplementation studies may provide additional insights. A systematic review found that the amount of caloric intake and type of diet may influence AD onset and progression (52). For example, reports show that a Mediterranean diet, which is naturally rich in metals, such as Mg, K, Ca, and Zn, (53), can have a neuroprotective effect that may be relevant to AD (54). In mouse models of AD, zinc supplementation prevented cognitive deficits (55). In humans, despite case-control and postmortem studies showing decreased systemic Zn levels in AD patients (47, 56), studies investigating the effects of zinc supplementation provided inconclusive results. Zinc supplementation also reduces copper uptake due to an antagonistic relationship between the two metals (57). A diet low in Cu has been suggested as beneficial for AD (58–60), but a recent study does not support this hypothesis (61).

Trace metal homeostasis in AD is affected at cellular and sub-cellular levels in the brain and, thus, therapeutic strategies involving their manipulation may require more targeted approaches than dietary supplementation/restriction. However, trace metal supplementation in AD is still understudied. In addition to their direct effects on physiological processes in the brain, they may exert indirect effects through the microbiota composition of the gut (62). Regardless, maintaining adequate trace metal status will affect general health positively and may modify AD phenotypes, for example, by lowering pro-inflammatory responses and oxidative stress (63).

TARGETED INTERVENTIONS: DELIVERY OF METAL-ATTENUATING COMPOUNDS

Altered homeostasis of certain essential and non-essential trace metals can be the cause, as well as the consequence of AD pathology. Abnormal trace metal levels have been associated with neuroinflammation, mitochondrial dysfunction, oxidative stress, and protein aggregation (19). The AD hallmark proteins A β and tau are metal-binding proteins. Therefore, metal chaperones and/or metal chelators may help modulate and counteract AD progression. Metal chaperones or metallochaperones are a class of compounds that shuttle metal ions to specific intracellular targets. In contrast, metal chelators or buffers function to exclude or deplete metals from discrete cellular compartments to limit biological or pathological interactions of essential metal ions. Cumulatively, these processes serve to maintain tight regulatory control over cellular metal ion homeostasis such that the intracellular concentration of freely available metal ions (like Cu and Zn) is close to zero, and sufficient metals are available for crucial metal-binding proteins. Among the metalfocused interventions (Table 1), metal protein attenuating compounds

TABLE 1

Metal interventions in preclinical and clinical trials

Drug	Targeted metal	Species	Effect of the drug	Reference
Clioquinol (CQ)	Fe, Cu, Zn	Mouse: Tg2576	Reduction of sedimentable A β (50%), plaques surface area and A β serum levels and increase in A β soluble fraction	(30)
Clioquinol (CQ)	Fe, Cu, Zn	Human: severe AD patients	Prevention of cognitive deterioration and reduction of $A\beta$ plasma levels	(64)
DP-109	Fe, Cu, Zn	Mouse: Tg2576	Reduction of A β plaques and CAA	(66)
PBT2	Fe, Cu, Zn	Mouse: APPswe/ PS1 dE9 and Tg2576	Decrease in soluble and insoluble Aβ levels and plaques burden and improvement in learning and memory in the MWM	(67)
PBT2	Fe, Cu, Zn	Human: early/mild AD patients	Reduction of Aβ plasma and CSF levels and improvement of executive tasks	(70, 71)
DFO	Al, Fe	Human: AD patients	Reduction of mental deterioration	(81)
Ferelex-G	Al, Fe	Human: postmortem AD brain	High Al/Fe removal bound to hyperphosphorylated tau	(79)
DFO	Al, Fe	Mouse: APP/PS1	Reduction of Aβ, improvement in learning and memory retention	(82)
TETA	Cu	Mouse: APP/PS1 and Tg2576	Reduction of Aβ deposition in APP/PS1 but not in Tg2576 mice	(73)
PDTC	Cu	Mouse: APP/PS1	Improvement of spatial learning but no reduction of $A\beta$ levels	(75)
Zn7MT3	Zn	Mouse: Tg2576	Reduction of behavioral deficits, Aβ aggregation, oxidative damage and neurodegeneration	(83)
XH1	Fe, Cu, Zn	Mouse: APP/PS1	Reduction of Aβ load	(87)

AD, Alzheimer's disease; CAA, cerebral amyloid angiopathy; CSF, cerebrospinal fluid; MWM, Morris water maze.

(MPACs) have been employed in both preclinical animal studies and human clinical trials. MPACs are metal chelators with a moderate affinity for metal ions and can cross the BBB. They correct abnormal metal interactions and have subtle effects on metal homeostasis. Therefore, they may limit metal binding to plaques, inhibit Zn and Cu-induced oligomerization of A β , and consequently reduce A β aggregation and toxicity (64).

Clioquinol (CQ, 5-chloro-7-iodo-8-hydroxyquinoline, MW = 305.5) is a hydrophobic 8-hydroxy (OH) quinoline derivative and an archetype of MAPCs. CQ freely crosses the BBB and is a moderate chelator of Fe, Cu, and Zn. Additionally, with its ionophore activity, it works as a metal chaperon relocating metal ions into cells (65). Originally developed as an oral antiparasitic agent, CQ treatment significantly prevents the cognitive decline of patients with moderately severe AD, as demonstrated in a phase II clinical trial (64). The slowed cognitive decline was also accompanied by a transient decrease in $A\beta_{1-42}$ plasma levels. These results indicate the potential of CQ and related compounds as treatments for AD and have sparked further interest in MPACs. Also, in animal models of AD such as Tg2576 mice, chronic treatment with CQ can decrease $A\beta$ loads without adverse effects (30). Treated mice showed significant improvements in general behavior and neurotoxicity, in accordance with the results obtained in severely affected AD patients (64).

DP-109 is a diester derivative of BAPTA (1,2-bis(2-aminophenyloxy)ethane-N,N,N',N'-tetra acetic acid), a widely-used calcium chelator, that is able to cross the BBB and is designed to selectively chelate transition metals such as Zn, Cu, and Fe within membrane compartments (increased efficacy for divalent metals). Like CQ, prolonged administration of DP-109 to female Tg2576 mice markedly reduced the amyloid plaque load and the degree of cerebral amyloid angiopathy (CAA) (66).

PBT2 belongs to the second-generation MPACs and is a highly soluble derivative of 8-OH quinoline that lacks iodine. With increased BBB permeability and ionophore activity compared to CQ, PBT2 reduced A β levels and improved cognitive abilities in Tg2576 and APP/PS1 AD mouse models (67, 68). In these mice, both acute and chronic dosing resulted in a significant decrease in A β plaque burden and interstitial fluid A β levels, a significant increase in synaptophysin levels, and restoration of hippocampal dendritic spine density (69). In a 12-week phase-II clinical trial of early AD patients, PBT2 was also found to lower cerebrospinal fluid A β_{42} concentrations and reverse frontal lobe functional deficits significantly, reducing executive dysfunction; however, no significant effects on cognition and memory were reported (70, 71).

The ionophore activity (the selective A β -metal-ion binding) of these compounds seems more promising than metal-chelation (metal-depletion) in counteracting AD pathology. While CQ and PBT2, possessing both features, are effective in reducing sedimented A β , triethylenetetramine (TETA), a classic hydrophilic Cu chelator, is not (72). TETA, a well-characterized anti-diabetic molecule, acts as a highly selective divalent copper (Cu(II)) chelator that prevents or reverses diabetic copper overload, thereby suppressing oxidative stress. Recent data showed that TETA disuccinate or trientine treatment also reduces A β production and deposition in APP/PS1 mice but not in Tg2576 mice in the same experimental paradigm in which CQ was effective (73). Given the

90

hybrid action of many of these compounds, the original proposed name shifted from MAPCs to "ionophores" and finally to "metallochaperones" (74). Other molecules with metal ion chaperone properties, like pyrrolidine dithiocarbamate (PDTC) and copper-bis(thiosemicarbazonoto) complexes, have also proved to ameliorate cognitive functions in AD mouse models, though they did not affect A β (75, 76). On the contrary, D-penicillamine chelates Cu by forming a stable complex with thiol groups and allowing renal copper excretion. A clinical pilot study indicates that D-penicillamine can reduce oxidative stress, but not the clinical progression of AD (77).

While Zn, Fe, and Cu mainly bind to A β (78), Al and Fe accumulate and bind to hyperphosphorylated tau in neurons with NTFs (79). As Fe binds to both misfolded proteins, Fe is a potential target for chelation therapy. A 2-year long, single-blind clinical study showed that sustained intramuscular administration of desferrioxamine mesylate (DFO), an Al/Fe metal chelator, slows the clinical progression of AD-associated dementia (80). Unfortunately, its hydrophilic nature and large molecular size limit its absorption across the gastrointestinal tract and prevent it from penetrating the BBB (80), inducing undesired side effects such as systemic metal depletion. All these features have prevented its final use in the clinics (81). More recent evidence showed that intranasal administration of DFO is accompanied by a reduction in A β burden/ load as well as improved learning and memory abilities in APP/PS1 mice (82) confirming, again, the beneficial effects of this drug on AD pathology.

Other Fe chelators have been developed in the last 20 years. Feralex-G, for instance, showed enhanced chelating activity for Al/Fe associated with hyperphosphorylated tau compared to DFO (79).

Chelators such as TETA, penicillamine, and DFO are routinely used in the treatment of metal overload disorders, such as Wilson's disease. However, these molecules are hydrophilic and exert their effects by systemic depletion of metals and are poorly absorbed through the BBB with limited bioavailability. The efficacy of chelation therapy is still a matter of debate (74). Several metal-binding proteins are currently being tested as AD treatment. Metallothionein 3 (MT-3), a key regulator of metal homeostasis in the brain, has been found to decrease in AD patients. Intraventricular injection of Zn-loaded MT-3 (Zn/MT3) attenuates behavioral deficits, A β aggregation, oxidative damage, and neurodegeneration in the Tg2576 AD mouse model. This drug allows the exchange of A β Cu with Zn making A β ROS inert (83).

Finally, another group of molecules has been characterized for their ability to modulate APP expression, leading to a novel therapeutic approach aimed at reducing the amyloidogenic process (84). A common feature of these molecules, to which the previously mentioned DFO belongs, is their ability to recognize an IRE stem-loop upstream to the start codon of the APP transcript at the APP-5'UTR region (84, 85). Like tetrathiomolybdate (TM) (a Cu chelator) and dimercapto-propanol (a Pb and Hg chelator), DFO reduces A β secretion and restrains the expression of APP holo-protein *in vitro* (84, 86). Similarly, XH1, a bifunctional compound capable of both APP binding and metal-chelation, was able to lower APP expression in the SH-SY5Y cell line and reduce A β burden in APP/PS1 mice (87). Several other chelators with multifunctional properties with potential applications in AD treatment exist (88). However, so far, they have mainly been tested *in vitro*, and the beneficial effects they exert cannot all be attributed to their interaction with metals.

IADLE 2	Novel app	roaches in AD using DD5s	
Delivery system	Function	Effect of the drug	Reference
G7-Zn-NP	Zinc delivery	Plaque size reduction and decrease in pro-inflammatory cytokine release <i>in vivo</i>	(92)
ZnONP	Zinc oxide delivery	Dose-dependent inhibition of fibrillar amyloid growth and β -sheet formation in vitro	(93)
CQ-MSNP	CQ metal chelator	Inhibition of Cu2+-induced Aβ40 aggregation in vitro	(94)
Nano-N2PY	Iron chelator	Inhibition of A β aggregation in human cortex in vitro	(95)
CQ-MSNP Nano-N2PY	CQ metal chelator Iron chelator	growth and β-sheet formation <i>in vitro</i> Inhibition of Cu2 ⁺ -induced Aβ40 aggregation <i>in vitro</i> Inhibition of Aβ aggregation in human cortex <i>in vitro</i>	(94) (95)

DDSs, drug delivery systems; AD, Alzheimer's disease.

TARGETED INTERVENTIONS: DELIVERY OF TRACE METALS

Although many studies support the efficacy of chelation therapy *in vitro*, the translation of metal chelators into clinical studies is currently limited due to their inability to cross the BBB easily: they are hydrophilic, and many are neurotoxic at high concentrations. The use of innovative drug delivery systems (DDSs) such as nanoparticles designed to overcome the BBB could be considered as a strategic approach in the prevention and treatment of neurological and neurodegenerative diseases such as AD (89–91). Several sustained-release nanocarriers have been studied for the treatment of AD; in particular, polymeric nanoparticles (NPs) allow a significant increase in bioavailability and effectiveness of the administered drugs. Although the efficacy and advantage of these systems have been supported by several experimental studies (Table 2), the translational efficacy is yet to be proven.

CONCLUSION

The role of trace metals in AD is complicated as they are involved in both pathological and physiological processes in the brain. In addition, trace metals such as zinc, which binds to more than 10% of proteins encoded in the human genome (96), regulate a plethora of processes whose effects are neither fully understood nor assessable in preclinical and clinical studies. Furthermore, levels of one trace metal affect the levels of other metals (19). Therefore, any therapeutic strategy targeting trace metal levels in AD needs to be specific in terms of drug delivery and consider its effect on the concentration of a particular metal. Preclinical and clinical studies attempting to manipulate trace metals, although not entirely successful, were among the most promising approaches for the treatment for AD. Metal chelation and chaperoning successfully reduced A β

92



Figure 2. Targeting trace metals as therapeutic strategy in AD. Metal chelators and ionophores can prevent metal binding to $A\beta$, inhibit oligomerization of $A\beta$, and consequently reduce its aggregation and toxicity. Altering metal homeostasis by depletion, delivery, and chaperoning was reported to decrease $A\beta$ plaque burden and interstitial fluid $A\beta$ levels, positively affect synaptic spine density, and suppress oxidative stress. Zinc delivery further reduces inflammation, while iron chelation modifies tau phosphorylation, and APP expression.

generation and aggregation, synaptic dysfunction, neuroinflammation, oxidative stress, and induced degradation of senile plaques (Figure 2). In addition, effects on cognitive abilities were observed in some studies. To further develop treatment strategies targeting metal homeostasis, it is necessary to clearly understand the role of trace metals in AD. It is important to explore the role of trace metals, such as zinc, in physiological processes that could be exploited as therapeutic targets apart from A β aggregates, such as synaptic plasticity mediated by zinc-dependent SHANK proteins (97) or reducing excitotoxicity and A β burden by zinc-binding S100B proteins (98–101). A re-distribution and balancing of metals may be necessary between neurons, astrocytes, and microglial cells to optimize effects on synapses and inflammatory processes. This could be achieved by targeted delivery of trace metals to A β aggregates and NFTs. To this end, novel compounds, combinations of novel metal chelators and chaperones, and novel delivery platforms will be needed.

Conflict of Interest: The authors declare no potential conflict of interest with respect to research, authorship and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- Hyman BT, Trojanowski JQ. Consensus recommendations for the postmortem diagnosis of Alzheimer disease from the National Institute on Aging and the Reagan Institute Working Group on diagnostic criteria for the neuropathological assessment of Alzheimer disease. J Neuropathol Exp Neurol. 1997;56(10):1095–7. http://dx.doi.org/10.1097/00005072-199710000-00002
- Hyman BT, Phelps CH, Beach TG, Bigio EH, Cairns NJ, Carrillo MC, et al. National Institute on Aging-Alzheimer's Association guidelines for the neuropathologic assessment of Alzheimer's disease. Alzheimers Dement. 2012;8(1):1–13. http://dx.doi.org/10.1016/j.jalz.2011.10.007
- 3. Wang P, Wang ZY. Metal ions influx is a double edged sword for the pathogenesis of Alzheimer's disease. Ageing Res Rev. 2017;35:265–90. http://dx.doi.org/10.1016/j.arr.2016.10.003
- Yehuda S, Youdim MB. Brain iron: A lesson from animal models. Am J Clin Nutr 1989;50(3 Suppl.): 618–25; discussion 625–9. http://dx.doi.org/10.1093/ajcn/50.3.618
- 5. Eisenstein RS. Iron regulatory proteins and the molecular control of mammalian iron metabolism. Annu Rev Nutr. 2000;20:627–62. http://dx.doi.org/10.1146/annurev.nutr.20.1.627
- Clardy SL, Wang X, Zhao W, Liu W, Chase GA, Beard JL, et al. Acute and chronic effects of developmental iron deficiency on mRNA expression patterns in the brain. J Neural Transm Suppl. 2006;71:173–96. http://dx.doi.org/10.1007/978-3-211-33328-0_19
- Kwik-Uribe CL, Gietzen D, German JB, Golub MS, Keen CL. Chronic marginal iron intakes during early development in mice result in persistent changes in dopamine metabolism and myelin composition. J Nutr. 2000;130(11):2821–30. http://dx.doi.org/10.1093/jn/130.11.2821
- Dallman PR. Biochemical basis for the manifestations of iron deficiency. Annu Rev Nutr. 1986;6: 13–40. http://dx.doi.org/10.1146/annurev.nu.06.070186.000305
- De Benedictis CA, Raab A, Ducie E, Howley S, Feldmann J, Grabrucker AM. Concentrations of essential trace metals in the brain of animal species – A comparative study. Brain Sci. 2020;10(7). http://dx.doi.org/10.3390/brainsci10070460
- 10. Takeda A. Zinc homeostasis and functions of zinc in the brain. Biometals. 2001;14(3–4):343–51. http://dx.doi.org/10.1023/a:1012982123386
- 11. Fukada T, Yamasaki S, Nishida K, Murakami M, Hirano T. Zinc homeostasis and signaling in health and diseases: Zinc signaling. J Biol Inorg Chem. 2011;16(7):1123–34. http://dx.doi.org/10.1007/ s00775-011-0797-4
- 12. Levenson CW, Morris D. Zinc and neurogenesis: Making new neurons from development to adulthood. Adv Nutr. 2011;2(2):96–100. http://dx.doi.org/10.3945/an.110.000174
- Tapiero H, Tew KD. Trace elements in human physiology and pathology: Zinc and metallothioneins. Biomed Pharmacother. 2003;57(9):399–411. http://dx.doi.org/10.1016/s0753-3322(03)00081-7
- Prasad AS. Clinical, immunological, anti-inflammatory and antioxidant roles of zinc. Exp Gerontol 2008;43(5):370–7. http://dx.doi.org/10.1016/j.exger.2007.10.013
- Peña MM, Lee J, Thiele DJ. A delicate balance: Homeostatic control of copper uptake and distribution. J Nutr 1999;129(7):1251–60. http://dx.doi.org/10.1093/jn/129.7.1251
- 16. Horning KJ, Caito SW, Tipps KG, Bowman AB, Aschner M. Manganese is essential for neuronal health. Annu Rev Nutr. 2015;35:71–108. http://dx.doi.org/10.1146/annurev-nutr-071714-034419
- Zheng W, Monnot AD. Regulation of brain iron and copper homeostasis by brain barrier systems: Implication in neurodegenerative diseases. Pharmacol Ther 2012;133(2):177–88. http://dx.doi. org/10.1016/j.pharmthera.2011.10.006

94 Vilella A et al.

- Sensi SL, Granzotto A, Siotto M, Squitti R. Copper and zinc dysregulation in Alzheimer's disease. Trends Pharmacol Sci. 2018;39(12):1049–63. http://dx.doi.org/10.1016/j.tips.2018.10.001
- Pfaender S, Grabrucker AM. Characterization of biometal profiles in neurological disorders. Metallomics. 2014;6(5):960–77. http://dx.doi.org/10.1039/c4mt00008k
- Walsh DM, Minogue AM, Sala Frigerio C, Fadeeva JV, Wasco W, Selkoe DJ. The APP family of proteins: Similarities and differences. Biochem Soc Trans. 2007;35(Pt2):416–20. http://dx.doi.org/10.1042/ BST0350416
- Barnham KJ, McKinstry WJ, Multhaup G, Galatis D, Morton CJ, Curtain CC, et al. Structure of the Alzheimer's disease amyloid precursor protein copper binding domain. A regulator of neuronal copper homeostasis. J Biol Chem. 2003;278(19):17401–7. http://dx.doi.org/10.1074/jbc.M300629200
- Bellingham SA, Lahiri DK, Maloney B, La Fontaine S, Multhaup G, Camakaris J. Copper depletion down-regulates expression of the Alzheimer's disease amyloid-beta precursor protein gene. J Biol Chem. 2004;279(19):20378–86. http://dx.doi.org/10.1074/jbc.M400805200
- Bush AI, Multhaup G, Moir RD, Williamson TG, Small DH, Rumble B, et al. A novel zinc(II) binding site modulates the function of the beta A4 amyloid protein precursor of Alzheimer's disease. J Biol Chem. 1993;268(22):16109–12.
- Bush AI, Pettingell WH, de Paradis M, Tanzi RE, Wasco W. The amyloid beta-protein precursor and its mammalian homologues. Evidence for a zinc-modulated heparin-binding superfamily. J Biol Chem. 1994;269(43):26618–21.
- Venti A, Giordano T, Eder P, Bush AI, Lahiri DK, Greig NH, et al. The integrated role of desferrioxamine and phenserine targeted to an iron-responsive element in the APP-mRNA 5'-untranslated region. Ann N Y Acad Sci. 2004;1035:34–48. http://dx.doi.org/10.1196/annals.1332.003
- Lovell MA, Robertson JD, Teesdale WJ, Campbell JL, Markesbery WR. Copper, iron and zinc in Alzheimer's disease senile plaques. J Neurol Sci. 1998;158(1):47–52. http://dx.doi.org/10.1016/ s0022-510x(98)00092-6
- 27. Cuajungco MP, Lees GJ. Zinc and Alzheimer's disease: Is there a direct link? Brain Res Brain Res Rev. 1997;23(3):219–36. http://dx.doi.org/10.1016/s0165-0173(97)00002-7
- Bush AI, Pettingell WH, Multhaup G, Paradis MD, Vonsattel JP, Gusella JF, et al. Rapid induction of Alzheimer A beta amyloid formation by zinc. Science. 1994;265(5177):1464–7. http://dx.doi. org/10.1126/science.8073293
- 29. Minicozzi V, Stellato F, Comai M, Dalla Serra M, Potrich C, Meyer-Klaucke W, et al. Identifying the minimal copper- and zinc-binding site sequence in amyloid-beta peptides. J Biol Chem. 2008;283(16):10784–92. http://dx.doi.org/10.1074/jbc.M707109200
- Cherny RA, Atwood CS, Xilinas ME, Gray DN, Jones WD, McLean CA, et al. Treatment with a copper-zinc chelator markedly and rapidly inhibits beta-amyloid accumulation in Alzheimer's disease transgenic mice. Neuron. 2001;30(3):665–76. http://dx.doi.org/10.1016/s0896-6273(01)00317-8
- Bush AI. Metal complexing agents as therapies for Alzheimer's disease. Neurobiol Aging. 2002;23(6):1031–8. http://dx.doi.org/10.1016/s0197-4580(02)00120-3
- 32. Telling ND, Everett J, Collingwood JF, Dobson J, van der Laan G, Gallagher JJ, et al. Iron biochemistry is correlated with amyloid plaque morphology in an Established Mouse Model of Alzheimer's disease. Cell Chem Biol. 2017;24(10):1205–15.e3. http://dx.doi.org/10.1016/j.chembiol.2017.07.014
- De Benedictis CA, Viella A, Grabrucker AM. The role of trace metals in Alzehimer's disease. In: Wisniewski T, editor. Alzheimer's disease. Brisbane: Codon Publications; 2019. Chapter 6. http:// dx.doi.org/10.15586/alzehimersdisease.2019.ch6
- Mo ZY, Zhu YZ, Zhu HL, Fan JB, Chen J, Liang Y. Low micromolar zinc accelerates the fibrillization of human tau via bridging of Cys-291 and Cys-322. J Biol Chem. 2009;284(50):34648–57. http:// dx.doi.org/10.1074/jbc.M109.058883
- 35. Kim I, Park EJ, Seo J, Ko SJ, Lee J, Kim CH. Zinc stimulates tau S214 phosphorylation by the activation of Raf/mitogen-activated protein kinase-kinase/extracellular signalregulated kinase pathway. Neuroreport. 2011;22(16):839–44. http://dx.doi.org/10.1097/ WNR.0b013e32834c0a2d
- Sayre LM, Perry G, Atwood CS, Smith MA. The role of metals in neurodegenerative diseases. Cell Mol Biol (Noisy-le-grand). 2000;46(4):731–41.
- Kitazawa M, Cheng D, Laferla FM. Chronic copper exposure exacerbates both amyloid and tau pathology and selectively dysregulates cdk5 in a mouse model of AD. J Neurochem. 2009;108(6): 1550–60. http://dx.doi.org/10.1111/j.1471-4159.2009.05901.x
- Smith MA, Harris PL, Sayre LM, Perry G. Iron accumulation in Alzheimer disease is a source of redoxgenerated free radicals. Proc Natl Acad Sci U S A. 1997;94(18):9866–8. http://dx.doi.org/10.1073/ pnas.94.18.9866
- Guo C, Wang P, Zhong ML, Wang T, Huang XS, Li JY, et al. Deferoxamine inhibits iron induced hippocampal tau phosphorylation in the Alzheimer transgenic mouse brain. Neurochem Int. 2013;62(2):165–72. http://dx.doi.org/10.1016/j.neuint.2012.12.005
- Cai T, Che H, Yao T, Chen Y, Huang C, Zhang W, et al. Manganese induces tau hyperphosphorylation through the activation of ERK MAPK pathway in PC12 cells. Toxicol Sci. 2011;119(1):169–77. http:// dx.doi.org/10.1093/toxsci/kfq308
- 41. Del Pino J, Zeballos G, Anadón MJ, Moyano P, Díaz MJ, García JM, et al. Cadmium-induced cell death of basal forebrain cholinergic neurons mediated by muscarinic M1 receptor blockade, increase in GSK-3β enzyme, β-amyloid and tau protein levels. Arch Toxicol. 2016;90(5):1081–92. http://dx.doi. org/10.1007/s00204-015-1540-7
- 42. Bihaqi SW, Bahmani A, Adem A, Zawia NH. Infantile postnatal exposure to lead (Pb) enhances tau expression in the cerebral cortex of aged mice: Relevance to AD. Neurotoxicology. 2014;44:114–20. http://dx.doi.org/10.1016/j.neuro.2014.06.008
- Yang GJ, Liu H, Ma DL, Leung CH. Rebalancing metal dyshomeostasis for Alzheimer's disease therapy. J Biol Inorg Chem. 2019;24(8):1159–70. http://dx.doi.org/10.1007/s00775-019-01712-y
- 44. Morley JE. Nutritional problems in the elderly. Bol Asoc Med P R. 1987;79(12):505–7.
- Prasad AS, Fitzgerald JT, Hess JW, Kaplan J, Pelen F, Dardenne M. Zinc deficiency in elderly patients. Nutrition. 1993;9(3):218–24.
- Haase H, Mocchegiani E, Rink L. Correlation between zinc status and immune function in the elderly. Biogerontology. 2006;7(5–6):421–8. http://dx.doi.org/10.1007/s10522-006-9057-3
- Loef M, von Stillfried N, Walach H. Zinc diet and Alzheimer's disease: A systematic review. Nutr Neurosci. 2012;15(5):2–12. http://dx.doi.org/10.1179/1476830512Y.0000000010
- Lopes da Silva S, Vellas B, Elemans S, Luchsinger J, Kamphuis P, Yaffe K, et al. Plasma nutrient status of patients with Alzheimer's disease: Systematic review and meta-analysis. Alzheimers Dement. 2014;10(4):485–502. http://dx.doi.org/10.1016/j.jalz.2013.05.1771
- de Wilde MC, Vellas B, Girault E, Yavuz AC, Sijben JW. Lower brain and blood nutrient status in Alzheimer's disease: Results from meta-analyses. Alzheimers Dement (NY). 2017;3(3):416–31. http:// dx.doi.org/10.1016/j.trci.2017.06.002
- 50. Du K, Liu M, Pan Y, Zhong X, Wei M. Association of serum manganese levels with Alzheimer's disease and mild cognitive impairment: A systematic review and meta-analysis. Nutrients. 2017;9(3):231. http://dx.doi.org/10.3390/nu9030231
- 51. Roohani N, Hurrell R, Kelishadi R, Schulin R. Zinc and its importance for human health: An integrative review. J Res Med Sci. 2013;18(2):144–57.
- Burgener SC, Buettner L, Coen Buckwalter K, Beattie E, Bossen AL, Fick DM, et al. Evidence supporting nutritional interventions for persons in early stage Alzheimer's disease (AD). J Nutr Health Aging. 2008;12(1):18–21. http://dx.doi.org/10.1007/BF02982159
- 53. Knight A, Bryan J, Murphy K. Is the Mediterranean diet a feasible approach to preserving cognitive function and reducing risk of dementia for older adults in Western countries? New insights and future directions. Ageing Res Rev. 2016;25:85–101. http://dx.doi.org/10.1016/j.arr.2015.10.005
- Solfrizzi V, Frisardi V, Seripa D, Logroscino G, Imbimbo BP, D'Onofrio G, et al. Mediterranean diet in predementia and dementia syndromes. Curr Alzheimer Res. 2011;8(5):520–42. http://dx.doi. org/10.2174/156720511796391809
- 55. Corona C, Masciopinto F, Silvestri E, Viscovo AD, Lattanzio R, Sorda RL, et al. Dietary zinc supplementation of 3xTg-AD mice increases BDNF levels and prevents cognitive deficits as well as mitochondrial dysfunction. Cell Death Dis. 2010;1:e91. http://dx.doi.org/10.1038/cddis.2010.73
- Brewer GJ, Kanzer SH, Zimmerman EA, Molho ES, Celmins DF, Heckman SM, et al. Subclinical zinc deficiency in Alzheimer's disease and Parkinson's disease. Am J Alzheimers Dis Other Demen 2010;25(7):572–5. http://dx.doi.org/10.1177/1533317510382283

- Brewer GJ. Copper excess, zinc deficiency, and cognition loss in Alzheimer's disease. Biofactors. 2012;38(2):107–13. http://dx.doi.org/10.1002/biof.1005
- Brewer GJ. The risks of copper toxicity contributing to cognitive decline in the aging population and to Alzheimer's disease. J Am Coll Nutr. 2009;28(3):238–42. http://dx.doi.org/10.1080/07315724. 2009.10719777
- Squitti R, Siotto M, Polimanti R. Low-copper diet as a preventive strategy for Alzheimer's disease. Neurobiol Aging. 2014;35(Suppl. 2):40–50. http://dx.doi.org/10.1016/j.neurobiolaging.2014.02.031
- Hsu HW, Bondy SC, Kitazawa M. Environmental and dietary exposure to copper and its cellular mechanisms linking to Alzheimer's disease. Toxicol Sci. 2018;163(2):338–45. http://dx.doi. org/10.1093/toxsci/kfy025
- Solioz M. Low copper-2 intake in Switzerland does not result in lower incidence of Alzheimer's disease and contradicts the Copper-2 Hypothesis. Exp Biol Med (Maywood). 2020;245(3):177–9. http://dx.doi.org/10.1177/1535370219899898
- Sauer AK, Grabrucker AM. Zinc deficiency during pregnancy leads to altered microbiome and elevated inflammatory markers in mice. Front Neurosci. 2019;13:1295. http://dx.doi.org/10.3389/ fnins.2019.01295
- Prasad AS. Zinc: An antioxidant and anti-inflammatory agent: Role of zinc in degenerative disorders of aging. J Trace Elem Med Biol. 2014;28(4):364–71. http://dx.doi.org/10.1016/j.jtemb.2014.07.019
- 64. Ritchie CW, Bush AI, Mackinnon A, Macfarlane S, Mastwyk M, MacGregor L, et al. Metal-protein attenuation with iodochlorhydroxyquin (clioquinol) targeting Abeta amyloid deposition and toxicity in Alzheimer disease: A pilot phase 2 clinical trial. Arch Neurol. 2003;60(12):1685–91. http://dx.doi. org/10.1001/archneur.60.12.1685
- Park MH, Lee SJ, Byun HR, Kim Y, Oh YJ, Koh JY, et al. Clioquinol induces autophagy in cultured astrocytes and neurons by acting as a zinc ionophore. Neurobiol Dis. 2011;42(3):242–51. http:// dx.doi.org/10.1016/j.nbd.2011.01.009
- Lee JY, Friedman JE, Angel I, Kozak A, Koh JY. The lipophilic metal chelator DP-109 reduces amyloid pathology in brains of human beta-amyloid precursor protein transgenic mice. Neurobiol Aging. 2004;25(10):1315–21. http://dx.doi.org/10.1016/j.neurobiolaging.2004.01.005
- 67. Adlard PA, Cherny RA, Finkelstein DI, Gautier E, Robb E, Cortes M, et al. Rapid restoration of cognition in Alzheimer's transgenic mice with 8-hydroxy quinoline analogs is associated with decreased interstitial Abeta. Neuron. 2008;59(1):43–55. http://dx.doi.org/10.1016/j.neuron.2008.06.018
- 68. Crouch PJ, Savva MS, Hung LW, Donnelly PS, Mot AI, Parker SJ, et al. The Alzheimer's therapeutic PBT2 promotes amyloid-β degradation and GSK3 phosphorylation via a metal chaperone activity. J Neurochem. 2011;119(1):220–30. http://dx.doi.org/10.1111/j.1471-4159.2011.07402.x
- 69. Adlard PA, Bush AI. Metal chaperones: A holistic approach to the treatment of Alzheimer's disease. Front Psychiatry. 2012;3:15. http://dx.doi.org/10.3389/fpsyt.2012.00015
- Lannfelt L, Blennow K, Zetterberg H, Batsman S, Ames D, Harrison J, et al. Safety, efficacy, and biomarker findings of PBT2 in targeting Abeta as a modifying therapy for Alzheimer's disease: A phase IIa, double-blind, randomised, placebo-controlled trial. Lancet Neurol. 2008;7(9):779–86. http:// dx.doi.org/10.1016/S1474-4422(08)70167-4
- Faux NG, Ritchie CW, Gunn A, Rembach A, Tsatsanis A, Bedo J, et al. PBT2 rapidly improves cognition in Alzheimer's disease: Additional phase II analyses. J Alzheimers Dis. 2010;20(2):509–16. http://dx.doi.org/10.3233/JAD-2010-1390
- 72. Bush AI. The metallobiology of Alzheimer's disease. Trends Neurosci. 2003;26(4):207–14. http:// dx.doi.org/10.1016/S0166-2236(03)00067-5
- 73. Wang CY, Xie JW, Xu Y, Wang T, Cai JH, Wang X, et al. Trientine reduces BACE1 activity and mitigates amyloidosis via the AGE/RAGE/NF-κB pathway in a transgenic mouse model of Alzheimer's disease. Antioxid Redox Signal. 2013;19(17):2024–39. http://dx.doi.org/10.1089/ars.2012.5158
- 74. Drew SC. The case for abandoning therapeutic chelation of copper ions in Alzheimer's disease. Front Neurosci. 2017;11:317. http://dx.doi.org/10.3389/fnins.2017.00317
- Malm TM, Iivonen H, Goldsteins G, Keksa-Goldsteine V, Ahtoniemi T, Kanninen K, et al. Pyrrolidine dithiocarbamate activates Akt and improves spatial learning in APP/PS1 mice without affecting beta-amyloid burden. J Neurosci. 2007;27(14):3712–21. http://dx.doi.org/10.1523/ JNEUROSCI.0059-07.2007

- Crouch PJ, Hung LW, Adlard PA, Cortes M, Lal V, Filiz G, et al. Increasing Cu bioavailability inhibits Abeta oligomers and tau phosphorylation. Proc Natl Acad Sci U S A. 2009;106(2):381–6. http:// dx.doi.org/10.1073/pnas.0809057106
- Squitti R, Rossini PM, Cassetta E, Moffa F, Pasqualetti P, Cortesi M, et al. d-penicillamine reduces serum oxidative stress in Alzheimer's disease patients. Eur J Clin Invest. 2002;32(1):51–9. http:// dx.doi.org/10.1046/j.1365-2362.2002.00933.x
- Hane F, Tran G, Attwood SJ, Leonenko Z. Cu(2+) affects amyloid-β (1-42) aggregation by increasing peptide-peptide binding forces. PLoS One 2013;8(3):e59005. http://dx.doi.org/10.1371/journal. pone.0059005
- Shin RW, Kruck TP, Murayama H, Kitamoto T. A novel trivalent cation chelator Feralex dissociates binding of aluminum and iron associated with hyperphosphorylated tau of Alzheimer's disease. Brain Res. 2003;961(1):139–46. http://dx.doi.org/10.1016/s0006-8993(02)03893-3
- Aouad F, Florence A, Zhang Y, Collins F, Henry C, Ward RJ, et al. Evaluation of new iron chelators and their therapeutic potential. Inorganica Chimica Acta. 2002;339:470–80. http://dx.doi.org/10.1016/ S0020-1693(02)01040-X
- Crapper McLachlan DR, Dalton AJ, Kruck TP, Bell MY, Smith WL, Kalow W, et al. Intramuscular desferrioxamine in patients with Alzheimer's disease. Lancet. 1991;337(8753):1304–8. http://dx.doi. org/10.1016/0140-6736(91)92978-b
- Guo C, Wang T, Zheng W, Shan ZY, Teng WP, Wang ZY. Intranasal deferoxamine reverses ironinduced memory deficits and inhibits amyloidogenic APP processing in a transgenic mouse model of Alzheimer's disease. Neurobiol Aging. 2013;34(2):562–75. http://dx.doi.org/10.1016/j. neurobiolaging.2012.05.009
- 83. Xu W, Xu Q, Cheng H, Tan X. The efficacy and pharmacological mechanism of Zn7MT3 to protect against Alzheimer's disease. Sci Rep. 2017;7(1):13763. http://dx.doi.org/10.1038/s41598-017-12800-x
- Rogers JT, Lahiri DK. Metal and inflammatory targets for Alzheimer's disease. Curr Drug Targets. 2004;5(6):535–51. http://dx.doi.org/10.2174/1389450043345272
- Bandyopadhyay S, Huang X, Cho H, Greig NH, Youdim MB, Rogers JT. Metal specificity of an ironresponsive element in Alzheimer's APP mRNA 5'untranslated region, tolerance of SH-SY5Y and H4 neural cells to desferrioxamine, clioquinol, VK-28, and a piperazine chelator. J Neural Transm Suppl. 2006;71:237–47. http://dx.doi.org/10.1007/978-3-211-33328-0_25
- Payton S, Cahill CM, Randall JD, Gullans SR, Rogers JT. Drug discovery targeted to the Alzheimer's APP mRNA 5'-untranslated region: The action of paroxetine and dimercaptopropanol. J Mol Neurosci. 2003;20(3):267–75. http://dx.doi.org/10.1385/JMN:20:3:267
- Dedeoglu A, Cormier K, Payton S, Tseitlin KA, Kremsky JN, Lai L, et al. Preliminary studies of a novel bifunctional metal chelator targeting Alzheimer's amyloidogenesis. Exp Gerontol. 2004;39 (11–12):1641–9. http://dx.doi.org/10.1016/j.exger.2004.08.016
- Nuñez MT, Chana-Cuevas P. New perspectives in iron chelation therapy for the treatment of neurodegenerative diseases. Pharmaceuticals (Basel). 2018;11(4):109. http://dx.doi.org/10.3390/ ph11040109
- Georganopoulou DG, Chang L, Nam JM, Thaxton CS, Mufson EJ, Klein WL, et al. Nanoparticle-based detection in cerebral spinal fluid of a soluble pathogenic biomarker for Alzheimer's disease. Proc Natl Acad Sci U S A. 2005;102(7):2273–6. http://dx.doi.org/10.1073/pnas.0409336102
- Andrieux K, Couvreur P. Nanomedicine as a promising approach for the treatment and diagnosis of brain diseases: The example of Alzheimer's disease. Ann Pharm Fr. 2013;71(4):225–33. http://dx.doi. org/10.1016/j.pharma.2013.04.001
- Brambilla D, Verpillot R, Le Droumaguet B, Nicolas J, Taverna M, Kóňa J, et al. PEGylated nanoparticles bind to and alter amyloid-beta peptide conformation: Toward engineering of functional nanomedicines for Alzheimer's disease. ACS Nano. 2012;6(7):5897–908. http://dx.doi.org/10.1021/ nn300489k
- Vilella A, Belletti D, Sauer AK, Hagmeyer S, Sarowar T, Masoni M, et al. Reduced plaque size and inflammation in the APP23 mouse model for Alzheimer's disease after chronic application of polymeric nanoparticles for CNS targeted zinc delivery. J Trace Elem Med Biol. 2018;49:210–21. http://dx.doi. org/10.1016/j.jtemb.2017.12.006

- Ban DK, Paul S. Nano zinc oxide inhibits fibrillar growth and uppresses cellular toxicity of lysozyme amyloid. ACS Appl Mater Interfaces. 2016;8(46):31587–601. http://dx.doi.org/10.1021/ acsami.6b11549
- 94. Yang L, Yin T, Liu Y, Sun J, Zhou Y, Liu J. Gold nanoparticle-capped mesoporous silica-based H2O2responsive controlled release system for Alzheimer's disease treatment. Acta Biomater. 2016;46: 177–90. http://dx.doi.org/10.1016/j.actbio.2016.09.010
- Liu G, Men P, Kudo W, Perry G, Smith MA. Nanoparticle-chelator conjugates as inhibitors of amyloidbeta aggregation and neurotoxicity: A novel therapeutic approach for Alzheimer disease. Neurosci Lett. 2009;455(3):187–90. http://dx.doi.org/10.1016/j.neulet.2009.03.064
- Andreini C, Banci L, Bertini I, Rosato A. Counting the zinc-proteins encoded in the human genome. J Proteome Res. 2006;5(1):196–201. http://dx.doi.org/10.1021/pr050361j
- Grabrucker AM, Schmeisser MJ, Udvardi PT, Arons M, Schoen M, Woodling NS, et al. Amyloid beta protein-induced zinc sequestration leads to synaptic loss via dysregulation of the ProSAP2/Shank3 scaffold. Mol Neurodegener. 2011;6:65. http://dx.doi.org/10.1186/1750-1326-6-65
- 98. Cristóvão JS, Figueira AJ, Carapeto A, Rodrigues MS, Cardoso I, Gomes CM. The S100B alarmin is a dual-function chaperone suppressing Aβ oligomerization through combined zinc chelation and inhibition of protein aggregation. ACS Chem Neurosci. 2020; 11(17):2753–60. http://dx.doi. org/10.1021/acschemneuro.0c00392
- 99. Cristóvão JS, Morris VK, Cardoso I, Leal SS, Martínez J, Botelho HM, et al. The neuronal S100B protein is a calcium-tuned suppressor of amyloid-β aggregation. Sci Adv 2018;4(6):eaaq1702. http://dx.doi.org/10.1126/sciadv.aaq1702
- Hagmeyer S, Cristóvão JS, Mulvihill JJE, Boeckers TM, Gomes CM, Grabrucker AM. Zinc binding to S100B affords regulation of trace metal homeostasis and excitotoxicity in the brain. Front Mol Neurosci. 2017;10:456. http://dx.doi.org/10.3389/fnmol.2017.00456
- 101. Hagmeyer S, Romão MA, Cristóvão JS, Vilella A, Zoli M, Gomes CM, et al. Distribution and relative abundance of \$100 proteins in the brain of the APP23 Alzheimer's disease model mice. Front Neurosci. 2019;13:640. http://dx.doi.org/10.3389/fnins.2019.00640

Atypical Protein Kinase C Hyperactivity in Insulin-Resistant and Insulin-Sensitive Forms of Alzheimer's Disease: A Potential Therapeutic Target

Robert V. Farese^{1,2} • Mark A. Kindy^{1,3} • Mini P. Sajan^{1,2}

¹James A. Haley Veterans Medical Center, Tampa, FL, USA; ²Department of Internal Medicine, University of South Florida College of Medicine, Tampa, FL, USA; ³School of Pharmacy, University of South Florida, Tampa, FL, USA

Author for correspondence: Robert V. Farese, James A. Haley Veterans Medical Center, 13000 Bruce B. Downs Blvd, Tampa, FL, USA. Email: rfarese@usf.edu

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch6

Abstract: Alzheimer's disease (AD) is commonly, not always, associated with insulin-resistant, hyperinsulinemic, and obesity/type-2-diabetic (O/T2D) states. Partial deficiencies of brain insulin receptor (IR) indeed occur in both O/T2D-AD and human AD, but these deficiencies can be bypassed by hyperinsulinemia, which activates atypical protein kinase *C* (aPK*C*) and β -secretase, increases A β -peptide and phospho-thr-231-tau levels, and induces memory impairments; importantly, these aberrations are reversed by reduction of liver/aPKC-dependent hyperinsulinemia or direct blockade of brain aPKC. New evidence shows that aPKC acts via nuclear factor kappa-B to increase β -secretase mRNA/protein levels in brain, where β -secretase acts on both β -amyloid precursor protein to increase AD risk and IR to limit beneficial (aPKC independent) insulin effects, particularly in normo/hypoinsulinemic AD, and liver, where β -secretase acts on IR to initiate or abet development of insulin resistance and compensatory hyperinsulinemia

In: Alzheimer's Disease: Drug Discovery. Huang X (Editor). Exon Publications, Brisbane, Australia.

ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020 **Copyright:** The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

that originates from diet-induced hepatic aPKC activation. Fortunately, agents that inhibit PKC- λ /t in brain, liver, or both effectively reduce β -secretase levels and adverse actions therein, and moreover, prevent/reverse O/T2D and AD development in mouse models. This chapter summarizes work implicating the critical role of atypical PKC in the development of liver-dependent hyperinsulinemia as a risk factor in O/T2D-associated AD and β -secretase-mediated pathological alterations in brains of O/T2D-associated and O/T2D-independent AD.

Keywords: atypical protein kinase C; β amyloid precursor protein; β secretase; insulin-resistant Alzheimer's disease; insulin-sensitive Alzheimer's disease

INTRODUCTION

In the United States, according to the Alzheimer Association, late onset (LO) Alzheimer's disease (AD) afflicts 1 in 5 women and 1 in 10 men over the age of 65, and 45% of people over the age of 85. Hyperinsulinemic forms of obesity, the associated metabolic syndrome (MetS) and type 2 diabetes mellitus (T2DM), which together afflict half of all adults in the United States, are thought to increase LO-AD risk. The extent and consequences of these interrelated disorders are enormous, in terms of morbidity, mortality, and cost. Although pathogenetic mechanisms of AD are still unsettled, both early-onset, inherited forms of AD and sporadic LO-AD are thought to be dependent on pathological increases of A β -plaques that arise from A β -peptides, whose production is initiated by the aspartyl peptidase, β -site β -amyloid precursor protein (β APP) cleaving enzyme-1 (BACE1), acting on specific amino acids of β APP (1–4). Indeed, BACE1 levels are increased in many or most, but not necessarily all, AD patients (5). In either case, BACE1 action is thought to be essential for the development of classical AD pathology, and this dependency spurred the widely heralded development of BACE1 inhibitors for AD treatment. Unfortunately, however, in human clinical trials, BACE1 inhibitors apparently produced cognitive/memory impairments, and this may reflect requirements for uncertain actions of BACE1 in cognitive/ memory functions, or development of compensatory increases in BACE1containing vesicles (6) that may be dysfunctional owing to inhibitor binding, with impairment in vesicle trafficking needed for cognitive/memory functions.

As to whether BACE1 is needed for normal cognition and memory, BACE1 knockout (KO) does not produce significant aberrations in mice (7, 8), but it is uncertain if humans have greater dependence on BACE1. It is also uncertain if vesicle trafficking needed for memory function is disrupted by BACE1 inhibitors. However, there is another approach that would avoid the use of BACE1 inhibitors in AD, that is, development of therapeutic measures to partially diminish BACE1 content to levels that sufficiently reduce A β -peptide production while maintaining essential BACE1 functions. In this regard, we recently found that the atypical protein kinase C (aPKC) signaling factor, PKC- λ/t , is critically needed for the production of BACE1 mRNA and protein (presumably BACE1 transcription), not only in the brain, where BACE1 acts upon β APP to generate A β -peptides in AD, but also in liver, as discussed later in greater detail. BACE1 diminishes insulin receptor (IR) levels (9) and thereby contributes to the development of hepatic and

systemic insulin resistance that, in turn, as alluded to, leads to hyperinsulinemia, which we postulate, increases AD risk in states of obesity, MetS and T2DM. Moreover, we have developed chemical agents that, when administered to mice orally or parenterally (subcutaneously [SC] or intravenously [IV]), inhibit both liver and brain PKC- λ/ι , and thereby reduce BACE1 levels in both organs; but, when given nasally, it inhibits brain, but not liver, PKC- λ/ι , and thereby reduce brain, but not liver, BACE1 levels. Furthermore, in very recent studies, we found that oral and nasal treatment of transgenic (Tg) AD mice with an inhibitor of PKC- λ/ι over several months diminished brain aPKC activity and BACE1 levels and, most importantly, largely prevented the development of AD pathology and memory impairment.

THE ESSENTIAL ROLE OF BACE1 IN THE PATHOGENESIS OF AD

It seems abundantly clear that the aspartyl protease, BACE1, acts upon β APP in the brain to initiate the production of $A\beta_{1-40/42}$ peptides that polymerize to form the characteristic A β -plaques of AD (1–4). Indeed, genetic KO of BACE1 blocks $A\beta_{1-40/42}$ production and A\beta-plaque formation in Tg AD mice carrying human mutations that increase susceptibility to actions of BACE1 (e.g., APP_{Swede}) and y-secretase (e.g., presenilin-1 [PS1]) (7, 8). This BACE1-KO-induced protection against the development of AD pathology in Tg AD mice is accompanied by the improvement of cognition/memory losses that occur in unprotected wild-type (WT) Tg AD mice, suggesting that (i) BACE1-dependent pathology is required and may account for cognitive/memory losses in Tg AD mice (and presumably other AD states) and (ii) BACE1 itself is not required for cognitive/memory functions, at least in mice. Moreover, delayed conditional KO of brain BACE1 in Tg AD mice, in which AD pathology and cognitive/memory losses are already well established, was found to reverse pathology and cognitive/memory losses therein (7), suggesting reversibility following BACE1 depletion, at least in Tg mouse AD models

Although elevations of brain BACE1 levels have been observed in some series of human AD patients (5), this is not always the case, and increases in BACE1-dependent A β -peptide production may also be due to enhanced trafficking of BACE1-containing vesicles (10) and interactions with β APP-containing vesicles in the Trans Golgi Network (TGN), where BACE1 is activated by increases in intravesicular acidity and perhaps other factors, for example, increased phosphorylation of ser-498 in the cytosolic, C-terminal tail of BACE1. Also, changes in other factors, for example, mutations in β APP, PS1, or γ - or α -secretase(s), or altered clearance of A β -peptides, may serve to increase A β -peptide levels. Regardless of the initial or later causes, BACE1 action is critically needed for A β -peptide production and A β -plaque formation in AD, and BACE1 is an attractive target for the treatment of AD and minimal cognitive impairment (MCI).

Unfortunately, as alluded to, the use of chemical agents that directly inhibit BACE1 in humans with AD or MCI resulted in impairments in cognition/memory, despite apparent reduction of A β production. This raises the possibility that increases in BACE1 levels, activity and/or action, do not explain cognition/

memory losses in human AD/MCI, or losses of cognition/memory reflect a requirement for BACE1 itself, or there are direct or indirect untoward effects of BACE1 inhibitors. With respect to the latter possibility, as discussed, BACE1 levels increase with BACE1 inhibitor treatment (6), and increases in vesicles containing chemically inhibited forms of BACE1 may interfere with the vesicular trafficking needed for neurite formation, cognition, and memory induction. Whether decreases in BACE1 levels that are elicited by inhibition of regulatory factors upstream of BACE1 transcription, in particular for the present discussion, PKC- λ/ι , or downstream factors linking PKC- λ/ι to BACE1, will adversely affect cognition/memory is uncertain, but this approach seems worthy of consideration, given the present dearth of effective treatments for MCI and AD.

SYSTEMIC VERSUS BRAIN INSULIN RESISTANCE IN DIABETES-ASSOCIATED AND NONDIABETIC AD

LO-AD and the insulin-resistant states of T2DM and its forerunners, that is, obesity and MetS, have frequently been found to coexist, for example, in a particularly important study, 80% of all AD patients seen at the Mayo Clinic were found to have, in half the cases, overt T2DM (fasting blood sugar [FBS] over 125 mg/dL) or, in the other half, pre-T2DM (FBS 110–125 mg/dL) (11). In view of this and other reports, it has been speculated that the systemic insulin resistance, which is generally, if not usually, seen in T2DM, pre-T2DM, obesity, and the MetS, serves as a risk factor for LO-AD. It has further been speculated or tacitly assumed that, as in peripheral tissues, most notably, muscle, where resistance to insulin is particularly severe, the brain is similarly insulin resistant, that is, essentially unresponsive to insulin, and therefore hypoinsulinized in the aforesaid insulin-resistant states. Accordingly, it is proposed that this hypoinsulinization *as such* provides an explanation for increases in AD risk in states of systemic insulin resistance.

In keeping with the idea that insulin action in brain is reduced in AD, brain IR activity and/or number is reportedly mild-moderately reduced in the brains of humans with *nondiabetic* AD (12); accordingly, the concept of brain hypoinsulinization has been reemphasized and expanded, and it has been proposed that the brain itself is insulin resistant and, moreover, hypoinsulinized, in both nondiabetic AD and obesity/MetS/T2DM-associated AD.

However, in view of our more recent findings, as discussed later, this proposal appears to be overstated and in need of modification. Thus, it is indeed reasonable to postulate that deficiencies of the brain IR would of necessity impair insulin effects in the brains of AD patients that have normal or low circulating levels of blood/plasma insulin, as in non-diabetic, normo-insulinemic AD, or in late forms of T2D, if and when insulin levels fully diminish to normal or subnormal levels. On the other hand, the same cannot be said in conditions of hyperinsulinemia, wherein elevated levels of blood insulin can bypass partial defects in the IR in many or most tissues by using "spare" IRs, that is, IRs in excess of those needed to produce maximal downstream effects.

In this respect, note that a more severe insulin resistance that exists in T2D muscle is largely a result of down-regulation of IRS-1 and its ability to activate phosphatidylinositol (PI) 3-kinase (3K), which is needed to produce

PI-3,4,5-(PO₄)₃ (PIP₃), which activates signaling factors required for insulinstimulated glucose transport in muscle, that is, Akt and aPKC. Also note that this defect in muscle IRS-1/PI3K is a "post-IR" impairment that cannot be bypassed by hyperinsulinemia and should not be confused with IR defects that can be bypassed via "spare" IRs, which, as will become evident in this chapter, are present and operative in both liver and brain (13–17).

And, in marked contrast to muscle, insulin action in liver and activation of hepatic Akt and aPKC in obesity, the MetS and *early* T2DM are excessive in many respects owing to hyperinsulin-emia-induced activation of "spare" hepatic IRs. Indeed, note that, unlike the post-IR defect in muscle IRS-1/PI3K which occurs early in high-fat-fed (HFF) mice (18), the post-IR impairments of hepatic IRS-1/PI3K and subsequent Akt activation by insulin occur only later as T2DM progresses, that is, only after 2-3 months of high-fat feeding in the classical mouse model of dietdependent obesity/MetS/T2DM (15). Also note that the activation of hepatic IRS-2/ PI3K and subsequent aPKC activation by insulin remains intact even in late T2DM when IRS-1/PI3K/Akt activation is impaired: in short, hepatic aPKC remains activated by hyperinsulinemia and continues to promote increases in hepatic production of glucose and lipids throughout the course of obesity, the MetS and T2DM. Moreover, as stated, the post-IR defect in muscle described earlier occurs very early in initial stages of the development of diet-induced obesity/MetS/T2DM in HFF mice (18), viz, and several months before impairments in hepatic IRS-1/PI3K and Akt develop. And this muscle defect in glucose metabolism undoubtedly contributes importantly to early elevations of blood insulin to levels (18) that are sufficient to bypass IR defects and strongly/maximally activate key processes in other tissues, most notably for the present discussion, liver and brain. In keeping with mouse studies, note that defective glucose transport in muscle occurs early in human obesity, well before the development of pre-T2D and overt T2DM.

INSULIN RECEPTOR DOWNREGULATION IN OBESITY/T2DM AND AD

The downregulation of the brain IR that is seen in nondiabetic AD (12) and diabetes-associated AD (see below) may well be at least partly due to increases in BACE1-dependent degradation of the IR, as this mechanism has recently been shown to clearly be operative in the liver in various murine and human forms of obesity and T2DM (9). Indeed, in initial studies, we too have found that there are decreases of IR levels in both brain and livers of hyperinsulinemic, HFF, and obese/MetS/T2DM mice, and these decreases in IR levels in both brain and liver are associated with, and apparently dependent on, increases in PKC- λ /t activity and aPKC-dependent increases in BACE1 levels and proteolytic activity, as aberrant elevations in BACE1 levels and subsequent deficiencies in IR levels are abrogated by the inhibition of aPKC.

In addition to the above-described aPKC/BACE1-dependent degradative mechanism for the decreases in IR levels, the activity of the IR may also be diminished by inhibitory phosphorylation of the IR that is provoked via negative feedback mechanisms that occur during actions of a wide variety of factors, many of which are likely to be increased and operative in damaged AD tissues.

These factors include lipids, such as ceramide and phosphatidic acid (PA), both of which directly activate aPKCs; cellular stress, inflammatory, and oxidative factors that activate phospholipases C and D (PLC and PLD) that produce diacylglycerol (DAG) and PA, respectively (note that DAG and PA can interconvert); DAG-sensitive conventional (c) and novel (n) PKCs ($\alpha,\beta,\gamma,\delta,\epsilon,\theta$) that phosphorylate and diminish activity of the IR and/or IRS-1 which activates PI3K. Also note that negative feedback on the IR and IRS-1 can be mediated by phosphorylation provoked by a variety of signaling factors, including Akt, aPKC, mammalian target of rapamycin (mTOR), and ERK, all of which are activated by insulin and various polypeptides that operate via PI3K. In this regard, note that aPKC, and Akt and ERK are all reportedly hyperactivated in brains of humans with nondiabetic AD (12), and any of these downstream factors may negatively feedback on the IR.

Finally, note that the *partial* resistance at the IR level in nondiabetic AD brain is in fact readily overcome by elevated insulin levels (12), which may be acting via "spare" IRs, that is, present in excess of that needed for maximal activation of downstream factors; alternatively or additionally, IGF-1 receptors may also be activated by elevated insulin levels and may contribute to the activation of IRS-1/PI3K, Akt, and aPKC. And, in this regard, note that, despite decreases in IR number that we recently observed in brains of obese/MetS/T2D HFF mice, the IR is nevertheless *maximally* activated by hyperinsulinemia in HFF mice (13), apparently by the activation of "spare" IRs. Further note that the treatment of HFF mice with inhibitors of brain aPKC (discussed later) diminished brain BACE levels and simultaneously increased brain IR levels.

HYPERINSULINEMIA IN INSULIN-RESISTANT FORMS OF OBESITY AND T2DM AS A RISK FACTOR FOR AD

With the prevailing assumption that the brain itself is insulin resistant and more importantly *hypo-insulinized* in AD, we were surprised to find (14), in whole brain and individual neurons of the anterior cortex and hippocampus of *systemically* insulin-resistant, hyperinsulinemic HFF mice, ob/ob mice, and, very importantly, monkeys with long-standing, diet-dependent obesity/T2D, that insulin signaling to both Akt and aPKC, and to all examined Akt substrates, viz, glycogen synthase kinase-2 β , mTOR, and forkhead homeobox factors (FoxO1, FoxO3a, and FoxO4), is uniformly increased (Figure 1). Indeed, increased Akt/aPKC signaling in obese/T2D HFF and ob/ob mice appears to be maximal, as insulin treatment (which rapidly, over 15 min, maximally increases brain Akt and aPKC activities in normal mice) did not elicit further increases in these activities in hyperinsulinemic HFF and ob/ob mice (13, 14).

As alluded to, we moreover found that the brain IR itself, despite a reduction in total levels, is nevertheless maximally activated (as per total phosphotyrosine content of the IR- β subunit) by the hyperinsulinemia that exists in HFF obese/ T2D mice (13). It therefore seems clear that any IR deficiency that is present in the brains of HFF obese/T2D mice is readily bypassed by the activation of "spare" IRs by hyperinsulinemia in HFF mice (13).

Finally, note that the hyperinsulinemia as such in insulin-resistant mice appears to be largely/fully responsible for increases in brain IR and Akt and aPKC signaling, as

105



Figure 1. Atypical PKC (aPKC)-dependent risk factors in Alzheimer's disease.

Activity of neuronal aPKC (encircled) can be excessively activated by (i) diet/liver/aPKCdependent hyperinsulinemia acting on insulin receptors (IRs) and IR substrates (IRS-1/2), (ii) other factors (left box) acting on noninsulin receptors (R) that similarly activate phosphatidylinositol 3-kinase (PI3K) by IRS-1/2 and/or other factors, and (iii) various agents and metabolites (right box) that operate directly or indirectly via phospholipases D and C to increase levels of ceramide and phosphatidic acid, which directly activate aPKC. In turn, aPKC acts via inhibitor of kappa-B kinase-beta (IKK β) to increase the activity of nuclear factor kappa-beta (NF κ B), which acts directly to increase β -secretase (aka, BACE1) transcription, and thus increase levels of A β -peptides and phospho-tau, that is, precursors of Alzheimer plaques and tangles. Note that, although not shown, NF κ B also increases transcription and levels of proinflammatory cytokines, tumor necrosis factor-alpha (TNF α) and interleukin 1- β (IL-1 β), and perhaps other cellular stress/oxidant/inflammatory factors, which, along with increases in A β 42 peptides, act, as depicted (right box), to increase aPKC activity, perhaps creating multiple vicious cycles.

the correction of hyperinsulinemia (elicited by improvement in hepatic abnormalities following treatment with aPKC inhibitor, aurothiomalate [ATM] (14), in doses that selectively act in liver [but not in brain] to fully ameliorate obesity/MetS/T2DM and correct hyperinsulinemia (16)) was attended by (i) return of all brain insulin signaling aberrations, including increases in activities of Akt, Akt substrate, and aPKC, to their normal resting/basal levels and (ii) restored ability of insulin to acutely and fully activate both Akt and aPKC in the brain (14). Obviously, unlike aPKC inhibitors

1H-imidazole-4-carboxamide, 5-amino-1-[2,3-dihydroxy-4-[(phosphono-oxy) methyl]cyclopentyl-[1R-(1a,2b,3b,4a)] (ICAPP), 1H-imidazole-4-carboxamide, 5-amino-1-[2,3-dihydroxy-4-[(hydroxyl)methyl]cyclopentyl-[1R-(1a,2b,3b,4a)] (ICAP), and 2-acetyl-cyclopentane-1,3-diketone (ACPD) (see later), ATM did not cross the blood–brain barrier (BBB) in these studies.

Accordingly, as depicted in Figure 1, we postulate that persistent hyperactivation of brain Akt in insulin-resistant states and phosphorylation/inhibition of activities of all brain FoxOs (1/3a/4/6) and PGC-1 α (14) may be problematic over time, as their transcriptional actions are needed to maintain cognitive functions and neuronal integrity (see discussion in (14)). On the other hand, Akt may also have more acute beneficial effects on memory and overall brain metabolism and may also function as an antiapoptotic agent. In any case, the hyperactivation of brain aPKC, and subsequent increases in BACE1, A $\beta_{1-40/42}$, and phospho-tau (p-tau) (13, 14) most likely have detrimental effects.

To summarize, systemic insulin resistance that originates in the liver due to diet-dependent caloric excess (15) or in the brain appetite/energy center as in leptin-deficient ob/ob mice (17), or in muscle that is defective in glucose transport (16) is accompanied by key abnormalities in liver that includes the hyperactivation of hepatic aPKC, aPKC-dependent increases in gluconeogenesis and lipogenesis, and development of hyperinsulinemia. In turn, hyperinsulinemia leads to chronic hyperinsulinization of the brain, which, by hyper-activating brain aPKC, increases levels of the factors that produce pathological plaques and tangles in AD, that is, BACE1 expression, BACE1 activity, and BACE1-dependent increases in A β -peptides and p-tau. Obviously, this hypothesis is at odds with postulates that the brain itself is uniformly insulin resistant, unresponsive to insulin, and hypoinsulinized in hyperinsulinemic states of obesity. MetS and T2DM, and brain insulin resistance per se increase AD pathology in these states (19–22). In any case, liver involvement plays a critical role in causing systemic insulin resistance, and liver abnormalities are dependent on excessive activation of *hepatic* aPKC (15-18). This dependence on hepatic aPKC explains how treatment with inhibitors of hepatic aPKC and correction of hyperinsulinemia can reduce brain aPKC activity and thereby improve BACE1, $A\beta_{1-40/42}$, p-tau, and memory alterations (13, 14).

On the other hand, in AD that is not associated with hyperinsulinemia, for example, in lean nondiabetic subjects, there may well be an impairment in brain IR activity or a deficiency of brain IR, which limits insulin action in the brain, that is, in response to normal or low blood insulin levels. As discussed, this, of course, may occur in nondiabetic, normo-insulinemic AD and possibly in later stages of T2D-related AD if and when insulin levels are reduced to normal or subnormal levels. These differences in ambient insulin levels may explain how some, but not all, studies suggest that nasal insulin treatment, as well as subsequent brain insulinization, has beneficial effects in AD (19–22).

EFFECTS OF DAG-SENSITIVE PKCs $(\alpha, \delta, \epsilon)$ VERSUS aPKC- λ/ι ON BACE1 AND AD

In addition to PIP₃-activated aPKCs, DAG-activated conventional (c) cPKCs and novel (n) nPKCs can influence AD pathology in a variety of ways. For example,

the activations of both DAG-sensitive cPKC- α and nPKC- ϵ diminish the amyloidogenic pathway and limit A $\beta_{1-40/42}$ production by activating α -secretase, which cleaves β APP between sites cleaved by β -secretase and γ -secretase, thus producing shortened, nonamyloidogenic peptides (1, 2, 23, 24). In this regard, note that insulin activates all DAG-sensitive PKCs in muscle, liver, and adipose tissues, but whether this occurs in brain is uncertain and is in need of study.

Additionally, the activation of PKC- α may diminish A $\beta_{1-40/42}$ production indirectly by restraining PKC- ϵ -dependent β APP expression, as it has been reported that phorbol-ester-induced *deficiency* of PKC- α is accompanied by increases in PKC- ϵ , β APP expression, β APP accumulation in the TGN, and subsequent increases in BACE1-dependent A $\beta_{1-40/42}$ production (25).

On the other hand, some LO-AD patients have gain-of-function mutations in PKC- α that diminishes cognitive/memory functions by altering synaptic activity (26). However, in the absence of this mutation, it is unknown if there are similar memory problems that result from PKC- α activation arising from (i) *de novo* synthesis of PA/DAG owing to increases in glucose and/or fatty acids, (ii) PLC-produced DAG, or (iii) increases in levels of blood insulin, which activates all c/nPKCs, including PKC- α , by a variety of mechanisms in muscle, liver, and adipose tissues (information on brain c/nPKCs during insulin action is lacking).

In marked contrast to the potentially beneficial effects on AD conferred by the activation of DAG-sensitive aPKC- α and nPKC- ε and subsequent activation of α -secretase, the effects of the DAG-sensitive nPKC, PKC- δ , are much different, in that PKC- δ overexpression increases, and PKC- δ knockdown decreases, BACE1 levels and A $\beta_{1-40/42}$ production in neuronal cells (27). Further, like the increases in activity or levels of PKC- λ t in brains of humans with AD (12), PKC- δ levels are increased and correlate with increases in BACE1 levels in brains of AD humans (27).

And, very interestingly, inhibition of PKC- δ by *rottlerin* simultaneously protects Tg/APP/PS1/AD mice from developing increases in BACE1 and A β -peptide/plaque production, and aberrations in cognition/memory functions (27). However, note that, like PKC- λ / ι , which activates NF κ B in liver (15–17, 28) and at least certain other tissues, for example, lymphocytes (29) (and presumably in brain, as discussed further later), PKC- δ similarly activates brain/neuronal NF κ B (27), which, in turn, increases the expression of proinflammatory cytokines, TNF- α , and IL-1 β , which, in turn, activate NF κ B by a *PKC-\lambda/\iota-dependent* mechanism (30). And, as PKC- δ is also activated by NF κ B (27), this suggests operation of a vicious cycle and raises the possibility that the activation of PKC- λ / ι by TNF- α and IL-1 β may explain or contribute to BACE1 increases and AD-abetting effects of PKC- δ overexpression (see Figure 2).



Figure 2. Potential interplay of aPKC, PKC- λ/ι , nPKC, and PKC- δ , to operate upstream and downstream of IKKβ/NFκB and act cooperatively in a vicious cycle to increase BACE1 transcription.

Also note that, in the previously cited PKC- δ study (27), BACE1 mRNA and protein levels were substantially increased in Tg/APP/PS1/AD mice, and inhibition of PKC- δ activity by rottlerin reduced BACE1 levels back to normal levels in Tg/ APP/PS1 mice, but had little or no effect on basal BACE1 levels in WT cells. This partial reduction of elevated BACE1 levels to normal in Tg AD mice contrasts with the marked (up to 80–90%) reductions of basal/resting levels of BACE1 protein and mRNA seen in both brain and liver of mice following chemical inhibition or haploinsufficiency of PKC- λ . Thus, PKC- λ may be more potent than PKC- δ in regulating BACE1 transcription.

In this regard, further note that, in adipocytes, PKC- λ and PKC- δ were found to physically interact, and the PKC- δ inhibitor, used in the aforesaid study (27), rottlerin, was found in another study (31) to block insulin-stimulated translocation of Glut4-containing vesicles to the plasma membrane *independently of PKC-* δ , which, in fact, is *not* required for this process in adipocytes (31). Thus, rottlerin may have also inhibited PKC- λ , which is clearly required for Glut4 translocation in adipocytes (31). Or, stated differently, rottlerin is not a specific inhibitor of PKC- δ , and the ability of rottlerin to improve AD pathology in Tg/APP/PS1/AD mice may reflect inhibition of PKC- λ (or other factors that operate via NF κ B), rather than PKC- δ . Alternatively, both PKC- λ and PKC- δ may work together, cooperatively or additively, to regulate BACE1 expression.

Finally, note that we have found that PKC- λ/ι activity is increased in brains of APP/PS1/AD mice, and inhibition of PKC- λ/ι with a relatively specific inhibitor, which clearly does not inhibit PKC- δ , is effective in improving memory impairments and for reducing A β -peptide and A β -plaque levels. This, of course, suggests that PKC- λ/ι is required for BACE1-dependent AD development in Tg AD mice, and AD-promoting effects of PKC- δ overexpression could conceivably involve PKC- λ/ι activation by the above-described circular pathway.

Nevertheless, as alluded to, it is possible that both PKC- δ and PKC- λ/ι are activated in the AD process, and, moreover, both may participate in the activation of NF κ B, which, in turn, via TNF- α and IL-1 β , may further activate both PKC- δ (32) and PKC- λ/ι (30) in the aforesaid circular "vicious" cycle. This possibility becomes even more interesting in light of the observation that increases in the activity of both PKC- δ and PKC- $\lambda/\iota\zeta$, but not PKC- α , are seen in T-lymphocytes isolated from AD humans, but only when stimulated *ex vivo* with A β_{1-42} (33,34); this suggests that neurotoxic A β_{1-42} (A β 42) activates both PKC- δ and PKC- $\lambda/\iota\zeta$, and this brings into play an interaction between the proliferative actions of PKC- λ/ι and the proapoptotic effects of PKC- δ , a combination that, when more fully activated, may enhance and hasten AD pathology development. The latter possibility, featuring a triad of increases in PKC- λ/ι , PKC- δ , and NF κ B that are put into high-gear when A β 42 levels reach a threshold may provide an explanation for the rapid downhill phase of AD that follows a slow inductive process.

aPKC INHIBITORS

By high throughput screening, a number of compounds have been identified, which target a site at or near the substrate and/or the ATP-binding site of aPKCs, and potently inhibit both recombinant aPKCs and aPKC activity of cultured

neurons and hepatocytes, but not myocytes. Among others yet to be studied, three such agents have been studied in detail: (i) ICAPP, which potently inhibits both recombinant PKC- λ/ι , and neuronal and hepatocyte PKC- λ/ι (IC50, 1–10 nM); (ii) ICAP, which is converted to active phosphorylated ICAPP intracellularly by adenosine kinase and is similarly potent for PKC- $i\Lambda$ in neurons and hepatocytes $(IC_{50}, 10-100 \text{ nM})$; and (iii) ACPD, which comparably inhibits recombinant forms of both PKC- λ *i* and PKC- ζ , and their activities in isolated neurons and hepatocytes (IC₅₀, 10–30 nM) (15–17, 28, 35). Note that, whereas ACPD inhibits PKC- \mathcal{N}_i and PKC- ζ with equal potency, ICAPP and ICAP preferentially inhibit PKC-NI. Also note that the full-length, 70 kDa aPKC in brain, is largely if not entirely PKC- λ/ι , as, unlike peripheral non-CNS PKC- ζ , brain PKC- ζ lacks an inhibitory regulatory domain and exists largely as a shortened, 50 kDa, constitutively-active molety, called PKM ζ , which is particularly important in longterm memory (LTM) functions, as discussed later. Further note that ICAPP, ICAP, and ACPD do not inhibit recombinant forms of conventional/novel (c/n) PKCs $(\alpha, \beta, \delta, \epsilon, \theta)$ (15–17, 28, 35).

During *in vivo* usage, in doses that are sufficient to effectively inhibit hepatic aPKC and largely, albeit not completely, reverse post-IR hepatic aberrations seen in mouse obesity/MetS/T2D models (15–17), the aforesaid inhibitors, ICAPP, ICAP, and ACPD, do not inhibit muscle, adipocyte, or brain aPKC, and, in all tissues, they have no inhibitory effects on Akt, or on AMPK. These inhibitors also have no effect on activities of an array of 35 other kinases independently tested by Life Technology Selectscreen Profiling, Madison, WI (15–17, 28, 34). We therefore believe that these agents have reasonable selectivity for aPKC, but as with all drugs, exclusive targeting cannot be assumed.

On the other hand, we have verified that partial heterozygous KO of total body PKC- λ in the mouse (total homozygous KO is embryonic lethal), and thus haploinsufficiency of PKC- λ , has biological effects similar if not identical to those of the aforesaid chemical aPKC inhibitors in liver (36) and in brain: for example, PKC- λ haploinsufficiency (i) in liver blocks all hepatic diabetic aberrations induced by high-fat-feeding (36) and (ii) in brain markedly diminishes BACE1 levels and insulin-stimulated production of A β -peptides. In addition, liver-specific KO of PKC- λ (by administration of adenovirus encoding Cre-recombinase to PKC- λ -floxed mice) or liver-specific inhibition of hepatic aPKC by adenovirus-mediated expression of kinase-inactive aPKC has biological effects in liver similar if not identical to those of the aforesaid chemical aPKC inhibitors on obesity/MetS/T2DM (18, 36, 37). We therefore are confidant that aPKC inhibition underlies all biological effects of ICAPP, ICAP, and ACPD that we have reported in both liver and brain.

However, at higher *in vivo* doses, ICAPP, ICAP, and ACPD inhibit brain, as well as liver, aPKC (13). Thus, in insulin-resistant, hyperinsulinemic states, these agents can reduce brain aPKC activity (i) at lower doses, *indirectly* via inhibition of liver aPKC and reduction of blood insulin levels, and (ii) at higher doses, by *directly* inhibiting brain aPKC. In this regard, note that, in published (38) and other pharmacokinetic/pharmacodynamic (PK/PD) studies, it was found that ICAP preferentially distributes to liver, but clearly passes the BBB to enter the brain. And, as found in other studies, ICAP is effective in liver and brain when given insufficiently increased doses by oral gavage (i.e., *per os* [PO]), as well when administered IV or SC. Moreover, when administered to mice intranasally, ICAP

inhibits brain, but not liver, aPKC, suggesting that ICAP reaches the brain through neurovascular bundles that traverse the sphenoid sinuses and the cribriform plate; this raises the possibility that nasally administered aPKC inhibitors (or other agents) may be used to selectively target brain aPKC during AD treatment.

Furthermore, in cultured human neuronal cells and/or mouse hippocampal slices, ICAPP fully inhibits insulin-stimulated increases in activities of 70 kDa PKC- t/λ and BACE1, and increases in the production of A $\beta_{1-40/42}$ by approximately 50% at 10 nM and 90–100% at 100 nM, without inhibiting the putative memory protein, that is, constituently-active 50 kDa PKM ζ (13), which was alluded to earlier and discussed more fully later. And, in cultured human neuronal cells, ICAPP, at low 10 nM concentrations, fully blocked 2–3-fold increases in BACE1 *levels* that were provoked by 24-h insulin treatment. It therefore seems clear that prolonged treatment with high-dose insulin acts directly via aPKC to induce remarkably strong increases in neuronal BACE1 levels. This bolsters our contention that hyperinsulinemia is an important risk factor for AD. It also seems very clear that aPKC inhibitors are working directly in neurons to alter BACE1 and BACE1-dependent alterations in A β -peptide production.

aPKC REQUIREMENTS FOR MEMORY FUNCTION

As alluded to, a number of findings suggest that the 50 kDa, constitutively active, brain-specific form of PKC- ζ , that is, PKM ζ (described earlier), functions in long-term potentiation (LTP) and LTM formation (39, 40). However, although PKM ζ KO in brain does not impair LTP/LTM, this appears to be explained by a compensatory increase in PKC- λ /t in PKM ζ -null mice, as suggested by the fact that LTP/LTM is impaired by selective inhibition of PKC- ι/λ by ICAP in PKM ζ -null mice, *but not in normal mice*, wherein ICAP does not inhibit PKM ζ (40). And, from the lack of effect of ICAP on LTP and memory function in normal mice (40), it may be argued that PKC- λ /t is not required for *on-going* LTP/LTM.

Indeed, in HFF mice, as reported in (13), aPKC inhibitor ACPD improved high-fat-diet-induced decreases in acute memory function (as per Novel Object Recognition [NOR]) while blocking insulin-stimulated increases in PKC- ι/λ activity, but largely sparing basal PKC- ι/λ activity. Moreover, in other studies, a 50% loss of brain PKC- λ in PKC- λ haplo-insufficient mice, and subsequent decreases in PKC- λ activity and BACE levels (80–90%) did not impair training and memory functions in Radial Arm Water Maze (RAWM) and NOR tests. It thus appears that brain aPKC activity can be partially diminished to levels that markedly reduce AD development, without impairing memory processes.

Nevertheless, it should be noted that, whereas a *marked and chronic loss* of PKC- ι/λ in a long-term, hippocampus/anterior-cortex-selective KO study showed essentially normal memory function and LTM/LTP, which was attributed to PKM ζ compensation, *acute KO* of hippocampal/anterior-cortical PKC- λ (and thus lacking time for compensatory changes) suggested a need for PKC- λ , in learning/memory processes, but only in more difficult (nonstandard) learning tasks (41). Furthermore, *acute 1-month knockdown* of PKC- λ produced specifically in the hippocampus by stereotactic injection of virus encoding shRNA, suggested that hippocampal PKC- λ is needed for the initiation of LTP, short-term memory (as per contextual and trace

fear testing), and consolidation of LTM (42). Thus, these inhibitory effects on learning and memory may be exaggerated by *acute*, and thus uncompensated, and *excessive* losses of PKC- λ , since (i) as discussed, *partial* inhibition of PKC- λ /t and simple reduction of elevated brain PKC- λ activity to normal by aPKC inhibitor treatment improve memory loss (in NOR) in hyperinsulinemic HFF mice (13) and (ii) chronic 50% loss of PKC- λ in PKC- λ -haploinsufficient mice did not impair learning/memory performance in RAWM and NOR tests.

It is, of course, interesting that brain PKC- λ t and PKM ζ (and perhaps other PKCs) can apparently compensate for losses of each other over time in the adult mouse brain, and these compensatory changes can satisfactorily maintain relatively normal memory functions. In any case, these caveats of aPKC inhibition or deficiency need to be kept in mind.

aPKC ACTIVATORS

Atypical PKCs are activated not only by insulin and other polypertides (IGF-1, NGF, BDNF, NMDA-receptor activators (43–47)) that activate PI3K and thereby produce PI-3.4.5-(PO₄)₃ (PIP₃) which directly activates aPKCs (48) but also by C:14 and C:16 ceramides which directly activate aPKCs (15) (higher C:20–24 ceramides may be inhibitory); PLD-derived phosphatidic acid (PA), which, like PIP₃, contains an acidic head group and directly activates aPKCs (49, 50); various factors, such as Aβ42 (33, 34) and certain oxidants (e.g., sorbitol (49)) and inflammatory factors (e.g., lipopolysaccharide (51)) which presumably activate PLCs and/or PLD to produce PA. Note that many of these metabolites and agonists are thought to abet AD development.

CONTROL OF BACE1 TRANSCRIPTION BY NFKB

There is clear evidence that NF κ B is a strong, if not the major, transactivator of the BACE1 gene in mouse brain (52–54). This is of considerable interest, as we have shown, that hyperinsulinemia in various O/MetS/T2D states (e.g., HFF, ob/ob, and HetM λ KO mice (15–18, 36, 37) and T2D humans (28)) leads to, in liver, aPKC-dependent increases in (i) phosphorylation/activation of inhibitor of kappa-B kinase- β (IKK- β), (ii) phosphorylation and thus dissociation of the inhibitor of NF κ B inhibitor (I κ B) from NF κ B (and, presumably, subsequent degradation of I κ B), (iii) transfer of the active p65/RelA subunit of NF κ B into the nucleus, (iv) p65/RelA phosphorylation and activation, as per electrophoretic mobility shift assay, and (v) NF κ B-dependent transcription of genes encoding various proinflammatory cytokines, including TNF- α and IL-1 β , that is, with increases in their mRNA and protein levels. These changes are clearly provoked by *an aPKC-dependent mechanism*, as shown by inhibition of aPKC by chemical inhibitors and expression of kinase-inactive-aPKC, and by various PKC- λ KO methods.

In particular, note, in normal WT mice, acute insulin treatment over 15 min increases nuclear levels of the p65/RelA subunit of NF κ B, and simple feeding-induced increases in insulin provoke increases in mRNA levels of TNF- α and

IL-1 β ; *and* all effects of insulin and feeding are blocked or markedly diminished in littermate mice haploinsufficient for PKC- λ (35). And, perhaps most importantly, we found in both hepatocytes harvested from humans with long-term preexisting T2DM, and following acute 4-h insulin treatment of normal human hepatocytes, that there were increases in TNF- α and IL-1 β mRNA levels by an aPKC-dependent mechanism (28). Note that others have also shown that aPKCs are major activators of NF κ B during inflammation and lymphocyte activation (29).

We therefore strongly suspect that aPKC increases BACE1 transcription by activating NF κ B in hepatocytes and neurons by the following mechanism and linear pathway:

- (i) In liver, dietary caloric excesses \rightarrow hepatic aPKC activity \rightarrow IKK β activity \rightarrow NF κ B activity \rightarrow BACE1 levels \rightarrow IR degradation \rightarrow systemic insulin resistance \rightarrow hyperinsulinemia \rightarrow activation of hepatic spare IRs \rightarrow post-IR hepatic aberrations \rightarrow hyperinsulinemia \rightarrow and so on in a vicious cycle
- (ii) and, in brain, hyperinsulinemia, cellular stress/oxidant/inflammatory factors, $A\beta_{42}$, and so on \rightarrow aPKC activity \rightarrow IKK β activity \rightarrow NF κ B activity \rightarrow BACE1 levels \rightarrow A β -peptides and p-tau, and IR degradation (that is partial and can be bypassed by hyperinsulinemia) \rightarrow AD pathology.

CONCLUSION

It seems abundantly clear that AD risk in obesity MetS/T2DM is real, and this risk at least partly results from hyperinsulinemia-induced increases in brain aPKC activity that provokes increases in BACE1 levels and production of A β -peptides and p-tau. Increases in brain aPKC activity and BACE1 activity/levels can also proteolytically diminish brain IR levels, but hyperinsulinemia in all mouse and monkey models of obesity/MetS/T2DM that we have examined does not diminish IR-dependent Akt activity; indeed, despite decreases in brain IR levels, brain IR and Akt activities are increased (apparently via "spare" IRs) by hyperinsulinemia and/or other factors.

On the other hand, increases in BACE1 activity and levels that occur in *nondiabetic* AD can not only increase A β -peptide and p-tau levels but presumably can also diminish brain IR levels, and this may diminish *insulin-stimulated* Akt activation in norm- or hypoinsulinemic subjects. On the other hand, activities of both Akt and aPKC are reportedly elevated by uncertain noninsulin factors in the brains of humans with nondiabetic AD, and this increase in aPKC activity can increase BACE1 levels and activity in these AD brains.

Accordingly, there is considerable evidence to suggest that brain aPKC activity is increased in both diabetes-related and nondiabetic forms of LO-AD. And initial findings in mice suggest the feasibility of using chemical aPKC inhibitor treatment to reduce BACE1, A β -peptides, and p-tau. Alternatively, treatments that target factors that activate aPKCs, or factors that operate between aPKC and BACE1, that is, at levels surrounding NF κ B, may also be effective in AD treatment and should be considered. In any event, the continued elucidation of signaling pathways that are operative in AD pathogenesis is likely to open new avenues for the development of effective treatments for AD.

Acknowledgments: This work was supported by VA Merit Review Awards, NIDDK RO1 065969-09 (RV Farese), VA Research & Education Foundation (MP Sajan), NIH (R01 ES016774-01 and R21AG043718), and AHA SFRN 15SFDRN25710468 (M Kindy). Dr. Robert V. Farese is the guarantor of this work, as such, had full access to all the data in the study and takes responsibility for the integrity of the data and the accuracy of the data analysis. This work does not represent the views of the Department of Veteran Affairs or the United States government.

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship, and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- Sinha S, Lieberburg I. Cellular mechanisms of β-amyloid production and secretion. Proc Natl Acad Sci U S A. 1999;96:11049–53. http://dx.doi.org/10.1073/pnas.96.20.11049
- 2. Nunan J, Small DH. Regulation of APP cleavage by α -, β and γ -secretases. FEBS Lett. 2000;483:6–10. http://dx.doi.org/10.1016/S0014-5793(00)02076-7
- Cole SI, Vassar R. The Alzheimer's disease enzyme, BACE1. Mol Degener. 2007;2:70. http://dx.doi. org/10.1186/1750-1326-2-22
- Kandalepas PC, Vassar R. The normal and pathological roles of the Alzheimer's β-secretase, BACE1. Curr Alzheimer Res. 2014;11:441–9. http://dx.doi.org/10.2174/1567205011666140604122059
- Yang L-B, Lindhol K, Yan R, Citron M, Xia W, Yang X-L, et al. Elevated β-secretase expression and enzymatic activity detected in sporadic Alzheimer's disease. Nat Med. 2003;9:3–4. http://dx.doi. org/10.1038/nm0103-3
- Liu L, Lauro BM, Ding L, Rovere M, Wolfe MS, Selkoe DJ. Multiple BACE1 inhibitors abnormally increase the BACE1 protein level in neurons by prolonging its half-life. Alzheimers Dement. 2019;15(9):1183–94. http://dx.doi.org/10.1016/j.jalz.2019.06.3918
- Hu X, Das B, Hou H, He W, Yan R. BACE1 deletion in the adult mouse reverses preformed amyloid deposition and improves cognitive functions. J Exp Med. 2018;215:927–40. http://dx.doi. org/10.1084/jem.20171831
- Kimura R, Devi L, Ohno M. Partial reduction of BACE1 improves synaptic plasticity, recent and remote memories in Alzheimer's disease transgenic mice. J Neurochem. 2010;113:248–61. http:// dx.doi.org/10.1111/j.1471-4159.2010.06608.x
- Meakin PJ, Mezzapesa A, Benabou E, Haas ME, Bonardo B, Grino M, et al. The beta secretase BACE1 regulates the expression of insulin receptor in the liver. Nat Commun. 2018;9:1306. http://dx.doi. org/10.1038/s41467-018-03755-2
- Tan J, Evin GG. B-site APP-cleaving enzyme 1 trafficking and Alzheimer's disease pathogenesis. J Neurochem. 2012;120:869–80. http://dx.doi.org/10.1111/j.1471-4159.2011.07623.x
- 11. Janson J, Laedtke T, Parisi JE, O'Brien P, Peterson RC, Butler PC. Increased risk of type 2 diabetes in Alzheimer's disease. Diabetes. 2004;53:478–81. http://dx.doi.org/10.2337/diabetes.53.2.474
- Talbot K, Wang Y, Kazi H, Han L-Y, Bakashi KP, Stucky A, et al. Demonstrated brain insulin resistance, IRS-1 dysregulation, and cognitive decline. J Clin Invest. 2012;122:1316–38. http://dx.doi. org/10.1172/JCI59903
- 13. Sajan MP, Ivey RA, III, Hansen BC, Higgs MG, Kahn CR, Braun U, et al. Atypical PKC, PKC λ t activates β -secretase and increases A $\beta_{1-40/42}$ and phospho-tau in mouse brain and isolated neuronal cells, and

may link hyperinsulinemia and other aPKC activators to development of pathological and memory abnormalities in Alzheimer's disease. Neurobiol Aging. 2018;43:62–73. http://dx.doi.org/10.1016/j. neurobiolaging.2017.09.001

- 14. Sajan MP, Hansen BC, Ivey RA, III, Sajan J, Ari A, Song S, et al. Brain insulin signaling is increased in insulin-resistant states and decreases in FoxOs and PGC-1 α and increases in A $\beta_{1-40/42}$ and phosphotau may abet Alzheimer's development. Diabetes. 2016;65:1892–903. http://dx.doi.org/10.2337/db15-1428
- Sajan MP, Acevedo-Duncan ME, Standaert ML, Ivey RA, III, Lee MC, Farese RV. Akt-dependent phosphorylation of hepatic FoxO1 is compartmentalized on a WD40/Propeller/FYVE scaffold and is selectively inhibited atypical PKC in early phases of diet-induced obesity. A mechanism for impairing gluconeogenic but not lipogenic enzyme expression. Diabetes. 2014;63:2690–701. http://dx.doi. org/10.2337/db13-1863
- Sajan MP, Nimal S, Mastorides S, Acevedo-Duncan M, Kahn CR, Leitges M, et al. Correction of metabolic abnormalities in a rodent model of obesity, metabolic syndrome and type 2 diabetes by inhibitors of hepatic protein kinase C-iota. Metabolism. 2012;61:459–69. http://dx.doi.org/10.1016/j. metabol.2011.12.008
- Sajan MP, Ivey RA, III, Lee MC, Farese RV. Hepatic insulin resistance in ob/ob mice involves increases in ceramide, atypical PKC activity and selective impairment of Akt-dependent FoxO1 phosphorylation. J Lipid Res. 2015;56:70–80. http://dx.doi.org/10.1194/jlr.M052977
- Sajan MP, Standaert ML, Nimal S, Varanasi U, Pastoor T, Mastorides S, et al. The critical role of atypical protein kinase C in activating hepatic SREBP-1c and NFκB in obesity. J Lipid Res. 2009;50:1133–45. http://dx.doi.org/10.1194/jlr.M800520-JLR200
- Craft S. Insulin resistance and Alzheimer's disease pathogenesis: Mechanisms and implications for treatment. Curr Alzheimer Res. 2007;4:147–52. http://dx.doi.org/10.2174/156720507780362137
- Craft S, Claxton A, Baker LD, Hanson AJ, Cholerton B, Trittschuh EH, et al. Effects of regular and long-acting insulin on cognition and Alzheimer's disease biomarkers: A pilot clinical trial. J Alzheimers Dis. 2017;57:1325–34. http://dx.doi.org/10.3233/JAD-161256
- Holscher C. First clinical data of the neuroprotective effects of nasal insulin application in patients with Alzheimer's disease. Alzheimers Dementia. 2014;10(Suppl):33–7. http://dx.doi.org/10.1016/j. jalz.2013.12.006
- Benedict C, Grillo CA. Insulin resistance as a therapeutic target in the treatment of Alzheimer's disease: A state-of-the-art review. Front Neurosci. 2018;12:215–28. http://dx.doi.org/10.3389/ fnins.2018.00215
- Buxbaum JD, Gandy SE, Cicchetti P, Ehrlich ME, Czernik AJ, Fracasso RP, et al. Processing of Alzheimer beta/A4 amyloid precursor protein: Modulation by agents that regulate protein phosphorylation. Proc Natl Acad Sci U S A. 1990;87:6003–6. http://dx.doi.org/10.1073/pnas.87.15.6003
- 24. Etcheberrigaray R, Tan M, Dewachter I, Kuipéri C, Van der Auwera I, Wera S, et al. Therapeutic effects of PKC activators in Alzheimer's disease transgenic mice. Proc Natl Acad Sci U S A. 2004;101: 11141–6. http://dx.doi.org/10.1073/pnas.0403921101
- da Cruz e Silva OA, Rebelo S, Vieira SI, Gandy S, da Cruz e Silva EF, Greengard P. Enhanced generation of Alzheimer's amyloid-beta following chronic exposure to phorbol ester correlates with differential effects on alpha and epsilon isozymes of protein kinase C. J Neurochem. 2009;108:319–30. http://dx.doi.org/10.1111/j.1471-4159.2008.05770.x
- Alfonso SI, Callender JA, Hooli B, Antal CE, Mullin K, Sherman MA, et al. Gain-of-function mutations in protein kinase Cα (PKCα) may promote synaptic defects in Alzheimer's disease. Sci Signal. 2016 May 10;9(427):ra47. http://dx.doi.org/10.1126/scisignal.aaf6209
- Du Y, Zhao Y, Zhang Q, Tian J, Li Z, Hyang TY, et al. Inhibition of PKCδ reduces amyloid-β levels and reverses Alzheimer disease phenotypes. J Exp Med. 2018;215(6):1665–77. http://dx.doi.org/10.1084/jem.20171193
- Sajan MP, Farese RV. Insulin signalling in hepatocytes of type 2 diabetic humans. Excessive expression and activity of PKC-1 and dependent processes and reversal by PKC-1 inhibitors. Diabetologia. 2012;55:1446–57. http://dx.doi.org/10.1007/s00125-012-2477-5
- Diaz-Meco MT, Moscat J. The atypical PKCs in inflammation: NF-κB and beyond. Immunol Rev. 2012;246:154–67. http://dx.doi.org/10.1111/j.1600-065X.2012.01093.x

- Anthonsen MW, Andersen S, Solhaug A, Johansen B. Atypical λ/PKC conveys 5-lipoxygenase/leucotriene B4-mediated cross-talk between phospholiase A₂s regulating NF-κB activation in response to tumor necrosis factor-α and interleukin-1β. J Biol Chem. 2001;276:35344–51. http://dx.doi. org/10.1074/jbc.M105264200
- Kyali AG, Austin JA, Webster NJG. Rottlerin inhibits insulin-stimulated glucose transport in 3T3/L1 adipocytes by uncoupling mitochondrial oxidative phosphorylation. Endocrinology. 2009;143:3884– 98. http://dx.doi.org/10.1210/en.2002-220259
- Vancurova A, Miskoloci V, Davidson D. NF kappa B activation in tumor necrosis factor-alpha-stimulated neutrophils is mediated by protein kinase Cdelta. Correlation to nuclear IkappaBalpha. J Biol Chem. 2001;276:19746–52. http://dx.doi.org/10.1074/jbc.M100234200
- 33. Ciccocioppo F, Lanuti P, Marchisio M, Gambi F, Santavenere E, Pierdomenico L, et al. Expression and phosphorylation of protein kinase isoforms in A β_{1-42} activated T lymphocytes from Alzheimer disease. Int J Immunopathol Pharmacol. 2008;21:23–33. http://dx.doi. org/10.1177/039463200802100104
- Miscia S, Ciccocioppo F, Lanuti P, Velluto L, Bascelli A, Pierdomenico L, et al. Abeta(1-42) stimulated T cells express PKC-delta and PKC-zeta/lambda/iota in Alzheimer disease. Neurobiol Aging. 2009;330:394–406. http://dx.doi.org/10.1016/j.neurobiolaging.2007.07.011
- 35. Sajan MP, Ivey RA, III, Farese RV. Metformin action in human hepatocytes. Co-activation of atypical protein kinase C alters 5'-AMP-activated protein kinase effects on lipogenic and gluconeogenic enzyme expression. Diabetologia. 2013;56:2507–3010. http://dx.doi.org/10.1007/s00125-013-3010-1
- Sajan MP, Ivey RA, III, Lee MC, Mastorides S, Jurzak MJ, Samuels VT, et al. PKCλ haplo-insufficiency prevents diabetes by a mechanism involving alterations in hepatic enzymes. Mol Endocrinol. 2014;28:1097–107. http://dx.doi.org/10.1210/me.2014-1025
- 37. Sajan MP, Standaert ML, Rivas J, Miura A, Kanoh Y, Soto J, et al. Role of atypical protein kinase C in activation of sterol regulatory element binding protein-1c and nuclear factor kappa B (NFκB) in liver of rodents used as model of diabetes, and relationships to hyperlipidaemia and insulin resistance. Diabetologia. 2009;52:1197–207. http://dx.doi.org/10.1007/s00125-009-1336-5
- Apostolatos AH, Apostolatos CA, Ratnayake WS, Neuger A, Sansil S, Bourgeois M, et al. Preclinical testing of 5-amino-1-((1R,2S,3S,4R)-2,3-dihydroxy-4-methylcyclopentyl)-1H-imidazole-4-carboxamide: A potent protein kinase C-t inhibitor as a potential prostate carcinoma therapeutic. Anticancer Drugs. 2019;30:65–71. http://dx.doi.org/10.1097/CAD.000000000000694
- Sacktor TC. PKMzeta, LTP maintenance, and the dynamic molecular biology of memory storage. Prog Brain Res. 2008;169:27–40. http://dx.doi.org/10.1016/S0079-6123(07)00002-7
- Tsokas P, Hsieh C, Yao Y, Lesburgueres E, Wallace EJC, Tcherepanov A, et al. Compensation for PKMζ in LTP and spatial long-term memory in mutant mice. eLife Sci. 2016;5:e14846. http://dx.doi. org/10.7554/eLife.14846
- Sheng T, Wang S, Qian D, Gao J, Ohno S, Lu W. Learning-induced suboptimal compensation for PKCι/λ function in mutant mice. Cerebral Cortex. 2017;27:3284–93. http://dx.doi.org/10.1093/ cercor/bhx077
- Wang S, Sheng T, Ren S, Tian T, Lu W. Distinct roles of PKCι/λ and PKMζ in the initiation and maintenance of hippocampal long-term potentiation and memory. Cell Rep. 2018;16:1954–61. http:// dx.doi.org/10.1016/j.celrep.2016.07.030
- Miranda C, Miele GA, Pierotti MA, Van Obberghen E. IRS-1 and IRS-2 are recruited by TrkA receptor and oncogenic TRK-T1. J Cell Physiol. 2001;186:35–46. http://dx.doi.org/10.1002/ 1097-4652(200101)186:1%3C35::AID-JCP1003%3E3.0.CO;2-X
- 44. Woolen MW, Selbenhener ML, Neidigh KB, Vandanplas ML. Mapping of atypical protein kinase C within the nerve growth factor signaling cascade: Relationship to differentiation and survival of PC12 cells. Mol Cell Biol. 2000;20:4494–504. http://dx.doi.org/10.1128/MCB.20.13.4494-4504.2000
- 45. Melemedjian OK, Tillu DV, Asiedu MN, Mandell EK, Moy JK, Blute VM, et al. BDNF regulates atypical PKC at spinal synapses to initiate and maintain centralized chronic pain state. Molec Pain. 2013;9: 12–26. http://dx.doi.org/10.1186/1744-8069-9-12
- Brennan-Minnella AM, Shen Y, Swanson RA. Phosphoinositide 3-kinase couples NMDA receptors to superoxide release in excitotoxic neuronal death. Cell Death Dis. 2013;4:e580. http://dx.doi. org/10.1038/cddis.2013.111

116 Farese R V et al.

- Koponen S, Kurkinen K, Akerman KE, Mochly-Rosen D, Chan PH, Koistinaho J. Prevention of NMDA-induced death of cortical neurons by inhibition of protein kinase Czeta. J Neurochem. 2003;86:442–50. http://dx.doi.org/10.1046/j.1471-4159.2003.01846.x
- Ivey RA, Sajan MP, Farese RV. Pseudosubstrate arginine residues are required for auto-inhibition and are targeted by phosphatidylinositol-3.4.5-(PO₄)₃ during aPKC activation. J Biol Chem. 2014;289:25021–30. http://dx.doi.org/10.1074/jbc.M114.565671
- 49. Sajan MP, Bandyopadhyay G, Kanoh Y, Standaert ML, Quon MJ, Reed BC, et al. Sorbitol activates atypical protein kinase C and GLUT4 glucose transporter translocation/glucose transport through proline-rich tyrosine kinase-2, the extracellular signal-regulated kinase pathway and phospholipase D. Biochem J. 2002;362:665–74. http://dx.doi.org/10.1042/bj3620665
- Bandyopadhyay G, Sajan MP, Kanoh Y, Standaert ML, Quon MJ, Reed BC, et al. Glucose activates protein kinase C-z/l through proline-rich tyrosine kinase-2, extracellular signal-regulated kinase, and phospholipase D. A novel mechanism for activating glucose transporter translocation. J Biol Chem. 2001;276:35537–45. http://dx.doi.org/10.1074/jbc.M106042200
- Thaler JP, Choi SJ, Sajan MP, Ogimoto K, Nguyen HT, Matsen M, et al. Atypical protein kinase C activity in the hypothalamus is required for lipopolysaccharide-mediated sickness responses. Endocrinol. 2009;150:5362–72. http://dx.doi.org/10.1210/en.2009-0509
- Buggia-Prevot V, Sevalle J, Rossner S, Checler F. NFκB-dependent control of BACE1 promoter transactivation by Ab42. J Biol Chem. 2008;283:10037–47. http://dx.doi.org/10.1074/jbc.M706579200
- Chen C-H, Zhou W, Liu S, Deng Y, Cai F, Tone M, et al. Increased NF-κ-B signaling up-regulates BACE1 expression and its therapeutic potential in Alzheimer's disease. Int J Neuropsychopharmacol. 2012;15:77–90. http://dx.doi.org/10.1017/S1461145711000149
- Wang R, Chen S, Liu Y, Diao S, Xue Y, You X, et al. All-trans-retinoic acid reduces BACE1 expression under inflammatory conditions via modulation pf nuclear factor kB (NFkB) signaling. J Biol Chem. 2015;290:22532–42. http://dx.doi.org/10.1074/jbc.M115.662908

Modeling the Blood-Brain Barrier to Understand Drug Delivery in Alzheimer's Disease

Joanna M. Wasielewska • Juliana C Da Silva Chaves • Anthony R. White • Lotta E. Oikari

Mental Health Program, Cell and Molecular Biology Department, QIMR Berghofer Medical Research Institute, Brisbane, QLD, Australia

Authors for correspondence: Anthony R. White, Mental Health Program, Cell and Molecular Biology Department, QIMR Berghofer Medical Research Institute, Brisbane, QLD, Australia. Email: tony.white@qimrberghofer.edu.au; Lotta E Oikari, Mental Health Program, Cell and Molecular Biology Department, QIMR Berghofer Medical Research Institute, Brisbane, QLD, Australia. Email: lotta.oikari@qimrberghofer.edu.au

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch7

Abstract: The blood–brain barrier is a semipermeable barrier structure that lines the walls of brain microvessels. Although the blood–brain barrier plays a key role in protecting the brain from unwanted molecules, it simultaneously challenges the delivery of drugs into the brain. In addition, the blood–brain barrier has been shown to be dysfunctional in Alzheimer's disease, the most common cause of dementia for which there is no cure. Mouse models of Alzheimer's disease have played a central role in investigating disease-specific changes in the blood–brain barrier, but the translation of findings from mouse models into the human system is hindered by interspecies differences. In an effort to develop new drug delivery techniques and/or understand changes in the human blood–brain barrier in Alzheimer's disease, several human blood–brain barrier *in vitro* models have been developed. These comprise primary and immortalized human endothelial cell-based models as well as human induced pluripotent stem cell-derived brain microvascular endothelial cell models. Both two- and three-dimensional

In: *Alzheimer's Disease: Drug Discovery.* Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

Copyright: The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

(2D and 3D) culture platforms have been established to better mimic the complexity of the brain. This chapter discusses the current blood–brain barrier models, their advantages and disadvantages as well as their potential to understand drug delivery in Alzheimer's disease.

Keywords: Alzheimer's disease; astrocyte cell culture model; blood–brain barrier; brain endothelial cell; induced pluripotent stem cell

INTRODUCTION

The blood–brain barrier (BBB), formed by tightly-sealed brain endothelial cells (BECs), serves as a selectively permeable membrane at the blood–brain interface. It allows for the delivery of oxygen and nutrients into the brain and at the same time, maintains a highly controlled brain milieu by preventing the entry of neuro-toxic blood components into the brain and transporting metabolic waste products from the brain to peripheral circulation (1). Furthermore, the brain vascular system supports the activity of neuronal networks through increased blood flow and oxygen supply, a mechanism termed as neurovascular coupling (2).

Integrity of the BEC layer is critical for the barrier function of the BBB and is achieved by the presence of tight and adherens junctions (AJ) between BECs. Tight junctions (TJ) consist of claudins-3,-5, and -12, occludin, and TJ associated *zona occludens* (ZO-1 and ZO-2) proteins (3). AJ are formed by vascular endothelial cadherin (VE-cadherin) and platelet endothelial cell adhesion molecule-1 (PECAM1) (3). Synergistically, TJ and AJ proteins ensure high transendothelial electrical resistance (TEER) of the BBB (1000–2000 ohm/cm² compared to 10 ohm/cm² in peripheral capillaries) and restrict paracellular permeability to molecules under 500 Da (4). Compared to peripheral endothelial cells, BECs also have more mitochondria and less fenestrations and pinocytic vesicles, further limiting the exchange of solutes across the BBB (4).

In addition to TJs and AJs, BECs express a unique selection of transporters, which either transport nutrients to the brain or transport molecules back to the blood. Solute carrier mediated transport (CMT) enables transendothelial exchange of organic cations and anions, carbohydrates, vitamins, fatty acids, nucleotides, amino acids, and hormones, whereas receptor-mediated transcytosis (RMT) facilitates the transport of peptides and proteins across the BBB, including insulin, transferrin, and apolipoproteins (1). ATP-binding cassette (ABC) transporters, including, P-glycoprotein (P-gp), multidrug resistance-associated proteins (MRPs), and breast cancer resistance protein (BCRP), protect the brain from the accumulation of xenobiotic compounds and drugs *via* their active efflux from BECs to the blood, and also greatly hinder drug delivery across the BBB (5).

Finally, to support BBB integrity, BECs are ensheathed by pericytes and astrocyte end-feet, and may create connections with microglia, neurons, and neural stem cells, which together form the so-called neurovascular unit (NVU) (Figure 1A) (1). Pericytes cover an approximate 1/3 of the BEC monolayer and can contract or relax cell membrane extensions to locally change cerebral blood flow in response to neuronal activity (6). Pericytes can also guide the polarisation of astrocyte end-feet, modulate TJ protein expression and permeability, participate in BBB immune responses, and regulate clearance of neurotoxic substances, such



Figure 1. Cellular structure and Alzheimer's disease (AD) specific changes in the blood–brain barrier (BBB). (A) Schematic representation of the BBB structure. The BBB is formed by brain endothelial cells (BECs) with astrocytes and pericytes functioning as key supporting cells. The BBB and other brain cells (neurons and microglia) form the neurovascular unit. (B) AD-specific changes in the BBB include altered tight junction and transporter expression in BECs, pericyte degradation, altered aquaporin-4 (AQP4) expression in astrocytes with these changes leading to increased permeability, reactive oxygen species (ROS) and neuroinflammation, and subsequent amyloid- β (A β) accumulation and neuronal loss. *Created using BioRender.com*.

as amyloid- β (A β) (6, 7). Astrocytes mediate innate immune responses on the brain barrier, provide nutrients from the blood to neurons, and regulate BBB barrier function by secreting protective factors, which modulate BEC TJ integrity (8). Astrocytes also play a primary role in the maintenance of parenchymal water and ionic homeostasis due to high expression of aquaporin 4 water (AQP4) channels

and ion transporters, contributing to clearance of interstitial solutes, including A β (9, 10). Together all the components comprise a highly specialized BBB unit, which plays a major role in maintaining homeostasis, demonstrates heterogeneity in disease, and poses a major hurdle for drug delivery.

ALZHEIMER'S DISEASE RELATED CHANGES IN THE BBB

Emerging evidence suggests that BBB dysfunction, together with A β plaques and hyperphosphorylated tau, is the third driving pathology underlying early Alzheimer's disease (AD) and corresponding cognitive decline (11–15). During preclinical asymptomatic stages of AD, cerebrovascular dysfunction is one of the first changes preceding any other detectable alterations characteristic of AD, including A β and tau (12, 16). The two-hit vascular hypothesis proposes that blood vessel impairment initiates the cascade of events by causing initial BBB dysfunction, decreased cerebral blood flow, and infiltration of neurotoxic molecules that in turn lead to neuronal loss (hit 1). BBB pathology subsequently causes impaired A β clearance and increased production of toxic A β species (hit 2), which synergistically with other vascular, genetic, and environmental risk factors, leads to progression of AD pathology (17).

Interestingly, major genetic risk factors of AD, including mutations is *APP* and *PSEN1* as well as *APOE* ϵ 4 polymorphism, have been shown to contribute to BBB leakage, brain microbleeds, BEC degeneration, pericyte injury, and abnormal A β clearance, linking multiple pathways of vascular- and neurodegeneration (13, 14, 18, 19). Increased BBB leakage in AD has been shown in both brain imaging studies as well as post-mortem brain tissue demonstrating an accumulation of blood-derived proteins in the brain (13, 20–23). Leakage of blood-borne factors further contributes to multiple pathological pathways in the AD brain, including BBB breakdown, pericyte dysfunction, neuronal death, neuroinflammation, and increased oxidative stress (24–27).

Defective function of nutrient and efflux transporters in the BBB has also been identified in human and animal studies of AD. Impaired glucose transport and reduced glucose transporter 1 (GLUT1) expression has been shown at the BBB of individuals with mild cognitive impairment (MCI) and in transgenic murine models of AD (28–30). The brain-to-blood efflux transporter P-gp has been implicated in AD pathology with clinical studies showing decreased P-gp activity in patients with mild AD, suggesting P-gp dysfunction and corresponding xenobiotic and A β build-up in the brain could contribute to AD pathogenesis (31, 32). Altered expression of efflux transporters in AD has further been demonstrated in a human induced pluripotent stem cell (hiPSC)-derived BEC model of familial AD (fAD) (33).

Cell-specific changes in the NVU have also been shown in AD. Human postmortem and cell model studies of AD have identified reduced BEC integrity, altered BEC TJ and AJ protein expression, and reduced expression in the brain endothelium of low-density lipoprotein receptor-related protein 1 (LRP1), the main receptor facilitating removal of A β (11, 33–35). Pericyte loss and decreased pericyte coverage of brain capillaries has also been observed in brain samples from AD individuals—an effect being amplified in *APOE* $\varepsilon 4$ carriers (11, 13, 36). In addition, pericytes contribute to the clearance of A β from the brain, and their loss has been shown to decrease the age of onset of AD and accelerate the development of A β and tau pathology in transgenic mice models (37). Astrocyte dysfunction is also implicated in AD, with altered APQ4 expression and localization as well as increased inflammatory and oxidative stress responses reported in human and mouse models of AD (38–40).

AD-specific changes in the BBB are summarized in Table 1 and Figure 1B. These alterations highlight severe AD-specific effects on the BBB with likely important consequences on disease progression and drug delivery.

TABLE 1	Alzheimer's disease related changed in the BBB		
Change	Details	Model	Ref
BBB breakdown	 Increased albumin ratio in CSF/blood Increased leakage in the hippocampus Increase in blood-borne factors in the brain 	Human brain imaging (MCI and APOE4) Human post-mortem tissue	(11, 13, 16, 36)
Vascular pathology	 Reduced cerebral blood flow Cerebral amyloid angiopathy Infarcts Haemorrhages Abnormal blood vessels Increased fibrin(ogen) deposition in cortical vessels 	Human brain images (LOAD) Human post-mortem tissue Transgenic (<i>PSEN1</i>) AD mouse model	(11, 12, 14, 15, 18, 19, 21)
Brain endothelium dysfunction	Reduced integrityReduced LRP1 expressionAltered TJ and AJ protein expressionAccumulation of CypA and MMP-9	Human post-mortem tissue Human fAD iBEC model	(11, 33–35)
Transporter dysfunction	 Reduced P-gp activity Reduced GLUT-1 expression Reduced glucose metabolism Altered efflux transporter expression 	Transgenic (APP) AD mouse model Human brain imaging (MCI and AD) Human fAD iBEC model	(28–33)
Pericyte dysfunction	 Loss of pericyte number and coverage in the hippocampus Increased PDGFβ in the CSF Accumulation of CypA and MMP-9 Upregulated calcineurin signalling 	Human brain imaging (MCI) Human post-mortem tissue Human APOE4 iBEC model	(11, 13, 16, 36, 64)
Astrocyte dysfunction	 Increased AQP4 expression Altered AQP4 distribution Altered inflammatory response Altered calcium signalling Increased oxidative stress 	Human post-mortem tissue Transgenic (<i>APP</i>) AD mouse model Human fAD iPSC-derived astrocytes	(38–40)

AJ, adherens junction; APOE4, apolipoprotein E allele ϵ 4 carrier; APP, amyloid precursor protein; AQP4, aquaporin 4; BBB, blood–brain barrier; CSF, cerebrospinal fluid; CypA, cyclophilin A; fAD, familial Alzheimer's disease; GLUT1, glucose transporter 1; iBEC, induced brain endothelial cell; iPSC, induced pluripotent stem cell; LRP1, low density lipoprotein receptor-related protein 1; LOAD, late onset Alzheimer's; MCI, mild cognitive impairment; MMP9, matrix metalloproteinase-9; PDGF β , platelet-derived growth factor β ; PSEN1, presenilin 1; TJ, tight junction.

DIFFERENCES BETWEEN ANIMAL AND HUMAN BBB AND IMPLICATIONS TO DRUG DELIVERY

Due to the challenges in studying the BBB in humans, animal models have played a central role in understanding AD-specific changes in the BBB and modeling delivery of brain targeting therapeutics (1, 41). Animal models have allowed for the in-depth investigation of BBB structure and biology, which is mostly only possible in humans using post-mortem tissue. The limitation of post-mortem tissue is that it does not allow for the analysis of BBB structure and function in a living person and at early stages of disease, critical for understanding the role of the BBB at different stages of disease progression.

Although extensive research has been conducted in animal models to identify novel therapeutics for AD, successful pre-clinical studies rarely translate into humans. In fact, over 99% of AD clinical trials with drugs which provided promising results in model animal systems, have not been successfully translated into humans (41, 42). The underlying problem is that although animal models of AD express key pathological hallmarks, including A β and hyper-phosphorylated tau, these models do not necessarily exhibit other biological features of AD and can be considered to not have AD (42). In addition, there is a lack of animal models for sporadic forms of AD, the most common type of AD in humans (42). Finally, interspecies differences are a central hindrance to translation of therapies from animal models to humans.

The mouse is one of the most commonly used animal models in BBB research and has been central to understanding BBB development and biology (43). Structurally mouse and human brains and BBB contain the same cell types, but distinct differences between mouse and human have been reported in properties of cells located in the neocortex, such as differential morphology and gene expression (43, 44). In addition, the human neocortex is vastly larger and more complex compared to that of the mouse, which complicates drug delivery in humans. The human neocortex also contains proportionally more astrocytes than the mouse cortex (45), which could potentially affect BBB formation and subsequent drug delivery.

Differences between the rodent and human BBB have been reported in TJ protein (TJP) and transporter expression and function. Key TJPs including claudin-5, occludin, and ZO-1 have been reported to have higher mRNA expression in mouse BECs compared to human BECs (46). In addition, comparison of protein level expression of transporters in brain microvessels identified a clearly higher expression of some transporters, including ABC (P-gp and MRP4) and solute carrier transporters (monocarboxylate transporter 1, L-type amino acid transporter, and organic anion transporter 3) in rats compared to humans (47). Real-time brain imaging also revealed differences in P-gp-dependent uptake of drugs between rat and human brains with brain concentrations of P-gp substrates found to be higher in humans than rats (48). These results indicate higher BBB permeability in humans compared to rodents and suggest that drug delivery experiments cannot be directly translated from rodents to humans. Interestingly, differences between humans and other primates have been shown to be smaller (47), with non-human primates potentially providing a more accurate model for human drug delivery than rodents.

Interspecies differences have also been found in other cell components of the NVU. Human astrocytes are larger and exhibit differences in process complexity than corresponding rat astrocytes (49). AQP4, the main water channel in the brain expressed by astrocytes, is polarized in astrocyte end-feet surrounding the BBB to a lesser extent in human astrocytes compared to mouse (50). Mislocalization of AQP4 has been linked to AD in humans (51), thus the differential expression and localization of AQP4 between mouse and human could have implications to how AD features in mouse models. Finally, primates are unique in terms of the presence of interlaminar astrocytes, which are not found in rodents (49). The disruption of processes of interlaminar astrocytes in AD has been reported (52), highlighting an important characteristic of AD humans that cannot be replicated in rodent models.

BBB IN VITRO CELL MODELS

To overcome species differences in BBB modeling, *in vitro* models of the human BBB are central to enhancing our understanding of BBB biology at a cellular level in health and disease. BBB cell models have traditionally been limited to primary and immortalized BECs (53), with hiPSC-derived BBB cells emerging as a novel approach for BBB modeling (54).

Primary and immortalized BECs

Human and mouse primary BEC isolation has been described from both fetal and adult brain tissue (Figure 2A) (55, 56). Primary BECs are reported to express BBB markers, such as TJPs and transporters (57), providing a tool for *in vitro* modeling of the BBB. BEC isolation from AD patient post-mortem tissue has also been described, revealing disease-specific differences compared to healthy BECs (56), suggesting a potential for using primary BECs for disease modeling.

The use of primary BECs is, however, associated with multiple limitations. Isolation of BECs from brain tissue is challenging as the proportion of BECs from all brain cell types is low (approximately 1-2%), easily resulting in contamination by unwanted cell types (55). In addition, the availability of tissue from patients is limited with donor-to-donor variability and ethical considerations posing challenges (57). Withdrawing BECs from their *in vivo* tissue and culturing them *in vitro* has also been reported to result in the loss of TJ markers and reduced transporter expression (58, 59). Passaging primary BECs has been reported to result in reduced TEER and an unstructured monolayer with loss of localized TJP expression (53), limiting the time that these cells can be utilized for experiments.

Immortalized human BEC lines (such as the hCMEC/D3 line) have helped overcome some of the challenges associated with primary BECs (Figure 2B) (60). Advantages of immortalized BEC lines include their ability to maintain cell properties over multiple passages, high viability, and expression of brain endotheliumspecific transport systems, making them an ideal model to perform high-throughput screening of new drugs targeting specific transporters and/or receptors (61). However, immortalization has been shown to affect the cell phenotype compared to primary BECs, including highly upregulated expression of genes related to



Figure 2. Brain endothelial cell (BEC) sources for blood–brain barrier (BBB) modeling. (A) Primary BECs are isolated directly from the brain with (B) immortalized BEC lines (e.g., hCMEC/D3) generated via the introduction of immortalization factors. (C) An alternative source of BECs are human induced pluripotent stem cells (hiPSC), which are generated by reprogramming patient-derived cells. hiPSCs can be used to generate all BBB cell types, including iBECs, pericytes and astrocytes. *Created using BioRender.com*.

nucleic acid processing and repair as well as interferon signaling (46). In addition, differential cell growth and altered gene expression of TJPs, receptors and transporters has been reported between immortalized and primary BECs (46). A major limitation of immortalized BECs is that they do not allow for the study of AD-specific changes. The overexpression of fAD mutations in *APP* and *PSEN1* in human immortalized neural progenitor cells has been described, resulting in the production of A β -plaques and pathological tau *in vitro* (62). A similar approach in human immortalized BECs could be possible, however, how this would alter BEC properties is unknown.

Human-induced pluripotent stem cell derived BECs

Due to the limitations associated with primary and immortalized BEC models, hiPSC-derived models have arisen as a promising approach for *in vitro* BBB modeling (Figure 2C) (54, 63). BECs can be generated from iPSCs (iBECs) with relative ease and AD patient-derived hiPSCs allow for the study of disease-specific differences (33, 63). iBECs exhibit key characteristics of BECs, including high TEER, high expression of BBB-specific TJPs as well as expression and function of efflux transporters (33, 63). Importantly, hiPSCs from the same patient can be differentiated into other cell types of the NVU, including pericytes and astrocytes, enabling the generation of an isogenic multi-cell BBB model (Figure 2C) (64, 65). In addition, CRISP/Cas9 gene editing allows for the further examination of the

contribution of AD risk genes on BBB function (33, 64). iBECs generated from patients with familial and sporadic AD have revealed key cellular differences giving insights into AD-specific effects on the BBB and potential implications to drug delivery (33, 64).

Although hiPSC-based technologies hold enormous potential for the development of preclinical BBB models, there are a few hindrances that need to be considered. hiPSC lines often exhibit a high level of variability and their generation and maintenance is expensive (66). The genetic editing of these cells also often results in loss of patient-specific epigenetic signatures and genetic instability. Other limitations include the lack of relevant genome matched-controls and lack of maturity in iPSC-derived cells, particularly important for modeling late onset diseases (67, 68). Finally, models based on hiPSC-derived iBECs allow only for a narrow experimental window, since cells tend to de-differentiate rapidly (54). Despite these limitations, continuous advancements in hiPSC research and commercially available reprogramming kits and cell culture reagents have the ability to ensure standardized culture conditions and the production of high quality hiPSC lines for BBB research.

IN VITRO CULTURE PLATFORMS FOR MODELING OF BBB STRUCTURE AND DRUG DELIVERY

In vitro platforms of the BBB are central to modeling the human BBB and to screening and developing BBB permeable drugs. BBB *in vitro* culture systems can broadly be divided into static 2D cultures and static or microfluidic 3D cultures.

2D models of the BBB

Static monolayer (i.e., 2D) culture systems are, to date, the most commonly used BBB *in vitro* model platforms. Most commonly, BECs (primary, immortalized or iPSC-derived) (69–71) are seeded inside a Transwell insert, in which cells are cultured on a permeable support as opposed to solid plastic (Figure 3A). The inside of the insert represents the luminal (blood) side, whereas the surrounding well, in which the insert is placed, represents the abluminal (brain) side (Figure 3A). The Transwell model allows for measurement of integrity as well as permeability of compounds through the BEC monolayer, from the luminal to the abluminal side (72).

BEC barrier integrity in the Transwell system can be increased via co-culture with pericytes, astrocytes, and other cells of the NVU usually resulting in increased TEER, higher TJP and transporter expression, promoting *in vivo*-like BBB phenotype (70, 73, 74). Co-cultures are achieved by culturing other BBB cells, such as pericytes or astrocytes, in the surrounding well, in which the BEC containing Transwell insert is placed, or on the underside of the BEC containing Transwell insert (Figure 3A) (63, 69). In this formation, cells are not in direct physical contact, as they are separated by a membrane, but will regulate each other via secreted factors. Direct contact Transwell models have also been described, where BECs and astrocytes, for example, are layered directly on top of each other, with this reported to result to higher TEER than indirect co-cultures (64, 70).



Figure 3. Two- (2D) and three- (3D) dimensional model platforms of the blood–brain barrier (BBB). (A) The Transwell system is the most common 2D platform for BBB modeling. In the Transwell system, brain endothelial cells (BECs) are cultured inside the Transwell insert with BBB co-cultures established by culturing other cell types (such as pericytes or astrocytes) in the surrounding well or on the underside of the Transwell insert. (B) 3D cultures can be established by embedding BBB cell types in a scaffolding matrix and allowing them to grow in a 3D conformation. In 3D and microfluidic models of the BBB, BECs are cultured in a tubular formation and the culture medium is allowed to flow through the BEC tube mimicking blood flow. Other BBB cell types (pericytes and astrocytes) are embedded in an extracellular matrix and allowed to grow adjacent to the BEC tube. *Created using BioRender.com*.

The Transwell assay has been widely used to study BBB integrity, permeability, and drug delivery (71, 72, 75). Although the Transwell assay is relatively easy to set-up and thus, widely used in BBB research, its limitations are that it lacks the complex 3D structure of the BBB *in vivo*, and BECs are cultured as a monolayer instead of a tubular structure, lacking complex cell interactions.

3D models of the BBB

Three-dimensional model systems of the BBB are emerging to better mimic the complexity of the BBB *in vivo*. In the brain, the BBB is a tubular structure, which is not accurately replicated using traditional 2D culture settings. Other features of the *in vivo* BBB include the complex interaction of BECs and other cells of the NVU with each other and with the extracellular matrix. In the body the BBB is also exposed to blood flow, which causes shear stress in BECs (4).

A central component of scaffold-based 3D models is using a supporting matrix (e.g., Matrigel or collagen I) that forms a gel, in which cells are able to grow in 3D conformation (Figure 3B) (76). In 3D BBB modeling, a central aim is to allow BECs to grow in a tubular formation using extracellular matrix (ECM) support (64, 77, 78).

Other cells, such as astrocytes and pericytes are then ideally cultured in direct contact with BECs, mimicking cell interactions in the body (64, 77). Vascular networks of both the healthy and AD BBB have been achieved *in vitro* by allowing BECs to self-assemble in an extracellular matrix containing 3D culture environment in coculture with astrocytes and pericytes (64, 79). When combining microfluidic technology with a 3D growth environment, it is possible to achieve tubular *in vitro* models of the BBB that also mimic blood flow (Figure 3B). These "organ-on-a-chip" models have shown promising outcomes in understanding cellular interactions and modeling drug transport, often allowing for minimal use of cells and culture reagents (79–81). Three-dimensional and microfluidic culture conditions have been used to model the AD BBB using immortalized human BECs and neural progenitor cells, with this model revealing AD-specific BBB dysfunction (78).

The limitation of many 3D and microfluidic BBB models is that they are complex in-house made platforms, often utilising proprietary materials, which are difficult to replicate in other laboratories. Commercial 3D and microfluidic platforms are emerging, providing a means for off-the-shelf systems for BBB modeling (82, 83). Other limitations include that most of the 3D and microfluidic BBB models have been established using cell lines, such as immortalized BECs. To be able to accurately model AD- or other neurodegenerative disease-specific effects, patient-derived cells, such as primary BECs or iPSC-derived iBECs would be ideal. Likely the challenge of culturing patient-derived BECs in co-culture with other cells as well as the poor long-term survival has hindered the development of patient-derived 3D and microfluidic models of the BBB.

BBB organoids provide an additional platform of studying BBB function in a 3D format without the need for a complex device. Previously generated BBB spheroids have consisted of primary or iPSC-derived BECs and other NVU cell types which spontaneously assemble under low-attachment conditions into a multicellular 3D structure (84, 85). These spheroids have been shown to demonstrate direct cell-tocell contacts, enhanced TJ and AJ expression, higher efflux transporters expression and reduced paracellular permeability, thus more closely mimicking the *in vivo* BBB when compared to traditional 2D cultures (84, 85). Human cortical spheroids containing BECs, pericytes, microglia, astrocytes, oligodendrocytes and neurons were shown to exhibit high expression of BBB markers as well as high viability, making them an attractive platform for drug discovery, disease modeling and long-term neuro- and cytotoxicity testing (85). The resemblance of organoid-like spheroids to the in vivo environment make them a promising platform for high-throughput screening of BBB penetrating drugs. Limitations of organoid research, however, include inter-sample variability, high processing time and technical difficulties in TEER measurements and permeability studies (86).

Permeability and drug delivery assays

Integrity of *in vitro* models of BBB is usually assessed via the permeability to fluorescently-conjugated molecules, such as dextran or sodium fluorescein, which have both been used in 2D Transwell and 3D microfluidic models of the BBB (33, 72, 83). For modeling drug delivery, various methods have been studied to transiently open the BBB *in vitro* cell model. Mannitol can be used to reversibly open the BBB, based on hyperosmosis, and has been used in an *in vitro* hiPSC-derived 3D BEC model to increase paracellular permeability (87). In the clinic, mannitol has been used to deliver antibodies to treat brain tumors (88), with its use otherwise not widely described due to possible side-effects. Human patient cell based AD BBB models may be important to allow the in-depth investigation of mannitol-dependent BBB opening (87), to develop its use for the delivery of therapeutic antibodies, such as to treat AD. Another reversible means to open the BBB is focused ultrasound (FUS) applied in-conjunction with gas-filled microbubbles (MB) (89). The safety of FUS+MB treatment in AD patients has been demonstrated, opening an avenue for potential A β clearance or therapeutic drug delivery (90). The effects of FUS+MB have also been investigated in a patient iPSC-derived fAD iBEC *in vitro* model, which demonstrated a differential response to FUS+MB treatment between patient and control cells, such as in FUS+MB-mediated permeability and A β clearance (33). These results highlight the importance of using patient-derived cell models to identify potential patient-specific differences that could affect drug delivery in the clinic.

ENHANCING DRUG DELIVERY IN ALZHEIMER'S DISEASE THROUGH *IN VITRO* MODELS OF BBB—CHALLENGES AND FUTURE PERSPECTIVES

BBB dysfunction in AD not only underlies disease pathogenesis and progression but also serves as the main burden for successful drug delivery. Additionally, BBB impairment in AD and subsequent physiological changes lead to disruption in drug delivery by diffusion, with pathological changes in transporters contributing to minimal (or no) bioavailability of the drug in the brain (1, 91). Furthermore, infiltrating toxic blood-derived products, reactive oxygen species and increased neuroinflammation may change the tightly controlled brain milieu and lead to undesired metabolism and/or interactions of delivered drugs (Figure 1B) (1). As such, to overcome challenges associated with drug delivery in AD, an in-depth understanding of AD-related changes at the BBB and subsequent effects on drug delivery are needed, which can be addressed using accurate model systems.

The basis of an accurate AD BBB cell model is that the cells, as closely as possible, recapitulate the disease phenotype. With challenges associated with all the described BEC sources, it is important to consider the benefits and limitations of each model. AD patient-derived primary BECs provide an opportunity to examine cells obtained directly from a patient (56), but patient brain tissue is difficult to obtain with these cells often only capturing late-stage of the disease. In addition, the lifespan of primary BECs in vitro is short, limiting their use for large-scale experiments (53). Patient-derived iPSCs on the other hand provide a scalable approach to generate BECs *in vitro* and one patient line can be used to generate all components of the NVU, providing an isogenic patient-specific BBB model (33, 64). In addition, using CRISPR/Cas9 gene editing, the role of specific AD mutations on BBB dysfunction can be investigated (33, 64). To ensure the translatable use of iPSC-derived cells, it is important to generate standardized differentiation protocols and utilize defined xeno-free reagents to minimize variability associated with iPSC culture.

Following selection of a good BBB cell source, it is vital to consider the culture environment for accurate BBB modeling. Successful screening for drug delivery across the BBB will be best achieved in an environment that as closely as possible mimics the brain environment in complexity, structure and chemistry. It has been shown that following a long-term culture (>3 months), AD-patient derived iPSC cerebral organoids exhibit hallmarks of AD, including A β plaques and accumulation of hyper-phosphorylated tau (92). Thus, to replicate physiological changes characteristic of AD, long-term cell culture models of the BBB are likely needed. Furthermore, to capture the complexity of the brain, standardized platforms that enable both 3D and microfluidic culture conditions are likely the future means to establishing an accurate BBB *in vitro* model. The challenge that still exists is developing a reproducible culture platform that is easy and cost-effective for laboratories to use, with emerging commercial platforms providing a potential solution (82, 83).

Finally, to accelerate drug discovery in AD, high-throughput testing in BBB *in vitro* models is important. Scaling down reagent and cell use (i.e., minimizing culture platform size) and scaling up the number of replicates, would help to achieve testing of a large number of drugs in a cost-effective manner. For this reason, small-scale "organ-on-a-chip" type platforms may be ideal (93). It is also vital to consider how easily drug delivery efficiency and downstream effects can be measured. Ideally, a BBB *in vitro* model that enables the simultaneous observation of drug delivery efficiency as well as effects on brain cell types and AD pathologies (A β and tau) would be a key step in AD drug discovery.

CONCLUSION

BBB dysfunction is associated with AD, likely playing a central role in AD progression and drug delivery. Considering inter-species differences, accurate human BBB *in vitro* models are needed to understand AD-specific changes on the BBB and subsequent effects on drug delivery. Human BBB cell models of AD, in particular using patient-derived cells in a culture environment that accurately mimics the AD brain, are an important step to enhance drug discovery in AD.

Conflict of Interest: The authors declare no potential conflict of interest with respect to research, authorship, and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s)

REFERENCES

- Sweeney MD, Sagare AP, Zlokovic BV. Blood-brain barrier breakdown in Alzheimer disease and other neurodegenerative disorders. Nat Rev Neurol. 2018;14(3):133–50. http://dx.doi.org/10.1038/ nrneurol.2017.188
- Kisler K, Nikolakopoulou AM, Sweeney MD, Lazic D, Zhao Z, Zlokovic BV. Acute ablation of cortical pericytes leads to rapid neurovascular uncoupling. Front Cell Neurosci. 2020;14:27. http://dx.doi. org/10.3389/fncel.2020.00027

- Tietz S, Engelhardt B. Brain barriers: Crosstalk between complex tight junctions and adherens junctions. J Cell Biol. 2015;209(4):493–506. http://dx.doi.org/10.1083/jcb.201412147
- Aday S, Cecchelli R, Hallier-Vanuxeem D, Dehouck MP, Ferreira L. Stem cell-based human bloodbrain barrier models for drug discovery and delivery. Trends Biotechnol. 2016;34(5):382–93. http:// dx.doi.org/10.1016/j.tibtech.2016.01.001
- Qosa H, Miller DS, Pasinelli P, Trotti D. Regulation of ABC efflux transporters at blood-brain barrier in health and neurological disorders. Brain Res. 2015;1628(Pt B):298–316. http://dx.doi.org/10.1016/j. brainres.2015.07.005
- Brown LS, Foster CG, Courtney JM, King NE, Howells DW, Sutherland BA. Pericytes and neurovascular function in the healthy and diseased brain. Front Cell Neurosci. 2019;13:282. http://dx.doi. org/10.3389/fncel.2019.00282
- Ma Q, Zhao Z, Sagare AP, Wu Y, Wang M, Owens NC, et al. Blood-brain barrier-associated pericytes internalize and clear aggregated amyloid-β42 by LRP1-dependent apolipoprotein E isoform-specific mechanism. Mol Neurodegener. 2018;13(1):57. http://dx.doi.org/10.1186/s13024-018-0286-0
- Michinaga S, Koyama Y. Dual roles of astrocyte-derived factors in regulation of blood-brain barrier function after brain damage. Int J Mol Sci. 2019;20(3):571. http://dx.doi.org/10.3390/ijms20030571
- 9. Iliff JJ, Wang M, Liao Y, Plogg BA, Peng W, Gundersen GA, et al. A paravascular pathway facilitates CSF flow through the brain parenchyma and the clearance of interstitial solutes, including amyloid β . Sci Transl Med. 2012;4(147):147ra11. http://dx.doi.org/10.1126/scitranslmed.3003748
- Mader S, Brimberg L. Aquaporin-4 water channel in the brain and its implication for health and disease. Cells. 2019;8(2):90. http://dx.doi.org/10.3390/cells8020090
- 11. Halliday MR, Rege SV, Ma Q, Zhao Z, Miller CA, Winkler EA, et al. Accelerated pericyte degeneration and blood-brain barrier breakdown in apolipoprotein E4 carriers with Alzheimer's disease. J Cereb Blood Flow Metab. 2016;36(1):216–27. http://dx.doi.org/10.1038/jcbfm.2015.44
- Iturria-Medina Y, Sotero RC, Toussaint PJ, Mateos-Pérez JM, Evans AC, Weiner MW, et al. Early role of vascular dysregulation on late-onset Alzheimer's disease based on multifactorial data-driven analysis. Nat Commun. 2016;7(1):11934. http://dx.doi.org/10.1038/ncomms11934
- Montagne A, Nation DA, Sagare AP, Barisano G, Sweeney MD, Chakhoyan A, et al. APOE4 leads to blood–brain barrier dysfunction predicting cognitive decline. Nature. 2020;581(7806):71–6. http:// dx.doi.org/10.1038/s41586-020-2247-3
- Niwa A, Matsuo K, Shindo A, Yata K, Shiraishi T, Tomimoto H. Clinical and neuropathological findings in a patient with familial Alzheimer disease showing a mutation in the PSEN1 gene. Neuropathology. 2013;33(2):199–203. http://dx.doi.org/10.1111/j.1440-1789.2012.01340.x
- Toledo JB, Arnold SE, Raible K, Brettschneider J, Xie SX, Grossman M, et al. Contribution of cerebrovascular disease in autopsy confirmed neurodegenerative disease cases in the National Alzheimer's Coordinating Centre. Brain. 2013;136(Pt 9):2697–706. http://dx.doi.org/10.1093/ brain/awt188
- Montagne A, Barnes SR, Sweeney MD, Halliday MR, Sagare AP, Zhao Z, et al. Blood-brain barrier breakdown in the aging human hippocampus. Neuron. 2015;85(2):296–302. http://dx.doi. org/10.1016/j.neuron.2014.12.032
- 17. Zlokovic BV. Neurovascular pathways to neurodegeneration in Alzheimer's disease and other disorders. Nat Rev Neurosci. 2011;12(12):723–38. http://dx.doi.org/10.1038/nrn3114
- Gama Sosa MA, Gasperi RD, Rocher AB, Wang AC, Janssen WG, Flores T, et al. Age-related vascular pathology in transgenic mice expressing presenilin 1-associated familial Alzheimer's disease mutations. Am J Pathol. 2010;176(1):353–68. http://dx.doi.org/10.2353/ajpath.2010.090482
- Zarranz JJ, Fernandez-Martinez M, Rodriguez O, Mateos B, Iglesias S, Baron JC. Iowa APP mutation-related hereditary cerebral amyloid angiopathy (CAA): A new family from Spain. J Neurol Sci. 2016;363:55–6. http://dx.doi.org/10.1016/j.jns.2016.02.029
- Cortes-Canteli M, Paul J, Norris EH, Bronstein R, Ahn HJ, Zamolodchikov D, et al. Fibrinogen and beta-amyloid association alters thrombosis and fibrinolysis: A possible contributing factor to Alzheimer's disease. Neuron. 2010;66(5):695–709. http://dx.doi.org/10.1016/j.neuron.2010.05.014
- Hultman K, Strickland S, Norris EH. The APOE ε4/ε4 genotype potentiates vascular fibrin(ogen) deposition in amyloid-laden vessels in the brains of Alzheimer's disease patients. J Cereb Blood Flow Metab. 2013;33(8):1251–8. http://dx.doi.org/10.1038/jcbfm.2013.76
- Ryu JK, McLarnon JG. A leaky blood-brain barrier, fibrinogen infiltration and microglial reactivity in inflamed Alzheimer's disease brain. J Cell Mol Med. 2009;13(9a):2911–25. http://dx.doi. org/10.1111/j.1582-4934.2008.00434.x
- Zipser BD, Johanson CE, Gonzalez L, Berzin TM, Tavares R, Hulette CM, et al. Microvascular injury and blood-brain barrier leakage in Alzheimer's disease. Neurobiol Aging. 2007;28(7):977–86. http:// dx.doi.org/10.1016/j.neurobiolaging.2006.05.016
- Altinoz MA, Guloksuz S, Schmidt-Kastner R, Kenis G, Ince B, Rutten BPF. Involvement of hemoglobins in the pathophysiology of Alzheimer's disease. Exp Gerontol. 2019;126:110680. http://dx.doi. org/10.1016/j.exger.2019.110680
- Di Pardo A, Castaldo S, Capocci L, Amico E, Vittorio M. Assessment of blood-brain barrier permeability by intravenous infusion of FITC-labeled albumin in a mouse model of neurodegenerative disease. J Visual Exp. 2017;129:56389. http://dx.doi.org/10.3791/56389
- Festoff BW, Sajja RK, van Dreden P, Cucullo L. HMGB1 and thrombin mediate the blood-brain barrier dysfunction acting as biomarkers of neuroinflammation and progression to neurodegeneration in Alzheimer's disease. J Neuroinflamm. 2016;13(1):194. http://dx.doi.org/10.1186/s12974-016-0670-z
- Merlini M, Rafalski VA, Rios Coronado PE, Gill TM, Ellisman M, Muthukumar G, et al. Fibrinogen induces microglia-mediated spine elimination and cognitive impairment in an Alzheimer's disease model. Neuron. 2019;101(6):1099–108.e6. http://dx.doi.org/10.1016/j.neuron.2019.01.014
- Mosconi L, Tsui WH, Herholz K, Pupi A, Drzezga A, Lucignani G, et al. Multicenter standardized 18F-FDG PET diagnosis of mild cognitive impairment, Alzheimer's disease, and other dementias. J Nucl Med. 2008;49(3):390–8. http://dx.doi.org/10.2967/jnumed.107.045385
- Niwa K, Kazama K, Younkin SG, Carlson GA, Iadecola C. Alterations in cerebral blood flow and glucose utilization in mice overexpressing the amyloid precursor protein. Neurobiol Dis. 2002;9(1):61–8. http://dx.doi.org/10.1006/nbdi.2001.0460
- Winkler EA, Nishida Y, Sagare AP, Rege SV, Bell RD, Perlmutter D, et al. GLUT1 reductions exacerbate Alzheimer's disease vasculo-neuronal dysfunction and degeneration. Nat Neurosci. 2015;18(4): 521–30. http://dx.doi.org/10.1038/nn.3966
- Deo AK, Borson S, Link JM, Domino K, Eary JF, Ke B, et al. Activity of P-glycoprotein, a β-amyloid transporter at the blood-brain barrier, is compromised in patients with mild Alzheimer disease. J Nucl Med. 2014;55(7):1106–11. http://dx.doi.org/10.2967/jnumed.113.130161
- van Assema DME, Lubberink M, Bauer M, van der Flier WM, Schuit RC, Windhorst AD, et al. Blood– brain barrier P-glycoprotein function in Alzheimer's disease. Brain. 2011;135(1):181–9. http://dx.doi. org/10.1093/brain/awr298
- Oikari LE, Pandit R, Stewart R, Cuní-López C, Quek H, Sutharsan R, et al. Altered brain endothelial cell phenotype from a familial Alzheimer mutation and its potential implications for amyloid clearance and drug delivery. Stem Cell Rep. 2020;14(5):924–39. http://dx.doi.org/10.1016/j. stemcr.2020.03.011
- Donahue JE, Flaherty SL, Johanson CE, Duncan JA, 3rd, Silverberg GD, Miller MC, et al. RAGE, LRP-1, and amyloid-beta protein in Alzheimer's disease. Acta Neuropathol. 2006;112(4):405–15. http://dx.doi.org/10.1007/s00401-006-0115-3
- Yamazaki Y, Shinohara M, Shinohara M, Yamazaki A, Murray ME, Liesinger AM, et al. Selective loss of cortical endothelial tight junction proteins during Alzheimer's disease progression. Brain. 2019;142(4):1077–92. http://dx.doi.org/10.1093/brain/awz011
- Sengillo JD, Winkler EA, Walker CT, Sullivan JS, Johnson M, Zlokovic BV. Deficiency in mural vascular cells coincides with blood-brain barrier disruption in Alzheimer's disease. Brain Pathol (Zurich, Switzerland). 2013;23(3):303–10. http://dx.doi.org/10.1111/bpa.12004
- 37. Sagare AP, Bell RD, Zhao Z, Ma Q, Winkler EA, Ramanathan A, et al. Pericyte loss influences Alzheimer-like neurodegeneration in mice. Nat Commun. 2013;4(1):2932. http://dx.doi.org/10.1038/ncomms3932
- Moftakhar P, Lynch MD, Pomakian JL, Vinters HV. Aquaporin expression in the brains of patients with or without cerebral amyloid angiopathy. J Neuropathol Exp Neurol. 2010;69(12):1201–9. http:// dx.doi.org/10.1097/NEN.0b013e3181fd252c
- Oksanen M, Petersen AJ, Naumenko N, Puttonen K, Lehtonen Š, Gubert Olivé M, et al. PSEN1 mutant iPSC-derived model reveals severe astrocyte pathology in Alzheimer's disease. Stem Cell Rep. 2017;9(6):1885–97. http://dx.doi.org/10.1016/j.stemcr.2017.10.016

- Yang J, Zhang R, Shi C, Mao C, Yang Z, Suo Z, et al. AQP4 association with amyloid deposition and astrocyte pathology in the Tg-ArcSwe mouse model of Alzheimer's disease. J Alzheimers Dis. 2017;57(1):157–69. http://dx.doi.org/10.3233/JAD-160957
- Cummings J, Feldman HH, Scheltens P. The "rights" of precision drug development for Alzheimer's disease. Alzheimers Res Ther. 2019;11(1):76. http://dx.doi.org/10.1186/s13195-019-0529-5
- 42. Drummond E, Wisniewski T. Alzheimer's disease: Experimental models and reality. Acta Neuropathol. 2017;133(2):155–75. http://dx.doi.org/10.1007/s00401-016-1662-x
- 43. O'Brown NM, Pfau SJ, Gu C. Bridging barriers: A comparative look at the blood-brain barrier across organisms. Genes Dev. 2018;32(7–8):466–78. http://dx.doi.org/10.1101/gad.309823.117
- 44. Hodge RD, Bakken TE, Miller JA, Smith KA, Barkan ER, Graybuck LT, et al. Conserved cell types with divergent features in human versus mouse cortex. Nature. 2019;573(7772):61–8. http://dx.doi. org/10.1038/s41586-019-1506-7
- Oberheim NA, Wang X, Goldman S, Nedergaard M. Astrocytic complexity distinguishes the human brain. Trends Neurosci. 2006;29(10):547–53. http://dx.doi.org/10.1016/j.tins.2006.08.004
- Urich E, Lazic SE, Molnos J, Wells I, Freskgård PO. Transcriptional profiling of human brain endothelial cells reveals key properties crucial for predictive in vitro blood-brain barrier models. PLoS One. 2012;7(5):e38149. http://dx.doi.org/10.1371/journal.pone.0038149
- Hoshi Y, Uchida Y, Tachikawa M, Inoue T, Ohtsuki S, Terasaki T. Quantitative atlas of blood-brain barrier transporters, receptors, and tight junction proteins in rats and common marmoset. J Pharmaceut Sci. 2013;102(9):3343–55. http://dx.doi.org/10.1002/jps.23575
- Syvänen S, Lindhe O, Palner M, Kornum BR, Rahman O, Långström B, et al. Species differences in bloodbrain barrier transport of three positron emission tomography radioligands with emphasis on P-glycoprotein transport. Drug Metab Disposit. 2009;37(3):635–43. http://dx.doi.org/10.1124/dmd.108.024745
- 49. Oberheim NA, Takano T, Han X, He W, Lin JH, Wang F, et al. Uniquely hominid features of adult human astrocytes. J Neurosci. 2009;29(10):3276–87. http://dx.doi.org/10.1523/JNEUROSCI.4707-08.2009
- Eidsvaag VA, Enger R, Hansson HA, Eide PK, Nagelhus EA. Human and mouse cortical astrocytes differ in aquaporin-4 polarization toward microvessels. Glia. 2017;65(6):964–73. http://dx.doi. org/10.1002/glia.23138
- Zeppenfeld DM, Simon M, Haswell JD, D'Abreo D, Murchison C, Quinn JF, et al. Association of perivascular localization of aquaporin-4 with cognition and Alzheimer disease in aging brains. JAMA Neurol. 2017;74(1):91–9. http://dx.doi.org/10.1001/jamaneurol.2016.4370
- 52. Colombo JA, Quinn B, Puissant V. Disruption of astroglial interlaminar processes in Alzheimer's disease. Brain Res Bull. 2002;58(2):235–42. http://dx.doi.org/10.1016/S0361-9230(02)00785-2
- Fujimoto T, Morofuji Y, Nakagawa S, Kovac A, Horie N, Izumo T, et al. Comparison of the rate of dedifferentiation with increasing passages among cell sources for an in vitro model of the bloodbrain barrier. J Neural Transm (Vienna, Austria). 2020;127(8):1117–24. http://dx.doi.org/10.1007/ s00702-0220-2020-1
- Lippmann ES, Azarin SM, Kay JE, Nessler RA, Wilson HK, Al-Ahmad A, et al. Derivation of bloodbrain barrier endothelial cells from human pluripotent stem cells. Nat Biotechnol. 2012;30(8): 783–91. http://dx.doi.org/10.1038/nbt.2247
- Navone SE, Marfia G, Invernici G, Cristini S, Nava S, Balbi S, et al. Isolation and expansion of human and mouse brain microvascular endothelial cells. Nat Protocols. 2013;8(9):1680–93. http://dx.doi. org/10.1038/nprot.2013.107
- 56. Yin X, Wright J, Wall T, Grammas P. Brain endothelial cells synthesize neurotoxic thrombin in Alzheimer's disease. Am J Pathol. 2010;176(4):1600–6. http://dx.doi.org/10.2353/ajpath.2010.090406
- 57. He Y, Yao Y, Tsirka SE, Cao Y. Cell-culture models of the blood-brain barrier. Stroke. 2014;45(8): 2514–26. http://dx.doi.org/10.1161/STROKEAHA.114.005427
- Couraud PO, Greenwood J, Roux F, Adamson P. Development and characterization of immortalized cerebral endothelial cell lines. Methods Mol Med. 2003;89:349–64. http://dx.doi.org/ 10.1385/1-59259-419-0:349
- Lyck R, Ruderisch N, Moll AG, Steiner O, Cohen CD, Engelhardt B, et al. Culture-induced changes in blood-brain barrier transcriptome: Implications for amino-acid transporters in vivo. J Cereb Blood Flow Metab. 2009;29(9):1491–502. http://dx.doi.org/10.1038/jcbfm.2009.72

- Weksler BB, Subileau EA, Perrière N, Charneau P, Holloway K, Leveque M, et al. Blood-brain barrierspecific properties of a human adult brain endothelial cell line. FASEB J. 2005;19(13):1872–4. http:// dx.doi.org/10.1096/fj.04-3458fje
- 61. Daniels BP, Cruz-Orengo L, Pasieka TJ, Couraud PO, Romero IA, Weksler B, et al. Immortalized human cerebral microvascular endothelial cells maintain the properties of primary cells in an in vitro model of immune migration across the blood brain barrier. J Neurosci Methods. 2013;212(1):173–9. http://dx.doi.org/10.1016/j.jneumeth.2012.10.001
- Choi SH, Kim YH, Hebisch M, Sliwinski C, Lee S, D'Avanzo C, et al. A three-dimensional human neural cell culture model of Alzheimer's disease. Nature. 2014;515(7526):274–8. http://dx.doi. org/10.1038/nature13800
- Lippmann ES, Al-Ahmad A, Azarin SM, Palecek SP, Shusta EV. A retinoic acid-enhanced, multicellular human blood-brain barrier model derived from stem cell sources. Sci Rep. 2014;4:4160. http:// dx.doi.org/10.1038/srep04160
- 64. Blanchard JW, Bula M, Davila-Velderrain J, Akay LA, Zhu L, Frank A, et al. Reconstruction of the human blood-brain barrier in vitro reveals a pathogenic mechanism of APOE4 in pericytes. Nat Med. 2020;26(6):952–63. http://dx.doi.org/10.1038/s41591-020-0886-4
- Canfield SG, Stebbins MJ, Morales BS, Asai SW, Vatine GD, Svendsen CN, et al. An isogenic bloodbrain barrier model comprising brain endothelial cells, astrocytes, and neurons derived from human induced pluripotent stem cells. J Neurochem. 2017;140(6):874–88. http://dx.doi.org/10.1111/ jnc.13923
- Berry BJ, Smith AST, Young JE, Mack DL. Advances and current challenges associated with the use of human induced pluripotent stem cells in modeling neurodegenerative disease. Cells Tissues Organs. 2018;205(5–6):331–49. http://dx.doi.org/10.1159/000493018
- 67. Doss MX, Sachinidis A. Current challenges of iPSC-based disease modeling and therapeutic implications. Cells. 2019;8(5):403. http://dx.doi.org/10.3390/cells8050403
- Lee J, Bayarsaikhan D, Bayarsaikhan G, Kim JS, Schwarzbach E, Lee B. Recent advances in genome editing of stem cells for drug discovery and therapeutic application. Pharmacol Ther. 2020;209:107501. http://dx.doi.org/10.1016/j.pharmthera.2020.107501
- Appelt-Menzel A, Cubukova A, Günther K, Edenhofer F, Piontek J, Krause G, et al. Establishment of a human blood-brain barrier co-culture model mimicking the neurovascular unit using induced pluri- and multipotent stem cells. Stem Cell Rep. 2017;8(4):894–906. http://dx.doi.org/10.1016/j. stemcr.2017.02.021
- Kulczar C, Lubin KE, Lefebvre S, Miller DW, Knipp GT. Development of a direct contact astrocytehuman cerebral microvessel endothelial cells blood-brain barrier coculture model. J Pharm Pharmacol. 2017;69(12):1684–96. http://dx.doi.org/10.1111/jphp.12803
- Wang Y, Wang N, Cai B, Wang GY, Li J, Piao XX. In vitro model of the blood-brain barrier established by co-culture of primary cerebral microvascular endothelial and astrocyte cells. Neural Regen Res. 2015;10(12):2011–17. http://dx.doi.org/10.4103/1673-5374.172320
- Stebbins MJ, Wilson HK, Canfield SG, Qian T, Palecek SP, Shusta EV. Differentiation and characterization of human pluripotent stem cell-derived brain microvascular endothelial cells. Methods (San Diego, Calif). 2016;101:93–102. http://dx.doi.org/10.1016/j.ymeth.2015.10.016
- Nakagawa S, Deli MA, Kawaguchi H, Shimizudani T, Shimono T, Kittel A, et al. A new blood-brain barrier model using primary rat brain endothelial cells, pericytes and astrocytes. Neurochem Int. 2009;54(3–4):253–63. http://dx.doi.org/10.1016/j.neuint.2008.12.002
- 74. Stone NL, England TJ, O'Sullivan SE. A novel transwell blood brain barrier model using primary human cells. Front Cell Neurosci. 2019;13:230. http://dx.doi.org/10.3389/fncel.2019.00230
- Ohshima M, Kamei S, Fushimi H, Mima S, Yamada T, Yamamoto T. Prediction of drug permeability using in vitro blood-brain barrier models with human induced pluripotent stem cell-derived brain microvascular endothelial cells. BioRes Open Access. 2019;8(1):200–9. http://dx.doi.org/10.1089/ biores.2019.0026
- Edmondson R, Broglie JJ, Adcock AF, Yang L. Three-dimensional cell culture systems and their applications in drug discovery and cell-based biosensors. Assay Drug Dev Technol. 2014;12(4):207–18. http://dx.doi.org/10.1089/adt.2014.573

- Herland A, van der Meer AD, FitzGerald EA, Park TE, Sleeboom JJ, Ingber DE. Distinct contributions of astrocytes and pericytes to neuroinflammation identified in a 3D human blood-brain barrier on a chip. PLoS One. 2016;11(3):e0150360. http://dx.doi.org/10.1371/journal.pone.0150360
- Shin Y, Choi SH, Kim E, Bylykbashi E, Kim JA, Chung S, et al. Blood-brain barrier dysfunction in a 3D in vitro model of Alzheimer's disease. Adv Sci (Weinheim, Baden-Wurttemberg, Germany). 2019;6(20):1900962. http://dx.doi.org/10.1002/advs.201900962
- 79. Campisi M, Shin Y, Osaki T, Hajal C, Chiono V, Kamm RD. 3D self-organized microvascular model of the human blood-brain barrier with endothelial cells, pericytes and astrocytes. Biomaterials. 2018;180:117–29. http://dx.doi.org/10.1016/j.biomaterials.2018.07.014
- Adriani G, Ma D, Pavesi A, Kamm RD, Goh EL. A 3D neurovascular microfluidic model consisting of neurons, astrocytes and cerebral endothelial cells as a blood-brain barrier. Lab on a chip. 2017;17(3):448–59. http://dx.doi.org/10.1039/C6LC00638H
- Griep LM, Wolbers F, de Wagenaar B, ter Braak PM, Weksler BB, Romero IA, et al. BBB ON CHIP: Microfluidic platform to mechanically and biochemically modulate blood-brain barrier function. Biomed Microdev. 2013;15(1):145–50. http://dx.doi.org/10.1007/s10544-012-9699-7
- Silvani G, Scognamiglio C, Caprini D, Marino L, Chinappi M, Sinibaldi G, et al. Reversible cavitationinduced junctional opening in an artificial endothelial layer. Small. 2019;15(51):1905375. http:// dx.doi.org/10.1002/smll.201905375
- Wevers NR, Kasi DG, Gray T, Wilschut KJ, Smith B, van Vught R, et al. A perfused human blood–brain barrier on-a-chip for high-throughput assessment of barrier function and antibody transport. Fluids Barr CNS. 2018;15(1):23. http://dx.doi.org/10.1186/s12987-018-0108-3
- Cho C-F, Wolfe JM, Fadzen CM, Calligaris D, Hornburg K, Chiocca EA, et al. Blood-brain-barrier spheroids as an in vitro screening platform for brain-penetrating agents. Nat Commun. 2017;8(1):15623. http://dx.doi.org/10.1038/ncomms15623
- Nzou G, Wicks RT, Wicks EE, Seale SA, Sane CH, Chen A, et al. Human cortex spheroid with a functional blood brain barrier for high-throughput neurotoxicity screening and disease modeling. Sci Rep. 2018;8(1):7413. http://dx.doi.org/10.1038/s41598-018-25603-5
- Bhalerao A, Sivandzade F, Archie SR, Chowdhury EA, Noorani B, Cucullo L. In vitro modeling of the neurovascular unit: Advances in the field. Fluids Barr CNS. 2020;17(1):22. http://dx.doi.org/10.1186/ s12987-020-00183-7
- Linville RM, DeStefano JG, Sklar MB, Chu C, Walczak P, Searson PC. Modeling hyperosmotic blood– brain barrier opening within human tissue-engineered in vitro brain microvessels. J Cereb Blood Flow Metab. 2020;40(7):1517–32. http://dx.doi.org/10.1177/0271678X19867980
- Burkhardt JK, Riina H, Shin BJ, Christos P, Kesavabhotla K, Hofstetter CP, et al. Intra-arterial delivery of bevacizumab after blood-brain barrier disruption for the treatment of recurrent glioblastoma: Progression-free survival and overall survival. World Neurosurg. 2012;77(1):130–4. http://dx.doi. org/10.1016/j.wneu.2011.05.056
- Aryal M, Arvanitis CD, Alexander PM, McDannold N. Ultrasound-mediated blood-brain barrier disruption for targeted drug delivery in the central nervous system. Adv Drug Deliv Rev. 2014;72: 94–109. http://dx.doi.org/10.1016/j.addr.2014.01.008
- Lipsman N, Meng Y, Bethune AJ, Huang Y, Lam B, Masellis M, et al. Blood–brain barrier opening in Alzheimer's disease using MR-guided focused ultrasound. Nat Commun. 2018;9(1):2336. http:// dx.doi.org/10.1038/s41467-018-04529-6
- Sekhar GN, Fleckney AL, Boyanova ST, Rupawala H, Lo R, Wang H, et al. Region-specific blood-brain barrier transporter changes leads to increased sensitivity to amisulpride in Alzheimer's disease. Fluids Barr CNS. 2019;16(1):38. http://dx.doi.org/10.1186/s12987-019-0158-1
- Gonzalez C, Armijo E, Bravo-Alegria J, Becerra-Calixto A, Mays CE, Soto C. Modeling amyloid beta and tau pathology in human cerebral organoids. Mol Psychiatry. 2018;23(12):2363–74. http://dx.doi. org/10.1038/s41380-018-0229-8
- 93. Wang DZ, Samanipour R, Kim K. Organ-on-a-chip platforms for drug screening and tissue engineering. 2016;9:209–33. Springer, Cham. https://doi.org/10.1007/978-3-319-21813-7_10

Blood–Brain Barrier Degradation and the Implication of SPARC Protein as a Potential Therapeutic Target for Alzheimer's Disease

Alexander Pilozzi¹ • Caitlin Carro¹ • Michael Whalen² • Xudong Huang¹

¹Neurochemistry Laboratory, Department of Psychiatry, Massachusetts General Hospital and Harvard Medical School, Charlestown, MA, USA; ²Neuroscience Center, Department of Pediatrics, Massachusetts General Hospital and Harvard Medical School, Boston, MA, USA

Author for correspondence: Xudong Huang, Neurochemistry Laboratory, Department of Psychiatry, Massachusetts General Hospital and Harvard Medical School, Charlestown, MA, USA. Email: Huang.Xudong@mgh.harvard.edu

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch8

Abstract: Alzheimer's disease is a progressive neurodegenerative disorder affecting a substantial portion of the older population, with the number of afflicted individuals expected to grow with time. Although numerous contributing factors to the disorder have been identified, there is currently no cure or effective prevention method. With the situation as dire as it is, many efforts have been made to shed light on the mechanisms tying diverse contributing factors to the pathogenesis of Alzheimer's disease. One common neuropathological feature of Alzheimer's disease is the dysfunction of the blood–brain barrier, which normally maintains brain homeostasis by isolating it from the peripheral circulation and mediating the transport of various bloodborne elements in and out of the brain. An increase in the blood–brain barrier permeability has been observed in Alzheimer's disease at a level considerably above normal aging. This chapter provides an overview of the effects of aging, the neuroimmune system, inflammation, traumatic brain injury, apolipoprotein E gene ε 4 allele, and secreted protein acidic and rich in cysteine (SPARC) protein on blood–brain barrier. The potential

In: Alzheimer's Disease: Drug Discovery. Huang X (Editor). Exon Publications, Brisbane, Australia.

ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020 **Copyright:** The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

of SPARC as a therapeutic target for Alzheimer's disease, and the application of deep-learning-based virtual screening tools against SPARC protein are explored.

Keywords: Alzheimer's disease; blood–brain barrier; Hevin; secreted protein acidic and rich in cysteine; virtual screening

INTRODUCTION

Alzheimer's disease (AD) is a progressive neurodegenerative disorder that has widespread detrimental effects on memory and cognitive abilities that worsen over time. The disease is ultimately fatal, often through complications associated with decreased cognition. It is the most common form of senile dementia, with an estimated 5.8 million Americans currently afflicted with AD, a number that is expected to increase dramatically with an aging population that is more consistently reaching the "oldest-old" phase where AD risk is at its highest (1). AD has a substantial economic impact, with projections indicating the global cost of dementia could balloon to 2 trillion US dollars by 2030 (2). Given the threat the disease poses, researchers have been tackling AD from different angles, but as of now attempts to develop treatments have been met with widespread clinical failure (3).

The blood–brain barrier (BBB) consists of endothelial cells, serving as a layer of separation between blood vessels and the brain. The endothelia that line the blood vessels of the brain serve to isolate the brain parenchyma from bloodborne molecules that lack corresponding transporters to mediate their entry, and maintain the equilibrium of the brain's environment. The barrier is also comprised of other elements interacting with endothelial cells, including astrocyte foot processes and pericytes, which together with neurons and microglia comprise the neurovascular unit. The BBB is also responsible for controlling immune surveillance within the brain, by restricting the flow of immune cells (4). Dysfunction of the BBB is implicated in AD pathogenesis. The BBB is partially responsible for the clearance of amyloid-beta (A β), which builds up and forms plaques in AD, and the BBB is a site of CNS inflammation, which is frequently observed in AD patients (5).

While the integrity of the BBB is tightly regulated, emerging evidence implicates the matricellular proteins, secreted protein acidic and rich in cysteine (SPARC), and Hevin, as having a role in regulating BBB permeability. This chapter discusses the effects of various AD risk factors on BBB permeability, with emphasis on SPARC, which is upregulated in AD brain tissue (6, 7). Since the SPARC protein enhances BBB permeability, promotes neuroinflammation, and prolongs pro-inflammatory M1 phase of microglia, its potential as a druggable target is also discussed.

THE BBB IN AD

The integrity of the BBB is critical to the maintenance of brain homeostasis in health. As mentioned, BBB dysfunction is commonly seen in cases of AD, and a variety of factors may contribute to the observed disruption.

Aging

Age is perhaps the most predominant risk factor for AD, with almost half of all individuals over the age of 85 suffering from it; conversely, less than 10% of cases under the age of 65 suffer from AD (8). BBB deterioration has been welldocumented in aging (9), with nearly all its components being affected (Table 1). Degradation of the BBB is known to start early, with notable permeability increases around the hippocampal region in individuals between 23 and 47 years of age, which worsens with increasing age (9, 10). Hormones such as insulin, which are associated with aging, can affect the permeability of the BBB and contribute to leakage (11). Additionally, transporters of certain molecules such as glucose, along with various proteins and hormones, may become defective in older individuals, reducing their availability to the brain (12). Outside of the barrier, the appearance of white matter hyperintensities, which are indicative of a loss of vascular integrity, also correlates with age. Damaged vasculature may cause a corresponding decrease in BBB integrity (13). Corroborating this, the vascular density of the brain appears to experience a significant age-related decline between the ages of 57 and 90 (14). Hypertension, which is more common in older individuals, can contribute to microvascular injury, thereby increasing the incidence of BBB disruption (15, 16).

TABLE 1	Alterations of BBB components during physiological aging			
BBB element	Property changes due to aging			
Endothelial cells	Increased capillary wall thickness			
	Decreased number of endothelial cells			
	Decreased number of mitochondria			
Tight junctions	Decreased expression of tight junction protein			
Basal lamina	Increased thickness of basement membrane			
	Increased concentration of collagen IV and arginase			
	Decreased concentration of laminin			
Astrocytes	Increased astrocyte proliferation			
	Increased GFAP expression			
Microglia	Changes to amoeboid morphology			
	Production of neurotoxic proinflammatory mediators			
Pericytes	Degeneration and loss of pericytes			
	Vesicular and lipofuscin-like inclusions			
	Increased size of mitochondria			
	Foamy transformation			
Neurons	Deterioration of synaptic plasticity			
	Deficit in long-term potentiation			
	Impaired neurogenesis			
	Increased apoptosis			
	Neuronal damage due to cytokine release			

Adapted from (9). GFAP, glial fibrillary acidic protein; BBB, blood-brain barrier.

Neuroimmunity

The brain exists in a state that is considered "immune privileged." Due to the existence of barriers between the brain and the rest of the body's circulation, the brain is insulated against many peripheral immune events (17). The presence of immune cells derived from main circulation, such as peripheral macrophages, neutrophils, and leukocytes, in the brain is an indication of BBB breakdown (18). In addition, the brain has intrinsic immune components, and these components interact with the BBB in such a way that peripheral immune events can also invoke a response in the brain (17). There are two primary types of neuroimmune cells, microglia and astrocytes, and both interact with the BBB.

Microglia are a variant of macrophage, though they do not develop and function as peripheral macrophages do. Microglial progenitors emerge from the yolk sac, and the development of microglia occurs in phases, with each phase being regulated by different transcription factors, and exhibiting differing gene expression profiles (19). Some of their key functions include phagocytosis, synapse pruning, and mediating immune signaling through the release of cytokines and other factors (20). They play a role in AD primarily by phagocytizing abnormal A β amyloid and forming a barrier between the plaques and the rest of the brain through plaque envelopment, thereby limiting the expansion of the plaque (21). Microglia are known to associate tightly with the BBB. Microglia exist in a resting state until they are activated due to brain injury or another immunological stimulus. Upon activation, they release a host of cytokines and other molecules that increase the permeability of the BBB; in the case of brain injury, this allows bloodborne agents like myeloid cells to cross the BBB. They also have been found to release reactive oxygen species that impair the function of the BBB (22). Perhaps, the most significant contribution of microglia to AD pathology is their involvement in evoking inflammatory responses within the brain (23). The role of inflammation in BBB disruption is discussed further in the following section.

Astrocytes are of epithelial origin and feature a wide array of morphologies and functions within the nervous system. Aside from their roles in neural immunity, they are responsible for ion transport, removal and catabolism of neurotransmitters, and neurogenesis. Some astrocytes are noted for their vascular end-feet, which are closely associated with brain vasculature and the BBB (24). They increase the permeability of the BBB through vascular endothelial growth factor A (VEGFA) and thymidine phosphorylase (TYMP). VEGFA, along with the TYMP product 2-deoxy-d-ribose, downregulates tight-junction proteins and promotes angiogenesis and BBB permeability (25). The release of these two factors is induced by interleukin-1 beta (IL-1 β), an inflammatory cytokine (25). Both astrocytes and microglia produce and react to inflammatory responses that can impact BBB health primarily through inflammatory cytokines (26).

Neuroinflammation

Neuroinflammation has been found to be relevant to AD pathology in a variety of ways. Inflammation in the brain has widespread effects on vasculature, cell signaling, neural function, and other immune responses. The effect of cytokines and other inflammatory mediators released during an inflammatory event involve some of the key components of the neuroimmune system and have been found to regulate the

clearance of A β (27). Many of these mediators have also been shown to influence BBB permeability. A summary of some of the mediators and their effects on the BBB are given in Table 2 (28). Neuroinflammation is a common effect of aging and notably includes an increase in the production of inflammatory cytokines by microglial cells (29). These mediators, in general, are not directly responsible for modulating the permeability of the BBB. Instead, they influence the expression or activation of other factors that, in turn, disrupt the BBB function (30).

Traumatic brain injury

Traumatic brain injury (TBI) has been shown to be a significant risk factor for AD. Individuals who had experienced a mild traumatic brain injury (mTBI) are more vulnerable to early-onset cognitive impairment (31) than those that have not experienced such an event. The effects of a TBI are often immediate, with force-induced injury resulting in what is considered secondary brain damage, which includes an increase in BBB permeability. Following TBI and mTBI, bloodborne substances accumulate in various regions of the brain, due to a breach in the BBB. While, in most cases, the effects of mTBI appear to be relatively short-lived, in rats with preexisting hypertension, mTBI can induce persistent disruption of the BBB (32). These rats experienced an increase in fibrin accumulation and neuronal expression of inflammatory cytokines (32). Generally, a focal breach following mTBI has been observed in rats to persist for

TABLE 2	Inflammatory mediators and their effect upon the BBB		
Inflammatory			
mediator	Observed effects on BBB		
TNF-α	Increase in BBB permeability in in vivo and in vitro models		
	Increased efflux of albumin from brain to blood		
	Decreased ZO-1 expression		
	Increased MMP-9 protein expression		
IL-1β	Increase in BBB permeability in in vivo and in vitro models		
	Decreased TEER of primary cultures of brain endothelial cells and human brain endothelial cells		
	Increased production of PGE and COX		
	Decreased ZO-1 expression		
IL-6	Decreased TEER in cerebrovascular endothelial cells from rats at higher doses but not at lower doses		
	Decreased BBB permeability in ischemic brain in rodents		
IL-17A	Increase in BBB permeability in in vivo and in vitro models		
CRP	Increase in BBB permeability in in vivo and in vitro models		
	Increase in ROS production in brain endothelial cells		

Adapted from (28). BBB, blood brain barrier; COX, cyclooxygenase; CRP, c-reactive protein; IL, interleukin; MMP-9, matrix metalloproteinase-9; PGE, prostaglandin E; ROS, reactive oxygen species; TEER, transepithelial/transendothelial electrical resistance; TNF-α, tumor necrosis factor alpha; ZO-1, zonula occludens-1

approximately 24–48 h (33, 34). In rare occurrences, singular instances of BBB disruption via mTBI in humans, typically measured by the cerebrospinal fluid/ serum albumin quotient, have been found to persist for months or even years (35). A meta-analysis of studies conducted from 1995 to 2012 found that TBI and mTBI events are substantial risk factors for AD (36). The BBB disruption at the onset of TBI is relatively short-lived; however, the subsequent events lead to structural degeneration in the brain causing long-lasting cognitive impairments (37). Disruption of the BBB has been observed to be a marker of mild cognitive impairment independently of the neurofibrillary tangles (NFTs), tau protein, and A β amyloid plaques, indicating that substantial breakdown of the BBB itself contributes to cognitive decline in addition to exacerbating other neurodegenerative processes in AD (38).

Apolipoprotein E gene ε4 allele (APOE ε4)

The APOE ε 4 allele has been identified as the most significant genetic risk factor for AD (39). It is also associated with other dementia subtypes, such as Parkinson's (40) and frontotemporal dementias (41, 42). Individuals homozygous for APOE4 $(\epsilon 4/\epsilon 4)$ experience a 10-fold higher risk of dementia, and individuals heterozygous $(\varepsilon_3/\varepsilon_4)$ for the variant experience a 1.7-fold higher risk of dementia (43). APOE4 is attributed to reduced clearance of $A\beta$ amyloid, which contributes to the formation of the A β amyloid plagues that are a hallmark of AD (44). Possession of at least one APOE4 allele increases the leakage of the BBB (45, 46). The role of APOE in maintaining the integrity of the BBB is confirmed by experiments involving APOE deficient (APOE-/-) mice, which exhibit signs of increased BBB permeability, such as the leakage of exogenous tracers, starting at 2 weeks old (18). The allele may also have relevance to TBI. When assessing the BBB repair ability of APOE3 and APOE4 mice, the APOE3 mice experienced a significant reduction in permeability between the 3-day and 10-day measurements, indicating substantial BBB repair. APOE4 mice, however, did not experience a significant reduction in permeability in the same time period; APOE4 was also expressed at lower levels than APOE3 at both 3 and 10 days (47). The role of APOE in BBB integrity is further reinforced by postmortem studies of both AD and normal humans with and without the APOE4 allele. AD-afflicted APOE4 carriers experienced a 3.1-fold increase in fibrin perivascular deposits in the brain relative to APOE3 carriers, indicating an increase in BBB permeability. The same study also found that APOE3 carriers still had a 6.9-fold increase in AB amyloid deposits relative to normal controls, indicating that BBB disruption is indeed a significant component of AD. In addition, pericytes, which are constituents of the BBB, have substantially reduced coverage in both AD and normal individuals (48).

SECRETED PROTEIN ACIDIC AND RICH IN CYSTEINE

SPARC belongs to a family of matricellular proteins that modulate cell interaction with the extracellular environment. There are currently six known members of the SPARC family. These members, along with some key features of them, are shown in Table 3. While their structures and functions are not identical, each member of the family possesses shared motifs and is secreted into the extracellular space where they influence the structure of the extracellular matrix and modulate various signaling pathways (58) such as the TGF- β pathway (30, 58). Two particular members of this family, SPARC and Hevin/SPARCL1, are notable in that they have collagen binding domains in addition to the calcium binding domains exhibited by all of the other members of the protein family (30). The SPARC collagen binding interaction is depicted in Figure 1. Nullification of SPARC expression decreases the expression of the proinflammatory cytokines IL-6, IP-10, and FAS/CD95 in rats (59). Hevin is a member of the SPARC (30). Studies have found that SPARC has an antiadhesive effect on brain endothelial cells and decreases cerebral endothelial transepithelial/transendothelial electrical resistance (TEER), indicating decreased BBB integrity (60).

SPARC expression in AD

Within the brain, SPARC and Hevin are attributed to a variety of functions, such as the regulation of synaptogenesis and tissue remodeling following an injury. The proteins are primarily expressed in immune cells. While both SPARC and Hevin are produced by microglia (7) and astrocytes (61), Hevin is produced only by some neurons (62). Postmortem examination of the brains of AD and control individuals found that there is a notable upregulation of SPARC and downregulation of Hevin in the AD brains. As indicated in Figure 2, SPARC is expressed by microglia found in close proximity to pathological A β amyloid plaques (7). Interestingly, while it seems that SPARC has a destructive effect on the BBB, which would exacerbate the AD condition, it appears to support the A β amyloid clearance process, which should have the opposite effect (7, 60). Hevin's role in BBB health is unclear, though it may be responsible for the initiation of the repair process by microglia (7).

TABLE 3	Members of the SPARC protein family			
SPARC family member	Significant brain expression	Binds to	Impact on cell adhesion	
SPARC	Yes (7, 30)	Collagen + calcium (30)	Antiadhesion (49, 50)	
Hevin/SPARC-like 1 (SPARCL1)	Yes (7, 30)	Collagen + calcium (30)	Antiadhesion (51)	
Smoc-1	Yes (52)	Calcium (30)	Unidentified	
Smoc-2	No (53)	Calcium (30)	No effect on non-epithelial cells (53)	
Testicans/spocks	Yes (54, 55)	Calcium (30)	Antiadhesion (testican-1) (56)	
Follastatin-like 1 (FSTL1)	Yes (57)	Calcium (30)	Unidentified	

SARC, secreted protein acidic and rich in cysteine.



Figure 1. Secreted protein acidic and rich in cysteine (SPARC)-collagen binding. The bottom part of the structure diagram represents a collagen alpha-1(III) chain; the top part of the structure diagram represents SPARC. The SPARC-collagen binding site represents a potential target for agents that modify SPARC activity. *Created with BioRender.com.*



Figure 2. Secreted protein acidic and rich in cysteine (SPARC) is expressed by microglia found in close proximity to pathological Alzheimer's $A\beta$ amyloid aggregates. Analysis of cortical tissue from Alzheimer's disease patients reveals the presence of SPARC (in green) in and around glial cells within $A\beta$ amyloid (ThioS in blue) plaques. These SPARC-associated glial cells were identified as microglia (IBA-1 in HRP brown). Adapted from (7) under CC BY-NC 4.0 (https://creativecommons.org/licenses/by-nc/4.0/) license. Reproduced with permission.

SPARC and the BBB

Expression of SPARC is also associated with inflammatory responses. When testing the effects of various cytokines on SPARC expression and BBB permeability in hCMEC/D3 cell culture, it was found that TNF- α caused an upregulation in SPARC only in the absence of IFN- γ that negated the effects of TNF- α (63).

In the brain, SPARC is typically localized to astrocytic end-feet and cerebral endothelium. SPARC was experimentally determined to increase transendothelial permeability and affect the differentiation of endothelial cells through protein tyrosine kinase signaling (63). A particular area of interest is the SPARC-collagen binding domain. Increased levels of collagen IV, as well as general thickening, in the basement membrane of brain microvessels are noted in cases of AD (64); increases in SPARC show a corresponding increase in the levels of collagen (65, 66), and SPARC acts as a chaperone for collagen IV (67). Furthermore, abnormalities in the vasculature that surround the A β amyloid plaques are associated with aberrant levels of collagen IV (68). The interactions between SPARC and collagen have been linked to inflammation and pathological fibrosis (69), as well as induction of a pro-inflammatory response in brain monocytes (64). The differential effects of SPARC may mean it can serve as an effective broad-spectrum therapeutic target.

Rationale for SPARC protein as a potential Alzheimer's therapeutic target

Although a better understanding of the disease and its mechanisms have provided avenues for druggable targets, attempts to develop effective ways to treat or reverse AD progression have been met with failure thus far. Perhaps targeting AD from multiple treatment angles may be the key. As vascular dysfunction is a substantial component of AD, SPARC and other members of the protein family may be druggable targets for AD (70). Information on SPARC modifiers on the central nervous system is limited; however, such modifiers have been studied to a degree in the context of cancers (71, 72). Currently, the translational aspects of these drugs for AD are largely speculative. The SPARC-collagen binding site represents a reasonable start to the search, given the detailed research surrounding the structure and mechanism of the SPARC-collagen binding domain and knowledge of collagen binding with other molecules (73). A general diagram of how SPARC and Hevin, in particular, interact with the BBB is shown in Figure 3.

FUTURE DIRECTIONS

Research on the SPARC protein and molecules that can modify its activity is limited. Given that the process of drug discovery is capital-intensive and time-consuming, it may be prudent to establish what molecules modify the activity of SPARC and its relative such as Hevin. High-throughput screening has been applied to other molecules, in which batteries of mini-scale experiments assaying the activity of the target when introduced to a library of molecules are conducted (74). However, conducting this procedure on a compound library that can contain hundreds of thousands of molecules, with a generally low hit rate of modifying compounds, is costly; samples of every compound must first be synthesized before it can be tested. The use of screening tools that conduct the filtering of molecules *in silico* has become increasingly popular, as large numbers of molecules can be processed quickly and cheaply, so long as one has access to sufficient computational power (75). Molecules must be first tested both *in vitro* and *in vivo* and, ultimately, in humans before a drug

144 Pilozzi A et al.



Figure 3. Secreted protein acidic and rich in cysteine (SPARC) and Hevin interactions with the blood-brain barrier (BBB). SPARC is produced primarily by microglia and astrocytes, while Hevin is primarily produced by astrocytes. SPARC and Hevin exert antiadhesive effects, and SPARC promotes the release of inflammatory cytokines in certain conditions. Both SPARC and Hevin are distinct from the other members of the SPARC protein family because they bind to collagen as well as calcium; both binding sites represent potential targets for Alzheimer's disease-modifying agents. *Created with BioRender.com*.

can be considered successful. The fast *in silico* screening compound hits can significantly reduce the time and costs of drug development (76).

We thus propose that such methods be used to identify inhibitors for SPARCcollagen binding. A variety of machine learning (ML) algorithms have been applied to the problem of drug discovery and molecular screening with considerable success; decision trees, support vector machines, and other classifiers have been applied to either structural or ligand-based virtual screening (VS) (77–79). Ligand-based approaches take the similarities of different molecules to other compounds that are known to be active against a target protein. Information on the molecules is generally taken from compound databases, which are filtered based on certain properties, such as those that influence pharmacokinetics and toxicity, in order to make the problem of screening a large number of molecules more computationally feasible (76). As the name implies, structure-based VS involves structural information, either obtained from techniques such as X-ray crystallography or, more commonly, data obtained from computational models. The molecular structures of the protein target and those of the structural databases are examined, in order to determine which will interact in the desired manner (76). Structural methods also encompass the development of novel molecules, as generative models, which are notable for their ability to use information gleaned from training data for the purpose of classification or prediction and create novel data for a novel sample of the given type.

Deep learning (DL), which involves ML algorithms that feature multiple neural network (NN) layers, has become prevalent in a variety of fields. NNs form the primary basic structure of DL models. Notable examples of NN-based drug discovery platforms include "AtomNet," developed by Wallach et al., a structure-based virtual screener based on a convolutional neural network (CNN) algorithm. Though typically applied to image processing and linguistic applications, a CNN model, which features layers of feature-reduction (convolution) and pooling operations, was trained on a set of molecular structures and tested against a set of benchmark decoy-structures. Performance of the CNN was found to be better than other ML methods (80). Thus, it may be the case that more research into some unorthodox NN strategies may provide a helpful performance boost for VS tools; such a tool could be useful in VS against SPARC protein target for AD drug discovery, as the limited data on SPARC target make *in silico* methods a practical predecessor to future *in vitro* and *in vivo* work and beyond.

CONCLUSION

Current evidence shows that the BBB plays a crucial role in a variety of neurological disorders, and its disruption is evident in AD. There is a great deal of interplay between the various known hallmarks of AD, such as the buildup of amyloid plaques, NFTs, and BBB degradation. Many risk factors tie into multiple facets of the disorder; APOE4, the most significant genetic risk factor for AD discovered to date, diminishes A β clearance and inhibits BBB repair. TBI/mTBI and neuroinflammation contribute to AD pathogenesis and BBB damage. Gradual erosion of the BBB is a common part of the aging process, increasing an individual's vulnerabilities to further breakdown and neurodegenerative diseases. Given the importance of the health and stability of the BBB, and the wide array of factors that can be detrimental to it, such as SPARC, more research into its mechanics, maintenance, and recovery pathways may be vital to understanding AD and how to treat it. DL-based VS tools may be employed to identify inhibitors of SPARC-collagen binding for AD drug discovery.

Acknowledgments: This work was partially supported by a grant from NIH National Institute on Aging (R01AG056614 to XH).

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship, or publication of this chapter.

Copyright and Permission Statement: We confirm that the materials included in this chapter do not violate copyright laws. Where relevant, appropriate permissions have been obtained from the original copyright holder(s). All original sources have been appropriately acknowledged and referenced.

REFERENCES

- 1. Alzheimer's-Association. 2019 Alzheimer's disease facts and figures. Alzheimers Dement. 2019;15(3):321–87. http://dx.doi.org/10.1016/j.jalz.2019.01.010
- El-Hayek YH, Wiley RE, Khoury CP, Daya RP, Ballard C, Evans AR, et al. Tip of the iceberg: Assessing the global socioeconomic costs of Alzheimer's disease and related dementias and strategic implications for stakeholders. J Alzheimers Dis. 2019;70(2):323–41. http://dx.doi.org/10.3233/JAD-190426
- Mehta D, Jackson R, Paul G, Shi J, Sabbagh M. Why do trials for Alzheimer's disease drugs keep failing? A discontinued drug perspective for 2010–2015. Expert Opin Investig Drugs. 2017;26(6):735–9. http://dx.doi.org/10.1080/13543784.2017.1323868
- Liebner S, Dijkhuizen RM, Reiss Y, Plate KH, Agalliu D, Constantin G. Functional morphology of the blood-brain barrier in health and disease. Acta Neuropathol. 2018;135(3):311–36. http://dx.doi. org/10.1007/s00401-018-1815-1
- Yamazaki Y, Kanekiyo T. Blood-brain barrier dysfunction and the pathogenesis of Alzheimer's disease. Int J Mol Sci. 2017;18(9):1965. http://dx.doi.org/10.3390/ijms18091965
- Kong W, Mou X, Liu Q, Chen Z, Vanderburg CR, Rogers JT, et al. Independent component analysis of Alzheimer's DNA microarray gene expression data. Mol Neurodegen. 2009;4(1):5. http://dx.doi. org/10.1186/1750-1326-4-5
- Strunz M, Jarrell JT, Cohen DS, Rosin ER, Vanderburg CR, Huang X. Modulation of SPARC/hevin proteins in Alzheimer's disease brain injury. J Alzheimers Dis. 2019;68(2):695–710. http://dx.doi.org/ 10.3233/JAD-181032
- Amemori T, Jendelova P, Ruzicka J, Urdzikova LM, Sykova E. Alzheimer's disease: Mechanism and approach to cell therapy. Int J Mol Sci. 2015;16(11):26417–51. http://dx.doi.org/10.3390/ ijms161125961
- Erdo F, Denes L, de Lange E. Age-associated physiological and pathological changes at the bloodbrain barrier: A review. J Cereb Blood Flow Metab. 2017;37(1):4–24. http://dx.doi.org/10.1177 /0271678X16679420
- Montagne A, Barnes SR, Sweeney MD, Halliday MR, Sagare AP, Zhao Z, et al. Blood-brain barrier breakdown in the aging human hippocampus. Neuron. 2015;85(2):296–302. http://dx.doi.org/ 10.1016/j.neuron.2014.12.032
- 11. Erickson MA, Banks WA. Age-associated changes in the immune system and blood(-)brain barrier functions. Int J Mol Sci. 2019;20(7):1632. http://dx.doi.org/10.3390/ijms20071632
- 12. Mooradian AD. Potential mechanisms of the age-related changes in the blood-brain barrier. Neurobiol Aging. 1994;15(6):751–5; discussion 61–2, 67. http://dx.doi.org/10.1016/0197-4580(94)90058-2
- 13. Young VG, Halliday GM, Kril JJ. Neuropathologic correlates of white matter hyperintensities. Neurology. 2008;71(11):804–11. http://dx.doi.org/10.1212/01.wnl.0000319691.50117.54
- 14. Brown WR, Moody DM, Thore CR, Challa VR, Anstrom JA. Vascular dementia in leukoaraiosis may be a consequence of capillary loss not only in the lesions, but in normal-appearing white matter and cortex as well. J Neurol Sci. 2007;257(1–2):62–6. http://dx.doi.org/10.1016/j.jns.2007.01.015
- Oparil S, Acelajado MC, Bakris GL, Berlowitz DR, Cifkova R, Dominiczak AF, et al. Hypertension. Nat Rev Dis Primers. 2018;4:18014. http://dx.doi.org/10.1038/nrdp.2018.14
- Zeevi N, Pachter J, McCullough LD, Wolfson L, Kuchel GA. The blood-brain barrier: Geriatric relevance of a critical brain-body interface. J Am Geriatr Soc. 2010;58(9):1749–57. http://dx.doi. org/10.1111/j.1532-5415.2010.03011.x
- Louveau A, Harris TH, Kipnis J. Revisiting the mechanisms of CNS immune privilege. Trends Immunol. 2015;36(10):569–77. http://dx.doi.org/10.1016/j.it.2015.08.006
- Montagne A, Zhao Z, Zlokovic BV. Alzheimer's disease: A matter of blood-brain barrier dysfunction? J Exp Med. 2017;214(11):3151–69. http://dx.doi.org/10.1084/jem.20171406
- Matcovitch-Natan O, Winter DR, Giladi A, Aguilar SV, Spinrad A, Sarrazin S, et al. Microglia development follows a stepwise program to regulate brain homeostasis. Science. 2016;353(6301):aad8670. http:// dx.doi.org/10.1126/science.aad8670
- Lenz KM, Nelson LH. Microglia and beyond: Innate immune cells as regulators of brain development and behavioral function. Front Immunol. 2018;9:698. http://dx.doi.org/10.3389/fimmu.2018.00698

- Condello C, Yuan P, Schain A, Grutzendler J. Microglia constitute a barrier that prevents neurotoxic protofibrillar Abeta42 hotspots around plaques. Nat Commun. 2015;6:6176. http://dx.doi. org/10.1038/ncomms7176
- da Fonseca AC, Matias D, Garcia C, Amaral R, Geraldo LH, Freitas C, et al. The impact of microglial activation on blood-brain barrier in brain diseases. Front Cell Neurosci. 2014;8:362. http://dx.doi. org/10.3389/fncel.2014.00362
- Spangenberg EE, Green KN. Inflammation in Alzheimer's disease: Lessons learned from microgliadepletion models. Brain Behav Immun. 2017;61:1–11. http://dx.doi.org/10.1016/j.bbi.2016.07.003
- Verkhratsky A, Nedergaard M. Physiology of astroglia. Physiol Rev. 2018;98(1):239–389. http:// dx.doi.org/10.1152/physrev.00042.2016
- Chapouly C, Tadesse Argaw A, Horng S, Castro K, Zhang J, Asp L, et al. Astrocytic TYMP and VEGFA drive blood-brain barrier opening in inflammatory central nervous system lesions. Brain. 2015;138(Pt 6):1548–67. http://dx.doi.org/10.1093/brain/awv077
- Sawikr Y, Yarla NS, Peluso I, Kamal MA, Aliev G, Bishayee A. Neuroinflammation in Alzheimer's disease: The preventive and therapeutic potential of polyphenolic nutraceuticals. Adv Protein Chem Struct Biol. 2017;108:33–57. http://dx.doi.org/10.1016/bs.apcsb.2017.02.001
- Heneka MT, Carson MJ, El Khoury J, Landreth GE, Brosseron F, Feinstein DL, et al. Neuroinflammation in Alzheimer's disease. Lancet Neurol. 2015;14(4):388–405. http://dx.doi.org/10.1016/S1474-4422(15)70016-5
- Hurtado-Alvarado G, Dominguez-Salazar E, Pavon L, Velazquez-Moctezuma J, Gomez-Gonzalez B. Blood-brain barrier disruption induced by chronic sleep loss: Low-grade inflammation may be the link. J Immunol Res. 2016;2016:4576012. http://dx.doi.org/10.1155/2016/4576012
- 29. Mattson MP, Arumugam TV. Hallmarks of brain aging: Adaptive and pathological modification by metabolic states. Cell Metab. 2018;27(6):1176–99. http://dx.doi.org/10.1016/j.cmet.2018.05.011
- Bradshaw AD. Diverse biological functions of the SPARC family of proteins. Int J Biochem Cell Biol. 2012;44(3):480–8. http://dx.doi.org/10.1016/j.biocel.2011.12.021
- Gardner RC, Yaffe K. Epidemiology of mild traumatic brain injury and neurodegenerative disease. Mol Cell Neurosci. 2015;66(Pt B):75–80. http://dx.doi.org/10.1016/j.mcn.2015.03.001
- Szarka N, Toth L, Czigler A, Kellermayer Z, Ungvari Z, Amrein K, et al. Single mild traumatic brain injury induces persistent disruption of the blood-brain barrier, neuroinflammation and cognitive decline in hypertensive rats. Int J Mol Sci. 2019;20(13):3223. http://dx.doi.org/10.3390/ ijms20133223
- Fehily B, Fitzgerald M. Repeated mild traumatic brain injury: Potential mechanisms of damage. Cell Transplant. 2017;26(7):1131–55. http://dx.doi.org/10.1177/0963689717714092
- Laurer HL, Bareyre FM, Lee VM, Trojanowski JQ, Longhi L, Hoover R, et al. Mild head injury increasing the brain's vulnerability to a second concussive impact. J Neurosurg. 2001;95(5):859–70. http://dx.doi.org/10.3171/jns.2001.95.5.0859
- Salehi A, Zhang JH, Obenaus A. Response of the cerebral vasculature following traumatic brain injury. J Cereb Blood Flow Metab. 2017;37(7):2320–39. http://dx.doi.org/10.1177/0271678X17701460
- Perry DC, Sturm VE, Peterson MJ, Pieper CF, Bullock T, Boeve BF, et al. Association of traumatic brain injury with subsequent neurological and psychiatric disease: A meta-analysis. J Neurosurg. 2016;124(2):511–26. http://dx.doi.org/10.3171/2015.2.JNS14503
- Tremblay S, Desjardins M, Bermudez P, Iturria-Medina Y, Evans AC, Jolicoeur P, et al. Mild traumatic brain injury: The effect of age at trauma onset on brain structure integrity. Neuroimage Clin. 2019;23:101907. http://dx.doi.org/10.1016/j.nicl.2019.101907
- Nation DA, Sweeney MD, Montagne A, Sagare AP, D'Orazio LM, Pachicano M, et al. Blood-brain barrier breakdown is an early biomarker of human cognitive dysfunction. Nat Med. 2019;25(2): 270–6. http://dx.doi.org/10.1038/s41591-018-0297-y
- Apostolova LG, Risacher SL, Duran T, Stage EC, Goukasian N, West JD, et al. Associations of the top 20 Alzheimer disease risk variants with brain amyloidosis. JAMA Neurol. 2018;75(3):328–41. http:// dx.doi.org/10.1001/jamaneurol.2017.4198
- Li J, Luo J, Liu L, Fu H, Tang L. The genetic association between apolipoprotein E gene polymorphism and Parkinson disease: A meta-analysis of 47 studies. Medicine (Baltimore). 2018;97(43):e12884. http://dx.doi.org/10.1097/MD.00000000012884

148 Pilozzi A et al.

- 41. Agosta F, Vossel KA, Miller BL, Migliaccio R, Bonasera SJ, Filippi M, et al. Apolipoprotein E epsilon4 is associated with disease-specific effects on brain atrophy in Alzheimer's disease and frontotemporal dementia. Proc Natl Acad Sci U S A. 2009;106(6):2018–22. http://dx.doi.org/10.1073/pnas.0812697106
- 42. Ji Y, Liu M, Huo YR, Liu S, Shi Z, Liu S, et al. Apolipoprotein E ε4 frequency is increased among Chinese patients with frontotemporal dementia and Alzheimer's disease. Dement Geriatr Cogn Disord. 2013;36(3–4):163–70. http://dx.doi.org/10.1159/000350872
- Slooter AJ, Cruts M, Kalmijn S, Hofman A, Breteler MM, Van Broeckhoven C, et al. Risk estimates of dementia by apolipoprotein E genotypes from a population-based incidence study: The Rotterdam Study. Arch Neurol. 1998;55(7):964–8. http://dx.doi.org/10.1001/archneur.55.7.964
- 44. Liu CC, Liu CC, Kanekiyo T, Xu H, Bu G. Apolipoprotein E and Alzheimer disease: Risk, mechanisms and therapy. Nat Rev Neurol. 2013;9(2):106–18. http://dx.doi.org/10.1038/nrneurol.2012.263
- Salloway S, Gur T, Berzin T, Tavares R, Zipser B, Correia S, et al. Effect of APOE genotype on microvascular basement membrane in Alzheimer's disease. J Neurol Sci. 2002;203–204:183–7. http:// dx.doi.org/10.1016/S0022-510X(02)00288-5
- Zipser BD, Johanson CE, Gonzalez L, Berzin TM, Tavares R, Hulette CM, et al. Microvascular injury and blood-brain barrier leakage in Alzheimer's disease. Neurobiol Aging. 2007;28(7):977–86. http:// dx.doi.org/10.1016/j.neurobiolaging.2006.05.016
- Main BS, Villapol S, Sloley SS, Barton DJ, Parsadanian M, Agbaegbu C, et al. Apolipoprotein E4 impairs spontaneous blood brain barrier repair following traumatic brain injury. Mol Neurodegener. 2018;13(1):17. http://dx.doi.org/10.1186/s13024-018-0249-5
- Halliday MR, Rege SV, Ma Q, Zhao Z, Miller CA, Winkler EA, et al. Accelerated pericyte degeneration and blood-brain barrier breakdown in apolipoprotein E4 carriers with Alzheimer's disease. J Cereb Blood Flow Metab. 2016;36(1):216–27. http://dx.doi.org/10.1038/jcbfm.2015.44
- Mayer U, Aumailley M, Mann K, Timpl R, Engel J. Calcium-dependent binding of basement membrane protein BM-40 (osteonectin, SPARC) to basement membrane collagen type IV. Eur J Biochem. 1991;198(1):141–50. http://dx.doi.org/10.1111/j.1432-1033.1991.tb15996.x
- Rosenblatt S, Bassuk J, Alpers C, Sage E, Timpl R, Preissner K. Differential modulation of cell adhesion by interaction between adhesive and counter-adhesive proteins: Characterization of the binding of vitronectin to osteonectin (BM40, SPARC). Biochem J. 1997;324(Pt 1):311. http://dx.doi.org/10.1042/ bj3240311
- Girard J-P, Springer TA. Modulation of endothelial cell adhesion by hevin, an acidic protein associated with high endothelial venules. J Biol Chem. 1996;271(8):4511–17. http://dx.doi.org/10.1074/jbc.271.8.4511
- Gersdorff N, Müller M, Schall A, Miosge N. Secreted modular calcium-binding protein-1 localization during mouse embryogenesis. Histochem Cell Biol. 2006;126(6):705–12. http://dx.doi.org/10.1007/ s00418-006-0200-7
- Maier S, Paulsson M, Hartmann U. The widely expressed extracellular matrix protein SMOC-2 promotes keratinocyte attachment and migration. Exp Cell Res. 2008;314(13):2477–87. http:// dx.doi.org/10.1016/j.yexcr.2008.05.020
- Vannahme C, Schübel S, Herud M, Gösling S, Hülsmann H, Paulsson M, et al. Molecular cloning of testican-2: Defining a novel calcium-binding proteoglycan family expressed in brain. J Neurochem. 1999;73(1):12–20. http://dx.doi.org/10.1046/j.1471-4159.1999.0730012.x
- Charbonnier F, Chanoine C, Cifuentes-Diaz C, Gallien C, Rieger F, Alliel P, et al. Expression of the proteoglycan SPOCK during mouse embryo development. Mech Dev. 2000;90(2):317–21. http:// dx.doi.org/10.1016/S0925-4773(99)00255-5
- Marr HS, Edgell C-JS. Testican-1 inhibits attachment of neuro-2a cells. Matrix Biol. 2003;22(3): 259–66. http://dx.doi.org/10.1016/S0945-053X(03)00036-2
- 57. Liu R, Yang Y, Shen J, Chen H, Zhang Q, Ba R, et al. Fstl1 is involved in the regulation of radial glial scaffold development. Mol Brain. 2015;8(1):1–14. http://dx.doi.org/10.1186/s13041-015-0144-8
- Atorrasagasti C, Peixoto E, Aquino JB, Kippes N, Malvicini M, Alaniz L, et al. Lack of the matricellular protein SPARC (secreted protein, acidic and rich in cysteine) attenuates liver fibrogenesis in mice. PLoS One. 2013;8(2):e54962. http://dx.doi.org/10.1371/journal.pone.0054962
- 59. Mazzolini G, Atorrasagasti C, Onorato A, Peixoto E, Schlattjan M, Sowa J-P, et al. SPARC expression is associated with hepatic injury in rodents and humans with non-alcoholic fatty liver disease. Sci Rep. 2018;8(1):725. http://dx.doi.org/10.1038/s41598-017-18981-9

- Tilling T, Korte D, Hoheisel D, Galla HJ. Basement membrane proteins influence brain capillary endothelial barrier function in vitro. J Neurochem. 1998;71(3):1151–7. http://dx.doi. org/10.1046/j.1471-4159.1998.71031151.x
- Kucukdereli H, Allen NJ, Lee AT, Feng A, Ozlu MI, Conatser LM, et al. Control of excitatory CNS synaptogenesis by astrocyte-secreted proteins Hevin and SPARC. Proc Natl Acad Sci U S A. 2011;108(32):E440–9. http://dx.doi.org/10.1073/pnas.1104977108
- 62. Mongrédien R, Erdozain AM, Dumas S, Cutando L, Del Moral AN, Puighermanal E, et al. Cartography of hevin-expressing cells in the adult brain reveals prominent expression in astrocytes and parvalbumin neurons. Brain Structure and Function. 2019;224(3):1219–44. http://dx.doi. org/10.1007/s00429-019-01831-x
- Alkabie S, Basivireddy J, Zhou L, Roskams J, Rieckmann P, Quandt JA. SPARC expression by cerebral microvascular endothelial cells in vitro and its influence on blood-brain barrier properties. J Neuroinflammation. 2016;13(1):225. http://dx.doi.org/10.1186/s12974-016-0657-9
- 64. Kalaria R. The blood-brain barrier and cerebrovascular pathology in Alzheimer's disease. Brain Res. 1999;814:164–70. http://dx.doi.org/10.1111/j.1749-6632.1999.tb07821.x
- Zhou X, Tan FK, Guo X, Arnett FC. Attenuation of collagen production with small interfering RNA of SPARC in cultured fibroblasts from the skin of patients with scleroderma. Arthr Rheum. 2006;54(8):2626–31. http://dx.doi.org/10.1002/art.21973
- McCurdy S, Baicu CF, Heymans S, Bradshaw AD. Cardiac extracellular matrix remodeling: Fibrillar collagens and Secreted Protein Acidic and Rich in Cysteine (SPARC). J Mol Cell Cardiol. 2010;48(3):544–9. http://dx.doi.org/10.1016/j.yjmcc.2009.06.018
- Bradshaw AD. The role of SPARC in extracellular matrix assembly. J Cell Commun Signal. 2009; 3(3–4):239. http://dx.doi.org/10.1007/s12079-009-0062-6
- 68. Kawai M, Kalaria R, Harik S, Perry G. The relationship of amyloid plaques to cerebral capillaries in Alzheimer's disease. Am J Pathol. 1990;137(6):1435.
- 69. Wang J-C, Lai S, Guo X, Zhang X, De Crombrugghe B, Sonnylal S, et al. Attenuation of fibrosis in vitro and in vivo with SPARC siRNA. Arthr Res Ther. 2010;12(2):R60. http://dx.doi.org/10.1186/ar2973
- Strickland S. Blood will out: Vascular contributions to Alzheimer's disease. J Clin Investig. 2018;128(2):556–63. http://dx.doi.org/10.1172/JCI97509
- Tai IT, Tang MJ. SPARC in cancer biology: Its role in cancer progression and potential for therapy. Drug Resist Updat. 2008;11(6):231–46. http://dx.doi.org/10.1016/j.drup.2008.08.005
- 72. Vaz J, Ansari D, Sasor A, Andersson R. SPARC: A potential prognostic and therapeutic target in pancreatic cancer. Pancreas. 2015;44(7):1024–35. http://dx.doi.org/10.1097/MPA.00000000000409
- Eichenbaum H, Dudchenko P, Wood E, Shapiro M, Tanila H. The hippocampus, memory, and place cells: Is it spatial memory or a memory space? Neuron. 1999;23(2):209–26. http://dx.doi. org/10.1016/S0896-6273(00)80773-4
- 74. Carnero A. High throughput screening in drug discovery. Clin Transl Oncol. 2006;8(7):482–90. http://dx.doi.org/10.1007/s12094-006-0048-2
- 75. Carpenter K, Huang X. Is it a prime time for AI-powered virtual drug screening? EC Pharmacol Toxicol. 2017;SI(1):16–17.
- Sliwoski G, Kothiwale S, Meiler J, Lowe EW, Jr. Computational methods in drug discovery. Pharmacol Rev. 2013;66(1):334–95. http://dx.doi.org/10.1124/pr.112.007336
- 77. Lavecchia A. Machine-learning approaches in drug discovery: Methods and applications. Drug Discov Today. 2015;20(3):318–31. http://dx.doi.org/10.1016/j.drudis.2014.10.012
- Carpenter KA, Cohen DS, Jarrell JT, Huang X. Deep learning and virtual drug screening. Fut Med Chem. 2018;10(21):2557–67. http://dx.doi.org/10.4155/fmc-2018-0314
- Carpenter KA, Huang X. Machine learning-based virtual screening and its applications to Alzheimer's drug discovery: A review. Curr Pharmaceut Design. 2018;24(28):3347–58. http://dx.doi.org/10.2174/ 1381612824666180607124038
- 80. Wallach I, Dzamba M, Heifets A. AtomNet: A deep convolutional neural network for bioactivity prediction in structure-based drug discovery. arXiv preprint arXiv:151002855. 2015.

9

The Therapeutic Potential of Epigenetic Modifications in Alzheimer's Disease

Enric Bufill¹ • Roser Ribosa-Nogué² • Rafael Blesa²

¹Neurology Department, Vic University Hospital, Vic, Spain; ²Memory Unit, Neurology Department, Hospital de la Santa Creu i Sant Pau, Biomedical Research Institute Sant Pau, Universitat Autònoma de Barcelona, Barcelona, Spain

Author for correspondence: Enric Bufill, Neurology Department, Vic University Hospital, Vic, Spain. Email: ebufill@telefonica.net

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch9

Abstract: Alzheimer's disease is characterized by the formation and deposit of abnormal peptides such as amyloid plaques and neurofibrillary tangles in the brain. Therapeutic strategies aimed at preventing the formation of such deposits have not been successful. Currently, there are no effective treatments for the disease. Since numerous epigenetic changes have been detected in Alzheimer's disease, treatments aimed at reversing these changes by intervening in DNA methylation, histone acetylation, and microRNA expression may constitute promising lines of research in the future. This chapter provides an overview of the epigenetic changes and the potential epigenetic therapies in Alzheimer's disease.

Keywords: Alzheimer's disease; DNA methyltransferase; epigenetic changes; histone acetylation; noncoding RNA

In: *Alzheimer's Disease: Drug Discovery.* Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

Copyright: The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

INTRODUCTION

Alzheimer's disease (AD) is the most common cause of dementia. It usually occurs in people over 60 years of age and presents with progressive loss of memory and cognitive capacity, language disorders, inability to translate ideas into actions (ideomotor apraxia), impaired planning and judgment, apathy, depression, and, in later stages, psychosis with paranoid delusions. AD is characterized by the presence of abnormal peptide deposits in the brain. The most characteristic lesions are neuritic extracellular plaques of the amyloid β (A β) peptide, which consists of 33–40 amino acids derived from the proteolysis of the transmembrane protein amyloid precursor protein or APP. These neuritic plaques contain a large number of distorted neuronal expansions, known as dystrophic neurites. Activated microglial cells are observed at their center.

Some evidence suggests that amyloid deposits may be neurotoxic and may cause neuronal dysfunction and even neuronal death. In the normal brain, APP is fragmented into functional segments by the α -, β -, and γ -secretase enzymes. Occasionally, there is an increase in β - and γ -secretase relative to α -secretase, leading to the accumulation of peptides with 40 and 42 amino acids, known as amyloid β 40 ($A\beta_{40}$) and amyloid β_{42} ($A\beta_{42}$). The $A\beta_{42}$ peptide appears to have greater neurotoxic properties. $A\beta$ oligomers, small aggregates of 2-12 peptides, appear to be especially toxic (1). Diffuse plaques, another kind of plaque, lack a dense center of amyloid and dystrophic neurites. Unlike neuritic plaques, they are not associated with either neuronal destruction or cognitive dysfunction (2).

Neurofibrillary tangles are twisted aggregates of abnormal intraneuronal fibers that have a helical structure, typically paired helical filaments, made up of hyperphosphorylated tau protein. The tau protein is involved in stabilizing microtubules, maintaining the integrity of the cytoskeleton and axoplasmic transport. Neurofibrillary tangles are found in the areas of association of the neocortex, hippocampus, limbic system, substantia nigra, raphe nuclei, locus coeruleus, and the nucleus basalis of Meynert (3). In AD, there is also a significant synaptic loss in certain areas of the neocortex and in the hippocampus, as well as the disappearance of dendritic spines.

AD occurs frequently in humans over 65 years of age. In those aged over 85, the prevalence of AD ranges between 20 and 40% in developed countries (3). In 2010, the prevalence of AD in China among people aged between 85 and 89 was 18.54% (4). In 2006, the number of patients with AD was 26.6 million world-wide. In the United States, the prevalence of AD in people over 70 years of age is 9.51%, and the incidence is 14.26 per 1000 person-years (4). However, AD is not an inevitable consequence of old age. A relatively high number of elderly people show neither cognitive decline nor lesions typical of AD with age. The causes of AD are still not well understood. In a small percentage of cases, AD can be attributed to mutations in genes located on chromosomes 1, 14, and 21. These cases are usually of early onset and are transmitted in an autosomal dominant manner. Most AD cases appear to be caused by the interaction of multiple genetic and environmental factors that are not yet well understood (1).

Areas of association, phylogenetically more recent areas of the human brain, have simpler organization and greater immaturity in the adult than phylogenetically older primary areas. Thus, in the neurons belonging to the areas of association, myelination occurs very slowly and many neurons belonging to these areas remain incompletely myelinated—that is, immature even in adulthood. Poorly myelinated neurons are chronically subjected to high-energy turnover, which makes them more vulnerable to the influence of oxidative stress. There are, therefore, extensive cortical areas in the human brain that remain structurally immature throughout life (5).

Various studies show that there has been an increase in the expression of genes related to aerobic metabolism and, more importantly, to synaptic plasticity and activity in the human cerebral cortex relative to nonhuman primate brains (6, 7). Learning and memory take place through the formation of new synapses and remodeling of preexisting synapses, suggesting that the increase in the expression of genes related to these functions has occurred in humans, as well as the selection of genes that encode proteins capable of increasing neuroplasticity. The apolipoproteins E (ApoEs) are proteins of 299 amino acids synthesized in the astrocytes of the central nervous system. They influence the transport and reuptake of cholesterol and the stabilization of the neuronal cytoskeleton, contributing to the preservation of synaptic integrity (3).

Humans present a polymorphism for ApoE with three alleles: $\epsilon 2$, $\epsilon 3$, and $\epsilon 4$. Possession of allele $\epsilon 4$ of the ApoE is the most important risk factor for the development of AD, after advanced age (3, 6). The most common allele is $\epsilon 3$, whose frequency is 60% or higher in all the populations studied. Possession of the $\epsilon 4$ allele is associated with lower neuroplasticity and lower synaptic repair capacity, and seems to promote the relatively early appearance of brain deposits of neurotoxins, such as A β and neurofibrillary tangles, whose excess is associated to AD.

There has been an increase in the expression of genes associated with neuronal plasticity in the human cerebral cortex, resulting in an increased capacity for learning and memory, neurotransmission, axonal transport, aerobic metabolism, and neuroprotection, all of which are adaptations that promote high neuronal activity over a long life (6, 7). The human brain has a high need for glucose, especially during its development. A child's brain consumes more than 40% of the body's basal energy requirements. Most of the glucose is oxidized to produce ATP. This process is upregulated in anaerobic conditions. Aerobic glycolysis is increased during childhood and is synonymous with high rates of synaptic formation and the growth and remodeling of synapses. Aerobic glycolysis is associated with the persistence of genetic expression associated with childhood, especially genes active in youth, and especially those related to the growth and formation of new synapses (transcriptional neoteny) (8). In the adult human brain, aerobic glycolysis is especially elevated in cortical areas related to cognitive functions that have undergone significant modifications during the evolution of the human species, such as the dorsolateral prefrontal cortex and the brain's default mode network (BDMN), related to the coordination of activity between different cortical areas and to planning and autobiographical memory capacities, which allow "mental travel in time," remembering and planning.

The brain regions where most of the A β deposits are located almost exactly match the regions that make up the BDMN, which suggests that the high synaptic turnover that occurs in these areas predisposes the formation of abnormal peptide deposits characteristic of AD (8). Multiple studies show that AD appears to be associated with oxidative stress (9). Increased aerobic metabolism in neurons that retain juvenile characteristics in adulthood could subject these neurons to high

oxidative stress. It appears that oxidative stress could induce epigenetic changes, reducing the expression of certain genes, including those related to synaptic plasticity.

EPIGENETIC CHANGES AND AD

Epigenetic changes modulate the expression of certain genes without altering the DNA sequence. Epigenetic factors include DNA methylation, histone modification, and the regulation and modification of chromatin by noncoding RNA (ncRNA) (10). DNA methylation modifies cytosine residues by adding methyl groups in regions rich in cytosine-guanine. DNA methyltransferases, such as DNA methyltransferase 1, DNA methyltransferase 2, DNA methyltransferase 3, and DNA methyltransferase 3,6, are involved in the process.

Some cytosines, for example those located in the promotor region of the APP gene, have been found to exhibit methylation with age, which can lead to the formation of A β deposits. Methylation of the gene coding for the microtubule-associated protein tau (MAPT) can lead to the suppression of MAPT, which can end up affecting the level of the tau protein. Further, methylation in the promotor region of the brain-derived neurotrophic factor (BDNF) gene seems to play a significant role in the appearance of mild cognitive impairment (11).

Methylation of certain loci of specific genes, such as sortilin-related receptor 1 (*SORL1*), ATP binding cassette subfamily A member 7 (*ABCA7*), HLA class II histocompatibility antigen DRB5 beta chain (*HLADRB5*), solute carrier family 24 member 4 (*SLC24A4*), and box-dependent-interacting protein 1 (*BIN1*), has also been associated with AD (11). The protein encoded by *SORL1* controls the production of A β , so the methylation of the DNA that codifies this protein could lead to increased levels of A β .

Reelin is an extracellular matrix glycoprotein that, together with ApoE, shares the LRP and VLDLR/ApoER2 membrane receptors. During embryonic development, this protein regulates neuronal migration and, in the adult brain, intervenes in synaptic plasticity, interacting with ApoE. Binding of this protein to the membrane receptors activates a series of proteins that constitute the signaling pathway of reelin, inducing changes in the neuronal cytoskeleton. In transgenic mice that have lesions similar to those of AD, reelin counteracts early-phase synaptic dysfunction induced by the A β peptide (12).

In vitro studies have shown that oxidative stress alters the activation of proteins that are part of the reelin signaling pathway, resulting in the hyperphosphorylation of tau, which precedes the formation of neurofibrillary tangles in AD (13). Depletion of brain reelin has been detected in patients with AD prior to the formation of A β deposits (14). Thus, there seems to be a relationship between dysfunction of the reelin signaling pathway and AD.

Some reelin genotypes have been found to be associated with mild cognitive impairment and AD. The reelin single nucleotide polymorphism 2299356 (RELN-rs2299356) guanine–guanine genotype is associated with cognitive decline, while the adenine–adenine genotype triples the risk of developing AD. The reelin single nucleotide polymorphism 528528 (RELN-rs528528) cytosine–cytosine genotype, on the other hand, reduces the probability of mild cognitive

impairment by two thirds. These variations are located in the promoter region of the gene, which seems to play a regulatory role in its expression (15).

Reelin is involved in neuroplasticity, a process that has increased during the evolution of the human brain. Oxidative stress and probably other factors seem to induce epigenetic changes capable of reducing the expression of genes involved in synaptic plasticity. In some cases, such as that of the carriers of certain reelin geno-types, dysfunction of the proteins involved in the reelin signaling pathway caused by a reduction in reelin related to epigenetic changes could increase the probability of having the abnormal peptide deposits that characterize AD (15). It cannot be ruled out that certain alleles are more vulnerable than others to oxidative stress, toxins, inflammation, and other factors possibly related to AD.

Histones are proteins that serve as structural support for the DNA of the cell nucleus. Nuclear DNA associates with histones to form nucleosomes. The distribution and compaction of nucleosomes determines the structure of the chromatin and the accessibility of DNA to factors involved in the transcriptional machinery. Histones are also susceptible to epigenetic changes that can cause an increase or decrease in genetic expression. Nucleosomes are mainly regulated by posttranslational modifications that occur in the N-terminal region of histones.

Both methylation and acetylation can occur in histones through the antagonistic action of histone acetyltransferases (HATs), histone deacetylases (HDACs), histone methyltransferases, and histone demethylases. The acetylation of histones results in increased genetic activity by reducing the compactness of the nucleosomes and thus facilitating access of the transcriptional machinery to DNA. During senescence, mammalian cell cultures develop highly condensed regions of chromatin that may be associated with transcriptional decline (16).

Various HDAC inhibitors, like valproic acid and sodium butyrate, seem to improve memory in animal models and some neurodegenerative diseases like Parkinson's and even AD (16). Among the epigenetic changes described are the alteration of expression of the ncRNA. ncRNA is involved in genetic silencing as well as other functions, including the regulation of the activity of retrotransposons, genes that are capable of moving from one location in the genome to another. Short fragments of ncRNA, such as microRNA (miRNA), are involved in transcriptional gene regulation. ncRNA is primarily expressed in the brain, where it is involved in neuronal development, control of regions of the genome, which are involved in neuronal migration, homeostasis, and plasticity (17).

Epigenetics has improved our understanding of the evolution of the human brain, synaptic plasticity and neuronal diversity. Several studies have identified DNA methylation, changes in histones and chromatin, and changes in ncRNA expression in various neurological diseases, including AD. A large proportion of the genes that compose our genome are expressed in the central nervous system, where a substantial amount of miRNA is also synthesized. Several factors that have been associated with AD, such as diabetes mellitus, high blood pressure, obesity, diet, excessive sedentary lifestyle, smoking, and even a low educational level, are capable of inducing epigenetic changes (18).

There is currently no effective treatment for AD. However, cholinesterase inhibitors, such as donepezil, rivastigmine, and galantamine, together with *N*-methyl-Daspartate (NMDA) receptor antagonists, such as memantine, produce moderate and transient symptomatic benefits in the early stages of the disease. Various treatments targeting the supposed causes of the disease are being developed, all still in the experimental phase. One such treatment, active immunotherapy with A β fragments, which has been effective in transgenic mice (19), has not only been clinically ineffective in human patients but has also caused encephalitis in some cases (20). Passive immunotherapy with antibodies to A β has shown some benefits in transgenic mice and is being tested in humans. These clinical trials have shown that the clearance of A β in humans does not appear to produce significant cognitive improvements, which has led to some researchers questioning the role that A β plays in the cognitive decline associated with AD (20).

Attempts are also being made to develop drugs that prevent hyperphosphorylation or aggregation of the tau protein, although less effort has been made to this end than in inhibiting the formation of $A\beta$ deposits. Most researchers support the hypothesis that amyloid plaques and the neurofibrillary tangles are neurotoxic. The amyloid cascade hypothesis has led to the development of treatments that promote $A\beta$ clearance or prevent the formation of plaques. Such treatment has thus far been ineffective. A relatively high number of elderly people develop $A\beta$ deposits without presenting with cognitive decline, which calls into question the amyloid cascade hypothesis.

Recent studies show that the A β peptide has antimicrobial properties, and that the absence of this peptide leads to an increased vulnerability to infection. Although the immune system has limited access to the central nervous system, it could fight invading pathogens with antimicrobial peptides like A β . The abnormal accumulation of A β observed in AD could be caused by persistent subacute infection or by noninfectious factors, such as trauma, ischemia, toxins, and anesthetics (21). Some researchers defend the hypothesis that A β acts as an antioxidant in response to the oxidative stress that takes place in regions of the brain subjected to high synaptic turnover, like that which occurs in the phenomenon of neuronal neoteny, where certain neurons retain a high synaptic plasticity in adulthood.

The generation of the A β peptide may have an adaptive function in its initial phases, and the same could be assumed about the hyperphosphorylation of tau. This would explain why drugs that reduce A β production have not been effective so far. Attempting to reverse the epigenetic changes that occur in AD could perhaps be of therapeutic value in the future.

EPIGENETIC THERAPIES IN AD

As previously discussed, the treatments currently approved for AD are acetylcholinesterase inhibitors (donepezil, rivastigmine, and galantamine) and the glutamate NMDA receptor antagonist memantine, drugs indicated for the specific treatment of memory disorders. Acetylcholinesterase inhibitors increase the levels of the neurotransmitter acetylcholine, which is decreased in brains with AD, and NMDA receptor antagonists prevent aberrant stimulation (22). These drugs achieve a discrete and transient improvement in cognitive and functional capacities, but do not delay the progression of the disease. Nevertheless, observational studies suggest that the combination of these treatments prolongs the time until patients need to be admitted to a residence (23). As a result, there is a great deal of interest in researching new treatments for the disease. The main line of research in AD is that of anti-amyloid therapies (24). Despite the serious complications associated with active immunotherapy and the repeated failures of passive immunotherapy, novel anti-amyloid antibodies such as aducanumab and BAN2401 have brought fresh hope in this line of research, since they have been shown to be capable of reducing amyloid load in preliminary clinical trials (25). Other treatments within the amyloid cascade hypothesis have been developed, which promote A β clearance or prevent plaque formation. However, not only have none of these treatments within this line of research been shown to be effective, but some of them have led to clinical worsening (26, 27).

Another line of research is that of anti-tau therapies, drugs that prevent the hyperphosphorylation or aggregation of the tau protein, or antibodies that reduce the levels of the protein in the cerebrospinal fluid. Lastly, other avenues of research that are currently unsuccessful or under investigation are anti-APOE4 drugs, anti-oxidants, anti-inflammatory drugs, cardiovascular drugs, mitochondrial protectors, hormone therapy, and antiviral drugs (28–30).

Due to the difficulty in finding effective drugs for AD, it is crucial that other possible therapeutic avenues are explored, such as that of epigenetic drugs. This line of research is based on the fact that epigenetic changes take place during neurodevelopment and aging, and that epigenetic alterations are common in various neurodevelopmental and neurodegenerative diseases.

In the case of AD, more than 20 epigenetic mechanisms have been identified, most of which involve direct DNA modifications (as in the case of methylation), modifications in chromatin structure (as in the case of histone modifications), or modification of mRNA-related processes, including ncRNA and miRNA.

With regard to changes in methylation in AD, a recent study has established reference maps of the genome-wide distribution of the three possible states of DNA methylation (5mC, 5hmC, and 5fC/caC) in this disease (31). The results of this study, based mainly on cortical neurons obtained from induced pluripotent stem cells, suggest that the changes detected in DNA may precede the appearance of the disease, rather than appear later as a consequence of its progression. These results could mean these markers could be very useful in reaching early molecular diagnosis and therapy.

In addition, it has been detected in AD and frontotemporal dementia that the levels of an important transcriptional repressor—repressor element 1-silencing transcription factor (REST)—do not increase adequately with age (32). Consequently, transcriptional changes occur, and decreases in the expression of neuroprotector genes are found, including forkhead box protein class O (FOXO), which contributes to resistance to oxidative stress. In contrast, increased expression levels of genes that promote AD pathology, such as presenilin 2, are found. Taken together, these changes would increase neuronal fragility in these diseases. Furthermore, in animal models, such as the K-p25 AD mouse model, an increase in the expression of genes associated with the immune response has been detected, along with decreases in the expression of genes involved in synaptic functions and learning (33).

Several changes in the histone acetylation process, which is heavily involved in the consolidation of memory, have been detected in AD. For example, the levels of histone H4 with acetylation at the 16th lysine residue protein (H4K16ac), a histone marker located in enhancers and promoters generally associated with active gene expression, are duplicated in the cerebral cortex in healthy aging but are barely detectable in the cerebral cortex of people with AD (34). Levels of histone deacetylase 2 (HDAC2), which increase in cultured cells after neurotoxic insults, are also found to be increased in the hippocampus and prefrontal cortex of AD mouse models and in the hippocampus of people with AD (35). Increases in deacetylase lead to worsening of synaptic function. It should be noted that blocking HDAC2 increases synaptic density and alleviates the loss of memory, but does not improve neuronal survival. This means that deficits in AD are caused not only by neuronal loss but also by epigenetic blocking of the functions of neuronal survivors (36). In addition to the reduction in expression of genes important for neuronal function, an increase in aberrant expression of genes that are normally silenced or expressed at low levels has also been observed in AD (37, 38).

Furthermore, the expression of miRNA in the brain is altered in AD. For example, reduced levels of miRNA-29a/b-1 and miRNA-132, and increased levels of miRNA-34c have been detected. The decrease in miRNA-29a/b-1, which is a betasite amyloid precursor protein cleaving enzyme 1 (BACE1) inhibitor, correlates with an increase in the production of A β (39). A decrease in miRNA-132, which targets the tau protein, HAT-associated protein 300 (EP300), sirtuin deacetylase 1, and FOXO1a, would jeopardize neuronal growth, the integration of newborn neurons, synaptic structure, and plasticity (40). Dysregulation of miRNA expression has also been detected in biofluids, suggesting that these molecules could be used as both biomarkers and therapeutic targets (39, 40).

Finally, an acceleration of epigenetic age has also been observed in AD, especially in the prefontal cortex. Epigenetic age is estimated from DNA methylation levels in 353 CpG sites, and its acceleration with respect to chronological age is associated with, in addition to AD, higher mortality, cognitive impairment, and other neurodegenerative diseases (41, 42). Epigenetic age could also explain differences in the onset age of AD in members of the same family that share the same gene mutation (43). That is, those who have an accelerated epigenetic age would develop AD symptoms at an earlier age.

The goal of epigenetic therapies is to reverse at least some of the epigenetic changes caused by AD. Such therapies have several advantages. First, specific drugs can be designed because the epigenetic changes are induced by enzymes that act at the DNA or histone level. Second, they act on reversible mechanisms since the epigenetic changes at the DNA and histone level are both regulated by enzymes. Finally, these therapies enable us to unite physiology and pathology, because epigenetics influence gene expression throughout life, and thus epigenetic drugs would be effective in both neurodevelopmental and neurodegenerative diseases. In addition, epigenetic therapies can target any component of the epigenetic machinery.

In the last decade, several epigenetic drugs have been designed for the treatment of neurological diseases. The most promising are DNA-demethylating agents and HDAC inhibitors (HDACis). In fact, there are already drugs of these two therapeutic groups that have been approved by the US Food and Drug Administration for the treatment of hematological cancer. In the former group, there is 5-azactydine and the 5-aza-2'-deoxycytidine (or decitabine), and in the latter group, suberoylanilide hydroxamic acid (SAHA or vorinostat), romidepsin, belinostat, panobinostat, and chidamide have been approved.

HDAC aims to regulate imbalances in protein acetylation levels and transcription. Their use in neurodegenerative diseases is based on their neuroprotective, neurotrophic, and anti-inflammatory properties. In the case of AD, HDACs play an important role in memory consolidation and could be useful as therapeutic targets. For example, HDAC2 and HDAC3 have been shown to play a repressive role in memory formation, while HDAC5 has a memory-enhancing effect (44). As a result, HDAC2-inhibiting drugs, such as CI-994, and HDAC5-enhancing drugs could be useful in AD. The therapeutic potential of HDACis has been demonstrated in studies with animals. In APP/PS1 mice, acute treatment with HADCi trichostatin A (TSA), sodium valproic acid, SAHA, sodium butyrate (NaB), butyrate, vorinostat, 4-phenylbutiric acid, MS-275, and crebinostat improved the cognitive performance of these animals (45–49).

Sulforaphane could also be useful in AD. Sulforaphane is an HDACi that decreased HDAC2 levels in the triple-transgenic mouse model of AD (3 × Tg-AD). This was accompanied by an increase in the acetylation of histones H3 and H4 in the BDNF promoter and resulting in an increase in its expression (50).

There are HDACis that affect multiple genes involved in AD, which could be advantageous given the multifactorial etiology of AD. The disadvantage of these compounds is that their wide spectrum theoretically broadens the possibilities of adverse effects with their use. This group includes M344 {4-(dimethylamino)-*n*-[7-(hydroxyamino)-7-oxoheptyl]benzamide}, CM-414 {3-[[4-ethoxy-3-(1-methyl-7-oxo-3-propyl-6*H*-pyrazolo[4,3-d]pyrimidin-5-yl)phenyl]methyl]-*N*-hydroxycyclobutane-1-carboxamide}, and RGFP-966 {(E)-*N*-(2-amino-4-fluorophenyl)-3-[1-[(E)-3-phenylprop-2-enyl]pyrazol-4-yl]prop-2-enamide}. Their chronic use in animal models showed cognitive benefits (51–57).

In addition to the HDACis described, there are other HDACis that have been specifically designed. In this way, HDACi W2 was obtained, which features a longer half-life and better penetration of the blood–brain barrier than the HDACis currently available. HDACi W2 has been shown to be capable of significantly reducing A β levels in hAPP 3×Tg AD mice by reducing the expression of genes involved in A β production and increasing the expression of A β degradation enzymes. This HDACi is also capable of decreasing phosphorylation of the tau protein, promoting the formation and growth of dendritic spines, and improving learning and memory in these mice, which makes it a potential candidate for the treatment of AD.

Due to the observed benefits of using HDACis in animal models of AD, there are several ongoing clinical studies with these compounds. One of the compounds under study is valproate, which is a class I HDAC inhibitor. It has already been approved for treating epilepsy, migraine, and bipolar disorder, and since its activity as a HDACi was discovered, it is also being studied to evaluate its effectiveness in neurodegenerative disease. In preclinical studies, it was shown to be effective in reversing cognitive impairment in a mouse model of AD (58, 59).

Another drug under study is vitamin B3 or nicotamide, which is also a class III NAD-dependent sirtuin HDAC inhibitor. This compound can delay aging in mouse oocytes and delay cognitive impairment in a mouse model of AD (60). A phase I clinical trial in patients with AD demonstrated its safety. It is currently in a phase II clinical trial. Another HDACi, vorinostat, is being evaluated in a phase I clinical trial in patients with AD. Finally, RDN-929, which is a CoREST-selective HDAC inhibitor that could reactivate neuronal gene expression, strengthen synaptic function, and promote new synapses, has been studied in two phase I clinical trials as a possible treatment for AD.

Another mechanism for enhancing acetylation is the use of drugs that enhance HATs. The increased expression of an enzyme of this type, Tip60, in drosophila overexpressing human APP was able to restore the benefits of environmental enrichment (61).

With regard to miRNA, utility of the drug gemfibrozil has been studied. Gemfibrozil is capable of modifying miR-107 levels, the reduction of which may accelerate the progression of AD by regulating the expression of BACE1. A phase I clinical trial showed the drug was safe and reduced miR-107 in plasma and in CSF to undetectable levels.

Other possible epigenetic therapeutic targets not yet explored in AD are drugs directed against HATs, ten-eleven-translocation methylcitosine-dioxygenases enzymes (which catalyze the conversion of 5-methylcytosine to 5-hydroxymethylcytosine), DNA demethylation, chromatine remodelers, and other histone modifications.

Finally, it should be noted that there are also nonspecific epigenetic therapies that can be useful in AD. The first is blood plasma therapy from young subjects. In a recent study, in which aged mice were treated with blood plasma from young mice, it was observed that the treatment halved the epigenetic ages of blood, heart, and liver tissue, and also rejuvenated the hypothalamus. The treatment also improved the functioning of these organs as well as cognitive functions (62). The second nonspecific epigenetic therapy is cognitive stimulation, which has been shown to be capable of causing epigenetic changes (63). The third nonspecific epigenetic therapy is physical exercise. In animals, physical exercise is capable of reversing age-related reduction of adult neurogenesis and cognitive function in the aged hippocampus (64). In addition, a recent study showed that the administration of circulating blood factors in the plasma of aged mice subjected to exercise was capable of transferring the beneficial effects to sedentary-aged mice. These investigations led to the discovery that glycosylphosphatidylinositol (GPI)-specific phospholipase D1 (Gpld1), a GPIdegrading enzyme derived from liver, was probably responsible for these effects.

Although epigenetic therapies have a promising future role in AD, there are still problems that must first be addressed. First, we need to bear in mind that perhaps not all epigenetic changes can be reversed with these types of therapy. Second, epigenetic changes are extremely complex, and the therapies could have any number of side effects that are difficult to control. Third, different regions of the same gene can have antagonistic epigenetic changes, so the effect of an epigenetic therapy could be unpredictable. With respect to pharmacological properties, current therapies lack specificity and are not selective for specific brain regions, cell types, or genes. This limitation could be addressed with the use of siRNA or the use of chromatin-modifying enzymes, transcription activation-like effectors or clustered regularly interspaced short palindromic repeats/ Cas System, and the use of artificial transcriptional factors of the silencing or promoter type.

The development of new study techniques will also be essential to better understand how epigenetic therapies work and to be able to design future drugs. First, laboratory techniques such as cell-type specific analysis of transcription and DNA methylation in the brain, single cell analysis of DNA–protein interactions, and chromatin 3D structure might be applied in future studies to uncover neuronal cell-type specific chromatin structure and interneuronal variations. Furthermore, new models for studying the effect of these therapies, such as neuronal cultures or other brain cell types derived from human stem cells or induced pluripotent stem cells, could be better than the animal models used so far. The application of innovative imaging techniques, such as positron emission tomography using radiochemical [¹¹C] martinostat, which binds specifically to certain HDAC isoforms, could help to discover the gene expression patterns regulated by chromatinmodifying enzymes in the live brain. Finally, establishing epigenetic biomarkers for diagnosis, prognosis, and therapy would be important to test the efficacy of epigenetic drugs and to classify patients according to the particular therapy they would most benefit from.

CONCLUSION

At present, there are no treatments for AD. Given that several epigenetic changes occur in this disease, treatments aimed at reversing these changes may constitute promising lines of research in the future.

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship, and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- Ballard C, Gauthier S, Corbett A, Brayne C, Aarscland D, Jones E. Alzheimer's disease. Lancet. 2011 Mar 19;377(9770):109–31. http://dx.doi.org/10.1016/S0140-6736(10)61349-9
- 2. Weiner MF, Lipton AM, editors. Enfermedad de Alzheimer. Manual de enfermedad de Alzheimer y otras demencias. Madrid: Editorial Médica Panamericana; 2009. p. 155–72.
- 3. Mesulam MM. Principles of behavioral and cognitive neurology. 2nd ed. New York: Oxford University Press; 2000.
- Niu H, Alvarez Alvarez I, Guillen Grima F, Arguinaga Ontosa I. Prevalence and incidence of Alzheimer's disease in Europe: A meta-analysis. Neurologia. 2017;32(8):525–532. Spanish. http:// dx.doi.org/10.1016/j.nrleng.2016.02.009
- 5. Braak H, Rüb U, Schultz C, del Tredici K. Vulnerability of cortical neurons to Alzheimer's and Parkinson diseases. J Alzheimers Dis. 2006;9:35–44. http://dx.doi.org/10.3233/JAD-2006-9S305
- Cáceres M, Lachuer J, Zapala M, Redmond J, Kado L, Geshwind D, et al. Elevated gene expression levels distinguish human from non-human primate brains. Proc Natl Acad Sci U S A. 2003;100:13030–5. http://dx.doi.org/10.1073/pnas.2135499100
- Uddin M, Wildman DE, Liu G, Xu W, Johnson RM, Patrick RH, et al. Sister grouping of chimpanzees and humans as revealed by genome-wide phylogenetic analysis of brain gene expression profiles. Proc Natl Acad Sci U S A. 2004;101(9):2957–62. http://dx.doi.org/10.1073/pnas.0308725100
- 8. Bufill E, Agusti J, Blesa R. Human neoteny revisited: The case of synaptic plasticity. Am J Hum Biol. 2011;23:729–39. http://dx.doi.org/10.1002/ajhb.21225

162 Bufill E et al.

- 9. Zawia NH, Lahiri DK, Cardozo-Pelaez E. Epigenetics, oxidative stress and Alzheimer disease. Free Radic Biol Med. 2009;46:1241–9. http://dx.doi.org/10.1016/j.freeradbiomed.2009.02.006
- Liu X, Jiao B, Shen L. The epigenetics of Alzheimer disease: Factors and therapeutic implications. Front Genet. 2018 Nov 30;9:579. http://dx.doi.org/10.3389/fgene.2018.00579
- Traynor EJ, Renton AE. Exploring the epigenetics of Alzheimer's disease. JAMA Neurol. 2015;72(1):8–9. http://dx.doi.org/10.1001/jamaneurol.2014.3057
- Rogers JT, Rusiana I, Trotter J, Zhao L, Donaldson E, Pak DTS, et al. Reelin supplementation enhances cognitive ability, synaptic plasticity and dendritic spine density. Learn Mem. 2011 Sep;18(9):558–64. http://dx.doi.org/10.1101/lm.2153511
- Patrick GN, Zukerberg L, Nikolic M, de la Monte S, Dikkes P, Tsai LH. Conversion of p35 to p25, deregulates Cdk5 and promotes neurodegeneration. Nature. 1999 Dec;9(402):615–22. http://dx.doi. org/10.1038/45159
- Stranahan AM, Haberman RP, Gallagher M. Cognitive decline is associated with reduced reelin expression in the entorhinal cortex of aged rats. Cereb Cortex. 2011 Feb;21(2):392–400. http://dx.doi. org/10.1093/cercor/bhq106
- Bufill E, Roura-Poch P, Sala-Matavera I, Anton S, Lleó A, Sanchez Saudinós B, et al. Reelin signaling pathway genotypes and Alzheimer disease in a Spanish population. Alzheimer Dis Assoc Disord. 2015 April–June;29(2):169–72.
- Sanchez-Mut JV, Gräff J. Epigenetic alterations in Alzheimer's disease. Front Behav Neurosci. 2015 Dec 17;9:347. http://dx.doi.org/10.3389/fnbeh.2015.00347
- 17. Qureshi IA, Mark F, Mehler MD. Understanding neurological disease mechanisms in the era of epigenetics. JAMA Neurol. 2013;70(6):703–10. http://dx.doi.org/10.1001/jamaneurol.2013.1443
- Sen P, Shah P, Nativio R, Berger SL. Epigenetic mechanisms for longevity and aging. Cell. 2016 Aug 11;166:822–39. http://dx.doi.org/10.1016/j.cell.2016.07.050
- Schenk D, Barbour R, Dunn W, Gordon G, Grajeda H, Guido T, et al. Immunization with amyloid-β attenuates Alzheimer disease like pathology in the PDAPP mouse. Nature. 1999;400:173–7. http:// dx.doi.org/10.1038/22124
- Arai H. [Immunotherapy and Alzheimer's disease]. Nihon Shinkei Seishin Yakurigaku Zasshi. 2010 Aug;30(4):173–6. Japanese.
- Soscia SJ, Kirby JE, Washicosky KJ, Tucker SM, Ingelsson M, Hyman B, et al. The Alzheimer's disease-associated amyloid beta-protein is an antimicrobial peptide. PLoS One. 2010 Mar 3;5(3):e9505. http://dx.doi.org/10.1371/journal.pone.0009505
- Cummings JL. Alzheimer's disease. N Engl J Med. 2004;351(1):56–67. http://dx.doi.org/10.1056/ NEJMra040223
- Lopez OL, Becker JT, Wahed AS, Saxton J, Sweet RA, Wolk DA, et al. Long-term effects of the concomitant use of memantine with cholinesterase inhibition in Alzheimer disease. J Neurol Neurosurg Psychiatry. 2009 Jun;80(6):600–7. http://dx.doi.org/10.1136/jnnp.2008.158964
- 24. Cummings J, Lee G, Ritter A, Sabbagh M, Zhong K. Alzheimer's disease drug development pipeline: 2020. Alzheimer's Dement (NY). 2020 Jul 16;6(1):e12050. http://dx.doi.org/10.1002/trc2.12050
- Logovinsky V, Satlin A, Lai R, Swanson C, Kaplow J, Osswald G, et al. Safety and tolerability of BAN2401—A clinical study in Alzheimer's disease with a protofibril selective Aβ antibody. Alzheimers Res Ther. 2016 Apr 6;8(1):14. http://dx.doi.org/10.1186/s13195-016-0181-2
- Egan MF, Kost J, Voss T, Mukai Y, Aisen PS, Cummings JL, et al. Randomized trial of verubecestat for prodromal Alzheimer's disease. N Engl J Med. 2019;380:1408–20. http://dx.doi.org/10.1056/ NEJMoa1812840
- 27. Novak G, Streffer JR, Timmers M, Henley D, Brashear HR, Bogert J, et al. Long-term safety and tolerability of atabecestat (JNJ-54861911), an oral BACE1 inhibitor, in early Alzheimer's disease spectrum patients: A randomized, double-blind, placebo-controlled study and a two-period extension study. Alzheimers Res Ther. 2020 May;12(1):58. http://dx.doi.org/10.1186/s13195-020-00614-5
- Lee HP, Zhu X, Casadesus G, Castellani RJ, Nunomura A, Smith MA, et al. Antioxidant approaches for the treatment of Alzheimer's disease. Expert Rev Neurother. 2010 Jul;10(7):1201–8. http://dx.doi. org/10.1586/ern.10.74
- Cole GM, Frautschy SA. Mechanisms of action of non-steroidal anti-inflammatory drugs for the prevention of Alzheimer's disease. CNS Neurol Disord Drug Targets. 2010 Apr;9(2):140–8. http://dx.doi. org/10.2174/187152710791011991

- Bezprozvanny I. The rise and fall of Dimebon. Drug News Perspect. 2010 Oct;23(8):518–23. http:// dx.doi.org/10.1358/dnp.2010.23.8.1500435
- Fetahu IS, Ma D, Rabidou K, Argueta C, Smith M, Liu H, et al. Epigenetic signatures of methylated DNA cytosine in Alzheimer's disease. Sci Adv. 2019 Aug 28;5(8):eaaw2880. http://dx.doi. org/10.1126/sciadv.aaw2880
- 32. Lu T, Aron L, Zullo J, Pan Y, Kim H, Chen Y, et al. REST and stress resistance in ageing and Alzheimer's disease. Nature. 2014 Mar 27;507(7493):448–54. http://dx.doi.org/10.1038/nature13163
- Gjoneska E, Pfenning AR, Mathys H, Quon G, Kundaje A, Tsai L-H, et al. Conserved epigenomic signals in mice and humans reveal immune basis of Alzheimer's disease. Nature. 2015 Feb;518(7539):365–9. http://dx.doi.org/10.1038/nature14252
- Nativio R, Donahue G, Berson A, Lan Y, Amlie-Wolf A, Tuzer F, et al. Dysregulation of the epigenetic landscape of normal aging in Alzheimer's disease. Nat Neurosci. 2018;21(4):497–505. http://dx.doi. org/10.1038/s41593-018-0101-9
- Gräff J, Rei D, Guan J-S, Wang W-Y, Seo J, Hennig KM, et al. An epigenetic blockade of cognitive functions in the neurodegenerating brain. Nature. 2012 Feb 29;483(7388):222–6. http://dx.doi. org/10.1038/nature10849
- 36. Liu D, Tang H, Li XY, Deng MF, Wei N, Wang X, et al. Targeting the HDAC2/HNF-4A/miR-101b/ AMPK pathway rescues tauopathy and dendritic abnormalities in Alzheimer's disease. Mol Ther. 2017 Mar 1;25(3):752–64. http://dx.doi.org/10.1016/j.ymthe.2017.01.018
- 37. Frost B, Hemberg M, Lewis J, Feany MB. Tau promotes neurodegeneration through global chromatin relaxation. Nat Neurosci. 2014 Mar;17(3):357–66. http://dx.doi.org/10.1038/nn.3639
- Sugeno N, Jäckel S, Voigt A, Wassouf Z, Schulze-Hentrich J, Kahle PJ. α-Synuclein enhances histone H3 lysine-9 dimethylation and H3K9me2-dependent transcriptional responses. Sci Rep. 2016 Nov 3;6:36328. http://dx.doi.org/10.1038/srep36328
- Hébert SS, Horré K, Nicolaï L, Papadopoulou AS, Mandemakers W, Silahtaroglu AN, et al. Loss of microRNA cluster miR-29a/b-1 in sporadic Alzheimer's disease correlates with increased BACE1/ beta-secretase expression. Proc Natl Acad Sci U S A. 2008 Apr 29;105(17):6415–20. http://dx.doi. org/10.1073/pnas.0710263105
- Grasso M, Piscopo P, Confaloni A, Denti MA. Circulating miRNAs as biomarkers for neurodegenerative disorders. Molecules. 2014 May 23;19(5):6891–910. http://dx.doi.org/10.3390/molecules19056891
- Denk J, Boelmans K, Siegismund C, Lassner D, Arlt S, Jahn H. MicroRNA profiling of CSF reveals potential biomarkers to detect Alzheimer's disease. PLoS One. 2015 May;10(5):e0126423. http:// dx.doi.org/10.1371/journal.pone.0126423
- Marioni RE, Shah S, McRae AF, Chen BH, Colicino E, Harris SE, et al. DNA methylation age of blood predicts all-cause mortality in later life. Genome Biol. 2015 Jan 30;16(1):25. http://dx.doi. org/10.1186/s13059-015-0584-6
- Marioni RE, Shah S, McRae AF, Ritchie SJ, Muniz-Terrera G, Harris SE, et al. The epigenetic clock is correlated with physical and cognitive fitness in the Lothian Birth Cohort 1936. Int J Epidemiol. 2015 Aug;44(4):1388–96. http://dx.doi.org/10.1093/ije/dyu277
- 44. Horvath S, Erhart W, Brosch M, Ammerpohl O, von Schönfels W, Ahrens M, et al. Obesity accelerates epigenetic aging of human liver. Proc Natl Acad Sci U S A. 2014 Oct 28;111(43):15538–43. http:// dx.doi.org/10.1073/pnas.1412759111
- 45. Yang SS, Zhang R, Wang G, Zhang YF. The development prospection of HDAC inhibitors as a potential therapeutic direction in Alzheimer's disease. Transl Neurodegener. 2017 Jul 10;6:19. http://dx.doi. org/10.1186/s40035-017-0089-1
- 46. Göttlicher M, Minucci S, Zhu P, Krämer OH, Schimpf A, Giavara S, et al. Valproic acid defines a novel class of HDAC inhibitors inducing differentiation of transformed cells. EMBO J. 2001 Dec 17;20(24):6969–78. http://dx.doi.org/10.1093/emboj/20.24.6969
- Morris MJ, Mahgoub M, Na ES, Pranav H, Monteggia LM. Loss of histone deacetylase 2 improves working memory and accelerates extinction learning. J Neurosci. 2013 Apr 10;33(15):6401–11. http://dx.doi.org/10.1523/JNEUROSCI.1001-12.2013
- McQuown SC, Barrett RM, Matheos DP, Post RJ, Rogge GA, Alenghat T, et al. HDAC3 is a critical negative regulator of long-term memory formation. J Neurosci. 2011 Jan 12;31(2):764–74. http://dx.doi. org/10.1523/JNEUROSCI.5052-10.2011

- Rogge GA, Singh H, Dang R, Wood MA. HDAC3 is a negative regulator of cocaine-context-associated memory formation. J Neurosci. 2013 Apr 10;33(15):6623–32. http://dx.doi.org/10.1523/ JNEUROSCI.4472-12.2013
- Kim J, Lee S, Choi BR, Yang H, Hwang Y, Park JH, et al. Sulforaphane epigenetically enhances neuronal BDNF expression and TrkB signaling pathways. Mol Nutr Food Res. 2017 Feb;61(2). http://dx.doi.org/10.1002/mnfr.201600194
- Francis YI, Fà M, Ashraf H, Zhang H, Staniszewski A, Latchman DS, et al. Dysregulation of histone acetylation in the APP/PS1 mouse model of Alzheimer's disease. J Alzheimer's Dis. 2009;18:131–9. http://dx.doi.org/10.3233/JAD-2009-1134
- Kilgore M, Miller CA, Fass DM, Hennig KM, Haggarty SJ, Sweatt JD, et al. Inhibitors of class 1 histone deacetylases reverse contextual memory deficits in a mouse model of Alzheimer's disease. Neuropsychopharmacology. 2010 Mar;35(4):870–80. http://dx.doi.org/10.1038/npp.2009.197
- Fischer A, Sananbenesi F, Wang X, Dobbin M, Tsai L-H. Recovery of learning and memory is associated with chromatin remodelling. Nature. 2007;447(7141):178–82. http://dx.doi.org/10.1038/ nature05772
- Rustenhoven J, Smith AM, Smyth LC, Jansson D, Scotter EL, Swanson MEV, et al. PU.1 regulates Alzheimer's disease-associated genes in primary human microglia. Mol Neurodegener. 2018 Aug 20;13(1):44. http://dx.doi.org/10.1186/s13024-018-0277-1
- Volmar CH, Salah-Uddin H, Janczura KJ, Halley P, Lambert G, Wodrich A, et al. M344 promotes nonamyloidogenic amyloid precursor protein processing while normalizing Alzheimer's disease genes and improving memory. Proc Natl Acad Sci U S A. 2017 Oct 24;114(43):E9135–44. http://dx.doi. org/10.1073/pnas.1707544114
- 56. Cuadrado-Tejedor M, Garcia-Barroso C, Sánchez-Arias JA, Rabal O, Pérez-González M, Mederos S, et al. A first-in-class small-molecule that acts as a dual inhibitor of HDAC and PDE5 and that rescues hippocampal synaptic impairment in Alzheimer's disease mice. Neuropsychopharmacology. 2017 Jan;42(2):524–39. http://dx.doi.org/10.1038/npp.2016.163
- 57. Janczura KJ, Volmar C-H, Sartor GC, Rao SJ, Ricciardi NR, Lambert G, et al. Inhibition of HDAC3 reverses Alzheimer's disease-related pathologies in vitro and in the 3xTg-AD mouse model. Proc Natl Acad Sci U S A. 2018 Nov 20;115(47):E11148–57. http://dx.doi.org/10.1073/pnas.1805436115
- Qing H, He G, Ly PT, Fox CJ, Staufenbiel M, Cai F, et al. Valproic acid inhibits Abeta production, neuritic plaque formation, and behavioral deficits in Alzheimer's disease mouse models. J Exp Med. 2008 Nov 24;205(12):2781–9. http://dx.doi.org/10.1084/jem.20081588
- Yao Z, Liang L, Liu Y, Zhang L, Zhu H, Huang L, et al. Valproate improves memory deficits in an Alzheimer's disease mouse model: Investigation of possible mechanisms of action. Cell Mol Neurobiol. 2014;34(6):805–12. http://dx.doi.org/10.1007/s10571-013-0012-y
- Green KN, Steffan JS, Martinez-Coria H, Sun X, Schreiber SS, Thompson LM, et al. Nicotinamide restores cognition in Alzheimer's disease transgenic mice via a mechanism involving sirtuin inhibition and selective reduction of Thr231-phosphotau. J Neurosci. 2008 Nov 5;28(45):11500–10. http:// dx.doi.org/10.1523/JNEUROSCI.3203-08.2008
- Xu S, Panikker P, Iqbal S, Elefant F. Tip60 HAT action mediates environmental enrichment induced cognitive restoration. PLoS One. 2016 Jul 25;11(7):e0159623. http://dx.doi.org/10.1371/journal. pone.0159623
- Villeda S, Plambeck K, Middeldorp J, Castellano JM, Mosher KI, Luo J, et al. Young blood reverses agerelated impairments in cognitive function and synaptic plasticity in mice. Nat Med. 2014;20:659–63. http://dx.doi.org/10.1038/nm.3569
- 63. Pulkett R, Lubin FD. Epigenetic mechanisms in experience-driven memory formation and behavior. Epigenomics. 2011Oct;3(5):649–64. http://dx.doi.org/10.2217/epi.11.86
- Horowitz AM, Fan X, Bieri G, Smith LK, Sánchez-Díaz CI, Schroer AB, et al. Blood factors transfer beneficial effects of exercise on neurogenesis and cognition to the aged brain. Science. 2020 Jul 10;369(6500):167–73. http://dx.doi.org/10.1126/science.aaw2622

Chimeric Conjugates for Alzheimer's Disease

Illham Dhala • Tabassum Khan • Arati Prabhu

SVKM's Dr. Bhanuben Nanavati College of Pharmacy, Mumbai, India

Author for correspondence: Arati Prabhu, Gate No. 1, Mithibai College Campus, V.M. Road, Navpada, Suvarna Nagar, Vile Parle West, Mumbai, Maharashtra 400056, India. Email: arati.prabhu@bncp.ac.in

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch10

Abstract: Alzheimer's disease is a complex, progressive, neurodegenerative disorder with a multifactorial etiology. More than one mechanism appears to be involved in its pathogenesis. Current treatment targeting only a single mechanism provides only symptomatic relief and is unable to stop the progression of the disease. There is a substantial unmet medical need to develop more efficacious drugs that can address all the causative factors that lead to the development and progression of Alzheimer's disease. One of the strategies which has emerged is the development of chimeric conjugate compounds, in which multiple bioactive components are combined to form novel molecular entities, that can simultaneously regulate multiple mechanisms effectively. This chapter presents an overview of the various factors contributing to the pathophysiology of Alzheimer's disease. Chimeric strategies that are being developed to supplement the single-mechanism targeting acetylcholinesterase drugs, which are currently available for the treatment of Alzheimer's disease, are also exemplified.

Keywords: Alzheimer's disease; chimeric compounds; cholinesterase inhibitors; hepatotoxic; neuroinflammation

In: *Alzheimer's Disease: Drug Discovery*. Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

Copyright: The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

INTRODUCTION

Dementia can be defined as a clinical syndrome characterized by cognitive impairment that often leads to dependence on others for carrying out basic functions of daily life (1, 2). Alzheimer's disease (AD) is one of the most prevalent neurodegenerative disorder seen in older individuals, accounting for over 80% of all dementia cases worldwide (1, 2). AD causes structural damage to the brain, which results in substantial functional loss (3). It is a neurological disorder, often characterized by short-term memory impairment, which progresses into cognitive and physical disabilities (4). The etiology of AD is multifactorial with genetic, environmental, behavioral, and developmental components playing a role (5). The greatest risk factor is advancing age, while others include positive family history, head trauma, female gender, history of depression, diabetes mellitus, hyperlipidemia, and vascular factors (5).

In developed countries, the incidence of AD is increasing rapidly along with the aging of populations. The prevalence of AD among 60-year-old individuals is about 1%. This frequency doubles approximately every 5 years, becoming 2% at the age of 65 years, 8% at 75 years, and 16 and 32% at 85 years (4). This disorder may be classified as early and late-onset AD. Early onset AD is typically seen between 30 and 60 years of age and accounts for less than 6% of all cases. Late onset AD accounts for approximately 90% of cases and has an age at onset of more than 60 years (5). The estimated number of patients is 7–8 million in Europe, 4–5 million in the USA, and 24 million worldwide (6). This number is expected to reach around 100 million (one out of every 85 people) by 2050 (7).

AD is a multifaceted disease related with multiple risk factors that have an impact on the emotional and financial status of the patients and their families (3, 8). As per the World Alzheimer Survey 2015, the total health care expenses of the disease including medical services, social support, and informal care were 818 billion dollars with a rise of 35.4% compared with the same survey conducted in 2010. The expense for the treatment of AD for the year 2018 was \$1 trillion, which is expected to rise about 2 trillion by the year 2030 (9).

DISEASE AND PATHOLOGY

AD is a complex, self-escalating neurodegenerative disease, which is marked by the presence of beta-amyloid (A β)-rich senile plaques and neurofibrillary tangles (NFTs) in the brain (10, 11). The disease is characterized by impairments of memory and cognition, depression, and psychiatric and behavioral changes (12). The diagnosis of AD is confirmed by brain histopathological examination and relies on many clinical factors (4). Cognitive and functional declines are spread over 5–8 years as the disease progresses clinically from mild to moderate to severe AD (4). The mild stage is marked by short-term memory loss and generally lasts for about 2–3 years, which is often followed by symptoms of anxiety and depression (4). Neuropsychiatric manifestations, such as visual hallucinations, false beliefs, and reversal of sleep patterns, are prominent during the moderate stage (4). Motor signs, such as motor rigidity, mark the severe stage of AD (4). Cognitive and functional declines are seen in all three stages of the disease (4). The gross pathology
of AD includes generalized cortical atrophy, usually most prominent in the medial temporal lobe and hippocampus (13). AD pathology includes positive and negative signs (14, 15). Positive signs manifest in the form of cerebral AB plaques (the major peptide component being A β 42) and NFTs of paired helical filaments made of hyperphosphorylated microtubule-associated protein tau (MAP τ) (4). These signs are quantified and shape the foundation for diagnostic criteria. Negative signs include neuronal and synaptic losses (14). The chronology of neuronal loss occurs on two levels. There is a local impact after the accumulation of AB and aggregation of tau, and there is selective impact on the regions affected by tau pathology (14). Synapse loss is another factor that contributes toward atrophy of the brain cortex. Synaptic loss has been demonstrated in AD patients through immunohistochemical studies, where immunoreactivity to antibodies of pre- or postsynaptic proteins (generally the presynaptic protein synaptophysin) is quantified using electron microscopy studies (16). Negative signs are difficult to evaluate and are not included in the diagnostic criteria, even though they have great physiopathological relevance (14). In recent years, several hypotheses have been proposed in an attempt to explain the pathogenesis of AD. These include the amyloid hypothesis, tau hypothesis, cholinergic hypothesis, oxidative stress hypothesis, and metal ion hypothesis, as depicted in Figure 1.

Amyloid hypothesis

Histopathologically, the two hallmarks of the disease process in AD are extracellular amyloid plaques and intraneuronal tau NFTs, which characterize the dominant amyloid cascade hypothesis (9, 17). A β is formed after the proteolytic cleavage of a larger protein, known as the amyloid precursor protein (APP) (3).



Figure 1. Pathophysiological mechanisms and multifunctional targeting as an approach for Alzheimer's disease (AD) treatment. This figure depicts the various hypotheses involved in AD pathology and the role of chimeric conjugates in multifunctional targeting as a new therapeutic strategy for AD.

APP is a glycoprotein comprising of about 770 amino acids and is found primarily in the CNS neurons (9). Activation of APP and subsequent cleavage leads to the formation of oligomers, fibrils, plaques, and β -sheets, ultimately resulting in A β aggregation, disruption of cellular communication, and generation of neuroinflammation (3, 18). There are several forms of A β , including lower and higher order oligomers and fibrils such as A β_{40} and A β_{42} (3). A β_{42} is the first form that accumulates in the brain and forms amyloid plaques and deposits (3, 19). The two enzymes responsible for APP cleavage are alpha-secretase (α -secretase) and beta-secretase (β -secretase) (3, 20). The α -secretase enzyme combines with the C-83 subunit (non-amyloidogenic pathway) and produces APP- α precursors, which is thought to have neuroprotective effects (3, 21). On the other hand, the β -secretase enzyme combines with the C-99 β -subunit and produces APP- β (amyloidogenic pathway) (3).

Genetic factors also influence A β production and deposition. Genetic mutation and polymorphism of presenilin (PSEN1 and PSEN2) elevate the production of A β (3, 22).

Amyloid plaques are extracellular deposits of $A\beta$, abundant in the cortex of AD patients (23). There are two forms of plaques: neuritic and diffuse. Neuritic plaques mainly consist of $A\beta$ and form part of tau-containing dystrophic neurites (13). Neuritic plaques are useful in the pathological diagnosis of AD because their appearance indicates the extent of cognitive impairment (23). Diffuse plaques consist mainly of A β -protein (13). Diffuse plaques are generally non-neuritic and are not linked with synaptic loss (23). These forms of plaques are also normally found in the brains of elderly patients with normal cognition, and their presence is not indicative of AD (23). In extracellular regions, A β accumulates and forms deposits in the parenchyma and vascular walls, which is denoted as cerebral amyloid angiopathy (CAA) (15). These deposits lead to a threefold increase in the concentration of soluble A β and comprise the visible part of A β aggregates (15). These soluble, oligomeric A β assemblies are highly toxic as well as difficult to identify and analyze (14, 15). These deposits, which are comprised of A β 42, can be further classified as diffuse, focal, or stellate (15).

Diffuse deposits are large, about 50 µm in size or larger, and are generally seen in patients where there is no cognitive impairment, leading to the conclusion that these lesions are not directly toxic (15). Such deposits are typically seen in the striatum and the molecular layer of the cerebellum (14, 16). Focal deposits are characterized by dense, spheroid aggregation of A β . Few microglial cells are located in the area around the focal deposit that contains the amyloid plaque core. Astrocytes are located far away from the core and are involved in the processes of differentiation of neuronal stem cells without affecting neuronal or oligodendrocyte differentiation (16, 24). Stellate deposits are generally linked to astrocytes and are seldom observed and rarely examined (15). A β may build up in the walls of blood vessels, primarily in the arteries and capillaries, but seldom in the veins leading to CAA. CAA is characterized mainly by the presence of A β -40, which is frequently seen in the parenchymal deposits and is more soluble than A β -42 (14). Some degree of CAA, usually mild, is present in approximately 80% of AD patients (23).

Tau hypothesis

Tau is an MAP found in the axon, where it attaches and stabilizes the microtubules, thereby physiologically promoting axonal transport (23). In AD, tau is translocated to the somato-dendritic component, where it hyperphosphorylates, misfolds, and forms aggregates, leading to the formation of NFTs and neuropil threads (23). The effects of tau in neurodegeneration are less well established (25). However, aggregation of phosphorylated tau in dendritic spines appears to disrupt synaptic plasticity (14). In the cellular body of the neurons, tau aggregates form NFTs and neuropil threads in the dendrites and axons, which surround the core of the senile plaque (14). NFTs are hyperphosphorylated and misfolded tau intraneuronal aggregates, which become extraneuronal or "ghost tangles" with the death of the neurons carrying the tangles. NFT progression occurs in a stereotypical spatiotemporal way, which is linked with the decline in cognitive function (23). These are axonal and dendritic segments consisting of aggregated and hyperphosphorylated tau and are typically related with the NFTs in the brain (23).

The tau protein has a significant role to play in microtubule stabilization, which is important for maintaining cell integrity (3). The major structural domains of the tau proteins include the N-terminal projection domain, a microtubulebinding domain at the C-terminus, and a short sequence encompassing the tail domain (3). Tau proteins are hyperphosphorylated and form insoluble intracellular NFTs in AD, which lose the tenacity to bind the microtubules of brain cells. These hyperphosphorylated forms bind to each other, tying themselves in knots called NFTs that disrupt neuronal plasticity and cause neurodegeneration. Tau–tau interactions and its hyperphosphorylation in AD trigger a cascade of events in the microglial cells and astrocytes, activating the NF-kB pathway and overproduction of proinflammatory mediators such as TNF- α and interleukins (ILs), resulting in inflammatory reactions in brain (3). The elimination of tau is far more complicated compared to others like A β (3). It has been shown via PET imaging that deposition of A β (26).

Cholinergic hypothesis

The cholinergic hypothesis was the first established theory proposed to explain the pathogenesis and development of AD (9). In addition to the histopathological markers, the brain of AD patients is usually characterized by atrophy, synaptic loss, and decline in central neurotransmission (9) along with degeneration of neurons of the basal forebrain (9, 27). Cholinergic neurons are affected in the initial phase of the disease with greater than 90% of cholinergic neurons being lost in the advanced stages (9). Per this theory, the development of all symptoms related to impaired cognition in AD is due to the disruption of cholinergic neurons in the basal forebrain, along with loss of central cholinergic transmission (9).

Oxidative stress hypothesis

Free radicals play an important role in the progression of neurodegeneration (3). Neuronal cells are more susceptible to free radical damage because of greater oxygen content and lack of antioxidant enzymes when compared with other organs (3). There is clear evidence that oxidative stress induced by $A\beta$ is critical to the pathogenesis and progression of AD, leading to exacerbation of inflammatory processes, which is a characteristic of many multifactorial diseases including AD (9). The mitochondrial membranes of the AD postmortem brain have

demonstrated that $A\beta$ and APP cause disruption of electron transport chain, thereby promoting irreversible neurodegeneration and cellular damage (3, 28).

Metal ion hypothesis

Metal dyshomeostasis is involved in the progression of AD (29). Development of A β plaques and NFT is aggravated by the aberrant accumulation of metals in the brains of AD patients (30). High concentrations of Cu and Fe in the brain trigger the production of reactive oxygen species, which further exacerbates oxidative stress, thereby leading to worsening of AD (31, 32). Thus, a useful therapeutic strategy to mitigate AD would be to decrease the abnormal load of metal ions in the brain by chelating them (33).

CURRENT LINE OF TREATMENT

AD is a complex disease, and hence difficult to treat with a single medication or therapy (34). The current line of treatment functions by modulating the levels of specific brain neurotransmitters such as acetylcholine and glutamate (34). These are helpful in retaining thoughts, cognitive functions, and social skills and can mitigate behavioral issues to a certain extent (34). However, these approaches do not address the root cause of the disease. The U.S. Food and Drug Administration (US-FDA) has approved several drugs to provide symptomatic relief in AD (34). Existing drugs employed for the symptomatic treatment of AD can be divided into two major classes: acetylcholinesterase (AChE) inhibitors such as donepezil, rivastigmine, galantamine, and tacrine and *N*-methyl-D-aspartate (NMDA) antagonists (glutamate inhibitor) such as memantine.

Acetylcholinesterase inhibitors

Cholinesterase inhibitors (ChEIs) are generally used for long-term symptomatic treatment for AD (35). ChEIs are the only class of drugs approved by FDA for the symptomatic treatment of AD that can alter cholinergic neurotransmission and these include donepezil, rivastigmine, galantamine, and tacrine (4). To date, ChEIs are the only drugs that have shown significant improvements in cognition of AD patients by improving the cholinergic transmission in neuronal synapses. ChEIs slow down the degradation of the choline neurotransmitters at the synaptic clefts by inhibiting the cholinesterase enzymes, AChE, and butyrylcholinesterase (BuChE), which are responsible for the choline neurotransmitter degradation (9). These enzymes are abundant in neuritic plaques and can be inhibited by ChEIs; this may alter the build-up of $A\beta$, which is a critical part of AD pathophysiology (4). ChEIs increase cholinergic functions in AD at the postsynaptic cholinergic neuron (35). This class of drugs decreases AChE-induced destruction of ACh in the synaptic cleft, elevates the intrasynaptic residence time of acetylcholine, and promotes interaction between acetylcholine and the postsynaptic cholinergic receptor (35). Thus, to inhibit them, ChEIs increase the availability of these neurotransmitters in the synaptic cleft, thereby reducing the symptoms of AD (9).

AChE is also partially involved in the production of amyloid plaques and neurofibrillary tangles (34). AChE acts as an influencer to help in aggregating clusters of A β peptides, resulting in the formation of complexes with mature fibrils (34). The newly formed complexes are more cytotoxic in comparison to A β fibrils alone. ChEIs increase the levels of ACh in the brain of AD patients (34). Evidence derived from clinical trials, imaging, and basic science studies indicates that ChEIs are useful for symptomatic treatment but have limited disease-modifying effects (35).

Donepezil is a piperidine-derivative AChE inhibitor drug, which increases the levels of acetylcholine in the CNS (9, 36). It has shown moderate benefit in the treatment of AD patients due to its modest and transient outcomes (36, 37). It is effective in managing the symptoms of AD-associated dementia. However, it does not alter the progression of AD (38). Donepezil is metabolized via the cytochrome P-450 system and has the tendency of being involved in drug–drug interactions, especially when used in combination (9).

Rivastigmine, a physostigmine-derived drug, is the only carbamate containing AChE inhibitor approved for the treatment of mild to moderate AD (9, 39). It improves cognition and shows neuroprotective effects, but does not alter the course of disease and only leads to a modest improvement in cognitive functions. This drug shows good activity and tolerance in AD patients and is not involved in the cytochrome P-450 system metabolism, thereby decreasing the chances of drug–drug interactions (9).

Galantamine is a tertiary alkaloid extracted from various species of *Amaryllidaceae* (9). It is a selective, competitive, and reversible inhibitor of AChE with nicotinic-modulating properties, has low hepatotoxicity (9), and reduces APP metabolism in animal models of AD (35). However, its involvement in cytochrome P-450 metabolism makes it prone to interaction with other drugs (9).

Tacrine, a dual AChE and BuChE inhibitor, was the first of its kind to get FDA approval for the treatment of AD. It was withdrawn from the market shortly after FDA approval due to serious hepatotoxicity (40). Tacrine is a noncompetitive, reversible inhibitor of AChE, which has a short half-life (9).

N-methyl D-aspartate receptor antagonism (NMDA antagonists)

Overstimulation of the NMDA receptor by the neurotransmitter glutamate is implicated in neurodegenerative disorders (4). Glutamate is the principal excitatory neurotransmitter in the brain (4). The overstimulation of glutamate has been known to contribute to neuronal damage, which is termed as excitotoxicity (4). Such excitotoxicity eventually contributes to neuronal calcium overload and has been implicated in neurodegenerative diseases (4). Glutamate activates several postsynaptic receptors, including the NMDA receptor, which have a direct impact on the memory processes, dementia, and in the pathogenesis of AD (4).

The FDA-approved NMDA antagonist memantine decreases glutaminergic excitotoxicity by influencing neuronal activity in the hippocampus and can be used in the treatment of moderate to severe AD (35). Memantine is approved as an alternative to ChEIs in treatment of moderate to severe AD (25). Memantine at high concentrations can suppress synaptic plasticity, which is believed to have an effect on learning and memory. However, at lower concentrations, memantine can

promote synaptic plasticity, thereby enhancing memory in animal models of AD (4). Memantine also has the potential to enhance long-term potentiation and decrease tau hyperphosphorylation (17). The neurobiological basis for the therapeutic action of memantine in AD is not clearly known. Memantine is a noncompetitive NMDA receptor antagonist with moderate affinity and fast on/off kinetics (4). These attributes are vital for memantine action as it is able to balance the effects of excessive glutamate levels while preserving physiologic activation of NMDA receptors necessary for learning and memory (4). Memantine inhibits the effects of excessive glutamate production, which contributes to cell death and cognitive impairment (4).

RATIONALE FOR CHIMERIC CONJUGATES

The current FDA-approved drugs addressing a single mechanism have turned out to be palliative rather than curative. By addressing just the cholinergic hypothesis, they only provide temporary relief for the patients by improving their cognitive functionality throughout the time period of usage (41, 42). There is a critical need to add antioxidant, metal chelation, neuroprotective, $A\beta_{1-42}$ amyloid antiaggregation, and anti-inflammatory activities into the compounds. Such molecules that can address all the hypotheses of AD will likely yield significant disease-modifying outcomes.

The multifaceted, complex nature of AD has limited the treatment options in the battle against the disease. One of the ways of tackling this problem is to expand the scope of single-mechanism targeting drugs to form multi-targeting chimeric entities (43). The multi-target directed ligand (MTDL) design strategy is a method where a molecule is designed by simultaneously integrating multiple functionalities into it that can target different mechanisms crucial to the disease pathology (44). These chimeric conjugates are created by molecular hybridization of different biologically relevant pharmacophores (45). Each pharmacophore of the new chimeric molecule retains the ability to interact with its own target while the chimeric molecule simultaneously modulates multiple molecular targets, thereby producing a range of diverse pharmacological responses (45-47). Such engineered chimeric compounds simultaneously target many of the implicated pathways of the disease, thereby yielding a disease-modifying effect (48, 49). In addition, these compounds potentially have a lower risk of triggering drug–drug interactions and facilitate pharmacodynamic and pharmacokinetic roles of drug administration (46, 47). AChE inhibitors like donepezil, rivastigmine, and tacrine have been used as starting points and combined with several bioactive molecules to generate chimeric series, which have demonstrated expanded efficacy and safety profiles. The hybrid series were designed so as to retain the key structural features essential for retaining AChE inhibition of the parent molecules.

Donepezil-related derivatives as multifunctional compounds for AD

The structure of donepezil shows the presence of benzyl piperidine and substituted indanone fragments linked via a methylene bridge (Figure 2, left panel). The binding pose of donepezil within the AChE pocket as seen in the X-ray



Donepezil

[pharmacophoric unit (boxed)]



Donepezil-aryl acyl hydrazone hybrids

Compound	R
1a	nitro
1b	N-piperidino
1c	F
1d	Br
1e	CI
1f	methoxy
1g	morpholino

Figure 2. Left panel—donepezil with its pharmacophoric unit boxed in blue; right panel donepezil-aryl acyl hydrazone hybrids. Various donepezil-aryl acyl hydrazone chimeras have been constructed by conjugating the pharmacophoric unit of donepezil with substituted phenyl acyl hydrazones.

diffraction-elucidated crystal structure (PDB id 4EY7) shows the interaction of the benzyl piperidine fragment with the catalytic site of AChE (50). This fragment has been nominated as a crucial pharmacophoric subunit of donepezil. The indanone moiety binds to the peripheral anionic site of the AChE pocket and forms aromatic stacking interactions (50). The benzyl piperidine fragment of donepezil has been hybridized with various bioactive groups in order to introduce other activities into the molecule so as to make them more effective for the holistic treatment of AD. Aryl acyl hydrazones possess anti-inflammatory activity, which has been shown to slow down the progression of AD. Donepezil-aryl acyl hydrazone chimeras were constructed (51) by attaching the aryl-acyl hydrazone sidechain to the benzyl end of *N*-benzyl piperidine of donepezil (Figure 2, right panel).

The donepezil-derived hybrid molecules were screened for AChE inhibition according to the spectrophotometric method developed by Ellman (52). Replacement of the dimethoxy indanone fragment of donepezil with various ring-substituted aryl acyl hydrazones gave molecules that exhibited comparable AChE inhibition. In-depth structure–activity correlation studies showed that compounds that had the 3-hydroxy piperidine moiety were more active than the molecules where the 3-hydroxy group was acetylated. Substitutions on the aromatic ring on the other side of the molecule with groups like nitro (1a) and piperidine (1b) resulted in molecules that were more potent relative to the unsubstituted compound (51). Halogen substitution gave a threefold increase in activity as compared to the unsubstituted molecule with the fluoro analogue (1c) being more potent than bromo (1d) and chloro (1e) derivatives (51). The placement of the benzyl piperidine fragment in the chimeric series with the piperidine ring placed at the terminal end was opposite to that in donepezil. It is possible that by reversing the placement of benzyl piperidine fragment, the hybrid molecules were able

to find an alternate binding orientation in the AChE binding pocket, which resulted in their enhanced activity profiles.

Suppression of the neuroinflammation process is an effective therapeutic approach against AD. The donepezil-aryl acyl hydrazone chimeric molecules were tested for their *in vivo* anti-inflammatory activities using classical animal models such as mechanical allodynia test, formalin-induced hyperalgesia, and carrageenan-induced paw oedema assays (51). Halogens (1c, 1e) and methoxy-substituted hybrid molecules (1f) as well as compounds substituted with rings such as piperidine (1b) and morpholine (1g) (Figure 2) significantly reduced mechanical hyperalgesia index, decreased licking time in the formalin test pointing to an analgesic effect, and reduced oedema volume, thereby confirming an anti-inflammatory effect (51).

The halogen-substituted compounds (1c, 1e) were found to inhibit the release of TNF- α and IL-1 β induced by lipopolysaccharides (LPS) in THP-1 cells, which are representative of human microglial cells (51). THP-1 cells were treated with 10 μ M of the compounds and LPS (1 μ g/ml) for 24 h. At the end of the treatment, it was concluded that both halogen-substituted compounds reduce evoked neuroinflammation (51).

Rivastigmine-related derivatives as multifunctional compounds for AD

Rivastigmine, a carbamate-containing AChE inhibitor, has been found to provide only mild to moderate benefit in patients with AD (39). The carbamate moiety is the pharmacophore subunit of rivastigmine (Figure 3, left panel), which binds to the catalytic site of AChE and is responsible for its ChEI activity (53). Rivastigmine chimeras with amino chalcones with promising cholinesterase inhibition activity are shown in Figure 3 (right panel). Xiao et al. worked on developing rivastigmine-4 amino chalcone hybrids and conducted in-depth structure–activity correlation studies for the series (54).

Several of the rivastigmine-derived compounds showed potent AChE inhibition. Structure–activity correlation studies revealed that compounds with cyclic amine groups (2b, 2c, 2d) at one or both extremities of the molecule show better inhibitory activity as compared with those with noncyclic amine groups (2a). Pyrrolidine ring-substituted compound 2b was twice as potent as rivastigmine, while morpholine ring-substituted compound 2e (Figure 3) was the weakest inhibitor of AChE (54). The authors surmised that the electron-withdrawing inductive effect of the oxygen atom on the morpholine ring may decrease the electron density of the ring nitrogen, thereby influencing its protonation at physiological pH, which could diminish the cation– π interaction between the nitrogen and residues of the catalytic active site of AChE (54).

The antioxidant activity for rivastigmine-amino chalcone hybrids was evaluated by the oxygen radical absorbance capacity assay involving fluorescein. Compounds containing a pyrrolidine ring (2b), *N*-methylethaneamine (2f), and benzyl piperazine ring (2g) as substituents were found to exhibit the most potent antioxidant activities. 4-Dimethylamine chalcone–rivastigmine hybrid molecule (2a) showed moderate antioxidant activity, while replacement with other amino alkyl groups resulted in loss of antioxidant activity. The authors concluded that dimethylamine substitution at para position is favorable for antioxidant potency (54).



Carbamate

Rivastigmine [pharmacophoric unit (boxed)]



Rivastigmine-aminochalcone hybrids



Figure 3. Left panel—rivastigmine with its pharmacophoric unit boxed in blue; right panel rivastigmine-aryl acyl hydrazone hybrids. Various rivastigmine-aminochalcone chimeras have been constructed by conjugating the pharmacophoric unit of rivastigmine with substituted amino chalcones.

The 2-hydroxy and ketone groups of rivastigmine-amino chalcone chimeras undergo intramolecular hydrogen bonding as they exhibit metal-chelating properties. Xiao et al. found that compound 2b (Figure 3) exhibited selective metal chelation for copper and aluminium but not iron and zinc metal ions (54).

The effects of the rivastigmine-amino chalcone hybrid molecules on A β aggregation were evaluated by performing thioflavin T (ThT) fluorescence assay using curcumin as a reference standard. Structural studies concluded that noncyclic amines 2a and 2f (Figure 3) exhibited the most potent inhibition effects on selfinduced A β aggregation, while the *N*-benzyl piperazine containing analogue (2g, Figure 3) showed the best Cu²⁺-induced A β aggregation inhibition. Besides this, bulky substituents on the hydroxy group at 2-position on the chalcone fragment lowered the anti-aggregation effects (54).

Tacrine analogues with decreased liability of hepatotoxicity

Tacrine, an acridine analogue, was the first centrally acting ChEI approved for the treatment of AD. It is a reversible, noncompetitive inhibitor of AChE and BuCHE. It also possesses the ability to reduce $A\beta$ -induced neurotoxicity. Despite it benefits, tacrine is poorly tolerated and often causes reversible abnormalities in liver enzymes. Nevertheless, its inherent efficacy and small molecular weight have attracted a lot of research directed toward the development of MTDLs. The whole molecule of tacrine (Figure 4, left panel) has been used as a starting point and fused with hepatoprotective scaffolds, leading to the development of safe, efficacious tacrine hybrids (55). Tacrine derivatives coupled to fragments that help counter its hepatotoxicity are shown in Figure 4 (right panel).

Zha et al. developed tacrine-benzofuran chimeric molecules in an attempt to combine the AChE inhibitory properties of tacrine and the *in vitro* inhibitory



Figure 4. Tacrine hybrids with reduced hepatotoxicity. Several tacrine chimeras have been constructed by conjugating tacrine with various hepatoprotective moieties.

effects on A β fibril formation and aggregation reported for the benzofuran nucleus (56). The most active designed molecule, compound 3 (Figure 4), showed nanomolar inhibitory potency against AChE, as well as an ability to partially inhibit AChE-induced A β fibril formation and amyloid self-aggregation (56). This compound demonstrated mixed inhibitory behavior in an enzyme kinetics study pointing to dual binding interaction sites (56). This was confirmed by the authors by determining the three-dimensional form of the AChE-bound structure of this compound through X-ray diffraction studies (56). The tacrine fragment was seen to occupy its place at the catalytic site, engaging in stacking interactions with Trp 84 and Phe 330 and hydrogen bonding interactions with His 440 of the catalytic triad of AChE, while the methyl benzofuran was found to make contact with the peripheral anionic site where it is accommodated within a pocket of hydrophobic residues (56). In addition, compound 3 had a better safety profile and showed significantly lower hepatotoxicity than tacrine when tested with the alanine aminotransferase and aspartate aminotransferase activity assays (56).

Mao et al. synthesized a number of tacrine-propargylamine derivatives, inspired by propargylamine-containing compounds, which exhibit neuroprotective effects (40). These compounds were tested for AChE inhibition and neurotoxicity (40). In addition, they were evaluated for hepatotoxicity in human hepatic stellate cells using the colorimetric MTT assay (40). It was reported that compound 4 (Figure 4) exhibited superior AChE inhibition and lower neurotoxicity than tacrine. In addition, it almost eliminated the hepatotoxicity of tacrine (40). Kinetic studies were also carried out on this compound, which pointed to a mixed type of enzyme inhibitory behavior. It is possible that extending the tacrine amine with the lipophilic propargyl group endowed the molecule with additional binding opportunities within the AChE enzyme, resulting in a twofold improvement in AChE inhibition. Such tacrine chimeras, which have excellent AChE inhibition and neuroprotective effects without the hepatotoxicity of tacrine, can be used as potential lead compounds for the treatment of AD (40).

Chioua et al. designed a series of tacripyrimidines by coupling the ChEI tacrine moiety to derivatives of 3,4-dihydro dihydropyrimidin-2(1H)-thiones, which are known calcium channel blockers (CCBs) (57). CCBs enhance cerebrovascular

perfusion and attenuate amyloid- β -induced neuronal decline and neurotoxicity, improve cell survival in the presence of A β *in vitro*, and show neuroprotective effects (57). Derivatives bearing halogens (Br, Cl) at meta and para position of the aromatic ring of the dihydropyrimidine-thiones demonstrated the highest inhibitory potencies toward AChE, while the presence of a 4-dimethylamino group or 3-nitro group was found to be the best CCBs, with potencies higher than that of the reference CCB drug nimodipine (57). Tacripyrimidine compound 5 (Figure 4) had the most balanced overall biological profile. It had low micromolar AChE inhibitory potency as well as calcium channel blocking activity and had no significant hepatotoxicity toward HepG2 cells up to 300 mM and excellent predicted oral absorption and BBB permeability (57).

CONCLUSION

The currently available FDA-approved drugs for the treatment of AD are limited by the fact that they target only a single mechanism in the development of this multifactorial disease with extremely complex pathophysiology. Several molecules with antioxidant, anti-inflammatory, and neuroprotective properties are known. Conjugation of these molecules with the currently available, FDA-approved ChEIs to form molecular chimeras has been shown to expand their anti-AD spectrum, thereby creating entities that have the potential for development as diseasemodifying therapies for AD.

Conflict of Interest: The authors declare no potential conflict of interest with respect to research, authorship, and publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- 1. Tobore Onojighofia Tobore. On the etiopathogenesis and pathophysiology of Alzheimer's disease: A comprehensive theoretical review. J Alzheimer's Dis. 2019;68(2):417–37. http://dx.doi.org/10.3233/ JAD-181052
- 2. Qiu C, Kivipelto M, Von Strauss E. Epidemiology of Alzheimer's disease: Occurrence, determinants, and strategies toward intervention. Dialogues Clin Neurosci. 2009;11(2):111–28.
- Sharma P, Srivastava P, Seth A, Tripathi PN, Banerjee AG. Comprehensive review of mechanisms of pathogenesis involved in Alzheimer's disease and potential therapeutic strategies. Prog Neurobiol. 2019;174(December 2018):53–89. http://dx.doi.org/10.1016/j.pneurobio.2018.12.006
- Silvestrelli G, Lanari A, Parnetti L. Treatment of Alzheimer's disease: From pharmacology to a better understanding of disease pathophysiology. Mech Ageing Dev. 2006;127:148–57. http://dx.doi.org/ 10.1016/j.mad.2005.09.018
- 5. Anand R, Gill KD, Mahdi AA. Neuropharmacology therapeutics of Alzheimer's disease: Past, present and future. Neuropharmacology. 2014;76(Pt A):27–50. http://dx.doi.org/10.1016/j.neuropharm.2013.07.004

178 Dhala I et al.

- Roland JR, Jacobsen H. Alzheimer's disease: From pathology to therapeutic approaches. Angew Chem Int Ed. 2009;48(17):3030–59. http://dx.doi.org/10.1002/anie.200802808
- González JF, Alcántara AR, Doadrio AL, María J, González JF, Alcántara AR. Developments with multi-target drugs for Alzheimer's disease: An overview of the current discovery approaches current discovery approaches. Expert Opin Drug Discov. 2019;14(9):879–91. http://dx.doi.org/10.1080/174 60441.2019.1623201
- 8. Huang Y, Mucke L. Alzheimer mechanisms and therapeutic strategies. Cell. 2012;148(6):1204–22. http://dx.doi.org/10.1016/j.cell.2012.02.040
- Ozela PF, Padilha EC, Braga FS. Alzheimer's disease: A review from the pathophysiology to diagnosis, new perspectives for pharmacological treatment. Curr Med Chem. 2018;25(26):3141–59. http:// dx.doi.org/10.2174/0929867323666161213101126
- Sun Y, Chen J, Chen X, Huang L, Li X. Inhibition of cholinesterase and monoamine oxidase-B activity by Tacrine-Homoisoflavonoid hybrids. Bioorganic Med Chem. 2013;21(23):7406–17. http://dx.doi. org/10.1016/j.bmc.2013.09.050
- Araújo JQ, Lima JA, Pinto ADC, De Alencastro RB, Albuquerque MG. Docking of the alkaloid geissospermine into acetylcholinesterase: A natural scaffold targeting the treatment of Alzheimer's disease. J Mol Model. 2011;17(6):1401–12. http://dx.doi.org/10.1007/s00894-010-0841-2
- Calcul L, Zhang B, Jinwal UK, Dickey CA, Baker BJ. Natural products as a rich source of tau-targeting drugs for Alzheimers disease. Future Med Chem. 2012;4(13):1751–61. http://dx.doi.org/10.4155/ fmc.12.124
- 13. Pressman P, Rabinovici GD, San C, Francisco S. Alzheimer's disease. Encycl Neurol Sci. 2014;1:91–6. http://dx.doi.org/10.1016/B978-0-12-385157-4.00475-9
- 14. Garciduenas-Ana Laura Calderon DC. Alzheimer's disease. Handb Clin Neurol. 2018;145 (3rd series):325–37. http://dx.doi.org/10.1016/B978-0-12-802395-2.00023-7
- Benoi CDÆ, Potier M. Classification and basic pathology of Alzheimer disease. Acta Neuropathol. 2009;118(1):5–36. http://dx.doi.org/10.1007/s00401-009-0532-1
- Jellinger KA. Alzheimer's disease. In: Gilman S, editors. Neurobiol Dis. 2007;1047–85. http://dx.doi. org/10.1016/B978-012088592-3/50008-6
- Tayeb HO, Duk H, Price BH, Tarazi FI. Pharmacology & therapeutics pharmacotherapies for Alzheimer's disease: Beyond cholinesterase inhibitors. Pharmacol Ther. 2012;134(1):8–25. http:// dx.doi.org/10.1016/j.pharmthera.2011.12.002
- O'Brien RJ, Wong PC. Amyloid precursor protein processing and Alzheimer's disease. Annu Rev Neurosci. 2011;34:185–204. http://dx.doi.org/10.1146/annurev-neuro-061010-113613
- Morgan C, Colombres M, Nuñez MT, Inestrosa NC. Structure and function of amyloid in Alzheimer's disease. Prog Neurobiol. 2004;74(6):323–49. http://dx.doi.org/10.1016/j.pneurobio.2004.10.004
- Thinakaran G, Koo EH. Amyloid precursor protein trafficking, processing, and function. J Biol Chem. 2008;283(44):29615–19. http://dx.doi.org/10.1074/jbc.R800019200
- Kojro E, Fahrenholz F. The non-amyloidogenic pathway: Structure and function of alpha-secretases. Subcell Biochem. 2005;38:105–27. http://dx.doi.org/10.1007/0-387-23226-5_5
- 22. Hardy J. The amyloid hypothesis for Alzheimer's disease: A critical reappraisal. J Neurochem. 2009;110(4):1129–34. http://dx.doi.org/10.1111/j.1471-4159.2009.06181.x
- Serrano-Pozo A, Frosch MP, Masliah E, Hyman BT. Neuropathological alterations in Alzheimer disease. Cold Spring Harb Perspect Med. 2011;1(1):1–23. http://dx.doi.org/10.1101/cshperspect. a006189
- 24. Guo Y, Wei Q, Huang Y, Xia W, Zhou Y, Wang S. The effects of astrocytes on differentiation of neural stem cells are influenced by knock-down of the glutamate transporter, GLT-1. Neurochem Int. 2013;63(5):498–506. http://dx.doi.org/10.1016/j.neuint.2013.08.003
- Jolla L, States U, Jolla L, States U. Alzheimer's disease. Handb Clin Neurol. 2019;167(3rd series): 231–55.
- 26. James OG, Doraiswamy PM, Borges-Neto S. PET imaging of tau pathology in Alzheimer's disease and tauopathies. Front Neurol. 2015;6(Mar):1–4. http://dx.doi.org/10.3389/fneur.2015.00038
- Racchi M, Mazzucchelli M, Porrello E, Lanni C, Govoni S. Acetylcholinesterase inhibitors: Novel activities of old molecules. Pharmacol Res. 2004;50(4):441–51. http://dx.doi.org/10.1016/j.phrs. 2003.12.027

- Reddy PH, Beal MF. Amyloid beta, mitochondrial dysfunction and synaptic damage: Implications for cognitive decline in aging and Alzheimer's disease. Trends Mol Med. 2008;14(2):45–53. http://dx.doi. org/10.1016/j.molmed.2007.12.002
- Thakur AK, Kamboj P, Goswami K, Ahuja K. Pathophysiology and management of Alzheimer's disease: An overview. J Anal Pharm Res. 2018;9(2):226–35. http://dx.doi.org/10.15406/japlr.2018.07.00230
- Barnham KJ, Bush AI. Metals in Alzheimer's and Parkinson's diseases. Curr Opin Chem Biol. 2008;12(2):222–8. http://dx.doi.org/10.1016/j.cbpa.2008.02.019
- Amit T, Avramovich-Tirosh Y, Youdim MBH, Mandel S. Targeting multiple Alzheimer's disease etiologies with multimodal neuroprotective and neurorestorative iron chelators. FASEB J. 2008;22(5): 1296–305. http://dx.doi.org/10.1096/fj.07-8627rev
- Zheng H, Youdim MBH, Fridkin M. Site-activated multifunctional chelator with acetylcholinesterase and neuroprotective-neurorestorative moieties for Alzheimer's therapy. J Med Chem. 2009;52(14): 4095–8. http://dx.doi.org/10.1021/jm900504c
- 33. Zhang C, Du QY, Chen LD, Wu WH, Liao SY, Yu LH, et al. Design, synthesis and evaluation of novel tacrine-multialkoxybenzene hybrids as multi-targeted compounds against Alzheimer's disease. Eur J Med Chem. 2016;116:200–9. http://dx.doi.org/10.1016/j.ejmech.2016.03.077
- Anand A, Patience AA, Sharma N, Khurana N. The present and future of pharmacotherapy of Alzheimer's disease: A comprehensive review. Eur J Pharmacol. 2017;815(August):364–75. http:// dx.doi.org/10.1016/j.ejphar.2017.09.043
- 35. Salloway S, Mintzer J, Weiner MF, Cummings JL. Disease-modifying therapies in Alzheimer's disease*. Alzheimers Dementia. 2008;4:65–79. http://dx.doi.org/10.1016/j.jalz.2007.10.001
- Wang ZM, Cai P, Liu QH, Xu DQ, Yang XL, Wu JJ, et al. Rational modification of donepezil as multifunctional acetylcholinesterase inhibitors for the treatment of Alzheimer's disease. Eur J Med Chem. 2016;123:282–97. http://dx.doi.org/10.1016/j.ejmech.2016.07.052
- 37. Darreh-Shori T. Effects of cholinesterase inhibitors on the activities and protein levels of cholinesterases in the cerebrospinal fluid of patients with Alzheimer's disease: A review of recent clinical studies. Curr Alzheimer Res. 2009;999:1–6. http://dx.doi.org/10.2174/1567209198607252050
- 38. Kumar A, Sharma S. Donepezil. StatPearls Treasure Island (FL): StatPearls Publishing. 2020.
- Sang Z, Li Y, Qiang X, Xiao G, Liu Q, Tan Z, et al. Multifunctional scutellarin-rivastigmine hybrids with cholinergic, antioxidant, biometal chelating and neuroprotective properties for the treatment of Alzheimer's disease. Bioorganic Med Chem. 2015;23(4):668–80. http://dx.doi.org/10.1016/j. bmc.2015.01.005
- Mao F, Li J, Wei H, Huang L, Li X. Tacrine-propargylamine derivatives with improved acetylcholinesterase inhibitory activity and lower hepatotoxicity as a potential lead compound for the treatment of Alzheimers disease. J Enzyme Inhib Med Chem. 2015;30(6):995–1001. http://dx.doi.org /10.3109/14756366.2014.1003212
- Abeysinghe AADT, Deshapriya RDUS, Udawatte C. Alzheimer's disease: a review of the pathophysiological basis and therapeutic interventions. Life Sci. 2020;256:117996. http://dx.doi. org/10.1016/j.lfs.2020.117996
- 42. Seltzer B. Is long-term treatment of Alzheimer's disease with cholinesterase inhibitor therapy justified? Drugs Aging. 2007;24(11):881–90. http://dx.doi.org/10.2165/00002512-200724110-00001
- Santos MA, Chand K, Chaves S. Recent progress in multifunctional metal chelators as potential drugs for Alzheimer's disease. Coord Chem Rev. 2016;327–328:287–303. http://dx.doi.org/10.1016/j. ccr.2016.04.013
- Singh M, Kaur M, Chadha N, Silakari O. Hybrids: A new paradigm to treat Alzheimer's disease. Mol Divers. 2016;20(1):271–97. http://dx.doi.org/10.1007/s11030-015-9628-9
- Gong CX, Liu F, Iqbal K. Multifactorial hypothesis and multi-targets for Alzheimer's disease. J Alzheimers Dis. 2018;64(Suppl. 1):S107–17. http://dx.doi.org/10.3233/JAD-179921
- Singh M, Kaur M, Kukreja H, Chugh R, Silakari O, Singh D. Acetylcholinesterase inhibitors as Alzheimer therapy: From nerve toxins to neuroprotection. Eur J Med Chem. 2013;70:165–88. http:// dx.doi.org/10.1016/j.ejmech.2013.09.050
- Schmitt B, Bernhardt T, Moeller HJ, Heuser I, Frölich L. Combination therapy in Alzheimer's disease: A review of current evidence. CNS Drugs. 2004;18(13):827–44. http://dx.doi.org/ 10.2165/00023210-200418130-00001

- Bajda M, Guzior N, Ignasik M, Malawska B. Multi-target-directed ligands in Alzheimer's disease treatment. Curr Med Chem. 2011;18(32):4949–75. http://dx.doi.org/10.2174/092986711797535245
- Pisani L, Catto M, Leonetti F, Nicolotti O, Stefanachi A, Campagna F, et al. Targeting monoamine oxidases with multipotent ligands: An emerging strategy in the search of new drugs against neurodegenerative diseases. Curr Med Chem. 2012;18(30):4568–87. http://dx.doi.org/10.2174/092986711797379302
- Cheung J, Rudolph MJ, Burshteyn F, Cassidy MS, Gary EN, Love J, et al. Structures of human acetylcholinesterase in complex with pharmacologically important ligands. J Med Chem. 2012; 55(22):10282–6. http://dx.doi.org/10.1021/jm300871x
- 51. Dias Viegas FP, de Freitas Silva M, Divino da Rocha M, Castelli MR, Riquiel MM, Machado RP, et al. Design, synthesis and pharmacological evaluation of N-benzyl-piperidinyl-aryl-acylhydrazone derivatives as donepezil hybrids: Discovery of novel multi-target anti-alzheimer prototype drug candidates. Eur J Med Chem. 2018;147:48–65. http://dx.doi.org/10.1016/j.ejmech.2018.01.066
- Ellman GL, Courtney KD, Andres V, Featherstone RM. A new and rapid colorimetric determination of acetylcholinesterase activity. Biochem Pharmacol. 1961;7(2):88–95. http://dx.doi. org/10.1016/0006-2952(61)90145-9
- Bacalhau P, San Juan AA, Goth A, Caldeira AT, Martins R, Burke AJ. Insights into (S)-rivastigmine inhibition of butyrylcholinesterase (BuChE): Molecular docking and saturation transfer difference NMR (STD-NMR). Bioorg Chem. 2016;67:105–9. http://dx.doi.org/10.1016/j.bioorg.2016.06.002
- 54. Xiao G, Li Y, Qiang X, Xu R, Zheng Y, Cao Z, et al. Design, synthesis and biological evaluation of 4'-aminochalcone-rivastigmine hybrids as multifunctional agents for the treatment of Alzheimer's disease. Bioorganic Med Chem. 2017;25(3):1030–41. http://dx.doi.org/10.1016/j.bmc.2016.12.013
- Girek M, Szymański P. Tacrine hybrids as multi-target-directed ligands in Alzheimer's disease: Influence of chemical structures on biological activities. Chem Pap. 2019;73(2):269–89. http:// dx.doi.org/10.1007/s11696-018-0590-8
- 56. Zha X, Lamba D, Zhang L, Lou Y, Xu C, Kang D, et al. Novel Tacrine-benzofuran hybrids as potent multitarget-directed ligands for the treatment of Alzheimers disease: Design, synthesis, biological evaluation, and X-ray crystallography. J Med Chem. 2016;59(1):114–31. http://dx.doi.org/10.1021/ acs.jmedchem.5b01119
- 57. Chioua M, Buzzi E, Moraleda I, Iriepa I, Maj M, Wnorowski A, et al. Tacripyrimidines, the first tacrine-dihydropyrimidine hybrids, as multi-target-directed ligands for Alzheimer's disease. Eur J Med Chem. 2018;155:839–46. http://dx.doi.org/10.1016/j.ejmech.2018.06.044

Multifunctional Ligand Approach: Search for Effective Therapy Against Alzheimer's Disease

Anna Pasieka • Dawid Panek • Barbara Malawska

Department of Physicochemical Drug Analysis, Faculty of Pharmacy, Jagiellonian University Medical College, Kraków, Poland

Author for correspondence: Barbara Malawska, Department of Physicochemical Drug Analysis, Faculty of Pharmacy, Jagiellonian University Medical College, Kraków 30-688, Poland. Email: mfmalaws@cyf-kr.edu.pl

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch11

Abstract: Alzheimer's disease is a progressive, incurable, and complex neurodegenerative disease. Currently, an effective treatment that can slow down or stop the damage and death of neurons, which is a characteristic of Alzheimer's disease, is lacking. Taking into account the complex nature of the disease, a multitarget design approach has been developed for the production of new potential anti-AD agents. The goal of this approach is to create a single molecule that can interact selectively with several desired molecular targets relevant to the disease. This strategy was successfully developed two decades ago and has been improved in recent years. This chapter describes the progress made in the discovery and design of selected multitargeted drugs based on molecular targets, which can be used for treating Alzheimer's disease. The most promising among these drugs are the molecules having properties that are valuable not only in the symptomatic therapy but also in the causal treatment of the disease. The main hypotheses of Alzheimer's disease, such as β -amyloid (A β), tau, and cholinergic, suggest that compounds capable of inhibiting the aggregation of neurotoxic Aβ-amyloid peptide and tau protein, and improving the cholinergic neurotransmission, may possess such properties.

In: *Alzheimer's Disease: Drug Discovery.* Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

Copyright: The Authors.

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

Examples of such multifunctional molecules, which have been recently reported in the literature, are presented in this chapter.

Keywords: Alzheimer's disease; disease-modifying strategies; multimodal compounds; polypharmacology; symptomatic strategies

INTRODUCTION

Alzheimer's disease (AD) is a progressive, incurable, and complex neurodegenerative disorder. The two main neuropathological hallmarks of AD are extracellular amyloid plaques composed of β -amyloid (A β), and intracellular neurofibrillary tangles (NFTs) containing the tau protein, leading to nerve cell death. In addition, AD is characterized by a low level of the neurotransmitter acetylcholine and loss of cholinergic neurons, excitotoxicity, impairment of other neurotransmitter systems, extensive oxidative stress, chronic neuroinflammation, mitochondrial dysfunction, calcium and metal dyshomeostasis, and other factors (1). This complex and unclear pathogenesis of AD has led to the creation of several theories. The oldest one is the cholinergic hypothesis, followed by others such as the A β hypothesis, the tau hypothesis, the oxidative stress theory, the metal imbalance theory, the mitochondrial cascade, and the inflammation hypothesis were developed. The available AD therapy is based only on medications that are capable of treating the cognitive symptoms: three cholinesterase inhibitors (donepezil, rivastigmine, and galantamine) and one N-methyl-D-aspartic acid (NMDA) receptor antagonist (memantine). Currently, an effective treatment that may slow down or stop the damage and death of neurons in AD is lacking.

Complex diseases such as AD require a more intricate treatment (2, 3), and hence, agents acting on only one particular target or modulating only one process are not sufficient. This problem can be overcome by two possible approaches. The first, polypharmacy, involves combination therapies with the use of two or more drugs (4). An alternative way is to use multifunctional compounds that exhibit many effects by interacting with more than one biological target (5, 6). Such molecules, which are referred to as multifunctional or multidirectional compounds, multitargeted ligands (MTLs), or multitarget-directed ligands (MTDLs), provide additive or synergistic effects, called polypharmacology (7).

The issues related to the use of combination therapies for AD have been recently described by Cummings et al. (8). Pharmacodynamic combination therapies are based on two or more symptomatic agents that can improve the behavioral and cognitive symptoms of AD and/or disease-modifying therapies (DMTs) that affect the causes of the disease. In 2014, Namzaric, a combination of two symptomatic medications—the cholinesterase inhibitor, donepezil and the NMDA receptor antagonist, memantine—was approved for the treatment of AD patients with moderate-to-severe dementia (9). Currently, various clinical trials are ongoing on combination therapies involving a standard-of-care medication like a cholinesterase inhibitor or memantine combined with another agent. Selected add-on clinical trials of combination treatment for AD, using a small molecule with a standard-of-care medication, are presented in Table 1.

TABLE 1Selected add-on clinical trials of combination
treatment for AD using a small molecule with a
standard-of-care medication

Agent	Biological target	Phase	Baseline Therapy	Clinical trials
Masitinib	Selective tyrosine kinase inhibitor	III	Rivastigmine and/or memantine	NCT01872598
Telmisartan	Angiotensin II receptor antagonist	II	AChE inhibitor and/or memantine	NCT02085265
Saracatinib	Src/Abl kinase family inhibitor	II	AChE inhibitor and/or memantine	NCT02167256
UE2343	β-Hydroxysteroid dehydrogenase inhibitor	II	AChE inhibitor and/or memantine	NCT02727699
ANAVEX2-73	Sigma-1 chaperone agonist	II	AChE inhibitor and/or memantine	NCT02244541
Neflamapimod	P38 MAPK alfa inhibitor	II	AChE inhibitor or memantine	NCT03402659
CT1812	Antagonist of the sigma-2 receptor	II	AChE inhibitor or memantine	NCT02907567
Idalopirdine	5-HT ₆ receptor antagonist	III	Donepezil or donepezil and memantine	NCT01955161
Citalopram	Selective serotonin reuptake inhibitor	III	AChE inhibitor and/or memantine	NCT00898807
Rasagiline	MAO-B inhibitor	II	AChE inhibitor or memantine	NCT02359552
Piromelatine	Melatonin and serotonin receptor agonist	II	AChE inhibitor or memantine	NCT02615002

AD, Alzheimer's disease; AChE, acetylcholinesterase; MAO, monoamine oxidase; MAPK, mitogen-activated protein kinase.

MULTITARGET LIGANDS APPROACH

Combination therapy has some disadvantages resulting from the possible accumulation of drug side effects, pharmacokinetic complexity, drug–drug interactions, and decreased compliance. Moreover, older patients might feel that using one tablet is more convenient than using two or more. Multifunctional compounds are free of the above-mentioned disadvantages. Since these agents can be considered as a form of multicomponent therapy, the creation of new molecules is based on similar principles. Multifunctional agents can be designed by combing two or more structures of active agents or their pharmacophore fragments that interact with the desired targets. Possible strategies for designing multifunctional agents include combining molecules that can interact with two symptomatic targets or symptomatic targets and disease-modifying targets or with two diseasemodifying targets. The targets of DMT are neurotoxic aggregates of A β and tau protein, as well as various processes associated with neuroprotection or neuroinflammation. The targets of symptomatic therapy are the cholinergic system, including acetylcholinesterase (AChE) and butyrylcholinesterase (BuChE) enzymes, NMDA neurotransmission, G protein-coupled receptors (GPCRs), and monoamine oxidase (MAO) enzymes (Table 2).

A multitarget design approach was successfully developed two decades ago and has been improved in recent years. There are basically three ways to combine pharmacophores to form an MTL structure, which can interact with appropriate biological targets (10). The simplest way is to connect two pharmacophores through a linker. Another way is to combine two pharmacophore fragments without a linker, forming condensed molecules. The third and best way is to create merged pharmacophores that result in small molecules with low molecular weight and thus with good drug-like properties. Over the years, various potential multifunctional anti-AD agents have been designed, synthesized, and developed (11–13). Based on the structures of currently used anti-AD drugs or their

TABLE 2

Therapeutic strategies of AD applied for the creation of multifunctional agents

Therapeutic strategies		
for AD treatment	Target for AD treatment	Agents
Disease modifying therapies	Aβ aggregates	β-Secretase inhibitor Aβ aggregation inhibitor Metal chelator
	Tau aggregates	GSK-3β inhibitor Tau aggregation inhibitor
	Neuroinflammation Neuroprotection Various	COX-2 inhibitor, LOX inhibitor, Antineuroinflammatory agent, S1R modulator, Antioxidant agent, PDE4 inhibitor
Symptomatic therapies	Cholinergic neurotransmission	AChE inhibitor BuChE inhibitor
	NMDA neurotransmission	NMDA receptor antagonist
	GPCRs Serotonin receptor	5-HT ₄ receptor agonist 5-HT ₆ receptor antagonist
	GPCRs Histamine receptor	H ₃ receptor antagonist/inverse agonist
	GPCRs Cannabinoid receptor	CB ₁ receptor antagonist CB ₂ receptor agonist FAAH enzyme
	Oxidative damage	MAO-A inhibitor MAO-B inhibitor

AD, Alzheimer's disease; AChE, acetylcholinesterase; BuChE, butyrylcholinesterase; COX, cyclooxygenase; FAAH, fatty acid amide hydrolase; GPCR, G protein-coupled receptors; GSK, glycogen synthase kinase; LOX, lypoxygenase; MAO, monoamine oxidase; NMDA, N-methyl-D-aspartic acid; PDE4, phosphodiesterase 4.

pharmacophoric fragments, multimodal compounds are produced. Thus, ligands containing fragments of tacrine (as a strong inhibitor of cholinesterases), donepezil, or rivastigmine are the most numerous. Of these, only ladostigil has entered phase II/III clinical trials in patients with mild cognitive impairment. Its structure represents a merged pharmacophore, which is formed by combining a propargylamine moiety derived from an MAO-A/B inhibitor rasagiline with a pharmacophoric carbamyl moiety derived from rivastigmine. Ladostigil is a dual AChE/BuChE and a brain-selective MAO-A/B inhibitor possessing *in vivo* neuroprotective properties (14) and, as such, is an example of a multifunctional agent created by combining two pharmacophores that act on the symptomatic targets of AD.

Among the various multifunctional agents, the largest group that focuses on the targets of symptomatic therapy is cholinesterase inhibitors combined with other targets (15). Dual binding site cholinesterase inhibitors capable of interacting with the catalytic active site and the peripheral anionic binding site of AChE can inhibit A β aggregation. Based on this action, many such ligands were obtained, which showed potency toward both symptomatic targets (AChE/BuChE) and disease-modifying targets (A β aggregation) (16–20). Herein, the recent advances made in the design of multifunctional agents for AD treatment based on molecular targets and selected examples of novel multimodal ligands are presented.

MULTIFUNCTIONAL AGENTS FOCUSING ON DISEASE-MODIFYING TARGETS: Aβ AND OTHERS

The A β hypothesis assumes that the primary cause of neuron loss is the formation of senile plaques due to abnormal processing of amyloid precursor protein (APP) (21). The key role in this process is played by the β -secretase enzyme (BACE-1, A β precursor protein-cleaving enzyme 1), which together with γ -secretase cleaves APP, producing A β peptides consisting of 38–42 amino acids. Because of their fibrillogenic and hydrophobic nature, AB peptides accumulate easily. These peptide aggregates induce oxidative stress, neuroinflammation, hyperphosphorylation, and the aggregation of tau protein, ultimately resulting in loss of neurons and dementia. Based on this theory, some biological disease-modifying targets have been identified and are used in the search for new anti-AD agents. Inhibition of BACE-1 and amyloid aggregation is considered as the most important objective of these agents. Currently, 38 (29%) Aβ-targeting agents are undergoing clinical trials, of which four BACE-1 inhibitors are gaining continued interest as biological targets in the field of drug discovery and development (22). BACE-1 inhibition has also been applied for the generation of MTDLs having other beneficial properties.

BACE-1 inhibitors with other properties

The structure of donepezil is widely used as a pharmacophoric moiety for developing multifunctional agents (19). Examples of compounds formed using the donepezil structure are *N*-benzylpiperidine analogs acting as BACE-1 and AChE inhibitors with antioxidant and antiaggregating properties, which were described by Sharma et al. (23). As a core group, *N*-benzylpiperidine moiety present in donepezil and capable of inhibiting BACE-1 was selected. Virtual screening led to the emergence of the hit compound SEW06622. Based on its structure, a series of new anti-AD agents was designed. The compounds differed in the substitution of benzylamine and the presence of a double bond close to the nitrogen atom. The design of these compounds is presented in Figure 1. Among them, compound **1**



Figure 1. Multifunctional agents focusing on disease-modifying targets: BACE-1, $A\beta$, and others.

was found to be the most promising. Besides inhibiting both BACE-1 and AChE enzymes with IC_{50} values in the submicromolar range, it showed blood–brain barrier (BBB) permeability in the parallel artificial membrane permeability assay (PAMPA) and inhibited self-induced and AChE-induced A β aggregation, while proving to be safe in SH-SY5Y neuroblastoma cell lines at a concentration of up to 80 μ M. Most importantly, it alleviated cognitive impairment in rat models of scopolamine-induced amnesia and by continuous intracerebroventricular infusion of a pathogenic dose of A β peptides. In addition, compound 1 displayed good pharmacokinetic parameters after oral administration. Therefore, it can be considered as a potential drug candidate for AD treatment (23).

The approach to obtain disease-modifying and symptomatic multifunctional ligands inspired the design and synthesis of a group of 1-benzylamino-2-hydroxyalkyl derivatives (24). The new multifunctional agents contained substituted benzylamine, responsible for inhibiting BuChE, connected to aromatic fragments: benzyl, phenyl, and benzhydryl through a 2-hydroxyethyleneamine linker. This linker was suggested to confer the agents with BACE-1 inhibitory activity. Among these ligands, the diphenylpropylamine derivative **2** (Figure 1) was the most interesting, which was capable of inhibiting BACE-1 and BuChE, as well as A β and tau aggregation. Due to its broad biological profile, it is regarded as a potential multifunctional compound (24).

Multifunctional agents combining Aβ antiaggregation effects with other activities

A series of tetrahydroisoquinoline–benzimidazole derivatives was developed, which represent multifunctional anti-AD agents focusing only on the disease-modifying effects (25). To design this group, two well-known pharmacophores, benzimidazole, present in BACE-1 and inflammation inhibitors, and a tetrahydroisoquinoline moiety possessing anti-inflammatory, antioxidation, and neuroprotection properties, were chosen. Benzene or pyridine ring was used as a linker. Among the obtained derivatives, compound **3** showed the most balanced profile (Figure 1). It inhibited nitric oxide (NO) production in lipopolysaccharide (LPS)induced BV2 microglia cells with an IC₅₀ value of 5.07 µM and BACE-1 activity by 65.7% at a concentration of 20 µM. Additionally, it showed a strong neuroprotective effect against glutamate-induced cell death at 5 µM (91.8% cell viability) and reduced the level of proinflammatory cytokines such as TNF- α and IL-1 β . Furthermore, the ability to penetrate the BBB demonstrated that compound **3** is a promising multiple anti-AD ligand.

Suppressing the formation of senile plaques is feasible not only by the inhibition of enzymes involved in the pathological A β cascade but also by direct inhibition of A β aggregation. Kong's group (26) designed compounds that displayed this property as well as other anti-AD activities, based on two structures: donepezil and Trolox, an antioxidative analog of vitamin E. The Trolox fragment was fused with a differently substituted *N*-benzylpiperidine moiety derived from donepezil by an amide bond. The designed compounds were supposed to bind to both active sites of AChE and inhibit AChE, A β aggregation, and MAOs, while also acting as antioxidants. The biological evaluation of the effect of these compounds on all proposed targets revealed that compound 4, which had a fluor atom in position 2 of benzene ring, was a promising MTDL (Figure 1). This compound inhibited both cholinesterases (IC₅₀ hAChE = 0.54 μ M, IC₅₀ hBuChE = 5.97 μ M). Moreover, it displayed nonselective inhibitory activity against MAO-A and MAO-B at a micromolar range. Its antioxidant property compared to Trolox was confirmed in three different experiments. Furthermore, compound 4 exhibited a metal-chelating effect, especially with copper ions, which are associated with AD pathogenesis. The results of self-induced and metal-induced A β aggregation assays also indicated that compound 4 can inhibit both types of A β aggregation. The promising outcomes of the *in vitro* assays were translated successfully to *in vivo* experiments, in which multifunctional ligand 4 was found to significantly improve cognitive decline in scopolamine-, D-galactose-, and AlCl₃-induced memory deficit models. All these results suggested that this compound is an excellent anti-AD candidate.

Another antioxidant and antiaggregating molecule that can serve as a multifunctional compound is the well-known natural derivative resveratrol. For designing new hybrid structures, deferiprone, metal-chelating drug was used (27). These two structures were merged by replacing one of the aromatic rings of resveratrol by deferiprone moiety (Figure 1). Among the newly synthesized hybrids, compound **5** possessing an ethoxy group at position 4 showed triple anti-AD functions—inhibition of A β aggregation, antioxidation, and metal chelation. Such properties make it a potential disease-modifying drug candidate.

MULTIFUNCTIONAL AGENTS FOCUSING ON DISEASE-MODIFYING TARGETS: TAU PROTEIN AND OTHERS

Besides the presence of senile plaques in the brain of patients suffering from AD, the occurrence of NFTs consisting of hyperphosphorylated tau protein aggregates is discerned as the second hallmark of the disease. Based on this observation, the tau hypothesis was formulated to explain the development of AD (28). Tau is a microtubule-associated protein, which, in the physiological condition, stabilizes the microtubules and takes part in axonal transport. During the pathological process, kinases, especially glycogen synthase kinase (GSK-3 β) and GSK-3 α , excessively phosphorylate this protein, resulting in a loss of function. Hyperphosphorylated tau proteins aggregate and easily create intracellular lesions—primarily paired helical filaments and further NFTs. As indicated by the tau theory, the presence of these aggregates contributes to all the processes related to the AD pathomechanism leading to dementia, including A β aggregation. Therefore, in drug discovery and development, the tau-centered approaches, which involve the interaction with kinases, mainly the GSK-3 β enzyme, and the direct inhibition of tau aggregation process, are especially important (29).

Tau and Aβ inhibitors

One of the valuable strategies applied in the search of new anti-AD agents is the dual inhibition of $A\beta$ and tau aggregation for reducing simultaneously the level of both lesions. An example of these agents is 1,2,3,4-tetrahydro-1-acridone

analogs, which were designed by combining the structure of tacrine with quinolone moiety, and the antiaggregating compound cinnamaldehyde containing α , β -unsaturated carbonyl fragment (30). Among the derivatives that differed in the length of linker, degree of saturation, substitution of cinnamaldehyde moiety, and methylation of the nitrogen atom of tacrine, five compounds inhibited A β_{1-42} and tau aggregation by 84.7–99.5% and 71.2–101.8%, respectively, at the screening concentration of 20 µM. All these compounds had a quaternary amine considered crucial for the observed inhibitory activities. Compound **6** with a naphthalene residue (Figure 2), which was identified as a noncovalent inhibitor of both neurotoxic proteins and predicted to penetrate the BBB as well as prevent tau aggregation in living cells, was selected as the most promising dual agent.

Due to its rich biological properties, the curcumin scaffold is highly preferred in the design of multiple antiaggregation agents. It is a natural polyphenol, reported to have antioxidant, antiaggregating, and anti-inflammatory activities, and is used as a yellow spice. Because of its poor pharmacokinetic parameters, especially bioavailability, the scaffold is not applied in the treatment of AD. However, its structure was included in a series of potential multifunctional ligands (31). Compound 7 (PE859) is a perfect example of such ligands (32) (Figure 2), which was found to display higher dual antiaggregating properties compared to the parent compound. Moreover, it penetrated the BBB, improved memory *in vivo*, reduced the amount of both aggregated lesions in mice brains, and showed a promising pharmacokinetic profile.

Dual Aβ and tau antiaggregating properties were also combined with anticholinesterase activity in order to achieve disease-modifying and symptomatic effects. However, it should be noted that such hybrid molecules were designed as dual binding site cholinesterase inhibitors with potential A β antiaggregating properties. The hybrids described by Muñoz-Torreo et al. (33) were formed by the fusion of 6-chlorotacrine, a strong AChE inhibitor, and the previously described tetrahydrobenzonaphthyridine derivatives capable of interacting with both active sites of AChE and displaying weak A β aggregation inhibitory activity. This combination allowed obtaining a group of compounds that can target tau, A β , and cholinesterases. The most potent derivative 8 (Figure 2) inhibited hAChE with an IC₅₀ value of 2.06 nM and hBuChE with 0.286 μ M, as well as A β_{42} and tau aggregation by 77.5% and 68.7%, respectively, in recombinant *Escherichia coli* cells at a concentration of 10 µM. However, a major disadvantage of this compound is its weak drug-likeness. Similarly, a series of shogaol-huprine hybrids displaying antioxidant properties, in addition to anticholinesterase and antiaggregating activities, was developed, which was characterized by poor drug-likeness (34).

GSK-3β inhibitors with other activities

The inhibition of tau protein phosphorylation by interaction with GSK-3 β is an important tau-centered approach. A group of researchers from the University of Bologna (35) developed a series of 2,4-thiazolidinedione derivatives, which were capable of inhibiting not only GSK-3 β but also directly the tau aggregation process. Compound **9** possessing a substituted indole moiety (Figure 2) was the most potent GSK-3 β inhibitor (IC₅₀ = 0.89 µM). It inhibited the AcPHF6 aggregation peptide (306VQIVYK311) up to 80% in a model system at 10 µM concentration.



Figure 2. Structures of small molecules targeting tau protein and other targets.

The BBB permeability of the derivatives revealed by PAMPA, their safety, and ability to inhibit truncated and full-length tau aggregation allow regarding them as tau-centric multitarget agents.

A new disease-modifying strategy that suppressed both amyloid and tau cascades by the inhibition of two enzymes, GSK-3 β and BACE-1, was proposed by Bolognesi's group (36, 37). A series of triazinones was designed based on two fragments: a guanidine motif interacting with BACE-1 and a cyclic amide group binding to GSK-3 β . Among the obtained derivatives, compound **10** (Figure 2) was recognized as a novel well-balanced multiple anti-AD agent (IC₅₀ GSK-3 β = 7.11 µM, IC₅₀ BACE-1 = 16.05 µM) because it showed neuroprotective and neurogenic effects and a good ADME (Adsorption, Distribution, Metabolism, Excretion) profile. Thus, it seems to be a promising lead structure for further development.

The inhibitory activity against GSK-3 β , together with AChE inhibition, was proposed as a new strategy by Chinese researchers (38). Figure 2 presents the most interesting example (compound **11**) for simple combination of tacrine scaffold with the structure of a selective GSK-3 β inhibitor using an alkyl linker. Among the ligands developed by the above team, compound **11** exhibited a well-balanced biological profile with the IC₅₀ values in the nanomolar range (IC₅₀ hAChE = 6.5 nM, IC₅₀ hGSK-3 β = 66 nM). The efficacy of this dual strategy was verified in the animal model where compound **11** ameliorated the cognitive decline. Moreover, the examined compound did not demonstrate any hepatotoxicity for tacrine.

Oukoloff et al. (39) also used the same AChE inhibitor fragment for creating hybrids, which showed a similar mechanism of action. They connected tacrine pharmacophore by a linker containing 1,2,3-triazole with the structure of the GSK- $3\alpha/\beta$ inhibitor valmerin. The *in vitro* results revealed that compound **12** in the form of an *R* enantiomer (Figure 2) was the most potent, which exhibited inhibitory potential with an IC₅₀ value of 9.5 and 7 nM for AChE and GSK- $3\alpha/\beta$, respectively. Additional advantages of the developed novel tacrine–valmerin multifunctional ligands were low cytotoxicity and good BBB permeability.

The metal hypothesis of AD indicates that excess levels and dysregulation of biometal ions, such as Cu²⁺, Fe²⁺, Zn²⁺, and Ca²⁺, in the brain cause A β aggregation, generation of reactive oxygen species (ROS), and oxidative stress, leading to cell death (40). Metal chelators are potential anti-AD agents and are also used in the development of multifunctional agents (41). A novel approach based on combing GSK-3 β inhibitors with agents exhibiting metal-chelating and A β -antiaggregating properties was recently presented. The newly formed compounds included 2,3-diaminopyridine derivatives (42) and hydroxy-substituted *trans*-cinnamoyl derivatives (43). The most promising representative compound **13** from the first group (Figure 2) inhibited the GSK-3 β enzyme with an IC₅₀ value of 49 nM and acted as a selective Cu²⁺ and Al³⁺ chelator, showing antioxidant and A β -antiaggregating properties. On the other hand, compound **14** from the second series (Figure 2) exhibited lower inhibitory potency against GSK-3 β (IC₅₀ = 24.36 µM), but displayed a broad profile of anti-AD activities.

In turn, two series of compounds, benzoxazinone and indole derivatives, were designed as dual kinase inhibitors (44). Besides inhibiting GSK-3 β , they targeted human adenosine kinase (hAK), which also induces the development of AD by

regulating the level of adenosine and cytoprotective effects. Based on three wellknown *h*AK/*h*GSK-3 β inhibitors, the structures of novel agents were designed and developed. Among them, **15** (Figure 2) displayed dual kinase inhibitory activity (IC₅₀ *h*AK = 13.6 μ M, IC₅₀ *h*GSK-3 β = 6.4 μ M) with antioxidant and neuroprotective properties, and thus identified as an appropriate lead structure for the development of multifunctional ligands that can exhibit an innovative mechanism of action.

MULTIFUNCTIONAL AGENTS FOCUSING ON SYMPTOMATIC THERAPIES WITH ADDITIONAL BENEFICIAL PROPERTIES

In the course of Alzheimer's disease, it is important to both treat the causes and symptoms of the disease. Dementia changes in patients are also associated with other disease symptoms, such as depression, anxiety, psychosis, and personality changes. According to the oldest theory of AD, dementia changes and memory disturbances are associated with damage to cholinergic neurotransmission. Other symptoms result from disturbances in other neurotransmitter systems, including GPCRs, such as serotonin, H₃ histamine receptors, and the cannabinoid system or with impaired functioning of enzymatic systems such as MAOs. Hence, the design of multifunctional molecules aimed at the treatment of disease symptoms is based on combining the inhibitory activity of cholinesterase with the influence on the other molecular targets mentioned earlier.

Multifunctional ligands influencing serotonin and cholinergic neurotransmission

Among all GPCRs, serotoninergic receptors are the most attractive as they are targeted for designing MTDLs as potential treatment of AD. Their particularly desired actions are the activation of $5\text{-}HT_4$ receptors ($5\text{-}HT_4\text{Rs}$) and inhibition of $5\text{-}HT_6$ receptors ($5\text{-}HT_6\text{Rs}$) (45). The first action leads to the modulation of APP metabolism and redirects the protein to a nonamyloid pathway. In turn, $5\text{-}HT_6$ blockage improves cognitive performance by increasing the release of glutamate, acetylcholine, and catecholamine in the cortical and limbic areas.

The first class of multifunctional ligands formed by combining 5-HT₄Rs with cholinesterases was presented by Dallemagne's research team (46). They described a compound named donecopride (compound **16**, Figure 3) as a new preclinical multifunctional drug candidate showing both *in vitro* AChE inhibitory activity and 5-HT₄R agonistic potency. Moreover, donecopride also displayed *in vivo* procognitive and antiamnestic activities in mice. The same research group (47) described novel MTLs with an interesting biological profile, which were capable of restoring cholinergic transmission through the activation of 5-HT₄R, blockage of 5-HT₆R, and inhibition of AChE. This class of compounds was developed by merging the previously developed dual compound with the benzyl analog of donecopride displaying *in vitro* 5-HT₄R agonist and 5-HT₆R antagonist properties. The most interesting compound was **17** (Figure 3), a fumaric acid salt exhibiting well-balanced activities toward all the three mentioned targets. Compound **17** was found to act



Figure 3. Multifunctional ligands influencing serotonin or histaminergic neurotransmission and cholinesterases.

as a partial agonist of h5-HT₄R (Ki = 210 nM), an inverse agonist of h5-HT₆R (Ki = 219 nM), and an inhibitor of hAChE (IC₅₀ = 33.7 nM). Additionally, the compound was tested *in vivo* in a model of scopolamine-induced working memory deficit, where it displayed antiamnestic effects at a dose of 0.3 mg/kg.

Several multifunctional ligands based on the combination of pharmacophores that are dedicated to $5\text{-HT}_6\text{R}$ blockage and cholinesterase inhibition were developed by Więckowska et al. (48, 49). These novel hybrid ligands were obtained by combining the pharmacophores directed against $5\text{-HT}_6\text{R}$ (1-(phenylsulfonyl)-4-(piperazin-1-yl)-1*H*-indole) and cholinesterases (tacrine or *N*-benzylpiperidine analogs). Among them, compound **18** (Figure 3) was the most interesting as it displayed potent and balanced antagonist activity toward $5\text{-HT}_6\text{R}$ (Ki = 27 nM) and inhibitory effect against both cholinesterases (IC₅₀ AChE = 12 nM, IC₅₀

BuChE = 29 nM). The compound also showed good *in vitro* BBB permeability (proved by PAMPA). Additionally, an *in vivo* study with the use of a scopolamineinduced hyperlocomotion rat model confirmed the central cholinomimetic activity of the compound (48). The further development of this series of hybrids resulted in multifunctional ligands with A β antiaggregation properties (49).

Indole-based multifunctional ligands were designed, synthesized, and evaluated for the treatment of AD as 5-HT₆R blockers with inhibitory abilities toward *eq*BuChE and antioxidant properties (50). Based on the biological screening of the 5-HT₆R antagonists against BuChE, two compounds were selected. In order to improve their BuChE inhibitory activity, structural modifications were introduced in the next stage in the two series of compounds. Of them, ligand **19** (Figure 3) displayed beneficial, dual activity targeting 5-HT₆R (K*i* = 41.8 nM) and *eq*BuChE (IC₅₀ = 5.07 µM), in addition to favorable antioxidant properties that were comparable with the reference ascorbic acid.

Multifunctional ligands combining H₃R antagonism and cholinesterase inhibition

Histamine H_3 receptor (H_3R) belongs to the GPCRs family. In the central nervous system (CNS), H_3R acts as a presynaptic autoreceptor and is involved in inhibiting the release of histamine and the modulation of other neurotransmitters such as acetylcholine. Numerous *in vivo* studies have proven that both antagonist and inverse agonists of H_3R improve cognitive deficits, memory, and spatial orientation (51). Based on their results, H_3R is frequently chosen as a biological target for anti-AD multifunctional ligands (52–54).

An international multiteam group discovered new multifunctional ligands with a broad and well-balanced spectrum of activities against H₃R, AChE, BuChE, and MAO-A/B (55, 56). These new indole derivatives were designed by combining the neuroprotectant ASS234 with cholinesterase- and MAO-inhibiting motifs and ciproxifan containing H₃R-blocking and MAO-inhibiting pharmacophore fragments (Figure 3). Among the newly formed hybrids, contilisant was identified as the most promising multifunctional agent. It inhibited hMAO-A (IC₅₀ = 1.85μ M, after 30 min of preincubation IC₅₀ = 0.145 μ M), hMAO-B (IC₅₀ = 1.94 μ M, after 30 min of preincubation IC₅₀ = 0.078 μ M), hAChE (IC₅₀ = 0.530 μ M), and hBuChE (IC₅₀ = 1.690 μ M), as well as blocked H₃R (Ki = 10.8 nM). Moreover, contilisant demonstrated in vitro antioxidant neuroprotective effects and the ability to penetrate the BBB in PAMPA. Due to its original in vitro biological profile, contilisant was selected for in vivo studies and tested using the novel object recognition test in mice with LPS-induced cognitive deficit. Based on the excellent in vitro data supported by positive in vivo activity, contilisant was studied further with the aim of exploring new pharmacological properties that may be potentially beneficial for AD therapy (57). The studies revealed another valuable activity of the compound: selective agonistic effect on Sigma 1 receptor (S1R) (Ki = 65.2 nM). S1R is associated with learning and memory processes, and hence, its agonists are used as antiamnestic agents in a variety of pharmacological models, probably due to the improvement of glutamatergic and cholinergic neurotransmissions. Additional in vivo studies including Y-maze and radical arm-maze tasks in mice with cognitive impairment induced by $A\beta_{1-42}$ oligomers showed that contilisant exhibited higher activity in comparison with the

commonly used anti-AD drug donepezil. Due to its multifunctional profile as well as *in vitro* activities, which were reflected in *in vivo* tests, contilisant thus seems to be an interesting multifunctional ligand for further development as a disease-modifying anti-AD agent (57).

A number of anti-AD multifunctional ligands have been developed based on naturally occurring substances. Wang et al. (58) synthesized a series of multifunctional ligands based on isoflavone, which was recently proven to exhibit inhibitory properties toward cholinesterases. *In vitro* studies revealed that some of these ligands showed a multifunctional profile including blockage of H₃R, neuroprotective effect, and antineuroinflammatory properties. Among them, compound **20** (Figure 3) is noteworthy since it displayed moderate H₃R antagonist property (K*i* = 270 nM) and potently inhibited AChE (IC₅₀ = 80 nM). In addition, the compound **20** demonstrated neuroprotective and antineuroinflammatory properties. The *in vivo* study also confirmed that it did not induce acute toxicity even at high doses (1000 mg/kg), but penetrated through the BBB into the CNS and caused significant improvements in mice with scopolamine-induced cognitive deficit, which were revealed by passive avoidance test (58).

Ismaili et al. (59) described new small molecules combining activities against three biological targets including hH_3R . The most promising lead showed high affinity toward hH_3R (Ki = 0.565 µM), Ca²⁺ channel blockade activity (IC₅₀ = 21 µM), and moderate selective *h*BuChE inhibition (IC₅₀ = 7.83 µM), besides strong antioxidant properties and ability to restore cognitive impairment induced by LPS.

Multifunctional ligands targeting endocannabinoid system and cholinesterases

Since the discovery that the activation of cannabinoid receptors (CB₁R, CB₂R) causes a reduction in the production of neurotoxic factors (ROS, proinflammatory mediators) leading to decreased neuroinflammatory processes, the receptors were considered as another biological target in the search for anti-AD agents as well as multipotent ligands (60, 61). Consequently, Decker's group (62) developed a series of benzimidazole-based dual-acting ligands that can activate CB₂R and inhibit BuChE. Among them, the authors highlighted compound **21** as the most promising hybrid (Figure 4) as it activated hCB₂R with an IC₅₀ value of 0.763 μ M and hBuChE with an IC₅₀ value of 1.6 μ M. It is worth noting that this compound **21** was selective over hCB₁R and AChE. Compound **21** was further evaluated *in vivo* which revealed that it improved cognitive functions in mice showing neuroinflammation and cognition deficits after A β_{25-35} administration at doses ranging from 1 to 3 mg/kg.

In turn, Montanari et al. (63) synthesized multifunctional ligands that indirectly enhanced endocannabinoid signaling by inhibiting fatty acid amide hydrolase (FAAH), an enzyme responsible for the degradation of crucial endocannabinoid signaling molecules: *N*-arachidonoylethanoloamine and 2-arachidonoyl glycerol. In addition to FAAH, the compounds inhibited cholinesterases. Some of the compounds exhibited very high potencies toward single targets; however, the most interesting MTL was compound **22** (Figure 4). It potently inhibited FAAH (IC₅₀ = 157.2 nM, after preincubation IC₅₀ = 27.9 nM), hAChE (IC₅₀ = 922 nM), and hBuChE (IC₅₀ = 42.7 nM).

Enzyme inhibitors—MAO-A/B and cholinoesterases

MAOs (MAO-A and MAO-B) are the enzymes responsible for the degradation and inactivation of monoamine neurotransmitters. The inhibition of MAOs leads to neuroprotective effects, not only due to an increase in monoaminergic neurotransmission but also due to limitations in the production of neurotoxic substances, which are by-products of a reaction catalyzed by these enzymes. Due to its multiactive nature, the chromone scaffold is commonly applied as a structural motive in the development of multimodal ligands. Based on its structure, Reis et al. (64) reported a series of dual MAO and cholinesterase inhibitors. Among the tested compounds, the most promising was found to be compound 23 (Figure 4), which selectively inhibited hAChE (IC₅₀ = 210 nM) over BuChE and nonselectively inhibited both the isoforms of MAO (IC_{50} MAO-A = 0.94 μ M, IC_{50} MAO-B = 3.81 μ M). Moreover, the additional pharmacokinetic and toxicological studies of compound 23 showed its lack of cytotoxicity at a concentration below 25 µM, and PAMPA predicted its penetration through the BBB. Thus, the established well-balanced activities, low toxicity, and predicted permeability of the tested compound make them interesting multifunctional ligands for AD therapy.

Another group of dual inhibitors of MAOs and cholinesterases was developed by connecting *N*-benzylpiperidine fragment derived from donepezil with di-*tert*-butylated hydroxytoluene, which is a fragment responsible for antioxidant, anticancer, and anti-inflammatory properties in numerous compounds (65). These inhibitors were analyzed by studies divided into two parts: the first aimed to find an appropriate linker to connect the two pharmacophores and the second aimed to decorate the aromatic ring of the benzylpiperidine scaffold. Among the tested compounds, **24** was recognized as a double AChE (IC₅₀ *ee*AChE = 75 nM, IC₅₀ hAChE = 750 nM) and MAO-B (IC₅₀ = 7.5 μ M) inhibitor with noticeable antioxidant properties and ability to reduce both self-induced and hAChE-induced Aβ aggregation (65).

MULTIFUNCTIONAL AGENTS FOCUSING ON VARIOUS DISEASE-MODIFYING AND SYMPTOMATIC THERAPIES

The complexity of AD etiopathogenesis forced a search for new biological targets in the CNS and even beyond. Among the numerous molecular targets discovered, some have been used to create multifunctional ligands.

In pathological conditions, S1R modulates regulatory proteins, restores calcium homeostasis, and controls the production of ROS, thereby contributing to the overall neuroprotection effect (66). Based on their previous study, Rui et al. (67) developed multifunctional agents capable of modulating S1R and inhibiting AChE. The chemical structure of the new multifunctional ligands was developed through an interesting combination of RRC-33 (S1R agonist), curcumin (having antioxidant properties), and donepezil. The newly developed compounds exhibited a high affinity to S1R, but very weak AChE inhibitory properties. Among them, compound **25** (Figure 4) showed binding affinity to S1R with a Ki value of 15 nM; however, it inhibited AChE only by 64.80% at a concentration of 50 µM. Its neuroprotective effect associated with S1R modulation was confirmed by the neurite outgrowth observed in the dorsal root ganglia in the *in vitro* model (67).



Figure 4. Multifunctional agents focusing on various targets.

Interesting biological activities were also presented by flurbiprofen–clioquinol hybrids (68). Flurbiprofen is a known potent nonsteroidal anti-inflammatory drug, which also inhibits platelet aggregation and A β aggregation and reduces tau phosphorylation. On the other hand, clioquinol is a metal chelator with proven antioxidant properties. Among the obtained hybrids, compound **26** (Figure 4) was able to inhibit both self-induced and Cu²⁺-induced A β aggregation and MAO-B. In addition, it presented biometal-chelating abilities as well as antioxidant and antineuroinflammatory activities and appropriate BBB permeability.

In turn, Hu et al. (69) developed a novel group of multifunctional compounds by combining clioquinol with rolipram or roflumilast. These compounds inhibited phosphodiesterase 4D (PDE4D), an enzyme participating in the process of memory consolidation and long-term potentiation. The most interesting of them was compound **27** (Figure 4), which inhibited not only the PDE4D with an IC_{50} value of 0.399 µM but also the metal-induced aggregation of A β , chelated metal ions, and exhibited antioxidant properties (ORAC [oxygen radical absorbance capacity] = 0.60 Trolox eq.). In addition to *in vitro* activity, compound **27** showed *in vivo* activity by demonstrating procognitive effects in the Morris water-maze test.

Another example of multifunctional compounds is hybrids formed by the combination of pharmacophore cyclooxygenase-2 and 15-lipoxygenase enzymes with the structure of tacrine (70). These compounds are endowed with antineuroinflammatory and anticholinesterase activities.

CONCLUSION

The lack of an effective treatment or DMT capable of influencing the causes of AD has led to a constant need to search for new solutions and drugs. Undoubtedly, an important issue here is the identification of new disease biomarkers, which can indicate early enough the pathological changes in the brain that may lead to irreversible processes and symptoms over time. The search for new anti-AD agents is challenging due to the complexity of the disease and the corresponding lack of appropriate animal models useful in preclinical studies. In addition, there is a need for properly planned clinical trials to demonstrate the effectiveness of the drugs. These problems concern small molecules targeting single biological target as well as combination treatment and multifunctional compounds. The clinical trials of combination therapy for AD focus on combining a cholinesterase inhibitor or memantine with other medications having various therapeutic indications. In this context, multifunctional ligands seem to be a much better strategy as they combine effects aimed at both causal and symptomatic treatment of the disease.

The examples presented in this chapter show that the search for new anti-AD drugs is based on the combination of pharmacophores acting not only on cholinesterases and A β aggregation but also on tau protein aggregation (GSK-3 β inhibitors), neuroinflammation, and antioxidation. The ligands that regulate the influence on GPCRs or MAOs are particularly interesting, taking into account that the symptoms of depression or other mental disorders often coexist in AD. This may indicate that the proper design of a multifunctional molecule can facilitate the discovery of an effective therapy for this devastating disease.

Acknowledgment: This work was supported by the National Science Center of Poland (grant UMO-2016/21/B/NZ7/01744).

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship, and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced. Where relevant, appropriate permissions have been obtained from the original copyright holder(s).

REFERENCES

- Querfurth HW, LaFerla FM. Alzheimer's disease. N Engl J Med. 2010;362(4):329–44. http://dx.doi. org/10.1056/NEJMra0909142
- Carreiras M, Mendes E, Perry M, Francisco A, Marco-Contelles J. The multifactorial nature of Alzheimer's disease for developing potential therapeutics. Curr Top Med Chem. 2014;13(15): 1745–70. http://dx.doi.org/10.2174/15680266113139990135
- 3. Morsy A, Trippier PC. Current and emerging pharmacological targets for the treatment of Alzheimer's disease. J Alzheimers Dis. 2019;72(Suppl 1):S145–76. http://dx.doi.org/10.3233/JAD-190744
- Peters J-U. Polypharmacology—Foe or friend? J Med Chem. 2013;56(22):8955–71. http://dx.doi. org/10.1021/jm400856t
- 5. Morphy R, Rankovic Z. Designing multiple ligands—Medicinal chemistry strategies and challenges. Curr Pharm Des. 2009;15(6):587–600. http://dx.doi.org/10.2174/138161209787315594
- Bansal Y, Silakari O. Multifunctional compounds: Smart molecules for multifactorial diseases. Eur J Med Chem. 2014;76:31–42. http://dx.doi.org/10.1016/j.ejmech.2014.01.060
- Proschak E, Stark H, Merk D. Polypharmacology by design: A medicinal chemist's perspective on multitargeting compounds. J Med Chem. 2019;62(2):420–44. http://dx.doi.org/10.1021/acs. jmedchem.8b00760
- Cummings JL, Tong G, Ballard C. Treatment combinations for Alzheimer's disease: Current and future pharmacotherapy options. J Alzheimers Dis. 2019;67(3):779–94. http://dx.doi.org/10.3233/ JAD-180766
- Deardorff WJ, Grossberg G. A fixed-dose combination of memantine extended-release and donepezil in the treatment of moderate-to-severe Alzheimer's disease. Drug Des Devel Ther. 2016;10:3267–79. http://dx.doi.org/10.2147/DDDT.S86463
- Morphy R, Rankovic Z. Designed multiple ligands. An emerging drug discovery paradigm. J Med Chem. 2005;48(21):6523–43. http://dx.doi.org/10.1021/jm058225d
- Prati F, Cavalli A, Bolognesi ML. Navigating the chemical space of multitarget-directed ligands: From hybrids to fragments in Alzheimer's disease. Molecules. 2016;21(4):466. http://dx.doi.org/10.3390/ molecules21040466
- 12. Oset-Gasque MJ, Marco-Contelles J. Alzheimer's disease, the "one-molecule, one-target" paradigm, and the multitarget directed ligand approach. ACS Chem Neurosci. 2018;9(3):401–3. http://dx.doi. org/10.1021/acschemneuro.8b00069
- Benek O, Korabecny J, Soukup O. A perspective on multi-target drugs for Alzheimer's disease. Trends Pharmacol Sci. 2020;41:1–12. http://dx.doi.org/10.1016/j.tips.2020.04.008
- Weinreb O, Amit T, Bar-Am O, Youdim MB. Ladostigil: A novel multimodal neuroprotective drug with cholinesterase and brain-selective monoamine oxidase inhibitory activities for Alzheimer's disease treatment. Curr Drug Targets. 2012;13(4):483–94. http://dx.doi.org/ 10.2174/138945012799499794
- 15. Zhang P, Xu S, Zhu Z, Xu J. Multi-target design strategies for the improved treatment of Alzheimer's disease. Eur J Med Chem. 2019;176:228–47. http://dx.doi.org/10.1016/j.ejmech.2019.05.020
- Viayna E, Sabate R, Muñoz-Torrero D. Dual inhibitors of β-amyloid aggregation and acetylcholinesterase as multi-target anti-Alzheimer drug candidates. Curr Top Med Chem. 2013;13(15):1820–42. http://dx.doi.org/10.2174/15680266113139990139
- Guzior N, Wieckowska A, Panek D, Malawska B. Recent development of multifunctional agents as potential drug candidates for the treatment of Alzheimer's disease. Curr Med Chem. 2015;22(3): 373–404. http://dx.doi.org/10.2174/0929867321666141106122628
- Ismaili L, Refouvelet B, Benchekroun M, Brogi S, Brindisi M, Gemma S, et al. Multitarget compounds bearing tacrine- and donepezil-like structural and functional motifs for the potential treatment of Alzheimer's disease. Prog Neurobiol. 2016;151:4–34. http://dx.doi.org/10.1016/j.pneurobio.2015.12.003
- Li Q, He S, Chen Y, Feng F, Qu W, Sun H. Donepezil-based multi-functional cholinesterase inhibitors for treatment of Alzheimer's disease. Eur J Med Chem. 2018;158:463–77. http://dx.doi.org/10.1016/j. ejmech.2018.09.031

- Panek D, Wichur T, Godyń J, Pasieka A, Malawska B. Advances toward multifunctional cholinesterase and β-amyloid aggregation inhibitors. Future Med Chem. 2017;9:1835–54. http://dx.doi. org/10.4155/fmc-2017-0094
- Selkoe DJ, Hardy J. The amyloid hypothesis of Alzheimer's disease at 25 years. EMBO Mol Med. 2016;8(6):595–608. http://dx.doi.org/10.15252/emmm.201606210
- Cummings J, Lee G, Ritter A, Sabbagh M, Zhong K. Alzheimer's disease drug development pipeline: 2019. Alzheimers Dement Transl Res Clin Interv. 2019;5:272–93. http://dx.doi.org/10.1016/j. trci.2019.05.008
- Sharma P, Tripathi A, Tripathi PN, Prajapati SK, Seth A, Tripathi MK, et al. Design and development of multitarget-directed N-Benzylpiperidine analogs as potential candidates for the treatment of Alzheimer's disease. Eur J Med Chem. 2019;167:510–24. http://dx.doi.org/10.1016/j.ejmech.2019.02.030
- Panek D, Więckowska A, Jończyk J, Godyń J, Bajda M, Wichur T, et al. Design, synthesis, and biological evaluation of 1-benzylamino-2-hydroxyalkyl derivatives as new potential disease-modifying multifunctional anti-Alzheimer's agents. ACS Chem Neurosci. 2018;9(5):1074–94. http://dx.doi. org/10.1021/acschemneuro.7b00461
- Fang Y, Zhou H, Gu Q, Xu J. Synthesis and evaluation of tetrahydroisoquinoline-benzimidazole hybrids as multifunctional agents for the treatment of Alzheimer's disease. Eur J Med Chem. 2019;167:133–45. http://dx.doi.org/10.1016/j.ejmech.2019.02.008
- Cai P, Fang SQ, Yang XL, Wu JJ, Liu QH, Hong H, et al. Rational design and multibiological profiling of novel donepezil-trolox hybrids against Alzheimer's disease, with cholinergic, antioxidant, neuroprotective, and cognition enhancing properties. ACS Chem Neurosci. 2017;8(11):2496–511. http:// dx.doi.org/10.1021/acschemneuro.7b00257
- Xu P, Zhang M, Sheng R, Ma Y. Synthesis and biological evaluation of deferiprone-resveratrol hybrids as antioxidants, Aβ₁₋₄₂ aggregation inhibitors and metal-chelating agents for Alzheimer's disease. Eur J Med Chem. 2017;127:174–86. http://dx.doi.org/10.1016/j.ejmech.2016.12.045
- Naseri NN, Wang H, Guo J, Sharma M, Luo W. The complexity of tau in Alzheimer's disease. Neurosci Lett. 2019;705:183–94. http://dx.doi.org/10.1016/j.neulet.2019.04.022
- Congdon EE, Sigurdsson EM. Tau-targeting therapies for Alzheimer disease. Nat Rev Neurol. 2018; 14(7):399–415. http://dx.doi.org/10.1038/s41582-018-0013-z
- Lv P, Xia CL, Wang N, Liu ZQ, Huang ZS, Huang SL. Synthesis and evaluation of 1,2,3,4-tetrahydro-1-acridone analogues as potential dual inhibitors for amyloid-beta and tau aggregation. Bioorg Med Chem. 2018;26(16):4693–705. http://dx.doi.org/10.1016/j.bmc.2018.08.007
- 31. Okuda M, Hijikuro I, Fujita Y, Teruya T, Kawakami H, Takahashi T, et al. Design and synthesis of curcumin derivatives as tau and amyloid β dual aggregation inhibitors. Bioorg Med Chem Lett. 2016;26(20):5024–8. http://dx.doi.org/10.1016/j.bmcl.2016.08.092
- 32. Okuda M, Fujita Y, Hijikuro I, Wada M, Uemura T, Kobayashi Y, et al. PE859, a novel curcumin derivative, inhibits amyloid-β and tau aggregation, and ameliorates cognitive dysfunction in senescence-accelerated mouse prone 8. J Alzheimers Dis. 2017;59(1):313–28. http://dx.doi.org/10.3233/JAD-161017
- 33. Muñoz-Torreo D, Di Pietro O, Pérez-Areales FJ, Juárez-Jiménez J, Espargaró A, Clos MV, Pérez B, et al. Tetrahydrobenzo[h][1,6]naphthyridine-6-chlorotacrine hybrids as a new family of anti-Alzheimer agents targeting β-amyloid, tau, and cholinesterase pathologies. Eur J Med Chem. 2014;84:107–17. http://dx.doi.org/10.1016/j.ejmech.2014.07.021
- 34. Pérez-Areales FJ, Di Pietro O, Espargaró A, Vallverdú-Queralt A, Galdeano C, Ragusa IM, et al. Shogaol-huprine hybrids: Dual antioxidant and anticholinesterase agents with β-amyloid and tau anti-aggregating properties. Bioorg Med Chem. 2014;22(19):5298–307. http://dx.doi.org/10.1016/j. bmc.2014.07.053
- 35. Gandini A, Bartolini M, Tedesco D, Martinez-Gonzalez L, Roca C, Campillo NE, et al. Tau-centric multitarget approach for Alzheimer's disease: Development of first-in-class dual glycogen synthase kinase 3β and tau-aggregation inhibitors. J Med Chem. 2018;61(17):7640–56. http://dx.doi.org/10.1021/ acs.jmedchem.8b00610
- Prati F, De Simone A, Bisignano P, Armirotti A, Summa M, Pizzirani D, et al. Multitarget drug discovery for Alzheimer's disease: Triazinones as BACE-1 and GSK-3β inhibitors. Angew Chem Int Ed. 2015;54(5):1578–82. http://dx.doi.org/10.1002/anie.201410456

- Prati F, De Simone A, Armirotti A, Summa M, Pizzirani D, Scarpelli R, et al. 3,4-Dihydro-1,3,5-triazin-2(1H)-ones as the first dual BACE-1/GSK-3β fragment hits against Alzheimer's disease. ACS Chem Neurosci. 2015;6(10):1665–82. http://dx.doi.org/10.1021/acschemneuro.5b00121
- 38. Jiang XY, Chen TK, Zhou JT, He SY, Yang HY, Chen Y, et al. Dual GSK-3β/AChE inhibitors as a new strategy for multitargeting anti-Alzheimer's disease drug discovery. ACS Med Chem Lett. 2018;9(3):171–6. http://dx.doi.org/10.1021/acsmedchemlett.7b00463
- Oukoloff K, Coquelle N, Bartolini M, Naldi M, Le Guevel R, Bach S, et al. Design, biological evaluation and X-ray crystallography of nanomolar multifunctional ligands targeting simultaneously acetylcholinesterase and glycogen synthase kinase-3. Eur J Med Chem. 2019;168:58–77. http://dx.doi. org/10.1016/j.ejmech.2018.12.063
- Kepp KP. Bioinorganic chemistry of Alzheimer's disease. Chem Rev. 2012;112(10):5193–239. http:// dx.doi.org/10.1021/cr300009x
- 41. Savelieff MG, Nam G, Kang J, Lee HJ, Lee M, Lim MH. Development of multifunctional molecules as potential therapeutic candidates for Alzheimer's disease, Parkinson's disease, and amyotrophic lateral sclerosis in the last decade. Chem Rev. 2018;119(2):1221–322. http://dx.doi.org/10.1021/acs. chemrev.8b00138
- Shi XL, Wu J-D, Liu P, Liu ZP. Synthesis and evaluation of novel GSK-3β inhibitors as multifunctional agents against Alzheimer's disease. Eur J Med Chem. 2019;167:211–25. http://dx.doi.org/10.1016/j. ejmech.2019.02.001
- 43. De Simone A, Bartolini M, Baschieri A, Apperley KYP, Chen HH, Guardigni M, et al. Hydroxysubstituted trans-cinnamoyl derivatives as multifunctional tools in the context of Alzheimer's disease. Eur J Med Chem. 2017;139:378–89. http://dx.doi.org/10.1016/j.ejmech.2017.07.058
- 44. Brogi S, Ramunno A, Savi L, Chemi G, Alfano G, Pecorelli A, et al. First dual AK/GSK-3β inhibitors endowed with antioxidant properties as multifunctional, potential neuroprotective agents. Eur J Med Chem. 2017;138:438–57. http://dx.doi.org/10.1016/j.ejmech.2017.06.017
- Claeysen S, Bockaert J, Giannoni P. Serotonin: A new hope in Alzheimer's disease? ACS Chem Neurosci. 2015;6(7):940–3. http://dx.doi.org/10.1021/acschemneuro.5b00135
- 46. Rochais C, Lecoutey C, Gaven F, Giannoni P, Hamidouche K, Hedou D, et al. Novel multitargetdirected ligands (MTDLs) with acetylcholinesterase (AChE) inhibitory and serotonergic subtype 4 receptor (5-HT4R) agonist activities as potential agents against Alzheimer's disease: The design of donecopride. J Med Chem. 2015;58(7):3172–87. http://dx.doi.org/10.1021/acs.jmedchem.5b00115
- 47. Hatat B, Yahiaoui S, Lecoutey C, Davis A, Freret T, Boulouard M, et al. A novel in vivo anti-amnesic agent, specially designed to express both acetylcholinesterase (AChE) inhibitory, serotonergic sub-type 4 receptor (5-HT4R) agonist and serotonergic subtype 6 receptor (5-HT6R) inverse agonist activities, with a potential inter. Front Aging Neurosci. 2019;11:1–14. http://dx.doi.org/10.3389/ fnagi.2019.00148
- Więckowska A, Kołaczkowski M, Bucki A, Godyń J, Marcinkowska M, Więckowski K, et al. Novel multi-target-directed ligands for Alzheimer's disease: Combining cholinesterase inhibitors and 5-HT6 receptor antagonists. Design, synthesis and biological evaluation. Eur J Med Chem. 2016;124:63–81. http://dx.doi.org/10.1016/j.ejmech.2016.08.016
- Więckowska A, Wichur T, Godyń J, Bucki A, Marcinkowska M, Siwek A, et al. Novel multitarget-directed ligands aiming at symptoms and causes of Alzheimer's disease. ACS Chem Neurosci. 2018;9(5):1195–214. http://dx.doi.org/10.1021/acschemneuro.8b00024
- Marcinkowska M, Bucki A, Panek D, Siwek A, Fajkis N, Bednarski M, et al. Anti-Alzheimer's multitarget-directed ligands with serotonin 5-HT6 antagonist, butyrylcholinesterase inhibitory, and antioxidant activity. Arch Pharm (Weinheim). 2019;352(7):e1900041. http://dx.doi.org/10.1002/ ardp.201900041
- Sadek B, Saad A, Sadeq A, Jalal F, Stark H. Histamine H3 receptor as a potential target for cognitive symptoms in neuropsychiatric diseases. Behav Brain Res. 2016;312:415–30. http://dx.doi. org/10.1016/j.bbr.2016.06.051
- 52. Darras FH, Pockes S, Huang G, Wehle S, Strasser A, Wittmann HJ, et al. Synthesis, biological evaluation, and computational studies of Tri- and tetracyclic nitrogen-bridgehead compounds as potent dual-acting AChE inhibitors and h H3 receptor antagonists. ACS Chem Neurosci. 2014;5(3):225–42. http://dx.doi.org/10.1021/cn4002126

- Łażewska D, Jończyk J, Bajda M, Szałaj N, Więckowska A, Panek D, et al. Cholinesterase inhibitory activity of chlorophenoxy derivatives—Histamine H3 receptor ligands. Bioorg Med Chem Lett. 2016;26(16):4140–5. http://dx.doi.org/10.1016/j.bmcl.2016.04.054
- Bajda M, Łażewska D, Godyń J, Zaręba P, Kuder K, Hagenow S, et al. Search for new multi-target compounds against Alzheimer's disease among histamine H3 receptor ligands. Eur J Med Chem. 2020;185:111785. http://dx.doi.org/10.1016/j.ejmech.2019.111785
- 55. Bautista-Aguilera ÓM, Hagenow S, Palomino-Antolin A, Farré-Alins V, Ismaili L, Joffrin PL, et al. Multitarget-directed ligands combining cholinesterase and monoamine oxidase inhibition with histamine H₃R antagonism for neurodegenerative diseases. Angew Chemie Int Ed Engl. 2017;56(41):12765–9. http://dx.doi.org/10.1002/anie.201706072
- Romero A, Marco-Contelles J, Ramos E. Highlights of ASS234: A novel and promising therapeutic agent for Alzheimer's disease therapy. Neural Regen Res. 2020;15(1):30–5. http://dx.doi.org/ 10.4103/1673-5374.262679
- 57. Bautista-Aguilera ÓM, Budni J, Mina F, Medeiros EB, Deuther-Conrad W, Entrena JM, et al. Contilisant, a tetratarget small molecule for Alzheimer's disease therapy combining cholinesterase, monoamine oxidase inhibition, and H3R antagonism with S1R agonism profile. J Med Chem. 2018;61(15): 6937–43. http://dx.doi.org/10.1021/acs.jmedchem.8b00848
- Wang D, Hu M, Li X, Zhang D, Chen C, Fu J, et al. Design, synthesis, and evaluation of isoflavone analogs as multifunctional agents for the treatment of Alzheimer's disease. Eur J Med Chem. 2019;168:207–20. http://dx.doi.org/10.1016/j.ejmech.2019.02.053
- Ismaili L, Malek R, Arribas RL, Palomino-Antolin A, Totoson P, Demougeot C, Kobrlova T, et al. New dual small molecules for Alzheimer's disease therapy combining histamine H3 receptor (H3R) antagonism and calcium channels blockade with additional cholinesterase inhibition. J Med Chem. 2019;62(24):11416–22. http://dx.doi.org/10.1021/acs.jmedchem.9b00937
- Rizzo S, Tarozzi A, Bartolini M, Da Costa G, Bisi A, Gobbi S, et al. 2-Arylbenzofuran-based molecules as multipotent Alzheimer's disease modifying agents. Eur J Med Chem. 2012;58:519–32. http:// dx.doi.org/10.1016/j.ejmech.2012.10.045
- González-Naranjo P, Pérez-Macias N, Campillo NE, Pérez C, Arán VJ, Girón R, et al. Cannabinoid agonists showing BuChE inhibition as potential therapeutic agents for Alzheimer's disease. Eur J Med Chem. 2014;73:56–72. http://dx.doi.org/10.1016/j.ejmech.2013.11.026
- 62. Dolles D, Hoffmann M, Gunesch S, Marinelli O, Möller J, Santoni G, et al. Structure-activity relationships and computational investigations into the development of potent and balanced dual-acting butyrylcholinesterase inhibitors and human cannabinoid receptor 2 ligands with pro-cognitive in vivo profiles. J Med Chem. 2018;61(4):1646–63. http://dx.doi.org/10.1021/acs.jmedchem.7b01760
- 63. Montanari S, Scalvini L, Bartolini M, Belluti F, Gobbi S, Andrisano V, et al. Fatty acid amide hydrolase (FAAH), acetylcholinesterase (AChE), and butyrylcholinesterase (BuChE): Networked targets for the development of carbamates as potential anti-Alzheimer's disease agents. J Med Chem. 2016;59(13):6387–406. http://dx.doi.org/10.1021/acs.jmedchem.6b00609
- Reis J, Cagide F, Valencia ME, Teixeira J, Bagetta D, Pérez C, et al. Multi-target-directed ligands for Alzheimer's disease: Discovery of chromone-based monoamine oxidase/cholinesterase inhibitors. Eur J Med Chem. 2018;158:781–800. http://dx.doi.org/10.1016/j.ejmech.2018.07.056
- 65. Cai P, Fang SQ, Yang HL, Yang XL, Liu QH, Kong LY, et al. Donepezil-butylated hydroxytoluene (BHT) hybrids as anti-Alzheimer's disease agents with cholinergic, antioxidant, and neuroprotective properties. Eur J Med Chem. 2018;157:161–76. http://dx.doi.org/10.1016/j.ejmech.2018.08.005
- Ryskamp DA, Korban S, Zhemkov V, Kraskovskaya N, Bezprozvanny I. Neuronal sigma-1 receptors: Signaling functions and protective roles in neurodegenerative diseases. Front Neurosci. 2019;13:862. http://dx.doi.org/10.3389/fnins.2019.00862
- Rui M, Rossino G, Coniglio S, Monteleone S, Scuteri A, Malacrida A, et al. Identification of dual Sigmal receptor modulators/acetylcholinesterase inhibitors with antioxidant and neurotrophic properties, as neuroprotective agents. Eur J Med Chem. 2018;158:353–70. http://dx.doi.org/10.1016/j. ejmech.2018.09.010
- Yang Z, Song Q, Cao Z, Yu G, Liu Z, Tan Z, et al. Design, synthesis and evaluation of flurbiprofenclioquinol hybrids as multitarget-directed ligands against Alzheimer's disease. Bioorg Med Chem. 2020;28(7):115374. http://dx.doi.org/10.1016/j.bmc.2020.115374
- 69. Hu J, Pan T, An B, Li Z, Li X, Huang L. Synthesis and evaluation of clioquinol-rolipram/roflumilast hybrids as multitarget-directed ligands for the treatment of Alzheimer's disease. Eur J Med Chem. 2019;163:512–26. http://dx.doi.org/10.1016/j.ejmech.2018.12.013
- AlFadly ED, Elzahhar PA, Tramarin A, Elkazaz S, Shaltout H, Abu-Serie MM, et al. Tackling neuroinflammation and cholinergic deficit in Alzheimer's disease: Multi-target inhibitors of cholinesterases, cyclooxygenase-2 and 15-lipoxygenase. Eur J Med Chem. 2019;167:161–86. http://dx.doi. org/10.1016/j.ejmech.2019.02.012

Natural Products for Neurocognitive Disorders

Adriana Servello • Vincenzo Leccese • Evaristo Ettorre

Department of Clinical Internal, Anesthesiological and Cardiovascular Sciences, University of Rome, Sapienza, Italy

Author for correspondence: Adriana Servello, Department of Clinical Internal, Anesthesiological and Cardiovascular Sciences, University of Rome, Sapienza, Italy. Email: adriana.servello@uniroma1.it

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ch12

Abstract: Neurocognitive disorders are devastating. In 2016, 43.8 million people were estimated to have Alzheimer's disease worldwide. By 2050, this figure is expected to rise by 56%. Despite the extreme importance of the disease, the weapons available to us to combat it are very scarce. Natural substances may be a worthwhile option for the treatment and management of neurocognitive disorders. Some of these natural products have been shown to be capable of positively impacting memory, behavior, and functions of patients with Alzheimer's disease. These substances act on the disease mainly through antioxidant properties, the ability to eliminate oxygen radicals, the capacity to influence cell survival and programmed cell death, and the potential to condition amyloidogenesis. This chapter provides an overview of our current knowledge on the potential of natural products to be effective neuroprotective agents for Alzheimer's disease. Current evidence on Ginkgo biloba, bacopia, resveratol, curcumin, quercetin, kaempferol, capsaicin, and berberine, along with their adverse effects and drug interactions are discussed.

Keywords: bacopa; ginkgo biloba; natural products for Alzheimer's disease; neurocognitive disorders; resveratrol

Copyright: The Authors.

In: *Alzheimer's Disease: Drug Discovery*. Huang X (Editor). Exon Publications, Brisbane, Australia. ISBN: 978-0-6450017-0-9; Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020

License: This open access article is licenced under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) https://creativecommons.org/licenses/by-nc/4.0/

INTRODUCTION

In 2013, the Diagnostic and Statistical Manual of Mental Disorders (DSM–5), introduced the term "Neurocognitive Disorders". It groups neurocognitive disorders into "major" and "mild" categories. A major neurocognitive disorder (MaND) is a disorder characterised by a decline in intellectual function due to disease of the brain, caused by a variety of acquired conditions such as cerebrovascular disease, Alzheimer's disease, infections, adverse drug reactions, and trauma. Alzheimer's disease is the most common form of MaND. The key distinction between a MaND and a mild neurocognitive disorder (MND) is that individuals with MaND experience substantial degradation in function resulting in loss of independence due to profound cognitive impairment, whereas subjects with MND experience only modest cognitive decline and can therefore function relatively independently. Reduced mental capacity may involve problems with complex attention, executive functioning, learning and memory, expressive and receptive language, perceptual-motor abilities, changes in behavior, and trouble performing everyday tasks (1).

A growing number of herbal remedies, dietary supplements, and "medical foods" are advertised as having beneficial neuroprotective effects. Compounds such as Ginkgo biloba, resveratrol, curcumin, capsaicin, berberine, kaeperol, quercetin as well as others are promoted as memory enhancers or as treatments to delay or prevent MaND (2). Many of these substances are found naturally in the diet, in vegetables and fruits, as well as in some spices. Some epidemiological investigations revealed that high consumption of certain foods was inversely associated with the incidence of Alzheimer's disease. These foods contain antioxidants, especially polyphenols. Polyphenols are excellent antioxidants both as reactive oxygen species (ROS) scavengers and transition metal chelators. While the neuroprotective effects of these foods in neurocognitive disorders are largely attributed to their antioxidant potential, some antioxidants derived from these foods go beyond modulating ROS (Figure 1). Several natural products are used alone or in combination with other treatment modalities to improve memory and cognition in both AD and MND (3). This chapter discusses the current scientific data on the potential of Ginkgo biloba, bacopa, resveratol, curcimun, quercetin, kaempferol, capsaicin, and berberine as effective neuroprotective agents for Alzheimer's disease along with their toxicities and drug interactions.

GINKGO BILOBA

Ginkgo biloba, also known as the Maidenhair tree, is an ancient plant whose origins date back to 250 million years ago to the Permian period. The extracts of its leaves have been used since antiquity in traditional Chinese medicine to treat various pathologies. Nowadays, extracts of this plant are used in Europe, especially in Germany and France, for the treatment of memory and concentration problems, depressive anxiety disorder, dizziness, headache, and many other issues. As highlighted by a systematic review and meta analysis, the extract of *Ginkgo biloba*, EGb761, which contains about 22% of glycosides



Figure 1. Natural products for neurocognitive disorders. The neuroprotective effects of natural products are largely attributed to their free radical scavenging abilities. In addition, they can chelate metals, inhibit amyolid beta formation, and enhance its clearance. In general, the natural products exert minimal or lower-grade adverse events.

and flavonoids and about 5–7% of terpene lactans (ginkgollids and biloballids), is capable of stabilizing or slowing down cognitive decline, functional decline, and behavior disorders at a dose of 240 mg/day. Furthermore, it is able to achieve global changes in cognitive impairment and neuro cognitive disorders in 22–26 weeks. These changes are significant for patients with neuropsychiatric symptoms (4, 5).

Neuroprotective Effects of Ginkgo Biloba

The *in vitro* effects of *Ginkgo biloba* extracts are manifold and are widely documented. They contain significant amounts of polyphenols. The effects of the extracts are largely attributed to their free radical scavenging abilities and metal chelation properties, especially chelation of copper and iron. *In vitro* studies conducted on PC12 cell lines have shown that *Ginkgo biloba* extracts

prevent the production of ROS and reduce the cytotoxicity of $A\beta_{(1-42)}$ by inhibiting apoptosis and cellular glucose absorption (6). *Ginkgo biloba* extracts were also found to prevent the formation of diffusible neurotoxic ligands derived from $A\beta_{(1-42)}$.

The *in vivo* activities of *Ginkgo biloba* have been studied in various animal models. In the nematode *Caenorhabditis elegans*, EGb761 alleviated the pathological behavior connected to the presence of A β , inhibited beta oligomerization and A β deposits, and attenuated basal and induced levels of ROS in models of neurodegenerative pathology. In a trial conducted on TgCRND8 AD mice (mice that overexpress the amyloid precursor protein (APP) particularly at a neuronal level), treatment with EGb 761 for 5 months significantly improved cognitive function, as measured by Barnes Maze test. The results clearly showed inhibition of neuroinflammation, reduction in cognitive deficits, reduction in synaptic damage, improvement of autophagy, and inhibition of A β_{1-42} -induced microglial inflammatory activity (7). The doses given were designed to maintain plasma concentrations comparable to those reached by humans who take 240 mg of product per day (recommended dose).

The effects of the treatment with EGb761 extract of Ginkgo biloba have been the subject of a meta-analysis that collected data from three clinical studies. The meta-analysis showed that a dose of 240 mg per day of *Ginkgo biloba* extract is able to delay the deterioration of basal activities of daily life. An interesting fact that emerged from this analysis was that treatment with EGb761 had the same efficacy as acetylcholinesterase inhibitors, at a lower cost (8). The weak point of the cited meta-analysis was that it took into account only daily life activities and treatment costs, not the behavioral and psychological symptoms of dementia. The behavioral and neuropsychiatric effects were explored in a randomized controlled doubleblind multicenter trial, which was designed to investigate the safety and efficacy of a daily dose of 240 mg of EGb761 in patients with mild to moderate neurocognitive disorders, using the neuropsychiatric inventory scale. Treatment with EGb761 improved cognitive function and the neuropsychological symptoms of dementia. Significantly, the quality of life of patients and their caregivers improved compared to placebo-treated patients (9). A randomized double-blind phase III trial conducted on 3069 community volunteers of at least 75 years of age with normal cognitive functions or mild cognitive impairment has shown that a dose of 120 mg twice daily of Ginkgo biloba is unable to prevent or delay the overall incidence rate of MaND or cases of Alzheimer's disease (10). On the contrary, a systematic review and meta-analysis investigating the clinical efficacy of EGb761 in minor and major cognitive decline, established that 240 mg per day was able to stabilize or slow down the decline in cognitive functions, functional skills, and behavior, resulting in overall positive effects, especially in patients with behavioral and psychological symptoms of dementia (11).

The Paquid study, a population-based cohort with a follow-up period of 22 years, examined the effect of *Ginkgo biloba* in the elderly. There were 3777 community participants, at least 65 years of age at the time of recruitment. The patients were visited at home by psychologists at baseline, and then every 2 years. They were grouped into three categories: consumers of *Ginkgo biloba* extracts, consumers of other medicines, and controls that did not receive any treatment. Consumers of *Ginkgo biloba* extracts showed lower mortality rates

and longer dementia-free lifespans than subjects taking other drugs for the same indications (12).

Ginkgo biloba extracts are mainly indicated for the treatment of "ageassociated" cognitive decline and for improving the quality of life in patients with mild neurocognitive disorders (MND). Long-term use of these extracts is most effective in patients over 50 years of age. According to European Pharmacopoeia (Ph. Eur.), the extract should contain 22.0–27.0% of flavonoids, 2.8–3.4% of ginkgolides A, B, and C, and 2.6–3.2% of bilobalide (active constituents), and a maximum of 5 ppm ginkgoid acids (potential allergens) (13). Treatment with *Ginkgo biloba* extracts is safe, but close attention should be paid to patients being treated with anticoagulants, as the pharmacokinetics of some may be altered by certain plant extracts (14).

Adverse Effects of Ginkgo Biloba

Ginkgo biloba is currently considered to have one of the highest rates of adverse side effects and interaction (Table 1) with conventional drugs (15). Cases of intoxication have been reported in Japan and China, where Ginkgo seeds and nuts have been a common food since ancient times. Poisoning causes tonic-clonic epileptic seizures, vomiting, and loss of consciousness. These effects are mainly related to a neurotoxic compound known as ginkgotoxin. Rare cases of death related to Ginkgo poisoning have been described, especially in the years following the first and second World Wars, probably due to food shortages. Now-a-days, cases of *Ginkgo* poisoning are much rarer, perhaps due to an increase in knowledge of the toxicological profile of these substances. Another adverse effect attributable to Ginkgo biloba extracts is spontaneous bleeding, especially in patients taking Ginkgo and warfarin together. Rarer adverse effects include a proarrhythmic effect, nocturnal palpitations, cerebrovascular events (ischemic stroke and transient ischemic attacks), and allergic reactions such as contact dermatitis. Drug interactions of Ginkgo biloba with conventional medicinal substances have also been reported. It interacts with drugs widely used in clinical practice such as omeprazole, midazolam, and tolbutamide.

BACOPA

Brahmi or *Bacopa monnieri* is a plant used in Ayurvedic medicine with neuroprotective and nootropic properties. The neuroprotective effect of this plant seems to derive from numerous bioactive components such as bacoside A, bacoside B, bacosaponins, and betulinic acid. The mechanisms by which these compounds exert a neuroprotective effect and offer improvement of cognitive and learning abilities include reduction of ROS, anti-inflammation, and inhibition of beta amyloid aggregation. Bacopa seems to play a relevant role in the treatment of not only of Alzheimer's disease but also of other neurological pathologies (16).

$^{-}\mathbf{A}$	R	F.	1
~	-		

Interactions and side effects of natural products

Natural product	Indications	Drug interactions and side effects
Ginkgo biloba	 Age associated cognitive decline Mild neurocognitive disorder Multi infarct dementia Traumatic brain injury Normal aging Stroke Tinnitus Intermittent claudication Senile macular degeneration 	 Pay attention in association with anticoagulants Intoxication: tonic clonic epileptic seizures, vomiting, loss of consciousness Spontaneous bleeding (when administered with warfarin) Rare: proarrhytmic effect, allergic reactions, acute generalized exhantematous pustolosis
Bacopa monnieri	 Improve cognition in the elderly and in patients with neuro cognitive disorders Beneficial effects in Parkinson disease, depression, neoplastic pathologies Beneficial properties on gastrointestinal tract 	No significant harmful side effects is currently known
Resveratrol	 Improves adult cognitive abilities Reduces cognitive decline in the healthy elderly Slows down the decline in pathological states such as Alzheimer's and Parkinson's disease 	 Leukopenia Decreases circulating levels of TNF and IL 6 Mild to moderate diarrhea, nausea, hypersensitivity and annoying itching in the anal area.
Curcumin	 Neurodegenerative diseases, multiple sclerosis, prion diseases, stroke Autism, Down syndrome, Amyotrophic Lateral Sclerosis, depression, anxiety and aging Beneficial properties in diabetes mellitus, arthritis, liver, kidney and cardiovascular pathologies 	 High doses can produce toxic and even carcinogenic effects Mild nausea and diarrhea Subclinical iron deficiency
Quercetin	• Prevention and treatment of Alzheimer's disease and other types of dementia	 Adverse effects are rare and usually very mild In animal models it shows toxic effects on kidney
Kaempferol	Protective action against cognitive declineBeneficial effects in experimental models of Alzheimer's disease	No significant harmful side effects is currently known

Table continued on following page

Natural product	Indications	Drug interactions and side effects
Capsaicin	 Beneficial effects both on cognitive functions of middle aged adults and advanced age subjects Beneficial effects on biomarkers of Alzheimer's disease Anti cancer properties Anti obesity properties Osteoarthritic pain Cannabinoid hyperemesis syndrome Useful as a dermal patch in peripheral neurotrophic pain 	 At very high doses, well above those clinically useful, capsaicin can lead to death due to respiratory paralysis It is hypothesized that capsaicin may have carcinogenic effects linked to overdose Capsaicin in the form of pepper spray is likely to induce a form of acute polyneuropathy that resembles Guillain Barrè syndrome
Berberine	 Neurodegenerative pathologies and, in particular, Alzheimer's disease Ischemic stroke Vascular dementia 	 When administered for prolonged periods of time, causes a deterioration of dopaminergic neurons Several drug interactions (tetrandine, levodopa, doxorubicin, beta lactam antibiotics and hydroxycamptotecin, Panax ginseng, cisplatin, fluconazole, cyclosporin A, warfarin and thiopental)

TABLE 1Interactions and side effects of natural
products (Continued)

Neuroprotective Effects of Bacopa

Bacopa's mechanisms of action (Table 2) have been studied on the microglial cell line N9. Infusions, teas, and extracts of bacopa and bacoside A are able to significantly inhibit the release of mediators such as TNF- α and interleukin 6 by microglial cells in vitro. In cell-free systems, teas, infusions and alkaloid extracts of Bacopa can inhibit caspases 1 and 3, and matrix metalloproteinases 3. Thus the fundamental mechanism of action of bacopa appears to be inhibition of the release of pro-inflammatory cytokines by microglia cells and inhibition of enzymes that perform pro-inflammatory actions in the central nervous system. The net effect of all these pharmacological actions of bacopa evidently consists of keeping inflammation under control in the brain (17). In animal models, *Bacopa monnieri* extracts improve cognitive functions mainly through protective action on hippocampal neurons and partly through promoting neuroregeneration in the dentate gyrus (18, 19). Furthermore, studies on APP/PS1 mice have shown that Bacopia monnieri extracts can reduce plaque load and enhance plaque clearance, possibly via phagocytosis (20). Bacopa monnieri prevents senescence and has an anti-apoptotic effect in brain astrocytes and, therefore, is thought to combat brain pollution and age-related neurological disorders (21).

TABLE 2	Mechanism of action
Natural products	Mechanisms of action
Ginkgo biloba	 Antioxidant power Copper chelating ability Radicals scavenging properties Prevents formation of diffusible neurotoxic ligands derived from Aβ₍₁₋₄₂₎ Prevents apoptosis
Bacopa monnieri	 Inhibits the release by microglial cells <i>in vitro</i> of mediators such as TNF α and IL 6 Inhibits caspases 1 and 3 and metalloproteinases 3 Overall, keeps inflammation under control into the brain Expresses amyloid clearance capacity Prevents senescence and has anti apoptotic effect in brain astrocytes Reduces the share of Reactive Oxygen Species
Resveratrol	 Improves metabolic functions Acts as a neuroinflammation modulator Shows protective effects against oxidative stress Seems to inhibit the mechanism of apoptosis in brain cells Anti aggregating activity against Aβ₍₁₋₄₂₎
Curcumin	 Powerful antioxidant, anti inflammatory, anti cancer, anti microbial actions Neurotrophic properties Stimulates amyloid clearance in Alzheimer's disease
Quercetin	 Anti inflammatory, anti oxidant, anti cancer properties Beneficial properties for dyslipidemia, hypercolesterolemia, cardiovascular diseases, diabetes ROS scavenging ability Neuroprotective properties
Kaempferol	 Antiamyloidogenic activity Destabilizing activity against amyloid fibrils Chelates metal ions Keeps oxidative stress under control Inhibits platelet aggregation and thrombosis
Capsaicin	 Mitigates amyloid induced synaptic loss Performs neuroprotective functions against stress induced cognitive impairment Able to reduce the hyperphosphorylation degree of tau protein
Berberine	 Anti inflammatory activities Anti oxidant properties Inhibits acetyl cholinesterase Inhibiting properties against amyloid formation

Taken together, the neuroprotective actions and positive effects of bacoside A on memory, mental, and intellectual functions can be largely attributed to its ability to reduce beta amyloid aggregation and toxicity (22), lower inflammation in the brain, scavenge ROS, and increase cerebral blood flow (23, 24). Promising indications for use in humans include improving cognition in the elderly and in patients with neurodegenerative disorders (25). Bacopa is considered by some authors as a kind of

"bulletproof vest" against Alzheimer's disease (26). It seems that *Bacopa monnieri* extracts are able to reduce the formation of transthyretin fibrils (TTR) by attenuating their disassembly from the tetrameric form to the monomeric form. This suggests bacopa has a role in preventing transthyretin amyloidosis, a very debilitating and often fatal disease in humans (27). The anti-inflammatory properties of bacopa have been beneficial in the treatment of many neurological diseases of the central nervous system. For example, they have shown promising results in some experimental models of Parkinson's disease (28). In addition, bacopa has proven beneficial in the treatment of depression, anhedonia, various inflammatory diseases, and neoplastic pathologies. In the rat, bacopa has exhibited beneficial properties on the gastrointestinal tract as an antidiarrheal, as a protector of the gastric mucosa, and as an anti-ulcer drug (29, 30). Currently, no significant harmful side effects of *Bacopa monnieri* are known.

RESVERATROL

Resveratrol is a phenol predominantly found in grapes and wine as well as in berries, peanuts, and soybeans (31). Resveratrol is effective in reducing the risk of dementia in mouse models; this perhaps is due to the fact that improving metabolic function improves brain health during senile age (32). Resveratrol acts as a brain neuroinflammation modulator (33), shows protective effects against oxidative stress (34), inhibits the mechanism of apoptosis in brain cells (35), and exerts antiaggregating activity against $A\beta_{(1-42)}$ (36). Resveratrol is a promising molecule for improving cognitive abilities in adults, reducing cognitive decline in healthy elderly, and slowing decline in pathological states such as Alzheimer's and Parkinson's disease (37, 38). The toxic effects of resveratrol seem to be closely associated with a hormetic effect; low doses are generally associated with antioxidant effects while higher doses can have a pro-oxidant effect. Resveratrol can cause leukopenia, decrease circulating levels of TNF and interleukin 6, and increase plasma levels of alanine aminotransferase. In addition, it can cause mild to moderate diarrhea, nausea, hypersensitivity, and irritation of the anal area. Through the dose-dependent ROS increase, resveratrol can also cause proteolysis and DNA damage (39).

CURCUMIN

Curcumin is an extract of turmeric widely used in Asia, especially in the culinary sector. This substance seems to have multiple beneficial effects, particularly with respect to various neurological pathologies and cancer. Curcumin exerts its effects through powerful antioxidant, anti-inflammatory, anti-cancer and antimicrobial actions. In addition, the molecule has neurotrophic properties and appears to stimulate amyloid clearance in Alzheimer's disease (40). Curcumin shows beneficial effects in neurodegenerative diseases such as Alzheimer's disease, Parkinson's disease, Huntington's chorea (41), multiple sclerosis, prion diseases, and stroke. In addition, the molecule can attenuate age-related severity of autism and Down syndrome, and prevent the progression of amyotrophic lateral sclerosis, depression, anxiety, and pathological aging by reducing the expression of IL-1 α , IL-6, and TNF- α , and by the activation of mitochondrial protection and anti-apoptotic

mechanisms (42). Currently, there is still insufficient clinical data to support the beneficial effects of curcumin in patients with Alzheimer's disease. In addition, curcumin administered orally is poorly bioavailable (43). The pharmacological effects, whether therapeutic or toxic, are dose-dependent. At high doses, the molecule can produce toxic and even carcinogenic effects (44). Sometimes curcumin, due to metabolic activation, can exert pro-oxidant effects (45). Other side effects include mild nausea and diarrhea (46). Curcumin also chelates iron and suppresses hepcidin, leading to a subclinical iron deficiency (47).

QUERCETIN

Quercetin is a flavonoid present in numerous plants, as well as in certain fruits and vegetables. It is also present in red wine and green tea. Quercetin has antioxidant, anti-inflammatory, anti-cancer, and ROS scavenging properties, and has been shown to be beneficial for the prostate, dyslipidemia hypercholesterolemia, cardiovascular diseases, diabetes, viral infections, kidney transplantation, asthma, and lung disease. Furthermore, it seems to have neuroprotective effects in schizophrenia, Alzheimer's disease, and other forms of dementia (48, 49). The adverse effects associated with the use of quercetin are rare and typically mild; however, animal studies have shown that quercetin has toxic effects on the kidney, promote tumor development especially in estrogen-dependent cancers, and can interact with various drugs to alter their bioavailability (50).

KAEMPFEROL

Kaempferol is a flavonoid present in many plants and in vegetables such as cabbage, spinach, beans, broccoli, and tea (51). Kaempferol has antiamyloidogenic activity. It can destabilize amyloid fibrils, chelate metals, and keep oxidative stress under control (52). Kaempferol has also been shown to inhibit platelet aggregation and thrombosis, an effect that could prove particularly interesting in vascular forms of cognitive decline (53). Like the other main dietary flavonoids (myricetin and quercetin), Kaempferol protects against cognitive decline (54). All flavonoids (myricetin, morin, rutin, quercetin, fisetin, kaempferol, apigenin, and glycitein) have shown beneficial effects in experimental models of Alzheimer's disease (55). There is currently insufficient data in the literature regarding the adverse effects of Kaempferol in humans.

CAPSAICIN

Capsaicin is a well-known alkaloid present in chilli pepper and is responsible for its spiciness. In experimental mouse models, capsaicin has been shown to mitigate amyloid-induced synaptic loss (56). Furthermore, capsaicin has proven to be capable of performing neuroprotective functions against stress-induced cognitive impairment (57). Finally, it has been shown in experimental animals that capsaicin is able to reduce the degree of hyperphosphorylation of the tau protein, a protein involved in the pathogenesis of Alzheimer's disease (58). A capsaicin-rich diet has been shown to have beneficial effects both on the cognitive functions of middle aged adults and advanced age subjects, and on the blood biomarkers of Alzheimer's disease (A β_{40} , A β_{42} , and their ratio) (59). In addition to its neuroprotective properties, capsaicin has anti-cancer (60), antiobesity properties (61) and can counteract metabolic syndrome (62). It is useful as a topical applicant to ease osteoarthritic pain (63), particularly in that of the knee (64). It can be used in the treatment of cannabinoid hyperemesis syndrome (65), and as a dermal patch for peripheral neuropathic pain (66). At very high doses (well above those clinically useful), capsaicin can lead to death due to respiratory paralysis (67, 68). Capsaicin in the form of pepper spray can induce a form of acute polyneuropathy that resembles Guillain-Barrè syndrome (69).

BERBERINE

Berberine is an alkaloid extracted from various plants of the Berberis species and is widely used in Asia. Berberine exhibits anti-inflammatory, antioxidative, and anti-amyloid activities (70). It can also inhibit acetylcholinesterase (70). Berberine is useful in neurodegenerative pathologies such as Alzheimer's disease (71), stroke (72), and vascular dementia (73, 74).

Berberine, when administered for prolonged periods of time, causes a deterioration of dopaminergic neurons due to the cytotoxic effect exerted by 6 hydroxydopamine (75–77).

Berberine stimulates uterine contractions and should therefore be used with caution in pregnancy. It also appears to exacerbate jaundice and kernicterus (Kernicterus or Bilirubin encephalopathy or Nuclear jaundice) in infants with glucose-6-phosphate dehydrogenase deficiency. Berberine is likely to be able to cross the placenta and harm the developing fetus. It can be transferred to breast milk and should therefore also be used with caution when breastfeeding. However, berberine does not have known genotoxic, cytotoxic, mutagenic actions, or other adverse effects at clinically relevant doses.

Various drug interactions (Table 1) of berberine have been reported: (i), tetrandineh worsens the hypoglycaemic properties of berberine; (ii), the activity of levodopa is antagonized by berberine; (iii), with doxorubicin, Panax ginseng, cisplatin, and fluconazole berberine exerts synergistic effect; and (iv), berberine increases the circulating levels of cyclosporin A, warfarin, and thiopental (78).

CONCLUSION

Neurocognitive disorders are multifactorial pathological conditions that require a multidisciplinary therapeutic approach. The pharmacological weapons at our disposal are extremely limited and any aid is a welcome measure for the patients, their families, and caregivers. Given the available evidence, it is reasonable to assume that natural products will gain an increasingly important role in the future alongside traditional and experimental pharmacological therapies for the management of patients with neurocognitive disorders, including Alzheimer's disease.

Conflict of Interest: The authors declare no potential conflicts of interest with respect to research, authorship, and/or publication of this chapter.

Copyright and Permission Statement: To the best of our knowledge, the materials included in this chapter do not violate copyright laws. All original sources have been appropriately acknowledged and/or referenced.

REFERENCES

- 1. American Psychiatric Association. Diagnostic and statistical manual of mental disorders (DSM-5®). American Psychiatric Publishing; 2013. http://dx.doi.org/10.1176/appi.books.9780890425596
- Habtemariam S. Natural products in Alzheimer's disease therapy: Would old therapeutic approaches fix the broken promise of modern medicines? Molecules. 2019;24(8):1519. http://dx.doi.org/10.3390/ molecules24081519
- Islam MA, Khandker SS, Alam F, Khalil MI, Kamal MA, Gan SH. Alzheimer's disease and natural products: Future regimens emerging from nature. Curr Top Med Chem. 2017;17(12):1408–28. http://dx.doi.org/10.2174/1568026617666170103163054
- 4. Birks J, Grimley Evans J. *Ginkgo biloba* for cognitive impairment and dementia. Cochrane Database Syst Rev. 2009;(1):CD003120. http://dx.doi.org/10.1002/14651858.CD003120.pub3
- Tan MS, Yu JT, Tan CC, Wang HF, Meng XF, Wang C, et al. Efficacy and adverse effects of *Ginkgo biloba* for cognitive impairment and dementia: A systematic review and meta-analysis. J Alzheimer's Dis. 2015;43(2):589–603. http://dx.doi.org/10.3233/JAD-140837
- Pohl F, Lin PKT. The potential use of plant natural products and plant extracts with antioxidant properties for the prevention/treatment of neurodegenerative diseases: In vitro, in vivo and clinical trials. Molecules. 2018;23(12):3283. http://dx.doi.org/10.3390/molecules23123283
- Liu X, Hao W, Qin Y, Decker Y, Wang X, Burkart M, et al. Long-term treatment with *Ginkgo biloba* extract EGb 761 improves symptoms and pathology in a transgenic mouse model of Alzheimer's disease. Brain Behav Immun. 2015;46:121–31. http://dx.doi.org/10.1016/j.bbi.2015.01.011
- Rainer M, Mucke H, Schlaefke S. *Ginkgo biloba* extract EGb 761 in the treatment of dementia: A pharmacoeconomic analysis of the Austrian setting. Wien Klin Wochenschr. 2013;125(1–2):8–15. http://dx.doi.org/10.1007/s00508-012-0307-x
- Herrschaft H, Nacu A, Likhachev S, Sholomov I, Hoerr R, Schlaefke S. *Ginkgo biloba* extract EGb 761 in dementia with neuropsychiatric features: A randomised, placebo-controlled trial to confirm the efficacy and safety of a daily dose of 240 mg. J Psychiatr Res. 2012;46(6):716–23. http://dx.doi. org/10.1016/j.jpsychires.2012.03.003
- DeKosky ST, Williamson JD, Fitzpatrick AL, Kronmal RA, Ives DG, Saxton JA, et al. *Ginkgo biloba* for prevention of dementia: A randomized controlled trial [published correction appears in JAMA. 2008 Dec 17;300(23):2730. http://dx.doi.org/10.1001/jama.300.23.2730
- Tan MS, Yu JT, Tan CC, Wang HF, Meng XF, Wang C, et al. Efficacy and adverse effects of *Ginkgo biloba* for cognitive impairment and dementia: A systematic review and meta-analysis. J Alzheimers Dis. 2015;43(2):589–603. http://dx.doi.org/10.3233/JAD-140837
- Dartigues JF, Grasset L, Helmer C, Féart C, Letenneur L, Jacqmin-Gadda H, et al. Ginkgo biloba extract consumption and long-term occurrence of death and dementia. J Prev Alzheimers Dis. 2017;4(1):16–20.
- Petrović S, Ušjak L. Herbal medicines from ginkgo leaf extract in the treatment of mild dementia. Arhiv Farm. 2020;70(2):81–97. http://dx.doi.org/10.5937/arhfarm2002081P
- Diamond BJ, Shiflett SC, Feiwel N, Matheis RJ, Noskin O, Richards JA, et al. *Ginkgo biloba* extract: Mechanisms and clinical indications. Arch Phys Med Rehabil. 2000;81(5):668–78. http://dx.doi. org/10.1016/S0003-9993(00)90052-2

- Di Lorenzo C, Ceschi A, Kupferschmidt H, Lüde S, De Souza Nascimento E, Dos Santos, et al. Adverse effects of plant food supplements and botanical preparations: A systematic review with critical evaluation of causality. Br J Clin Pharmacol. 2015;79(4):578–92. http://dx.doi.org/10.1111/ bcp.12519
- Dubey T, Chinnathambi S. Brahmi (Bacopa monnieri): An ayurvedic herb against the Alzheimer's disease. Arch Biochem Biophys. 2019;676:108153. http://dx.doi.org/10.1016/j. abb.2019.108153
- Nemetchek MD, Stierle AA, Stierle DB, Lurie DI. The ayurvedic plant *Bacopa monnieri* inhibits inflammatory pathways in the brain. J Ethnopharmacol. 2017;197:92–100. http://dx.doi. org/10.1016/j.jep.2016.07.073
- Pham HTN, Phan SV, Tran HN, Phi XT, Le XT, Nguyen KM, et al. *Bacopa monnieri* (L.) ameliorates cognitive deficits caused in a trimethyltin-induced neurotoxicity model mice. Biol Pharm Bull. 2019;42(8):1384–93. http://dx.doi.org/10.1248/bpb.b19-00288
- Pham HTN, Tran HN, Nguyen PT, Le XT, Nguyen KM, Phan SV, et al., *Bacopa monnieri* (L.) Wettst. Extract improves memory performance via promotion of neurogenesis in the hippocampal dentate gyrus of adolescent mice. Int J Mol Sci. 2020 May 9;21(9):3365. http://dx.doi.org/10.3390/ijms21093365
- Li Y, Yuan X, Shen Y, Zhao J, Yue R, Liu F, et al. Bacopaside I ameliorates cognitive impairment in APP/ PS1 mice via immune-mediated clearance of β-amyloid. Aging (Albany NY). 2016;8(3):521. http:// dx.doi.org/10.18632/aging.100913
- 21. Saha S, Mahapatra KK, Mishra SR, Mallick S, Negi VD, Sarangi I, et al. *Bacopa monnieri* inhibits apoptosis and senescence through mitophagy in human astrocytes. Food Chem Toxicol. 2020;141:111367. http://dx.doi.org/10.1016/j.fct.2020.111367
- Malishev R, Shaham-Niv S, Nandi S, Kolusheva S, Gazit E, Jelinek R. Bacoside-A, an Indian traditional-medicine substance, inhibits β-amyloid cytotoxicity, fibrillation, and membrane interactions. ACS Chem Neurosci. 2017;8(4):884–91. http://dx.doi.org/10.1021/acschemneuro.6b00438
- Chaudhari KS, Tiwari NR, Tiwari RR, Sharma RS. Neurocognitive effect of nootropic drug brahmi (*Bacopa monnieri*) in Alzheimer's disease. Ann Neurosci. 2017;24(2):111–22. http://dx.doi. org/10.1159/000475900
- Ayati Z, Chang D, Lake J. Advances in treatment of mild cognitive impairment (MCI) and dementia: A review of promising non-pharmaceutical modalities. Front Clin Drug Res Dementia. 2020;1(1):78. http://dx.doi.org/10.2174/9789811410949120010005
- Stough C, Scholey A, Cropley V, Wesnes K, Zangara A, Pase M, et al. Examining the cognitive effects of a special extract of *Bacopa monniera* (CDRI08: Keenmnd): A review of ten years of research at Swinburne University. J Pharm Sci. 2013;16(2):254–8. http://dx.doi.org/10.18433/J35G6M
- Abdul Manap AS, Vijayabalan S, Madhavan P, Chia YY, Arya A, Wong EH, et al. *Bacopa monnieri*, a neuroprotective lead in Alzheimer disease: A review on its properties, mechanisms of action, and preclinical and clinical studies. Drug Target Insights. 2019;13:1177392819866412. http://dx.doi. org/10.1177/1177392819866412
- Eze FN, Ingkaninan K, Prapunpoj P. Transthyretin anti-amyloidogenic and fibril disrupting activities of *Bacopa monnieri* (L.) Wettst (Brahmi) extract. Biomolecules. 2019;9(12):845. http://dx.doi. org/10.3390/biom9120845
- Singh B, Pandey S, Rumman M, Mahdi AA. Neuroprotective effects of *Bacopa monnieri* in Parkinson's disease model. Metab Brain Dis. 2020;35(3):517–25. http://dx.doi.org/10.1007/s11011-019-00526-w
- Saha PS, Sarkar S, Jeyasri R, Muthuramalingam P, Ramesh M, Jha S. In vitro propagation, phytochemical and neuropharmacological profiles of *Bacopa monnieri* (L.) Wettst.: A review. Plants. 2020;9(4):411. http://dx.doi.org/10.3390/plants9040411
- Micheli L, Spitoni S, Di Cesare Mannelli L, Bilia AR, Ghelardini C, Pallanti S. Bacopa monnieri as augmentation therapy in the treatment of anhedonia, preclinical and clinical evaluation. Phytother Res. 2020;34(9):2331–40. http://dx.doi.org/10.1002/ptr.6684
- Breuss JM, Atanasov AG, Uhrin P. Resveratrol and its effects on the vascular system. Int J Mol Sci. 2019 Mar 27;20(7):1523. http://dx.doi.org/10.3390/ijms20071523
- 32. Wahl D, Solon-Biet SM, Cogger VC, Fontana L, Simpson SJ, Le Couteur DG, et al. Aging, lifestyle and dementia. Neurobiol Dis. 2019;130:104481. http://dx.doi.org/10.1016/j.nbd.2019.104481

- Szczechowiak K, Diniz BS, Leszek J. Diet and Alzheimer's dementia Nutritional approach to modulate inflammation. Pharmacol Biochem Behav. 2019;184:172743. http://dx.doi.org/10.1016/j. pbb.2019.172743
- 34. Cosín-Tomàs M, Senserrich J, Arumí-Planas M, Alquézar C, Pallàs M, Martín-Requero A, et al. Role of resveratrol and selenium on oxidative stress and expression of antioxidant and anti-aging genes in immortalized lymphocytes from Alzheimer's disease patients. Nutrients. 2019;11(8):1764. http:// dx.doi.org/10.3390/nu11081764
- Zhang Y, Li Y, Wang Y, Wang G, Mao L, Zhang D, et al. Effects of resveratrol on learning and memory in rats with vascular dementia. Mol Med Rep. 2019;20(5):4587–93. http://dx.doi.org/10.3892/ mmr.2019.10723
- 36. Saini RK, Shuaib S, Goyal D, Goyal B. Insights into the inhibitory mechanism of a resveratrol and clioquinol hybrid against Aβ42 aggregation and protofibril destabilization: A molecular dynamics simulation study. J Biomol Struct Dyn. 2019;37(12):3183–97. http://dx.doi.org/10.1080/07391102.2018.1511475
- Devi SA, Chamoli A. Reviews on new drug targets in age-related disorders. Cham: Springer; 2020. p. 159–174. http://dx.doi.org/10.1007/978-3-030-42667-5_7
- Mathew B, Parambi DG, Mathew GE, Uddin MS, Inasu ST, Kim H, et al. Emerging therapeutic potentials of dual-acting MAO and AChE inhibitors in Alzheimer's and Parkinson's diseases. Arch Pharm (Weinheim). 2019;352(11):e1900177. http://dx.doi.org/10.1002/ardp.201900177
- Shaito A, Posadino AM, Younes N, Hasan H, Halabi S, Alhababi D, et al. Potential adverse effects of resveratrol: A literature review. Int J Mol Sci. 2020;21(6):2084. http://dx.doi.org/10.3390/ ijms21062084
- Teter B, Morihara T, Lim GP, Chu T, Jones MR, Zuo X, et al. Curcumin restores innate immune Alzheimer's disease risk gene expression to ameliorate Alzheimer pathogenesis. Neurobiol Dis. 2019;127:432–48. http://dx.doi.org/10.1016/j.nbd.2019.02.015
- Elifani F, Amico E, Pepe G, Capocci L, Castaldo S, Rosa P, et al. Curcumin dietary supplementation ameliorates disease phenotype in an animal model of Huntington's disease. Hum Mol Genet. 2019;28(23):4012–21. http://dx.doi.org/10.1093/hmg/ddz247
- 42. Bhat A, Mahalakshmi AM, Ray B, Tuladhar S, Hediyal TA, Manthiannem E, et al. Benefits of curcumin in brain disorders. BioFactors. 2019;45(5):666–89. http://dx.doi.org/10.1002/biof.1533
- 43. Farkhondeh T, Samarghandian S, Pourbagher-Shahri AM, Sedaghat M. The impact of curcumin and its modified formulations on Alzheimer's disease. J Cell Physiol. 2019;234(10):16953–65.
- 44. Cianfruglia L, Minnelli C, Laudadio E, Scirè A, Armeni T. Side effects of curcumin: Epigenetic and antiproliferative implications for normal dermal fibroblast and breast cancer cells. Antioxidants. 2019;8(9):382. http://dx.doi.org/10.3390/antiox8090382
- Kawanishi S, Oikawa S, Murata M. Evaluation for safety of antioxidant chemopreventive agents. Antioxid Redox Signal. 2005;7(11–12):1728–39. http://dx.doi.org/10.1089/ars.2005.7.1728
- Hsu CH, Cheng AL. The molecular targets and therapeutic uses of curcumin in health and disease. Boston, MA: Springer; 2007. pp. 471–80.
- Jiao Y, Wilkinson IV J, Di X, Wang W, Hatcher H, Kock ND, et al. Curcumin, a cancer chemopreventive and chemotherapeutic agent, is a biologically active iron chelator. Blood. 2009;113(2):462–9. http:// dx.doi.org/10.1182/blood-2008-05-155952
- Zaplatic E, Bule M, Shah SZA, Uddin MS, Niaz K. Molecular mechanisms underlying protective role of quercetin in attenuating Alzheimer's disease. Life Sci. 2019;224:109–19. http://dx.doi.org/10.1016/j. lfs.2019.03.055
- Babaei F, Mirzababaei M, Nassiri-Asl M. Quercetin in food: Possible mechanisms of its effect on memory. J Food Sci. 2018;83(9):2280–87. http://dx.doi.org/10.1111/1750-3841.14317
- Andres S, Pevny S, Ziegenhagen R, Bakhiya N, Schäfer B, Hirsch-Ernst KI, et al. Safety aspects of the use of quercetin as a dietary supplement. Mol Nutr Food Res. 2018;62(1):1700447. http://dx.doi. org/10.1002/mnfr.201700447
- Holland TM, Agarwal P, Wang Y, Leurgans SE, Bennett DA, Booth SL, et al. Dietary flavonols and risk of Alzheimer dementia. Neurology. 2020;94(16):e1749–56. http://dx.doi.org/10.1212/ WNL.000000000008981
- Simunkova M, Alwasel SH, Alhazza IM, Jomova K, Kollar V, Rusko M, et al. Management of oxidative stress and other pathologies in Alzheimer's disease. Arch Toxicol. 2019:1–23. http://dx.doi. org/10.1007/s00204-019-02538-y

- Choi JH, Park SE, Kim SJ, Kim S. Kaempferol inhibits thrombosis and platelet activation. Biochimie. 2015;115:177–186. http://dx.doi.org/10.1016/j.biochi.2015.06.001
- Root M, Ravine E, Harper A. Flavonol intake and cognitive decline in middle-aged adults. J Med Food. 2015;18(12):1327–32. http://dx.doi.org/10.1089/jmf.2015.0010
- Beg T, Jyoti S, Naz F, Ali F, Ali SK, Reyad AM, et al. Protective effect of kaempferol on the transgenic Drosophila model of Alzheimer's disease. CNS Neurol Disord Drug Targets. 2018;17(6):421–9. http:// dx.doi.org/10.2174/1871527317666180508123050
- Chen L, Huang Z, Du Y, Fu M, Han H, Wang Y, et al. Capsaicin attenuates amyloid-β-induced synapse loss and cognitive impairments in mice. J Alzheimers Dis. 2017;59(2):683–94. http://dx.doi. org/10.3233/JAD-170337
- Jiang X, Jia LW, Li XH, Cheng XS, Xie JZ, Ma ZW, et al. Capsaicin ameliorates stress-induced Alzheimer's disease-like pathological and cognitive impairments in rats. J Alzheimers Dis. 2013;35(1):91–105. http://dx.doi.org/10.3233/JAD-121837
- Xu W, Liu J, Ma D, Yuan G, Lu Y, Yang Y. Capsaicin reduces Alzheimer-associated tau changes in the hippocampus of type 2 diabetes rats. PLoS One. 2017;12(2):e0172477. http://dx.doi.org/10.1371/ journal.pone.0172477
- Liu CH, Bu XL, Wang J, Zhang T, Xiang Y, Shen LL, et al. The associations between a capsaicin-rich diet and blood amyloid-β levels and cognitive function. J Alzheimers Dis. 2016;52(3):1081–8. http:// dx.doi.org/10.3233/JAD-151079
- Chapa-Oliver AM, Mejía-Teniente L. Capsaicin: From plants to a cancer-suppressing agent. Molecules. 2016;21(8):931. http://dx.doi.org/10.3390/molecules21080931
- Zheng J, Zheng S, Feng Q, Zhang Q, Xiao X. Dietary capsaicin and its anti-obesity potency: From mechanism to clinical implications. Biosci Rep. 2017;37(3). http://dx.doi.org/10.1042/BSR20170286
- Panchal SK, Bliss E, Brown L. Capsaicin in metabolic syndrome. Nutrients. 2018;10(5):630. http:// dx.doi.org/10.3390/nu10050630
- 63. Guedes V, Castro JP, Brito I. Topical capsaicin for pain in osteoarthritis: A literature review. Reumatol Clin. 2018;14(1):40–5. http://dx.doi.org/10.1016/j.reuma.2016.07.008
- 64. Rodriguez-Merchan EC. Topical therapies for knee osteoarthritis. Postgrad Med. 2018;130(7):607–12. http://dx.doi.org/10.1080/00325481.2018.1505182
- McConachie SM, Caputo RA, Wilhelm SM, Kale-Pradhan PB. Efficacy of capsaicin for the treatment of cannabinoid hyperemesis syndrome: A systematic review. Ann Pharmacother. 2019;53(11):1145–52. http://dx.doi.org/10.1177/1060028019852601
- Blair HA. Capsaicin 8% dermal patch: A review in peripheral neuropathic pain. Drugs. 2018;78(14):1489–500. http://dx.doi.org/10.1007/s40265-018-0982-7
- 67. Johnson Jr W. Final report on the safety assessment of *Ricinus communis* (castor) seed oil, hydrogenated castor oil, glyceryl ricinoleate, glyceryl ricinoleate se, ricinoleic acid, potassium ricinoleate, sodium ricinoleate, zinc ricinoleate, cetyl ricinoleate, ethyl ricinoleate, glycol ricinoleate, isopropyl ricinoleate, methyl ricinoleate, and octyldodecyl ricinoleate. Int J Toxicol. 2007;26:31–77. http://dx.doi. org/10.1080/10915810701663150
- Hwang MK, Bode AM, Byun S, Song NR, Lee HJ, Lee KW, et al. Cocarcinogenic effect of capsaicin involves activation of EGFR signaling but not TRPV1. Cancer Res. 2010;70(17):6859–69. http:// dx.doi.org/10.1158/0008-5472.CAN-09-4393
- Özçora GDK, Çıraklı S, Canpolat M, Doğanay S, Kumandaş S. Pepper spray inhalation-induced acute polyneuropathy mimicking Guillain-Barre syndrome. Turk Arch Ped/Türk Pediatri Arşivi. 2019;54(1):53.
- Yuan NN, Cai CZ, Wu MY, Su HX, Li M, Lu JH. Neuroprotective effects of berberine in animal models of Alzheimer's disease: A systematic review of pre-clinical studies. BMC Complement Altern Med. 2019;19(1):109. http://dx.doi.org/10.1186/s12906-019-2510-z
- 71. Velmurugan BK, Rathinasamy B, Lohanathan BP, Thiyagarajan V, Weng CF. Neuroprotective role of phytochemicals. Molecules. 2018;23(10):2485. http://dx.doi.org/10.3390/molecules23102485
- 72. Kim M, Shin MS, Lee JM, Cho HS, Kim CJ, Kim YJ, et al. Inhibitory effects of isoquinoline alkaloid berberine on ischemia-induced apoptosis via activation of phosphoinositide 3-kinase/protein kinase B signaling pathway. Internat Neurourol J. 2014;18(3):115. http://dx.doi.org/10.5213/inj.2014.18.3.115
- 73. Yin S, Bai W, Li P, Jian X, Shan T, Tang Z, et al. Berberine suppresses the ectopic expression of miR-133a in endothelial cells to improve vascular dementia in diabetic rats. Clin Exp Hypertens. 2019;41(8):708–16. http://dx.doi.org/10.1080/10641963.2018.1545846

- 74. Aski ML, Rezvani ME, Khaksari M, Hafizi Z, Pirmoradi Z, Niknazar S, et al. Neuroprotective effect of berberine chloride on cognitive impairment and hippocampal damage in experimental model of vascular dementia. Iran J Basic Med Sci. 2018;21(1):53.
- 75. Kwon IH, Choi HS, Shin KS, Lee BK, Lee CK, Hwang BY, et al. Effects of berberine on 6-hydroxydopamine-induced neurotoxicity in PC12 cells and a rat model of Parkinson's disease. Neurosci Lett. 2010;486(1):29–33. http://dx.doi.org/10.1016/j.neulet.2010.09.038
- Kim M, Cho KH, Shin MS, Lee JM, Cho HS, Kim CJ, et al. Berberine prevents nigrostriatal dopaminergic neuronal loss and suppresses hippocampal apoptosis in mice with Parkinson's disease. Int J Mol Med. 2014;33(4):870–8. http://dx.doi.org/10.3892/ijmm.2014.1656
- Shin KS, Choi HS, Zhao TT, Suh KH, Kwon IH, Choi SO, et al. Neurotoxic effects of berberine on long-term L-DOPA administration in 6-hydroxydopamine-lesioned rat model of Parkinson's disease. Arch Pharm Res. 2013;36(6):759–67. http://dx.doi.org/10.1007/s12272-013-0051-4
- Kumar A, Chopra K, Mukherjee M, Pottabathini R, Dhull DK. Current knowledge and pharmacological profile of berberine: An update. Eur J Pharmacol. 2015;761:288–97. http://dx.doi.org/10.1016/j. ejphar.2015.05.068

Index 221

Index

A

AADvac1, 75 ABBV-8E12, 76 Abeotaxane (TPI-287), 73 ABvac40, 29 Acetylcholine, 11, 156, 170, 171, 182, 192, 194 Acetylcholinterase (AChE) AD treatment, use in, 170 amyloid plaques, production of, 171, 176 amyloid-beta accumulation, relationship between, 11 binding pocket, 174 catalytic site of, 174, 176 enzymes, 183 inhibitors, 71, 156, 171, 172, 173, 174, 175-176, 177, 185-187, 189, 192, 196-197, 198, 208 NFTs, role in producing, 171 retaining, 172 ACI-24, 29, 31 ACI-35, 75 Aducanumab (BIIB037), 24, 39–40, 157 Affitope AD02, 31 Aging, as rick factor for AD, 137 Akt, 103, 104, 105, 106, 112 Aluminum (Al), 31, 84, 86 Alzheimer Association, 100 Alzheimer, Alois, 2 Alzheimer's disease blood-brain barrier. role of (see blood-brain barrier)

characteristics of, 2, 151 decline, 6, 108 diabetes, relationship between (see diabetes) environmental factors, 152 epigenetic modifications in (see epigenetics in AD) etiology (see etiology of Alzheimer's) genetics, role of (see genetics, Alzheimer's) inflammation, role in Alzheimer's (see inflammation, role in Alzheimer's) manifestations of, 2, 3 neurodegeneration, 8, 52, 71, 73, 75, 77, 90, 120, 169 neuroinflammatory changes, 70, 92 neuropsychiatric manifestations, 166-167 nondiabetic, 102, 103, 104, 106, 112 (see also diabetes) overview, 70, 135, 136, 165, 166, 181 pathogenesis (see pathogenesis, Alzheimer's disease) pathogenic mechanisms, 62 pathology, 166, 167 preclinical, 29, 71, 120, 159, 198 preclinical drug candidate, 34, 192 prevalence (see prevalence, Alzheimer's disease) progression, 160 risk factors, 2 stages, 2, 6, 77, 155 synaptic losses, 168

AMPK, 109 Amyloid cascade hypothesis, 11, 24, 51, 52, 167-168, 181, 185 Amyloid precursor protein (APP), 83, 85-86, 90, 108, 120, 167-168 Amyloid β40 (Aβ40), 29, 36–37, 52, 54, 57-58, 152, 168, 215 Amyloid β42 (Aβ42), 58, 60–61, 89, 108, 111, 112, 152, 167-168 Amyloid- β , 85, 119. *See also* amyloid- β plaques amyloid hypothesis (see amyloid cascade hypothesis) antiaggregation effects, 187–188 antimicrobial properties, 156 inhibitors, 188-189 metal-binding properties, 87 peptides, 25, 108, 112, 156, 181 Amyloid- β plaques, 1, 5 aggregation in brain, 52, 87, 166 amyloid hypothesis (see amyloid cascade hypothesis) binding affinity, 37 brain, accumulation in, 24, 51-52, 153 diffuse, 166 immunotherapy, 77 increases of, in diabetics, 100 metals, relationship between, 86 reducing, 38, 40, 52 role of, in AD, 41, 140, 151, 152 tau proteins, combination therapy directed to, 79 vasculature surrounding, 143 AN1792 vaccine, 28, 29 APKC, 99, 103, 104, 105, 106, 107, 110, 112. See also hepatic aPKC activators, 100, 111 brain activity, 101 inhibitors, 108-110

memory function, relationship between, 110-111 recombinant, 108 Apolipoprotein E gene ε 4 allele (APOE ε4), 120, 140, 145, 157 Apolipoproteins E (ApoEs), 70, 140, 153, 154 AstraZeneca, 41 Astrocyte cell culture model, 118 Astrocytes, 4, 120, 123, 125, 127, 138, 141, 153, 168, 169, 211 ATP-binding cassette (ABC) transporters, 118 Atypical protein kinase C, 99, 100–101 Aurothiomalate [ATM], 105, 106 Autophagy lysosome pathway, 11 AZP2006, 72 A β plaques. *See* amyloid- β plaques A β precursor protein (APP), 3, 24, 25

B

BACE1 mRNA, 108 Bacopa, 205, 209, 211-213 BAN2401, 38-39, 157 Bapineuzumab, 34 BAPTA, 89 Benzyl piperidone, 173-174 Berberine, 205, 206, 215 Beta-site amyloid precursor protein cleaving enzyme 1 (BACE1), 100, 103, 106, 107, 108 AD, role in pathogenesis of, 101–102 inhibitors, 158, 185-186, 187 interactions with, 191 levels, 101 processing APPs, role in, 185 regulating, 160 Bioavailability of drugs, 128

Index 223

Biogen, 40 Blood-brain barrier, 73, 84, 89, 106 AD-related changes, 120-121 brain endothelial cells (BECs) (see brain endothelial cells (BECs)) cognitive impairment, as marker of, 140 damage, 145 defining, 117 description, 136 deterioration, 137 disruption, 137 drug delivery across, 118, 127–128, 128-129 dysfunction, in AD, 127, 128, 129 focused ultrasound use, 128 human versus rodent, 122 immune responses, 118 integrity of, 118, 125, 136 leakage, 120, 140 models, 2D, 125 models, 3D, 126-127 neuroimmunity, 138 organoids, 127 overview, 117, 118-120 penetrating, 187 permeability, 122, 135, 139, 194, 197 preclinical models, 125 SPARC effect on, 141, 142-143 in vitro model development, 117, 123, 125, 126, 129 Brain endothelial cells (BECs), 118 degeneration, 120 iBECs. 124 immortalized, 123, 124 primary, 123, 124 Brain's default mode network (BDMN), 153 Breast cancer resistance protein (BCRP), 118 Butyrylcholinesterase (BuCHE), 184

С

C-terminal fragments (CTFs), 55, 57 Cadmium (Cd), 84, 86-87 Calcium (Ca), 87 Calcium channel blockers (CCBs), 176-177 Cannabinoid receptors, 195 Canosine, 59 Capsaicin, 205, 214-215 Carrier mediated transport (CMT), 118 Cascade hypothesis, 24 Ceramide, 104 Cerebellum, 168 Chelators, metal, 52, 59, 86, 89, 90, 91, 170 Chimeric conjugates, 165, 172, 175-176 Cholinergic hypothesis, 11, 52, 169 Cholinergic neurotransmission, 192–194 Cholinesterase inhibitors, 24, 155, 165, 170, 182, 189 Cholinoesterases, 196 Clinical trials, ethical standards for, 78 Clinioquinol (CQ), 89 Cobalt (Co), 87, 90 Conventional PKCs (cPKCs). 107–108 Convolutional neural network (CNN) algorithm, 145 Copper (Cu), 84, 86, 87, 88, 89, 170 Copper-bis(thiosemicarbazonoto), 90 Crenzumab, 35 CRISPR. 128 Curcumin, 205, 206, 213-214 Cysteine, 86

D

DAG-activated cPKCs, 107–108 Davunetide peptide, 81 Dementia. *See also* Alzheimer's disease Dementia forms, 1 Diabetes copper overload, 89 hyperinsulinemia, 99-100, 101, 104-105, 106, 112 insulin receptors (IR), 102, 103, 104.112 insulin-resistant, 99, 101, 102, 106 insulin-resistant Alzheimer's disease, 100 liver, insulin action in, 103 metabolic syndrome (MetS) (see metabolic syndrome (MetS)) obesity/type-2-diabetic (O/T2D) states, 99, 102, 103, 105, 109, 111, 112 (see also obesity) pre-diabetes, 102, 103 type 2, 100, 102, 103 Diagnosis, Alzheimer's disease, 53 Disability Assessment for Dementia (DAD), 34 DNA methyltransferases, 151, 154 DNA-demethylating agents, 158 Dominantly Inherited Alzheimer Network (DIAN) trial, 38 Donecopride, 192 Donepezil, 170, 171, 172-174, 182, 185, 187, 195, 196 Down Syndrome, 4, 31, 213 DP-109,89 Drug delivery systems (DDSs), 91 Drug reactions, adverse, 206 Dystrophic neurites, 152

E

Early-onset Alzheimer's disease (EOAD), 3, 24 Endothelial cells, brain. *See* brain endothelial cells (BECs) Enzyme inhibitors, 196 Epigenetics in AD anti-amyloid-β therapies, 157 brain evolution, role in, 155 changes in, 151, 156, 160 changes, reversing, 156 gene modulation, 154, 158 therapies, 156–161 Epothilone D, 73 Etiology of Alzheimer's, 2, 3–5. *See also* genetics, Alzheimer's Excitotoxicity, 92

F

Flavinoids, 214 Flubiprofen, 197 Fluorination, 53 Focused ultrasound (FUCs), 128 Free radicals, 170 Fyn tyrosine kinase, 72

G

G protein-coupled receptors (GPCRs), 184, 192, 194, 198 Galantamine, 155, 170, 182 Gantenerumab, 31–32, 36, 38 Genentech, 35 Genetics, Alzheimer's environmental factors, interaction between, 152 factor in AD cases, 2 gene mutations, 24–25 inherited forms, 1, 3, 70 Mendelian inheritance, 3 mutations, 3 risk factors, 3–4 Ginkgo biloba, 205, 206–207, 207–208 Glucose transport, 137 Glycogen synthase kinase-3 (GSK-3), 72 GSK-3β, 86, 189, 190, 191

H

HDAC inhibitors (HDACis), 158-159 Hepatic aPKC, 106. See also aPKC Hevin, 136, 141, 143 Hippocampus, 152, 158, 160, 167 Histone acetylation, 151, 157–158 Histone acetyltransferases (HATs), 155.160 Histone deacetylases (HDACs), 155 Histone H4 with acetylation at the 16th lysine residue protein (H4K16ac), 157 - 158Homeostasis, brain, 136 Homeostasis, metal. See metal homeostasis Huntington's chorea, 213 Hyper-phosphorylated tau, 129 Hyperinsulinization, of brain, 106 Hyperosmosis, 127 Hyperphosphorylation of tau proteins. See Tau proteins

I

IBECs, 124, 127 ICAP, 109, 110 ICAPP, 109 IL-1 β , 108, 111, 187 Immunotherapy strategies ABvac40, 29–30 ACI-24, 29, 31 active-A β -directed strategies, 28–29 Affitope AD02, 31 anti-A β strategies, 24, 25–28 antibodies, clinical trials, 24

BAN2401, 38-39 bapineuzumab, 34 crenzumab, 35 gantenerumab, 36, 38 MEDI1814, 40-41 monoclonal antibodies, passive immunotherapy with, 31-32 ponezumab, 36-37 SAR228810, 41 solanezumab, 37-38 UB 311. 31 vaccines (see vaccines, for AD) Induced pluripotent stem cell, 118, 124 - 125Inflammation, role in Alzheimer's, 4, 10, 75, 104 Inherited forms of Alzheimer's. See genetics, Alzheimer's Insulin resistance, brain, 102 Insulin, role in AD, 9, 10 Interlaminar astrocytes, 123 Intravenous immunoglobulin (IVIg), 75,76 Ionophores IPSCs, 124, 128 Iron (Fe), 84, 86, 87, 90, 170

K

Kaempferol, 205, 214 Keyhole limpet cyanine (KHL) carrier proteins, 29

L

L-peptides, 57 Late-onset Alzheimer's disease (LOAD), 3 Lead (Pb), 84 Limbic system, 152

M

Machine learning (ML) algorithms, 144 Magnesium (Mg), 87 Mammalian target of rapamycin (mTOR), 104 Manganese (Mn), 84, 86, 87 Mannitol. 127 MAOs (MAO-A and MAO-B), 196 MAP kinases, 86 MEDI1814. 40-41 Memantine, 156, 170, 172, 182 Mendelian inheritance, 3 Met 35, 53-54 Metabolic syndrome (MetS), 100, 103, 105. 106. 111. See also diabetes Metal chelators. See chelators, metal Metal homeostasis, 83, 87-88 Metal hypothesis, 191 Metal imbalance hypothesis, 182 Metal ion hypothesis, 170 Metal protein attenuating compounds (MPACs), 88-89, 90 Metal-binding proteins, 84 Metal-focused interventions. See trace metals Metallochaperones, 84, 87-88, 90 Metals, trace. See trace metals; specific metals Meynert, 152 Microglia, 138, 141, 187 MicroRNA expression, 151 Microtubule-associated protein tau (MAPT), 5, 154 Mild traumatic brain injury (mTBI), 139-140 MiRNA, 160 Molybdenum (Mo), 87 Monoclonal antibodies, 31-32

MRNA, 111 Multidrug resistance-associated proteins (MRPs), 118 Multitarget-directed ligands (MTDLs). *See* multitargeted ligands (MTLs) Multitargeted ligands (MTLs), 182, 183–185 cholinergic inhibition, 194–195 cholinergic neurotransmission, 192–194 cholinesterases, targeting, 195 endocannabinoid system, targeting, 195 H3R antagonism, 194–195 serotonic, influence on, 192

Ν

N-benzylpiperidine, 185-186, 187, 193, 196 N-methyl-D-aspartate (NMDA) antagonists (glutamate inhibitor), 170, 171-172 N-methyl-D-aspartate (NMDA) neurotransmissions, 184 N-methyl-D-aspartate (NMDA) receptor antagonists, 71, 182 N-methyl-D-aspartate receptor agonist, 24 N-methylation, 53 Namzaric, 182 Nanomedicine, 83 Natural substances, use in AD. See also specific substances drug interactions, 205 overview, 205 Paquid study, 208–209 Neocortex, 6 Neurocognitive disorders, definition, 206, 215

Index 227

Neurodegeneration in Alzheimer's, 1, 169.170 Neurofibrillary tau tangles (NFTs), 1, 2, 5-6, 86, 92, 140, 151, 152, 166, 167.169 formation, 8 Neuroimmunity, 138 Neuroinflammation, 138-139 Neuropathology of Alzheimer's autophagy lysosome pathway, 11 A β plaques (see amyloid- β plaques) cholinergic hypothesis, 11 hypoglycemia, 10 inflammation, role in Alzheimer's (see inflammation, role in Alzheimer's) insulin, 9, 10 mitochodrial dysfunction, 9 neurofibrillary tau tangles (NFTs) (see neurofibrillary tau tangles (NFTs)) overview. 5 oxidative stress. 9 ubiquitin-proteasome system (UPS), 10 - 11vascular dysfunction, 10 Neuroplasticity, 153 Neurovascular unit (NVU), 118 NFKB, 108, 111-112 NMDA receptors, 7 NOR tests, 111 Novartis, 29 Nutritional interventions, AD, 87 NVU, 124

0

Obesity, 102, 103, 104, 106. See also diabetes Oligodendrocytes, 168 Oligomers, 7 Oxidative stress, 87, 104, 153, 169–170, 182

P

P-glycoprotein (P-gp), 118 Paired helical fragments (PHFs), 6 PAMPA, 196 Parkinson's disease, 213 Pathogenesis, Alzheimer's disease APP, 7-8 Aβ, 7–8 complexity, 196 hallmarks of, 182 NFTs, 8-9 overview. 7 pathogenic mechanisms, 62 tau, 8–9 Penicillamine, 90 Peptides Alzheimer's disease, use as strategy for. 52 inhibitors (see peptidic inhibitors) library screening, 59-61 Peptidic inhibitors A β peptide inhibitors, 54, 57–58, 61 - 62non-A β -based peptide inhibitors, 59 overview, 53-54 Pericytes, 118, 120-121, 125, 127 Peripheral endothelial cells, 118 Phage display, 59 Phagocytes, 25 Phosphatidic acid (PA), 104 PKC-δ, 108 PKC-λ, 108, 109, 111 PKMζ, 109, 110, 111

Polymeric nanoparticles, 91 Polypharmacology in AD. See also specific drugs drugs used, 182 multifunctional agents, 185-188 multitargeted ligands (MTLs) (see multitargeted ligands (MTLs)) Ponezumab. 36-37 Post-mortem histopathology of Alzheimer's, 6 Potassium (K), 87 Pre-diabetes. See diabetes Presenilin 1 (PSEN1), 3, 24, 120, 168 Presenilin 2 (PSEN2), 3, 24, 168 Prevalence, Alzheimer's disease, 2, 24, 52, 70, 100, 166 Proline (Pro), 86 Protein-fragment complementation assay (PCA), 81 Pyrrolidine dithiocarbamate (PDTC), 90

Q

Quercetin, 205, 206, 214

R

RAWM tests, 111 Reactive oxygen species (ROS), 206 Receptor-mediated transcytosis (RMT), 118 Reelin, 154–155 Repressor element 1-silencing transcription factor (REST), 157 Resveratrol, 205, 206, 213 Retro-inverso cyclization, 53 Rivastigmine, 170, 171, 172, 174–175, 182 Rottlerin, 107

S

Sanofi, 41 SAR228810, 41 Saracatinib, 72 Secreted protein acidic and rich in cysteine (SPARC) protein, 135-136, 140-141, 142-143, 144, 145 Serine (Ser), 86 Serotonin, 192-193 SHANK proteins, 92 Shogaol-huprine hybrids, 189 Short-term memory decline, 1 Sodium (Na), 87 Solanezumab, 37-38 SPARC. See secreted protein acidic and rich in cysteine (SPARC) protein Sulforaphane, 159 Synaptophysin, 89

T

T-cells, 25-26 Tacrine, 170, 171, 172, 175-176, 191, 193, 198 Tau hypothesis, 168-169, 182 Tau proteins aggregation prevention, 73-74, 188.189 Alzheimer's disease, as factor in, 71 amyloid, combination therapy directed to, 79 antibodies, 76, 77 hyperphosphorylation, 86, 90 immunotherapy, 74-76, 77 impact of incorrect levels of, 8-9 inhibitors, 188-189 isoforms of, 8 metal-binding properties, 87

Index 229

neuropathophysiological functions, 77 p-Tau, 34, 106 PET imaging of, 74 phosphorylation, 72 protein hypothesis, 52 protein therapy, 69 role of, 8 T-Tau. 34 trace metals, relationship between, 83.85 zinc, interactions with, 86 TEER, 124, 125 Tetrahydroisoquinoline-benzimidazole derivatives. 187 Threonine (Thr), 86 Tideglusib, 72 TJ proteins (TJPs), 122, 124 TNF-α, 108, 111–112, 187 Trace metals. See also specific metals Alzheimer's disease, relationship between, 84, 85, 91 deficiencies, 87 disregulation of, 84 homeostasis (see metal homeostasis) imbalance of, hypothesis (see metal hypothesis)

metal-focused interventions, 88–89 supplementation, 87 tau proteins, relationship between, 84 Transcriptional neoteny, 153 Transwell models, 125, 126 Traumatic brain injury (TBI), 139–140 Triethylenetetramine (TETA), 89, 90 Tryptophan hydroxylase, 84 Tyrosine hydroxylase, 84

U

UB 311, 31 Ubiquitin-proteasome system (UPS), 10–11

V

Vaccines, for AD, 23, 69, 74, 75. See also specific vaccines Vascular dementia, 4 Virtual screening, 136 Vitamin B3, 159

Z

Zinc (Zn), 84, 86, 87, 88, 89, 90, 91, 92

Doi: https://doi.org/10.36255/exonpublications.alzheimersdisease.2020.ind